

No d'Ordre : 00008/98

# THESE PRESENTEE POUR OBTENIR LE GRADE DE DOCTEUR EN SCIENCES DE L'UNIVERSITE NATIONALE DU BENIN

Option: Physique Mathématique par

Mama FOUPOUAGNIGNI

Laguerre-Hahn Orthogonal Polynomials with respect to the Hahn Operator: Fourth - Order Difference Equation for the rth Associated and the Laguerre-Freud Equations for the Recurrence coefficients.

Soutenue le 16 Décembre 1998 devant le JURY :

Président: Augustin BANYAGA (Pennsylvania State University, Pennsylvanie USA)

Rapporteurs Saïd BELMEHDI (Université des Sciences et Technologies de Flandres, Lille I)

M. Norbert HOUNKONNOU (Université Nationale du Bénin, Bénin)

W. KOEPF (Hochschule für Technik Wirtschaft und Kultur, Leipzig, Allemangne)

A. RONVEAUX (Facultés Universitaires Notre Dame de la Paix Namur-Belgique)

Examinateurs Jean-Pierre EZIN (Université Nationale du Bénin, Bénin)

Côme GOUDJO (Université Nationale du Bénin, Bénin)

Co-Directeurs M. Norbert HOUNKONNOU André RONVEAUX Laguerre-Hahn Orthogonal Polynomials with respect to the Hahn Operator: Fourth-order Difference Equation for the rth Associated and the Laguerre-Freud Equations for the Recurrence Coefficients

Mama FOUPOUAGNIGNI

Institut de Mathématiques et de Sciences Physiques Porto-Novo. Bénin A mon épouse Adjara et ma fille Samihra. A ma famille, mes amis et à tous ceux qui croient à l'effort et œuvrent pour la justice, la paix et la dignité humaine.

#### REMERCIEMENTS

Tout d'abord, je remercie le Professeur Augustin BANYAGA qui me fait un grand honneur en acceptant de présider le jury de cette thèse. Je remercie également les Professeurs Saïd BELMEHDI, Jean-Pierre EZIN. Wolfram KOEPF et Côme GOUDJO pour avoir accepté de faire partie du jury.

J'exprime ici ma reconnaissance aux professeurs M. Norbert HOUNKONNOU et André RONVEAUX pour les efforts fournis et les sacrifices énormes consentis pour la co-direction de cette thèse.

Mes remerciements vont aussi à l'endroit du Professeur Jean-Pierre Ezin, Directeur de l'IMSP, pour sa constante sollicitude.

Je remercie profondément le Service Allemand d'Echanges Universitaires (DAAD) qui, en m'octroyant une bourse doctorale, a rendu possible la réalisation de ce travail. Mes remerciements vont aussi à l'endroit de l'Université Nationale du Bénin et en particulier de l'Institut de Mathématiques et de Sciences Physiques pour l'hospitalité, le souvien financier et les sacrifices consentis tout au long de ma formation.

Le séjour en Europe, de septembre 1997 à mars 1998, a été déterminant pour la finalisation de ce travail. Pour cela, je tiens a remercier le DAAD pour avoir financé ce séjour, le Centre "Konrad-Zuse-Zentrum für Informationstechnik Berlin" (ZIB) pour m'avoir acceuilli en son sein et aussi pour m'avoir offert d'excellentes conditions de travail. C'est l'occasion ici de remercier le Professeur Wolfram KOEPF pour son hospitalité pendant mon séjour à Berlin et aussi pour la formation qu'il m'a donnée en calcul symbolique.

Harald BOING et le Professeur Winfried NEWN, tous de ZIB, m'ont aussi aidé notamment pour la programmation et je les en remercie.

Il me plait ici de remercier très sincèrement le Professeur André RONVEAUX qui, malgré son admission à la retraite, a financé à ses propres frais mon séjour à Namur et aussi son séjour à Berlin. Ces rapprochements ont permis une évolution très significative du travail.

Je remercie également Gérard LAGMAGO pour l'acheminement des copies de cette thèse à Namur. John TITANTAH pour la lecture du manuscrit de ce travail. Ivan AREA de l'Université de Vigo en Espagne, le Professeur Francisco MARCELLÁN de l'Universidad Carlos III de Madrid en Espagne pour les discussions fructueuses que nous avons eues, le Professeur Antonio J. DURAN de l'Université de Séville en Espagne pour avoir financé partiellement mon séjour à Seville et, le professeur Walter VAN ASSCHE de l'Université de Leuven en Belgique pour l'invitation, l'hospitalité, l'encadrement et le support financier dont j'ai bénéficiés pendant mon séjour à Leuven.

Mes remerciements vont aussi à l'endroit de mon père Moussa MOWOUM, ma mère Kentouma MAPIEMFOU, mon épouse Adjara FOUPOUAGNIGNI, ma fille Samihra FOUPOUAGNIGNI, mes frères Mamouda NJUTAPMVOUI, Mama NJOYA, Ferdinand NGAKEU, Issofa MOUNDI, Ousmanou PATOUOSSA, Abdou NGOUHOUO..., mes sœurs, pour les sacrifices qu'ils ont consentis pendant mon absence et aussi pour leur soutien inconditionnel.

J'exprime aussi ma reconnaissance aux «nseignants des Départements de Mathématiques de l'Ecole Normale Supérieure de Yaoundé et de l'Université de Yaoundé I, au Professeur Moïse KWATO NJOCK pour m'avoir encouragé à me présenter au concours d'admission à l'IMSP et aussi pour leur soutien et leurs conseils.

Mon séjour à l'IMSP a failli prendre fin en 1995, ceci à cause des difficultés administratives rencontrées au Ministère de l'Education Nationale du Cameroun. Ces problèmes ont été résolus heureusement et pour cela je remercie sincèrement les Professeurs Jean Pierre EZIN, Mahouton Norbert HOUNKONNOU et Moïse KWATO NJOCK pour leur compréhension et leur support pendant cette période difficile. Le soutien du Professeur Alphonse ELONG Proviseur du Lycée du Manengouba de Nkongsamba au Cameronn ainsi que les interventions de Youssouf NCHARE et Marie LOUISE du Ministère de l'Education Nationale

du Cameroun ont été aussi importantes pour la continuation de mes études à l'IMSP et je les en remercie.

Je remercie enfin le personnel enseignant et administratif de l'IMSP ainsi que les étudiants pour le bon climat de collaboration et de travail, sans oublier tous ceux qui de près ou de loin ont contribué à ma formation et à la soutenance de cette thèse.

## Contents

1	Intr	oduction	7
	1.1	Historical introduction	7
		1.1.1 The fourth-order differential and difference equation	7
		1.1.2 The non-linear difference equations	S
	1.2	Summary of the main results	10
		1.2.1 The fourth-order difference equation	10
		1.2.2 The non-linear recurrence equations	11
	1.3	Outline of dissertation	11
2	Prel	liminaries	13
		2.0.1 The notion of topology	13
		2.0.2 Notations	14
	2.1	Orthogonality and quasi-orthogonality	14
			14
			16
		2.1.3 Other definitions	16
		2.1.4 Dual basis	17
	2.2	Associated orthogonal polynomials	18
		2.2.1 Three-term recurrence relation	18
		2.2.2 The first associated orthogonal polynomials	19
		2.2.3 The $r$ th associated orthogonal polynomials	19
	2.3	Operators $\mathcal{D}, \ \mathcal{T}_{\omega}, \ D_{\omega}, \ \mathcal{G}_q \ \text{and} \ \mathcal{D}_q \ \dots $	20
		2.3.1 Operator $\mathcal{D}$	20
		2.3.2 Class of the $\mathcal{D}$ -semi-classical linear functional	21
		2.3.3 Characterisation of $\mathcal{D}$ -classical orthogonal polynomials	22
		2.3.4 Operators $\mathcal{T}_{\omega}$ and $D_{\omega}$	23
		2.3.5 Class of the $D_{\omega}$ -semi-classical linear functional	25
		2.3.6 Characterisation of $\Delta$ -classical orthogonal polynomials	26
		2.3.7 Operators $\mathcal{G}_q$ and $\mathcal{D}_q$	27
		2.3.8 Class of the $\mathcal{D}_q$ -semi-classical linear functional	28
		2.3.9 Characterisation of $\mathcal{D}_q$ -classical orthogonal polynomials	29
	2.4	The $q$ -integration	31
		2.4.1 The q-integration on the interval $[0, a], a > 0 \dots \dots \dots \dots$	31
		2.4.2 The q-integration on the interval $[a, 0]$ , $a < 0$	31
		2.4.3 The q-integration on the interval $[a, \infty[, a > 0 \dots \dots \dots \dots \dots]]$	31
		2.4.4 The q-integration the interval $]-\infty,a]$ . $a<0$	32
3	The	e $D_{q,\omega}$ -semi-classical orthogonal polynomials	33
	3.1	Introduction	33
		3.1.1 Operators $A_{q,\omega}$ and $D_{q,\omega}$	33
		3.1.2 Class of the $D_{q,\omega}$ -semi-classical linear functional	38
	3.2	Characterisation theorems for $D_{q,\omega}$ -semi-classical OP	41
		3.2.1 $D_{a,a}$ -classical orthogonal polynomials	41

CONTENTS 5

		3.2.2	$D_{q,\omega}$ -semi-classical orthogonal polynomials	48				
4	4.1	The St 4.1.1 4.1.2	al Stieltjes function tieltjes function and the Riccati difference equation Some definitions Some properties	51 51 51 52				
	4.2	$D_{q,\omega}$ -I	Laguerre-Hahn OP as $\mathcal{D}_q$ -Laguerre-Hahn OP	57				
5			equations for the first associated OP	59				
	5.1		luction	59				
	5.2		sical weight	59				
	$\begin{array}{c} 5.3 \\ 5.4 \end{array}$		n-order $q$ -difference equation for $P_{n-1}^{(1)}(x;q)$	60 61				
	9.1	5.4.1	The first associated Little and Big q-Jacobi polynomials	61				
		5.4.2	The first associated $\mathcal{D}$ -classical orthogonal polynomials	62				
		5.4.3	The first associated $D_{q,\omega}$ -classical orthogonal polynomials	62				
		5.4.4	The first associated $\Delta$ -classical orthogonal polynomials	63				
6	Diff	erence	e equations for the $r$ th associated OP	65				
	6.1		luction	65				
	6.2		ssociated $\mathcal{D}_q$ -Laguerre-Hahn linear functional	65				
		6.2.1	The associated $\mathcal{D}_q$ -Laguerre-Hahn linear functional is a $\mathcal{D}_q$ -Laguerre-Hahn linear	C.F				
	6.3	Fourtl	functional	65 68				
	5.5	6.3.1	Fourth-order differential equation for $P_n^{(r)}$	74				
		6.3.2	Fourth-order difference equation for the $r$ th associated $D_{q,\omega}$ -Laguerre-Hahn orthog-	1-1				
			onal polynomials	75				
		6.3.3	Fourth-order difference equation for the $r$ th associated $\Delta$ -Laguerre-Hahn orthogo-					
			nal polynomials	76				
	6.4		cation of difference equations to classical situations	76				
		$6.4.1 \\ 6.4.2$	Coefficients $E_r$ , $F_r$ and $H_r$ for classical situations	76 77				
		6.4.2	Fourth-order differential equation for the rth associated $\mathcal{D}$ -classical orthogonal poly-	• •				
		0.2.0	nomials	78				
		6.4.4	Fourth-order difference equation for the $r$ th associated $\Delta$ -classical orthogonal poly-					
			nomials	78				
7	Thr	ee-ter	m recurrence relation coefficients	81				
-	7.1		duction	81				
	7.2		e-term recurrence relation coefficients for $\mathcal{D}_q$ -classical situations					
		7.2.1	Coefficients $T_{n,1}$ and $T_{n,2}$					
	7.3	7.2.2 Three	Coefficients $\beta_n$ and $\gamma_n$ for $\mathcal{D}_q$ -classical orthogonal polynomials					
	1.5	7.3.1	Coefficients $\tilde{T}_{n,1}$ and $\tilde{T}_{n,2}$					
	7.4							
8	Lag	'iierre-	Freud equations for semi-classical OP of class 1	84				
Ŭ	8.1		duction					
	8.2	0 0						
	8.3		mediate coefficients					
		8.3.1	Coefficients $T_{n,j}$					
		8.3.2	Coefficients $B_n^k$					
	8.4	8.3.3 Final	Structure relations					
	3.4		Laguerre-Freud equations for $\mathcal{D}_{\circ}$ -classical orthogonal polynomials					

6 CONTENTS

	8.5	Applic	ations to $\mathcal{D},D_\omega$ and $D_{q,\omega}$ -semi-classical OP of class $1$	91						
		8.5.1	Laguerre-Freud equations for $\mathcal{D}$ -semi-classical OP of class $1 \dots \dots \dots$	91						
		8.5.2	Laguerre-Freud equations for $D_{\omega}$ -semi-classical OP of class $1 \dots \dots \dots$	91						
		8.5.3	Laguerre-Freud equations for $D_{\omega}$ -classical orthogonal polynomials	92						
	8.6	Applie	ations to generalised Charlier and generalised Meixner polynomials of class one	93						
		8.6.1	Laguerre-Freud equations for the generalised Meixner polynomial of class one	93						
		8.6.2	Laguerre-Freud equations for generalised Charlier polynomial of class one	94						
		8.6.3	Asymptotic behaviour	95						
9	Conclusion and perspectives 9									
	9.1	9.1 Conclusion								
			ectives							
10	Арј	pendic	es	99						
	10.1	Apper	ndix I	99						
		10.1.1	About $\mathcal{D}$ -classical orthogonal polynomials	99						
		10.1.2	About $\Delta$ -classical orthogonal polynomials	99						
		10.1.3	About q polynomials	100						
	10.2	Apper	ndix II	100						
		10.2.1	Results on general associated classical discrete polynomials	100						
	10.3	Apper	ndix III	105						

## Chapter 1

## Introduction

#### 1.1 Historical introduction

#### 1.1.1 The fourth-order differential and difference equation

Consider the family of monic polynomials  $\{P_n\}_{n\in\mathbb{N}}$ , orthogonal with respect to a linear functional  $\mathcal{L}$  (see (2.5)). It satisfies a three-term recurrence relation (which we denote TTRR) [Chihara, 1978]

$$\begin{cases} P_{n+1}(x) = (x - \beta_n)P_n(x) - \gamma_n P_{n-1}(x), & n \ge 1, \\ P_0(x) = 1, P_1(x) = x - \beta_0, \end{cases}$$

where  $\beta_n$  and  $\gamma_n$  are complex numbers with  $\gamma_n \neq 0 \quad \forall n \in \mathbb{N}$ .

The rth associated of  $\{P_n\}_{n\in\mathbb{N}}$  is the family of monic polynomials  $\{P_n^{(r)}\}_{n\in\mathbb{N}}$ , defined by the previous relation in which  $\beta_n$ ,  $\gamma_n$  and  $P_n$  are replaced by  $\beta_{n+r}$ ,  $\gamma_{n+r}$  and  $P_n^{(r)}$ , respectively,

$$\begin{cases}
P_{n+1}^{(r)}(x) = (x - \beta_{n+r}) P_n^{(r)}(x) - \gamma_{n+r} P_{n-1}^{(r)}(x), & n \ge 1, \\
P_0^{(r)}(x) = 1, P_1^{(r)}(x) = x - \beta_r.
\end{cases}$$

The rth associated of the regular linear functional  $\mathcal{L}$  is, by Favard Theorem [Favard, 1935], the unique linear functional  $\mathcal{L}^{(r)}$  with respect to which  $\{P_n^{(r)}\}_{n\in\mathbb{N}}$  is orthogonal and satisfices  $\langle \mathcal{L}^{(r)}, 1 \rangle = \gamma_r$ .

Let  $\{P_n\}_{n\in\mathbb{N}}$  be a family of polynomials, orthogonal with respect to the linear functional  $\mathcal{L}$  and  $S(\mathcal{L})$ , the Stieltjes function of  $\mathcal{L}$  given by

$$S(\mathcal{L})(x) = S(x) = -\sum_{n>0} \frac{M_n}{x^{n+1}},$$

where  $M_n$  is the moment of order n of  $\mathcal{L}$ :  $M_n = (\mathcal{L}, x^n)$ .

When S satisfies a Riccati differential equation

$$\phi(x)S(x)' = B(x)S(x)^{2} + A(x)S(x) + D(x),$$

where  $\phi$ , A, B and D are polynomials, then  $\{P_n\}_{n\in\mathbb{N}}$  are called Laguerre-Halm orthogonal polynomials [Magnus, 1984], [Dzoumba, 1985]. It is well-known [Magnus, 1984] that these polynomials satisfy a fourth-order linear differential equation.

Classical and semi-classical (continuous) orthogonal polynnomials are particular cases of Laguerre-Hahn orthogonal polynomials, and they satisfy a second-order linear differential equation.

The rth associated Laguerre-Hahn orthogonal polynomials are Laguerre-Hahn orthogonal polynomials, therefore they satisfy a fourth-order linear differential equation.

The search for these differential equations has been very intensive during the past few years. For r=1, Grosjean (1985, 1986) found them for Legendre and Jacobi families, and Ronveaux (1988), has given a single equation valid for the first associated classical (continuous) orthogonal polynomials.

For an arbitrary r, computer algebra packages have been very useful due to the heavy computations involved. In this context we mention that Wimp (1987) has used the MACSYMA [ref] package to construct the fourth-order differential equations satisfied by the rth associated Jacobi polynomials (r in this case is integer or not). Belmehdi and Ronveaux (1989) devised a REDUCE program in order to obtain these differential equations for the associated classical orthogonal polynomials of integer (and fixed) order r.

Differential equations valid for the rth associated Laguerre-Hahn orthogonal polynomials and for any integer r were given by Belmehdi et al. (1991) using the properties of the Stieltjes function of the associated functional (see [Magnus, 1984]. [Dzoumba, 1985]). Then, followed some papers giving, in a simple way, the single fourth-order differential equation for the associated classical orthogonal polynomials of any integer order r (see for instance [Ronveaux, 1991], [Zarzo et al., 1993], [Lewanowicz, 1995]).

As it was the case for the associated orthogonal polynomial of a continuous variable, many works have been done to give the fourth-order difference equation satisfied by the associated classical orthogonal polynomials of a discrete variable.

Atakishiyev et al. (1996) have derived the relation (already known for classical continuous orthogonal polynomials [Ronveaux, 1988]) giving the link between the first associated classical discrete orthogonal polynomials and the starting polynomials, and used this relation to prove that the first associated of the classical discrete orthogonal polynomials are solutions of a fourth-order linear difference equation which can be factored as product of two second-order linear difference equations.

Using the explicit representation of the associated Meixner polynomials (with the real association parameter  $r \geq 0$ ) in terms of hypergeometric functions, Letessier et al.(1996) gave the fourth-order difference equation satisfied by the rth associated Meixner polynomials and deduced by an appropriate limit process the difference equation for the rth associated Charlier, Laguerre and Hermite polynomials.

This equation, thanks to the computer algebra system MATHEMATICA [Wolfram, 1993] and the relation proved in [Atakishiyev et al., 1996] is given explicitly for the first associated of Charlier, Meixner, Krawtchouk and Hahn polynomials [Ronveaux et al., 1998a].

The question one can ask is whether it is possible to give one fourth-order difference equation valid for the rth associated Laguerre-Hahn orthogonal polynomials including orthogonal polynomials of continuous, discrete variable and also q-polynomials? The answer is yes and the first part of this dissertation aimed at answering this question.

#### 1.1.2 The non-linear difference equations

Here, we consider that the polynomials  $\{P_n\}_{n\in\mathbb{N}}$  orthogonal with respect the semi-classical linear functional  $\mathcal{L}$  is orthonormal  $(\langle \mathcal{L}, P_n P_n \rangle = 1 \quad \forall n \in \mathbb{N}$ , thus, satisfying

$$x P_n = a_{n+1} P_{n+1} + b_n P_n + a_n P_{n-1}, \ n \ge 0, \ a_0 P_{-1} = 0,$$

where  $a_n$  and  $b_n$  are complex numbers with  $a_n \neq 0$ .

The coefficients  $a_n$  and  $b_n$  can be given explicitly for classical (continuous) orthogonal polynomials in terms of the polynomials  $\phi$  and  $\psi$  appearing in the Pearson differential equation,  $\frac{d}{dx}(\phi \mathcal{L}) = \psi \mathcal{L}$ , satisfied by the linear functional  $\mathcal{L}$  with respect to which  $\{P_n\}_{n\in\mathbb{N}}$  is orthogonal (see for instance [Nikiforov et al., 1983] [Chihara, 1978], [Szegő, 1939], [Lesky, 1985], [Koepf et al., 1996]...).

These coefficients are also known for classical orthogonal polynomials of a discrete variable and for q-classical orthogonal polynomials ([Nikiforov et al., 1991], [Szegő, 1939], [Lesky, 1985], [Koepf et al., 1996], [Medem, 1996]...).

When the polynomials are semi-classical (instead of classical), except for some particular cases, it is difficult to give, in general situation, the coefficients  $a_n$  and  $b_n$ .

The properties of the coefficients  $a_n$  and  $b_n$  as well as those of the polynomials  $P_n$  have been investigated by many authors.

• Firstly, we cite for example Laguerre, who, in 1885, explored the properties of the orthogonal polynomials related to the weight function  $\rho$  satisfying

$$\frac{\rho'(x)}{\rho(x)} = R(x),$$

where R(x) is a rational function of x. He also studied Padé approximations and continued fraction expansions of functions satisfying a differential equation of the form

$$W(x)f'(x) = 2V(x)f(x) + U(x),$$

where U, V and W are polynomials; and recovered orthogonal polynomials  $P_n$  as denominators of the approximants of f. He succeeded in showing that the orthogonal polynomials  $P_n$  satisfy the remarkable differential equation,

$$W\Theta_n y'' + [(2V + W')\Theta_n - W\Theta_n']y' + K_n y = 0,$$

where  $\Theta_n$  and  $K_n$  are polynomials with bounded degrees, whose coefficients are solutions of certain (usually) non-linear equations which provide non-linear equations for  $a_n$  and  $b_n$  (see [Magnus, 1991] for more details about Laguerre equations).

• Secondly, we cite the works by Freud (see [Freud, 1976, 1977, 1986]) who investigated the asymptotic behaviour of the recurrence coefficients for special families of measures by a technique producing an infinite system of (usually non-linear) equations (called Freud equations) for these coefficients (see [Magnus, 1991] for more details about Freud equations). For example, if the polynomials  $P_n$  are related to the weight  $\rho(x) = \exp(-x^4)$  on the whole real line, then the Freud equations are reduced to [Nevai, 1983]

$$\begin{cases} 4a_n^2(a_{n+1}^2 + a_n^2 + a_{n-1}^2) = n, \ n \ge 2, \ a_0 = 0, \ a_1^2 = \frac{\Gamma(3/4)}{\Gamma(1/4)}, \\ b_n = 0, \quad n \ge 0. \end{cases}$$

It should be noted that other people found similar non-linear equations and identities (see for instance [Laguerre, 1885], [Perron, 1929], [Shohat, 1939], see also [Nevai et al., 1986], [Magnus, 1991] for more details), but these authors did not study their solutions when no simple form could be found.

Using the Freud equations, Freud (1976) gave a conjecture about the asymptotic behaviour of recurrence coefficients when the polynomials  $P_n$  are related to the weight function  $\rho(x) = |x|^{\ell} \exp(-|x|^{\alpha})$  stating that :

Let  $a_n$  and  $b_n$  be the coefficients of the following recurrence relation

$$x P_n = a_{n+1} P_{n+1} + b_n P_n + a_n P_{n-1}, \ n \ge 0, \ a_0 P_{-1} = 0,$$

satisfied by the polynomials  $\{P_n\}_{n\in\mathbb{N}}$ , orthogonal with respect to the weight  $\rho(x)=|x|^{\ell}\exp(-|x|^{\alpha})$ ,  $\ell>-1$ ,  $\alpha>0$ , on the whole real line. Then  $a_n$  and  $b_n$  obey:

$$\lim_{n\to\infty} \frac{a_n}{[n/C(\alpha)]^{1/\alpha}} = 1, \ C(\alpha) = \frac{2\Gamma(\alpha)}{\Gamma(\alpha/2)^2}.$$

Important investigations have been devoted to the proof of Freud conjecture as well as to the study of the asymptotics for  $\{P_n\}_{n\in\mathbb{N}}$ , the distribution of zeros, the sharp estimates of the extreme zeros ... ([Chihara, 1978], [Freud, 1976, 1977, 1986], [Lubinsky, 1984, 1985a, 1985b], [Lubinsky et al. 1986, 1988], [Magnus, 1984. 1985a, 1985b, 1986], [Bonan, 1984], [Máté et al., 1985], [Mhaskar et al., 1984a, 1984b, 1985a, 1985a, 1984a, 1984b, 1985, 1986], [Sheen, 1984] ..., for more details see [Magnus, 1984, 1985a, 1985b, 1986]).

Later, Belmehdi and Ronveaux (1994) gave a systematic way to obtain non-linear equations for the recurrence coefficients, valid for any semi-classical orthogonal polynomial of a continuous variable. In fact, given a semi-classical linear functional  $\mathcal{L}$  satisfying  $\frac{d}{dx}(\phi\mathcal{L}) = \psi\mathcal{L}$ , where  $\phi$  and  $\psi$  are polynomials, they were able to provide two non-linear equations for the coefficients  $a_n$ ,  $b_n$  of the recurrence relation satisfied by the polynomials  $\{P_n\}_{n\in\mathbb{N}}$  associated to  $\mathcal{L}$ , called Laguerre-Freud equations (denomination borrowed from Magnus [Magnus 1985b, 1986]).

In the second part of this dissertation, we give a generalisation of the previous results [Belmehdi et al., 1994] by giving the system of two non-linear difference equations satisfied by the recurrence coefficients: equations which are valid for semi-classical orthogonal polynomials of a continuous and discrete variable, and also for q-semi-classical orthogonal polynomials (both of class 1).

#### 1.2 Summary of the main results

#### 1.2.1 The fourth-order difference equation

1. Using the result in [Suslov, 1989], we prove the following:

Consider  $\mathcal{L}$  a regular linear functional satisfying  $\mathcal{D}_q(\phi\mathcal{L}) = \psi\mathcal{L}$ , where  $\psi$  is a first degree polynomial and  $\phi$  a polynomial of degree at most two.  $\mathcal{D}_q$  is the Hahn operator defined by

$$\mathcal{D}_q f(x) = \frac{f(qx) - f(x)}{(q-1)x}, \quad x \neq 0, \quad q \neq 0, \quad q \neq 1, \mathcal{D}_q f(0) := f'(0).$$

Then, if  $\{P_n\}_{n\in\mathbb{N}}$  is the monic family of polynomials, orthogonal with respect to  $\mathcal{L}$ , then, the first associated  $P_n^{(1)}$  of  $P_n$  satisfies the fourth-order difference equation

$$Q_{2,n-1}^{**} \frac{Q_{2,n-1}^*}{q^2 (q-1)^2 x^2} \left[ P_{n-1}^{(1)}(x;q) \right] = 0.$$

Operators  $\mathcal{Q}_{2,n-1}^{**}$  and  $\mathcal{Q}_{2,n-1}^{*}$  are given by:

$$\begin{split} \mathcal{Q}_{2,n-1}^{\star} &= \phi_{(2)} \, \mathcal{G}_{q}^{2} - ((1+q)\phi_{(1)} + \psi_{(1)} \, t_{1} - \lambda_{n,0} \, t_{1}^{2}) \mathcal{G}_{q} + q \, (\phi + \psi \, t) \, \mathcal{I}_{d}, \\ \mathcal{Q}_{2,n-1}^{\star\star} &= (\phi_{(3)} + \psi_{(3)} \, t_{3}) [q^{2} \, A_{1} + (1+q) \, \phi_{(2)} + \psi_{(2)} \, t_{2}] \mathcal{G}_{q}^{2} \\ &- [q^{3} \, A_{1} \, (\phi_{(2)} + \psi_{(2)} \, t_{2}) + A_{3} \, (\phi_{(2)} + q \, A_{1})] \, \mathcal{G}_{q} \\ &+ q \, \phi_{(1)} \, [q^{2} \, A_{2} + (1+q) \, \phi_{(3)} + \psi_{(3)} \, t_{3})] \, \mathcal{I}_{d}, \end{split}$$

with

$$\lambda_{n,0} = -[n]_q \{ \psi' + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \}. \ [n]_q = \frac{q^n - 1}{q - 1}. \ q \neq 1, \ n \geq 0, \ \mathcal{G}_q P(x) = P(qx) \quad \forall P \in \mathbb{P},$$

$$\phi_{(i)} \equiv \phi(q^i x), \ \psi_{(i)} \equiv \psi(q^i x), \ t_i \equiv t(q^i x), \ t(x) = (q - 1)x,$$

$$A_j = (1 + q)\phi_{(j)} + \psi_{(j)} t_j - \lambda_{n,0} t_j^2.$$

This result [Foupouagnigni et al., 1998d], is used to deduce the factored form of the difference equations satisfied by the first associated classical orthogonal polynomials of a discrete variable [Ronveaux et al., 1998a] and also the factored form of the differential equation satisfied by the first associated classical continuous orthogonal polynomials [Ronveaux, 1988]. We have used, also, this result to prove that under certain conditions on the parameters, the first associated of little and big q-Jacobi polynomials are still classical. Moreover, we deduce that if  $p_n(x;a,b|q)$  (respectively  $P_n(x;a,b,c;q)$ ) denotes the monic little q-Jacobi polynomials (respectively monic big q-Jacobi polynomials), then they are related with their respective first associated by:

$$p_n^{(1)}(x; a, \frac{1}{q a} | q) = a^n q^n p_n(\frac{x}{a q}; \frac{1}{a}, a q | q),$$

$$P_n^{(1)}(x; a, \frac{1}{q a}, c; q) = a^n P_n(\frac{x}{a}; \frac{1}{a}, a q, c q; q).$$

2. We prove that the rth associated  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials satisfy the single fourth-order difference equation [Foupouagnigni et al., 1998e]

$$\sum_{j=0}^{4} I_{j}(n,r,q,x) \mathcal{D}_{q}^{j} P_{n}^{(r)} = 0,$$

where  $I_i(n, r, q, x)$  are polynomials in x.

We use suitable change of variable and limit processes to extend the above result to the rth associated Laguerre-Hahn orthogonal polynomials of a continuous and a discrete variable, respectively [Foupouagnigni et al 1998b].

We apply this result to compute explicitly the coefficients  $I_j(n, r, q, x)$  for the rth associated classical orthogonal polynomials (including classical continuous, classical discrete and q-classical polynomials) [Foupouagnigni et al., 1998b, 1998c, 1998e].

#### 1.2.2 The non-linear recurrence equations

We prove the following theorem (see 8.1) which is the main result of the second part of this Dissertation.

#### Theorem

The coefficients  $\beta_n$  and  $\gamma_n$  of the three-term recurrence relation

$$P_{n+1}(x) = (x - \beta_n)P_n(x) - \gamma_n P_{n-1}(x), n \ge 1, P_0(x) = 1, P_1(x) = x - \beta_0,$$

satisfied by the  $\mathcal{D}_q$ -semi-classical monic orthogonal polynomials of class at most one,  $\{P_n\}_{n\in\mathbb{N}}$ , can be computed recursively from the two non-linear equations

$$\begin{cases} (\psi_2 + [2n]_{\frac{1}{q}} \frac{\phi_3}{q})(\gamma_n + \gamma_{n+1}) = F_1(q; \beta_0, \dots, \beta_n; \gamma_1, \dots, \gamma_n), \\ (\psi_2 + [2n+1]_{\frac{1}{q}} \frac{\phi_3}{q})\beta_{n+1}\gamma_{n+1} = F_2(q; \beta_0, \dots, \beta_n; \gamma_1, \dots, \gamma_{n+1}). \end{cases}$$

 $\phi_j$  and  $\psi_j$  are the coefficients of the polynomials  $\phi$  and  $\psi$  ( $\phi(x) = \sum_{j=0}^3 \phi_j x^j$ ,  $\psi(x) = \sum_{j=0}^2 \psi_j x^j$ )

appearing in the  $\mathcal{D}_q$ -Pearson equation,  $\mathcal{D}_q(\phi\mathcal{L}) = \psi\mathcal{L}$ , satisfied by the regular linear functional  $\mathcal{L}$ .  $F_1$  is a polynomial of 2n+1 variables and of degree 2; and  $F_2$  a polynomial of 2n+2 variables and of degree 3, with the initial conditions

$$eta_0 = rac{\langle \mathcal{L}, x 
angle}{\langle \mathcal{L}, 1 
angle}, \ \psi_2 \, \gamma_1 = -\psi(eta_0).$$

We use suitable change of variable and limit processes to extend the previous theorem to the  $\mathcal{D}$  and  $\Delta$ -semi-classical orthogonal polynomials of class at most one [Foupouagnigni et al., 1998a]. We then give the Laguerre-Freud equations for the generalised Charlier and generalised Meixner of class one and use these equations (numerical and symbolic computation with Maple V Release 4) to give a conjecture about the asymptotic behaviour of the coefficients  $\beta_n$  and  $\gamma_n$  of the generalised Charlier and generalised Meixner polynomials of class one:

#### Conjecture

The coefficients  $\beta_n$  and  $\gamma_n$  of the three-term recurrence relation satisfied by the monic generalised Meixner polynomials of class one obey:

$$\lim_{n \to \infty} \left( \beta_n - \frac{1+\mu}{1-\mu} \, n - \frac{\mu \, (\alpha_1 + \alpha_2 - 1)}{1-\mu} \right) = 0, \ \lim_{n \to \infty} \left( \gamma_n - \frac{\mu (n+\alpha_1 - 1)(n+\alpha_2 - 1)}{(1-\mu)^2} \right) = 0,$$

and those of the three-term recurrence relation satisfied by the monic generalised Charlier polynomials of class one obey:

$$\lim_{n\to\infty}(\beta_n-n)=0,\ \lim_{n\to\infty}(\gamma_n-\mu)=0.$$

#### 1.3 Outline of dissertation

In Chapter 2 we give some results and definitions on orthogonal and associated orthogonal polynomials. We also prove some characterisation theorems for classical orthogonal polynomials.

Chapter 3 gives some useful properties of the operators  $A_{q,\omega}$  and  $D_{q,\omega}$  and the proof of some characterisation theorems for  $D_{q,\omega}$ -classical and  $D_{q,\omega}$ -semi-classical orthogonal polynomials; characterisation theorem which are valid (by limit processes) for the operators  $\frac{d}{dx}$ ,  $\mathcal{D}_q$  and  $\Delta$ .

Chapter 4 is devoted to the study of the  $D_{q,\omega}$ -Riccati difference equation satisfied by the Stieltjes function of the given associated linear functional. In particular, we prove that the affine  $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials are the  $D_{q,\omega}$ -semi-classical orthogonal polynomials and conversely. In this chapter, it is also proved that the  $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials can be obtained from the  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials by a change of variable.

In Chapter 5 we give the factored form of the fourth-order difference equation satisfied by the first associated  $\mathcal{D}_q$ -classical orthogonal polynomials and we deduce the difference equation for classical orthogonal polynomials of continuous and of discrete variable. We also consider the situations for which the first associated of the little and big q-Jacobi polynomials are still classical.

Chapter 6 describes the method used to obtain, for the general situation, the single fourth-order difference equation satisfied by the rth associated  $\mathcal{D}$ ,  $\mathcal{D}_q$  and  $\Delta$ -Laguerre-Hahn orthogonal polynomials. The coefficients of the fourth-order difference equation for classical situations are also given explicitly.

Chapter 7 gives useful coefficients for classical orthogonal polynomials like  $\beta_n$ ,  $\gamma_n$ ,  $T_{n,1}$  and  $T_{n,2}$ .

Chapter 8 presents the method used to obtain the two non-linear equations for the coefficients of the TTRR satisfied by the  $\mathcal{D}_q$ -semi-classical orthogonal polynomials of class at most one. We also show how these equations can be used to obtain the two non-linear equations for the coefficients of the TTRR satisfied by the  $\mathcal{D}$  and  $\Delta$ -semi-classical orthogonal polynomials of class at most one. The conjecture about the asymptotic behaviour of the coefficients of the TTRR satisfied by the generalised Charlier and the generalised Meixner polynomials of class one (conjecture obtained thanks to the two-non-linear equations) is also given.

The appendices I, II and III contain the data for classical orthogonal polynomials as well as the results on the fourth-order difference equations for classical situations.

#### It should be mentioned that:

- Chapter 2, devoted to the preliminaries, is based on [Chihara, 1978], [Guerfi, 1988], [Belmehdi, 1990a], [Salto, 1995] and [Medem, 1996].
- Chapters 3 and 4 generalise to the operator  $D_{q,\omega}$  certain results given in the above mentioned references.
- The original results obtained in the framework of this thesis are presented in chapters 5, 6 and 8.

## Chapter 2

## **Preliminaries**

#### 2.0.1 The notion of topology

We recall the notion of topology on polynomials and linear functional vector spaces. These notions have been defined in [Trèves, 1967], [Maroni, 1985, 1988], [Guerfi, 1988] and [Belmehdi, 1990a]. For these preliminaries, we shall exploit the works by Maroni [Maroni, 1988], Guerfi [Guerfi, 1988] and [Belmehdi, 1990a].

Let  $\mathbb{P}$  be a vector space of polynomials in one real variable with complex coefficients, endowed with the strict inductive limit topology of the spaces  $\mathbb{P}_n$ .  $\mathbb{P}_n \subset \mathbb{P}$  is the vector space of polynomials of degree at most n. It satisfies

$$\mathbb{P}_n \subset \mathbb{P}_{n+1}, \ n \ge 0, \ \mathbb{P} = \bigcup_{n=0}^{\infty} \mathbb{P}_n,$$

and is endowed with its natural topology which makes it a Banach space.

Let  $\mathbb{P}'$  be the dual of  $\mathbb{P}$ , equiped with its topology which is defined by the system of semi-norms:

$$||\mathcal{L}||_n = \sup_{k \le n} |M_k|,$$

where  $M_k$  denotes the moments of the functional  $\mathcal{L}$  with respect to the sequence  $\{x^n\}_n$ :  $M_k = (\mathcal{L})_k = (\mathcal{L}, x^k)$ .  $\mathbb{P}$  and  $\mathbb{P}'$  are Fréchet spaces.

We consider  $\mathbb{V}$  the vector space generated by the elements  $\{\frac{(-1)^n}{n!}\mathcal{D}^n\delta\}_n$   $(\mathcal{D}=\frac{d}{dx})$  with its inductive limit topology.  $\delta$  denotes the Dirac measure:  $\langle \delta, f \rangle = f(0), f \in C^{\infty}(\mathbb{R})$ .

Let  $\mathcal{F}$  be the linear application:

$$\mathcal{F}: \mathbb{V} \longrightarrow \mathbb{P}$$

$$d = \sum_{j=0}^{n} d_{j} \frac{(-1)^{j}}{j!} \mathcal{D}^{j} \delta \longrightarrow \mathcal{F}(d) = \sum_{j=0}^{n} d_{j} x^{j}.$$
(2.1)

 $\mathcal{F}$  verifies the following properties:

- i)  $\mathcal{F}$  is an isomorphism defined on  $\mathbb{V}$  into  $\mathbb{P}$ .
- ii) The transpose  ${}^t\mathcal{F}$  of  $\mathcal{F}$ , is an isomorphism defined on  $\mathbb{P}'$  into  $\mathbb{V}'$ .
- iii)  ${}^t\mathcal{F} = \mathcal{F}$  on  $\mathbb{P}'$ .

Thus,

$$\langle \mathcal{F}(\mathcal{L}), d \rangle = \langle \mathcal{L}, \mathcal{F}(d) \rangle, \forall \mathcal{L} \in \mathbb{P}', \ \forall d \in \mathbb{V}.$$
 (2.2)

Since  $\{\frac{(-1)^n}{n!}\mathcal{D}^n\delta\}_n$  forms a basis of  $\mathbb{P}'$  [Maroni, 1988], that is, any element  $\mathcal{L}$  of  $\mathbb{P}'$  can be expressed as

$$\mathcal{L} = \sum_{n \ge 0} (\mathcal{L})_n \frac{(-1)^n}{n!} \mathcal{D}^n \delta, \tag{2.3}$$

it follows that

$$\mathcal{F}(\mathcal{L}) = \sum_{n \ge 0} (\mathcal{L})_n x^n. \tag{2.4}$$

V' is therefore the vector space of formal series.

**Remark 2.1** Let  $L(\mathbb{P}, \mathbb{P})$  (respectively  $L(\mathbb{P}', \mathbb{P}')$ ) be the vector space of continuous linear applications defined on  $\mathbb{P}$  into  $\mathbb{P}$  (respectively on  $\mathbb{P}'$  into  $\mathbb{P}'$ ). The transpose of any element of  $L(\mathbb{P}, \mathbb{P})$  is an element of  $L(\mathbb{P}', \mathbb{P}')$ . We shall use this process to define certain elements of  $L(\mathbb{P}', \mathbb{P}')$  basically by transposing those of  $L(\mathbb{P}, \mathbb{P})$ .

#### 2.0.2 Notations

We understand by linear functional any element  $\mathcal{L}$  of  $\mathbb{P}'$  and denote by  $(\mathcal{L}, P)$  the action of  $\mathcal{L} \in \mathbb{P}'$  on  $P \in \mathbb{P}$ . We also denote by  $\mathbb{R}$  the field of real numbers,  $\mathbb{C}$  the field of complex numbers and by  $\mathbb{N}$  the set of integers. Henceforth, we will use interchangeably  $\deg(\phi)$  and  $\deg \phi$  to denote the degree of the polynomial  $\phi$ . The operator  $\mathcal{D}$  represents the usual derivative operator  $(\mathcal{D} = \frac{d}{dx})$  while the Kronecker symbol  $\delta_{n,j}$  is defined by

$$\delta_{n,j} = \left\{ egin{array}{ll} 1 & \mbox{if} & n=j, \\ 0 & \mbox{if} & n 
eq j \end{array} 
ight. .$$

#### 2.1 Orthogonality and quasi-orthogonality

#### 2.1.1 Orthogonal polynomials

**Definition 2.1** A set of polynomials  $\{P_n\}_{n\in\mathbb{N}}$  is said to be an orthogonal polynomial sequence (OPS) associated to the linear functional  $\mathcal{L}\in\mathbb{P}'$  if

$$\begin{cases}
\deg(P_n) = n, & \forall n \in \mathbb{N}, \\
\langle \mathcal{L}, P_n P_m \rangle = 0 & \forall m, n \in \mathbb{N}, m \neq n, \\
\langle \mathcal{L}, P_n P_n \rangle \neq 0 & \forall n \in \mathbb{N}.
\end{cases}$$
(2.5)

**Definition 2.2** A polynomial P is said to be monic if its leading coefficient is equal to one  $(P = x^n + b_n x^{n-1} + \ldots)$ ; and a monic polynomial family is a one in which any element is monic.

**Definition 2.3** A linear functional  $\mathcal{L} \in \mathbb{P}'$  is said to be regular if there exists an OPS associated to  $\mathcal{L}$ .

Remark 2.2 We state the following properties.

- If L is a regular linear functional, then there exists a unique monic (OPS) associated to L.
- 2. If  $\{P_n\}_{n\in\mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}$ , then  $\{P_n\}_{n\in\mathbb{N}}$  forms a basis of  $\mathbb{P}$ .
- 3. Any polynomial family  $\{P_n\}_{n\in\mathbb{N}}$  with  $\deg(P_n)=n$   $\forall n\in\mathbb{N}$  forms a basis of  $\mathbb{P}$ .

**Remark 2.3** If  $\{P_n\}_{n\in\mathbb{N}}$  is a set of polynomials with  $\deg(P_n)=n \quad \forall n\in\mathbb{N}$  and  $\mathcal{L}$  a given linear functional then the following properties are equivalent:

i) 
$$(\mathcal{L}, P_n P_m) = 0$$
  $\forall m, n \in \mathbb{N}, n \neq m \text{ and } (\mathcal{L}, P_n P_n) \neq 0$   $\forall n \in \mathbb{N}.$ 

ii) 
$$\langle \mathcal{L}, x^m P_n \rangle = 0$$
  $\forall m, n \in \mathbb{N}, \ 0 < m < n \ and \ \langle \mathcal{L}, x^n P_n \rangle \neq 0$   $\forall n \in \mathbb{N}.$ 

The following theorem, proved in [Chihara, 1978], gives a necessary and sufficient condition for the regularity of a given linear functional.

**Theorem 2.1 (Chihara, 1978)** Let  $\mathcal{L}$  be a linear functional and  $M_n$  the moment of order n of  $\mathcal{L}$  defined by  $M_n = \langle \mathcal{L}, x^n \rangle$ .

A necessary and sufficient condition for the existence of an orthogonal polynomial sequence for  $\mathcal L$  is

$$\Delta_n \neq 0 \quad \forall n \in \mathbb{N},$$

where the determinant  $\Delta_n$  is defined by

$$\Delta_{n} = \det(M_{j+k})_{0 \le j,k \le n} = \begin{vmatrix} M_{0} & M_{1} & \dots & M_{n-1} & M_{n} \\ M_{1} & M_{2} & \dots & M_{n} & M_{n+1} \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ M_{n-1} & M_{n} & \dots & M_{2n} & M_{2n-1} \\ M_{n} & M_{n+1} & \dots & M_{n} & M_{2n} \end{vmatrix}.$$

**Definition 2.4 (Chihara, 1978)** A linear functional  $\mathcal{L}$  is called positive-definite if  $\langle \mathcal{L}, \pi(x) \rangle > 0$  for every polynomial  $\pi$  that is not identically zero and is non-negative for all real x.

**Theorem 2.2 (Chihara, 1978)** The linear functional  $\mathcal{L}$  is positive-definite if and only if its moments are all real and  $\Delta_n > 0 \quad \forall n \in \mathbb{N}$ .

The following theorem, taken from [Belmehdi, 1990a] gives in a more general situation some characterisations of a regular linear functional.

**Theorem 2.3** (Maroni, 1987, Belmehdi, 1990a) Let  $\mathcal{L}$  be any linear functional; then the following properties are equivalent:

- i) The linear functional L is regular.
- ii) There exists a polynomial sequence  $\{P_n\}_{n\in\mathbb{N}}$  (with  $\deg(P_n)=n \quad \forall n\in\mathbb{N}$ ) such that

$$\det(\langle \mathcal{L}, P_j P_k \rangle)_{0 \le j, k \le n} \ne 0 \quad \forall n \in \mathbb{N}.$$

iii) For any polynomial sequence  $\{Q_n\}_{n\in\mathbb{N}}$  (with  $\deg(Q_n)=n \quad \forall n\in\mathbb{N}$ ),

$$\det(\langle \mathcal{L}, Q_j Q_k \rangle)_{0 < j,k < n} \neq 0 \quad \forall n \in \mathbb{N}.$$

Theorem 2.4 (Szegö, 1939, Belmehdi, 1990a) Given a regular linear functional  $\mathcal{L}$ , the monic orthogonal polynomials (O.P.) associated to  $\mathcal{L}$  are given by

$$P_{n}(x) = \frac{1}{\Delta_{n-1}^{\star}} \begin{vmatrix} \langle \mathcal{L}, Q_{0}Q_{0} \rangle & \langle \mathcal{L}, Q_{0}Q_{1} \rangle & \dots & \langle \mathcal{L}, Q_{0}Q_{n-1} \rangle & \langle \mathcal{L}, Q_{0}Q_{n} \rangle \\ \langle \mathcal{L}, Q_{1}Q_{0} \rangle & \langle \mathcal{L}, Q_{1}Q_{1} \rangle & \dots & \langle \mathcal{L}, Q_{1}Q_{n-1} \rangle & \langle \mathcal{L}, Q_{1}Q_{n} \rangle \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ \langle \mathcal{L}, Q_{n-1}Q_{0} \rangle & \langle \mathcal{L}, Q_{n-1}Q_{1} \rangle & \dots & \langle \mathcal{L}, Q_{n-1}Q_{n-1} \rangle & \langle \mathcal{L}, Q_{n-1}Q_{n} \rangle \\ Q_{0} & Q_{1} & \dots & Q_{n-1} & Q_{n} \end{vmatrix}$$

$$(2.6)$$

where  $\{Q_n\}_{n\in\mathbb{N}}$  is any monic polynomial family (with  $\deg(Q_n)=n \quad \forall n\in\mathbb{N}$ ); and,

$$\Delta_n^* = \det(\langle \mathcal{L}, Q_i Q_k \rangle)_{0 \le i,k \le n}, n \ge 0,$$

with the convention  $\Delta_{-1}^* = 1$ .

#### 2.1.2 Quasi-orthogonal polynomials

The notion of quasi-orthogonal polynomials was introduced in [Riesz. 1923] and extended by Maroni and Van Rossum (for more information see [Belmehdi, 1990a]).

**Definition 2.5 (Belmehdi, 1990a)** Let  $\mathcal{L}$  be any linear functional and  $\{P_n\}_{n\in\mathbb{N}}$  a polynomial family with  $\deg(P_n)=n$   $\forall n\in\mathbb{N}$ .  $\{P_n\}_{n\in\mathbb{N}}$  is said to be quasi-orthogonal of order s with respect to  $\mathcal{L}$  if

$$\begin{cases}
\langle \mathcal{L}, P_n P_m \rangle = 0, & |n - m| > s, \\
\exists m \in \mathbb{N}, & \langle \mathcal{L}, P_m P_{m+s} \rangle \neq 0.
\end{cases}$$
(2.7)

 $\{P_n\}_{n\in\mathbb{N}}$  is said to be strictly quasi-orthogonal with respect to  $\mathcal{L}$  if

$$\begin{cases}
\langle \mathcal{L}, P_n P_m \rangle = 0. |n - m| > s, \\
\langle \mathcal{L}, P_m P_{m+s} \rangle \neq 0 \quad \forall m \in \mathbb{N}.
\end{cases}$$
(2.8)

Remark 2.4 (Belmehdi, 1990a) 1. Conditions (2.7) are equivalent to

$$\begin{cases}
\langle \mathcal{L}, x^m P_{m+t-s} \rangle = 0, & \forall m \in \mathbb{N} \quad \forall t \ge 1, \\
\exists m \in \mathbb{N}. & \langle \mathcal{L}. x^m P_{m+s} \rangle \ne 0,
\end{cases} (2.9)$$

while (2.8) is equivalent to

$$\begin{cases}
\langle \mathcal{L}, x^m P_{m+t+s} \rangle = 0, & \forall m \in \mathbb{N}, \quad \forall t \ge 1, \\
\langle \mathcal{L}, x^m P_{m+s} \rangle \ne 0, & \forall m \in \mathbb{N}.
\end{cases}$$
(2.10)

- 2. It follows from the definition 2.5 that if  $\{P_n\}_{n\in\mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}$ , then  $\{P_n\}_{n\in\mathbb{N}}$  is strictly quasi-orthogonal of class s=0 with respect to  $\mathcal{L}$  (see also [Shohat, 1937]).
- 3. Notice that quasi-orthogonality of class s=1 was investigated in [Dickinson, 1961] and that the definition 2.5 was also given in [Chihara, 1957] and [Ronveaux, 1979] but without the second condition:  $\exists m \in \mathbb{N}, \langle \mathcal{L}, P_m P_{m-s} \rangle \neq 0$ .

#### 2.1.3 Other definitions

**Definition 2.6** Given a polynomial  $f \in \mathbb{P}$  and a linear functional  $\mathcal{L} \in \mathbb{P}'$ , the product of f and  $\mathcal{L}$ ,  $f\mathcal{L}$ , is defined as

$$\begin{array}{rcl} f \, \mathcal{L} & : & \mathbb{P} \to \mathbb{C} \\ \langle f \, \mathcal{L}, P \rangle & = & \langle \mathcal{L}, f P \rangle & & \forall P \in \mathbb{P}. \end{array}$$

Given f an element of  $\mathbb{P}$ , the application  $\mathcal{L} \longrightarrow f\mathcal{L}$  belongs to  $L(\mathbb{P}', \mathbb{P}')$  and is the transpose of the following element of  $L(\mathbb{P}, \mathbb{P}): P \longrightarrow fP$ .

**Definition 2.7** Given a polynomial  $g \in \mathbb{P}$  and a linear functional  $\mathcal{L} \in \mathbb{P}'$ , the product of  $\mathcal{L}$  and g,  $\mathcal{L}g$ , is a polynomial defined as

$$\mathcal{L}g(x) = \sum_{j=0}^{n} \sum_{k=j}^{n} g_k \langle \mathcal{L}, x^{k-j} \rangle x^j.$$
 (2.11)

where

$$g(x) = \sum_{j=0}^{n} g_j x^j.$$

Given a functional  $\mathcal{L}$ , the application  $P \longrightarrow \mathcal{L}P$  belongs to  $L(\mathbb{P}, \mathbb{P})$ . By transposition, we define the product of two linear functionals  $\mathcal{L}$  and  $\mathcal{M}$  as:

**Definition 2.8** The product of two linear functionals  $\mathcal{L}$  and  $\mathcal{M}$  is defined by

$$\langle \mathcal{LM}, P \rangle = \langle \mathcal{L}, \mathcal{M}P \rangle, \ \forall P \in \mathbb{P}.$$

Definition 2.9 (Belmehdi, 1990a, Dini, 1988) The operator  $\theta_c$  is defined as

$$\theta_{c} : \mathbb{P} \to \mathbb{P}$$

$$(\theta_{c}P)(x) = \begin{cases} \frac{P(x) - P(c)}{x - c}, & x \neq c \\ P'(c), & x = c \end{cases}$$
(2.12)

where c is a complex number.

The application  $\theta_c$  belongs to  $L(\mathbb{P}, \mathbb{P})$ .

**Definition 2.10** Consider the linear functional  $\mathcal{L}$ . From the above definition and by transposition (see remark 2.1), we define the linear functional  $(x-c)^{-1}\mathcal{L}$ , as

$$(x-c)^{-1}\mathcal{L} : \mathbb{P} \to \mathbb{C}$$

$$\langle (x-c)^{-1}\mathcal{L}, P \rangle = \langle \mathcal{L}, \theta_c P \rangle \quad \forall P \in \mathbb{P},$$
(2.13)

where  $c \in \mathbb{C}$ .

Corollary 2.1 (Belmehdi, 1990a) For any complex number c, and for any linear functional  $\mathcal{L}$  the following holds:

$$(x-c)[(x-c)^{-1}\mathcal{L}] = \mathcal{L}, (x-c)^{-1}[(x-c)\mathcal{L}] = \mathcal{L} - \langle \mathcal{L}, 1 \rangle \delta_c,$$
 (2.14)

where  $\delta_c$  is the Dirac measure at the point c.

#### 2.1.4 Dual basis

**Definition 2.11 (Maroni, 1988)** Let  $\{P_n\}_{n\in\mathbb{N}}$  be a monic polynomial family with  $\deg(P_n)=n$   $\forall n\in\mathbb{N}$ . Then  $\{P_n\}_{n\in\mathbb{N}}$  forms a basis of  $\mathbb{P}$  and therefore generates a unique basis of  $\mathbb{P}'$ , called dual basis associated to  $\{P_n\}_{n\in\mathbb{N}}$ , denoted by  $\{P_n\}_{n\in\mathbb{N}}$  and satisfying

$$\langle \mathbf{P}_n, P_m \rangle = \delta_{n,m} \quad \forall m, n \in \mathbb{N}.$$
 (2.15)

Any element  $\mathcal{L}$  of  $\mathbb{F}'$  can be expressed in this basis as (see [Roman et. al., 1978], [Maroni, 1988]):

$$\mathcal{L} = \sum_{n \ge 0} \langle \mathcal{L}, P_n \rangle \mathbf{P}_n. \tag{2.16}$$

**Proposition 2.1** Let  $\mathcal{L}$  be a regular linear functional,  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family and  $\{\mathbf{P}_n\}_{n\in\mathbb{N}}$  the dual basis associated to  $\{P_n\}_{n\in\mathbb{N}}$ . We have

$$\mathbf{P}_{n} = \frac{P_{n}}{\langle \mathcal{L}, P_{n} P_{n} \rangle} \mathcal{L} \quad \forall n \in \mathbb{N}.$$
(2.17)

*Proof:* Let us write  $P_n \mathcal{L} = \sum_j c_{n,j} \mathbf{P}_j$ . We obtain

$$c_{n,j} = \langle \mathcal{L}, P_n P_j \rangle = \langle \mathcal{L}, P_n P_n \rangle \delta_{n,j}$$

by the fact that  $\{P_n\}_{n\in\mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}$ . Thus,

$$\mathbf{P}_n = \frac{P_n}{\langle \mathcal{L}, P_n P_n \rangle} \mathcal{L}.$$

#### 2.2 Associated orthogonal polynomials

#### 2.2.1 Three-term recurrence relation

We first give the following theorems which we shall use further to define associated orthogonal polynomials. The first is taken from [Chihara, 1978] and the second from [Favard,1935] (see also [Wintner,1929],[Stone,1932],[Sherman,1933],[Shohat,1938], [Peron,1957]).

**Theorem 2.5 (Chihara,1978)** Let  $\mathcal{L}$  be a regular linear functional and  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal polynomials.  $\{P_n\}_{n\in\mathbb{N}}$  satisfy a three-term recurrence relation

$$\begin{cases}
P_{n+1}(x) = (x - \beta_n)P_n(x) - \gamma_n P_{n-1}(x), & n \ge 1, \\
P_0(x) = 1, P_1(x) = x - \beta_0,
\end{cases}$$
(2.18)

where  $\beta_n$  and  $\gamma_n$  are complex numbers with  $\gamma_n \neq 0$   $\forall n \in \mathbb{N}$ .

*Proof*: Since  $\{P_n\}_{n\in\mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}$ , it forms a basis of  $\mathbb{P}$  (see Remark 2.2). We therefore expand the polynomial  $xP_n$  on the basis  $\{P_n\}_{n\in\mathbb{N}}$  and obtain

$$xP_n = P_{n+1} + \beta_n P_n + \gamma_n P_{n-1} + \sum_{i=0}^{n-2} \eta_{n,j} P_j, \ n \ge 1,$$
 (2.19)

where  $\gamma_n$ ,  $\beta_n$  and  $\eta_{n,j}$  are complex numbers.

To compute  $\eta_{n,j}$ , we apply the linear functional  $\mathcal{L}$  to both sides of the equation obtained after multiplying the previous one by  $P_j$ ,  $j \leq n-2$  to get

$$\eta_{n,j} I_{0,j} = \langle \mathcal{L}, x P_n P_j \rangle = 0, \ j < n-1,$$

with  $I_{\mathfrak{I},n} = \langle \mathcal{L}. P_n P_n \rangle$ .

Considering the fact that  $I_{0,n} \neq 0 \quad \forall n \in \mathbb{N}$  (see (2.5)), it follows from the above equation that  $\eta_{n,j} = 0, j < n-1$ . Therefore equation (2.19) becomes

$$xP_n = P_{n+1} + \beta_n P_n + \gamma_n P_{n-1}$$
.  $n \ge 1$ .

Mimicking the approach used above to compute  $\eta_{n,j}$ , but with the previous equation, we express  $\gamma_n$  as

$$\gamma_n I_{0,n-1} = \langle \mathcal{L}, x P_{n-1} P_n \rangle = \langle \mathcal{L}, P_n P_n \rangle = I_{0,n} \neq 0, n > 1.$$

Hence  $\gamma_n \neq 0$   $n \geq 1$ .

By convention one takes  $\gamma_0 = \langle \mathcal{L}, 1 \rangle$ .

The converse of the above theorem is due to Favard (1935) (see also [Chihara,1978]) .

Theorem 2.6 (Favard's Theorem) Let  $\{\beta_n\}_{n\in\mathbb{N}}$  and  $\{\gamma_n\}_{n\in\mathbb{N}}$  be two sequences of complex numbers and let  $\{P_n\}_{n\in\mathbb{N}}$  be the family of polynomials defined by the recurrence formula

$$\left\{ \begin{array}{l} P_{n+1}(x) = (x - \beta_n) P_n(x) - \gamma_n P_{n-1}(x), & n \ge 1, \\ P_0(x) = 1, P_1(x) = x - \beta_0. \end{array} \right.$$

Then, there exists a unique linear functional L such that

$$\langle \mathcal{L}, 1 \rangle = \gamma_0 \text{ and } \langle \mathcal{L}, P_n P_m \rangle = 0 \quad \forall m, n \in \mathbb{N}, n \neq m.$$

 $\mathcal{L}$  is regular and  $\{P_n\}_{n\in\mathbb{N}}$  are the corresponding monic orthogonal polynomials if and only if

$$\gamma_n \neq 0 \quad \forall n \in \mathbb{N},$$

while L is positive-definite if and only if

$$\beta_n, \gamma_n \in \mathbb{R} \quad \forall n \in \mathbb{N}, \text{ and } \gamma_n > 0 \quad \forall n \in \mathbb{N}.$$

#### 2.2.2 The first associated orthogonal polynomials

**Definition 2.12** Given a regular linear functional  $\mathcal{L}$  and the corresponding monic orthogonal polynomials  $\{P_n\}_{n\in\mathbb{N}}$ , the first associated of the polynomial  $P_n$  is a monic polynomial of degree n, denoted by  $P_n^{(1)}$  and defined by

$$P_n^{(1)}(x) = \frac{1}{\gamma_0} \langle \mathcal{L}, \frac{P_{n+1}(x) - P_{n+1}(t)}{x - t} \rangle \quad \forall n \in \mathbb{N},$$
 (2.20)

with  $\gamma_0 = \langle \mathcal{L}, 1 \rangle$ . It is understood that the linear functional  $\mathcal{L}$  acts on the variable t.

**Lemma 2.1** The monic polynomial family  $\{P_n^{(1)}\}_{n\in\mathbb{N}}$  satisfies the three-term recurrence relation

$$\begin{cases}
P_{n+1}^{(1)}(x) = (x - \beta_{n+1})P_n^{(1)}(x) - \gamma_{n+1}P_{n-1}^{(1)}(x), & n \ge 1, \\
P_0^{(1)}(x) = 1, P_1^{(1)}(x) = x - \beta_1,
\end{cases}$$
(2.21)

where  $\beta_n$  and  $\gamma_n$  are defined in (2.18).

Proof: Using the three-term recurrence relation satisfied by  $\{P_n\}_{n\in\mathbb{N}}$  (see (2.18)) and (2.20) we obtain

$$\begin{split} P_{n+1}^{(1)}(x) &= \frac{1}{\gamma_0} \langle \mathcal{L}, \frac{P_{n+2}(x) - P_{n-2}(t)}{x - t} \rangle \\ &= \frac{1}{\gamma_0} \langle \mathcal{L}, \frac{(x - \beta_{n+1}) P_{n+1}(x) - \gamma_{n+1} P_n(x)}{x - t} \\ &= \frac{(t - \beta_{n+1}) P_{n+1}(t) - \gamma_{n+1} P_n(t)}{x - t} \rangle \\ &= (x - \beta_{n-1}) \frac{1}{\gamma_0} \langle \mathcal{L}, \frac{P_{n+1}(x) - P_{n+1}(t)}{x - t} \rangle \\ &= -\gamma_{n+1} \frac{1}{\gamma_0} \langle \mathcal{L}, \frac{P_n(x) - P_n(t)}{x - t} \rangle - \frac{1}{\gamma_0} \langle \mathcal{L}, P_{n+1}(t) \rangle \\ &= (x - \beta_{n+1}) P_n^{(1)}(x) - \gamma_{n+1} P_{n-1}^{(1)}(x) \quad \forall n \in \mathbb{N}. \end{split}$$

We deduce from Theorem 2.6 and Lemma 2.1 that there exists a unique regular linear functional  $\mathcal{L}^{(1)}$  with respect to which  $\{P_n^{(1)}\}_{n\in\mathbb{N}}$  is orthogonal with  $\langle \mathcal{L}^{(1)}, 1 \rangle = \gamma_1$ .

Iterating the above process, we define the general associated orthogonal polynomials.

#### 2.2.3 The rth associated orthogonal polynomials

**Definition 2.13** Let  $\mathcal{L}$  be a regular linear functional and  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal polynomials satisfying (2.18).

The rth associated of the orthogonal polynomial  $P_n$  is a polynomial of degree n, denoted  $P_n^{(r)}$  and defined by

$$\gamma_{r-1} P_n^{(r)}(x) = \langle \mathcal{L}^{(r-1)}, \frac{P_{n+1}^{(r-1)}(x) - P_{n+1}^{(r-1)}(t)}{x - t} \rangle, \quad n \ge 0, \quad r \ge 1,$$
(2.22)

with

$$\langle \mathcal{L}^{(r)}, 1 \rangle = \gamma_r, \ r > 1.$$

assuming that  $\gamma_0 \equiv \langle \mathcal{L}, 1 \rangle$ ,  $P_n^{(0)} \equiv P_n$ , and  $\mathcal{L}^{(0)} \equiv \mathcal{L}$ ; where  $\mathcal{L}^{(r-1)}$  is the regular linear functional with respect to which  $\{P_n^{(r-1)}\}_{n\in\mathbb{N}}$  is orthogonal; and it is understood that  $\mathcal{L}^{(r-1)}$  acts on the variable t.

7

**Lemma 2.2** If  $\mathcal{L}$  is a regular linear functional and  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal polynomials, then, the rth associated polynomials  $\{P_n^{(r)}\}_{n\in\mathbb{N}}$  of  $\{P_n\}_{n\in\mathbb{N}}$  satisfy the three-term recurrence relation

$$\begin{cases}
P_{n+1}^{(r)}(x) = (x - \beta_{n+r})P_n^{(r)}(x) - \gamma_{n+r}P_{n-1}^{(r)}(x), & n \ge 1, \\
P_0^{(r)}(x) = 1, P_1^{(r)}(x) = x - \beta_r, r \ge 0.
\end{cases}$$
(2.23)

*Proof:* We shall prove the lemma by induction on r. For r = 1, (2.23) is satisfied thanks to Lemma 2.1. We suppose that (2.23) is satisfied up to a fixed integer r. Then using (2.22) we obtain

$$\begin{split} P_{n+1}^{(r+1)}(x) &= \frac{1}{\gamma_r} \langle \mathcal{L}^{(r)}, \frac{P_{n+2}^{(r)}(x) - P_{n+2}^{(r)}(t)}{x - t} \rangle \\ &= \frac{1}{\gamma_r} \langle \mathcal{L}^{(r)}, \frac{(x - \beta_{n+r+1}) P_{n+1}^{(r)}(x) - \gamma_{n+r+1} P_n^{(r)}(x)}{x - t} \\ &- \frac{(t - \beta_{n+r+1}) P_{n+1}^{(r)}(t) - \gamma_{n+r+1} P_n^{(r)}(t)}{x - t} \rangle \\ &= (x - \beta_{n+r+1}) \frac{1}{\gamma_r} \langle \mathcal{L}^{(r)}, \frac{P_{n+1}^{(r)}(x) - P_{n+1}^{(r)}(t)}{x - t} \rangle \\ &- \gamma_{n+r+1} \frac{1}{\gamma_r} \langle \mathcal{L}^{(r)}, \frac{P_n^{(r)}(x) - P_n^{(r)}(t)}{x - t} \rangle - \frac{1}{\gamma_r} \langle \mathcal{L}^{(r)}, P_{n+1}^{(r)}(t) \rangle \\ &= (x - \beta_{n+r+1}) P_n^{(r+1)}(x) - \gamma_{n+r+1} P_{n-1}^{(r+1)}(x) \quad \forall n \in \mathbb{N}. \end{split}$$

Thus  $\{P_n^{(r)}\}_{n\in\mathbb{N}}$  satisfies (2.23)  $\forall r\in\mathbb{N}$ .

As consequence of the previous lemma, we claim the following known result (see [Magnus, 1984], [Belmehdi, 1990b]).

Lemma 2.3 (Magnus, 1984, Belmehdi, 1990b) The associated polynomials  $P_n^{(r)}$  satisfy

$$P_n^{(r)} P_n^{(r+1)} - P_{n+1}^{(r)} P_{n-1}^{(r+1)} = \prod_{k=1}^n \gamma_{r-k} \equiv \pi_{n,k} \quad \forall n \in \mathbb{N}, \quad \forall r \in \mathbb{N}.$$
 (2.24)

*Proof:* In the first step we write (2.23) for  $P_{n+1}^{(r)}$  and  $P_n^{(r+1)}$ 

$$P_{n+1}^{(r)}(x) = (x - \beta_{n+r})P_n^{(r)}(x) - \gamma_{n+r}P_{n-1}^{(r)}(x), \qquad (2.25)$$

$$P_n^{(r+1)}(x) = (x - \beta_{n+r}) P_{n-1}^{(r+1)}(x) - \gamma_{n+r} P_{n-2}^{(r+1)}(x).$$
 (2.26)

In the second step we subtract the two equations obtained after multiplying (2.25) and (2.26) by  $P_{n-1}^{(r+1)}$  and  $P_n^{(r)}$ , respectively,

$$P_n^{(r)}P_n^{(r+1)} - P_{n+1}^{(r)}P_{n-1}^{(r+1)} = \gamma_{n+r}(P_{n-1}^{(r)}P_{n-1}^{(r+1)} - P_n^{(r)}P_{n-2}^{(r+1)}).$$

Then relation (2.24) results by iterating the latter.

### 2.3 Operators $\mathcal{D}, \ \mathcal{T}_{\omega}, \ D_{\omega}, \ \mathcal{G}_q \ ext{and} \ \mathcal{D}_q$

#### 2.3.1 Operator $\mathcal{D}$

The application  $P \longrightarrow \mathcal{D}P$  belongs to  $L(\mathbb{P},\mathbb{P})$ . By transposition, we define derivative of the linear functional as:

**Definition 2.14** Let  $\mathcal{L}$  be a given linear functional, we define the  $\mathcal{D}$ -derivative of  $\mathcal{L}$ ,  $\mathcal{DL}$ , as

$$\mathcal{DL} : \mathbb{P} \to \mathbb{C}$$

$$\langle \mathcal{DL}, P \rangle = -\langle \mathcal{L}, \mathcal{D}P \rangle \quad \forall P \in \mathbb{P}.$$
(2.27)

**Proposition 2.2** Let  $\mathcal{L}$  be a regular linear functional,  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family and  $\{\mathbf{P}_n\}_{n\in\mathbb{N}}$  the dual basis associated to  $\{P_n\}_{n\in\mathbb{N}}$ . If  $\{\mathbf{Q}_{n,1}\}_{n\in\mathbb{N}}$  is the dual basis associated to the monic family  $\{Q_{n,1}\}_{n\in\mathbb{N}}$  defined by

 $Q_{n,1} = \frac{\mathcal{D}P_{n+1}}{n+1},$ 

then we have

$$\mathcal{D}\mathbf{Q}_{n,1} = -(n+1)\mathbf{P}_{n+1}.$$

*Proof:* This follows from Proposition 3.5.

**Definition 2.15** The regular linear functional  $\mathcal{L}$  and the corresponding monic orthogonal polynomials are said to be  $\mathcal{D}$ -semi-classical (or scmi-classical continuous) if there exist two polynomials  $\psi$  of degree at least one, and  $\phi$  such that

$$\mathcal{D}(\phi \mathcal{L}) = \psi \mathcal{L}. \tag{2.28}$$

Moreover, if  $\phi$  is a polynomial of degree at most two and  $\psi$  a first-degree polynomial, then, the linear functional and the corresponding orthogonal polynomials are called  $\mathcal{D}$ -classical (classical continuous). For more details about  $\mathcal{D}$ -semi-classical orthogonal polynomials can be found in [Maroni. 1985, 1987], [Marcellàn, 1988], [Belmehdi, 1990a] and references therein.

#### 2.3.2 Class of the $\mathcal{D}$ -semi-classical linear functional

Let  $\mathcal{L}$  be a  $\mathcal{D}$ -semi-classical linear functional satisfying

$$\mathcal{D}(\phi \mathcal{L}) = \psi \mathcal{L},\tag{2.29}$$

where  $\phi$  is any non-zero polynomial and  $\psi$  a polynomial of degree at least one.  $\mathcal{L}$  satisfies  $\mathcal{D}(f\phi\mathcal{L}) = (\phi\mathcal{D}f + \psi f)\mathcal{L}$ , for any polynomial f.

**Definition 2.16** We define the class  $cl(\mathcal{L})$  of the  $\mathcal{D}$ -semi-classical linear functional  $\mathcal{L}$  as

$$\operatorname{cl}(\mathcal{L}) = \min_{(f,g) \in \mathcal{R}_1} \{ \max(\operatorname{deg}(f - 2, \operatorname{deg}(g) - 1)) \},$$

where

$$\mathcal{R}_1 = \{ (f, g) \in \mathbb{P}^2 / \deg(g) \ge 1 \text{ and } \mathcal{D}(f\mathcal{L}) = g\mathcal{L} \}.$$

**Proposition 2.3 (Belmehdi, 1990a)** If  $\mathcal{L}$  is a  $\mathcal{D}$ -semi-classical linear functional satisfying (2.29), then  $\mathcal{L}$  is of class  $s = \max(\deg(\phi) - 2, \deg(\psi) - 1)$  if and only if

$$\prod_{c \in \mathbb{Z}_{\phi}} ||r_c| + |\langle \mathcal{L}, \psi_c \rangle|| \neq 0, \tag{2.30}$$

where  $Z_{\phi}$  is the set of zeros of  $\phi$ . The complex number  $r_c$  and the polynomials  $\phi_c$ ,  $\psi_c$  are defined by

$$(x-c)\phi_c = \phi, \ \psi - \phi_c = (x-c)\psi_c + r_c. \tag{2.31}$$

Proof: For a proof see Proposition 3.4.

**Remark 2.5** It follows from the definition of the class of the linear functional that the  $\mathcal{D}$ -classical linear functional is a  $\mathcal{D}$ -semi-classical linear functional of class s = 0.

Lemma 2.4 Let  $\mathcal{L}$  be a regular linear functional.

i) If there exist two polynomials  $\psi \neq 0$ , and  $\phi$  such that

$$\mathcal{D}(\phi \mathcal{L}) = \psi \mathcal{L} \tag{2.32}$$

then  $\phi$  is a non-zero polynomial.

ii) Conversely, if there exist two polynomials  $\phi \neq 0$  and  $\psi$  such that (2.32) holds, then  $\psi$  is of degree at least one.

Proof: For a proof see Lemma 3.1.

#### 2.3.3 Characterisation of $\mathcal{D}$ -classical orthogonal polynomials

The following theorem which is a corollary of the theorem 3.1 gives some characterisations of classical continuous orthogonal polynomials (see [Chihara, 1978].[Nikiforov et al., 1983], [Al-salam, 1990], [Marcellán et al., 1994], ...).

**Theorem 2.7** Let  $\mathcal{L}$  be a regular linear functional,  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family and  $Q_{n,m}$  the monic polynomial of degree n defined by

$$B_{n,m} Q_{n,m} = \mathcal{D}^m P_{n+m},$$

with

$$B_{n,m} = \frac{(n+m)!}{n!}, \ Q_{n,0} \equiv P_n.$$

The following properties are equivalent:

i) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that

$$\mathcal{D}(\phi\mathcal{L}) = \psi\mathcal{L}.$$

ii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m,

$$\mathcal{D}(\phi \mathcal{L}_m) = \psi_m \mathcal{L}.$$

$$\langle \mathcal{L}_m, Q_{I,m} Q_{n,m} \rangle = k_{\perp} \delta_{I,n}, \quad \forall j, n \in \mathbb{N}, \ (k_n \neq 0 \ \forall n \in \mathbb{N}),$$

with the linear functional  $\mathcal{L}_m$  and the polynomial  $\chi_m$  defined, recursively, by

$$\psi_{m+1} = \mathcal{D}\phi + \psi_m, \ \psi_0 \equiv \psi,$$
  
 $\mathcal{L}_{m+1} = \phi \mathcal{L}_m. \ \mathcal{L}_0 \equiv \mathcal{L}$ 

and given explicitly by

$$\psi_m(x) = mo'(x) + \psi(x), \tag{2.33}$$

$$\mathcal{L}_m = \phi^m \, \mathcal{L}. \tag{2.34}$$

iii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m, the following second-order difference equation holds:

$$\phi \mathcal{D}^2 Q_{n,m} + \psi_m \mathcal{D} Q_{n,m} + \lambda_{n,m}^* Q_{n,m} = 0 \quad \forall n \in \mathbb{N},$$

with the polynomial  $\psi_m$  given by (2.33) and the constant  $\lambda_{n,m}^*$  given by

$$\lambda_{n,m}^{*} = -n\left\{\psi_{m}' + (n-1)\frac{\phi''}{2}\right\} = -n\left\{\psi' + (2m+n-1)\frac{\phi''}{2}\right\}. \tag{2.35}$$

iv) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that, for any integer m, the following relation holds:

$$n\mathcal{D}\left[Q_{n-1,m+1}\mathcal{L}_{m+1}\right] = -\lambda_{n,m}^{\star}Q_{n,m}\mathcal{L}_{m} \quad \forall n \in \mathbb{N}, \tag{2.36}$$

with the polynomial  $\psi_m$ , the linear functional  $\mathcal{L}_m$  and the constant  $\lambda_{n,m}^*$  given, respectively, by (2.33). (2.34) and (2.35).

v) There exist a polynomial  $\phi$  of degree at most two and three constants  $c_{n,n+1}$ ,  $c_{n,n}$ ,  $c_{n,n-1}$  with  $c_{n,n-1} \neq 0$  such that

$$\phi \mathcal{D} P_n = c_{n,n+1} P_{n+1} + c_{n,n} P_n + c_{n,n-1} P_{n-1}, \ n \ge 1.$$

vi) For any non-zero integer m, there exist a sequence of complex numbers  $\{u_{n,m}\}_{n\in\mathbb{N}}$  such that

$$Q_{n,m+1} = Q_{n,m} + u_{n-1,m}Q_{n-1,m} + v_{n-2,m}Q_{n-2,m}, \quad \forall n \in \mathbb{N} - \{0,1\}.$$

Remark 2.6 Let us comment on the above properties.

For all  $m \in \mathbb{N}$ , the derivative of order m,  $\{Q_{n,m}\}_{n \in \mathbb{N}}$ , of the family  $\{P_{n+m}\}_{n \in \mathbb{N}}$  is classical and orthogonal with respect to the classical linear functional  $\mathcal{L}_m$ .

The functional version of the generalised Rodrigues formula [Nikiforov et al, 1983], [Belmehdi, 1990c]. given below, is obtained by iterating the relation (2.36):

$$Q_{n,m}\phi^{m}\mathcal{L} = \prod_{j=0}^{n-1} \frac{1}{\psi' + (j+2m+n-1)\frac{\phi''}{2}} \mathcal{D}^{n}(\phi^{n+m}\mathcal{L}).$$

#### 2.3.4 Operators $\mathcal{T}_{\omega}$ and $D_{\omega}$

**Definition 2.17** The arithmetic shift operator  $\mathcal{T}_{\omega}$  is defined by

$$\mathcal{T}_{\omega} : \mathbb{F} \longrightarrow \mathbb{P}$$

$$P \longrightarrow \mathcal{T}_{\omega}P, \ \mathcal{T}_{\omega}P(x) = P(x+\omega), \ \omega \in \mathbb{R}.$$
(2.37)

We denote  $\mathcal{T}_1 = \mathcal{T}$ .

**Definition 2.18** The difference operator  $D_{\omega}$  is defined by

$$D_{\omega}: \mathbb{P} \longrightarrow \mathbb{P}$$

$$P \longrightarrow D_{-}P, \ D_{\omega}P(x) = \frac{P(x+\omega) - P(x)}{\omega}, \ \omega \in \mathbb{R}, \ \omega \neq 0.$$
(2.38)

We denote  $D_1 = \Delta$  and  $D_{-1} = \nabla$ .  $\Delta$  and  $\nabla$  denote the forward and the backward difference operators. respectively.

The applications  $P \longrightarrow \mathcal{T}_{\omega}P$  and  $P \longrightarrow D_{\omega}P$  belong to  $L(\mathbb{P},\mathbb{P})$ . We, therefore, use their transposes to define the action of the operators  $\mathcal{T}_{\omega}$  and  $D_{\omega}$  on the linear functionals.

**Definition 2.19** The action of the arithmetic shift operator  $T_{\omega}$  on the functional  $\mathcal{L}$  is defined by

$$\langle \mathcal{T}_{\omega} \mathcal{L}. P \rangle = \langle \mathcal{L}, \mathcal{T}_{-\omega} P \rangle \quad \forall P \in \mathbb{P}.$$
 (2.39)

**Definition 2.20** Given a linear functional  $\mathcal{L}$ , we define the  $D_{\omega}$  derivative of  $\mathcal{L}$ .  $D_{\omega}\mathcal{L}$ , as

$$D_{\omega}\mathcal{L} : \mathbb{P} \to \mathbb{C}$$

$$\underline{D_{\omega}\mathcal{L}, P} = -\langle \mathcal{L}, D_{-\omega}P \rangle \quad \forall P \in \mathbb{P}.$$
(2.40)

 $\Box$ 

**Definition 2.21** The regular linear functional  $\mathcal L$  and the corresponding monic orthogonal polynomials are said to be  $D_{\omega}$ -semi-classical if there exist two polynomials  $\psi$  of degree at least one, and  $\phi$  such that

$$D_{\omega}(\phi \mathcal{L}) = \psi \mathcal{L}. \tag{2.41}$$

Moreover, if  $\phi$  is a polynomial of degree at most two and  $\psi$  a first-degree polynomial, then, the linear functional and the corresponding orthogonal polynomials are called classical discrete.

Using the above definitions, we obtain the following properties:

#### Proposition 2.4 (Salto, 1995)

$$\mathcal{T}_{-\omega}D_{\omega} = D_{\omega}\mathcal{T}_{-\omega} = D_{-\omega}, \ D_{\omega}D_{-\omega} = D_{-\omega}D_{\omega}. \tag{2.42}$$

$$\mathcal{T}_{\omega}(fg) = \mathcal{T}_{\omega}f\mathcal{T}_{\omega}g, \quad \mathcal{T}_{\omega}(f\mathcal{L}) = \mathcal{T}_{\omega}f\mathcal{T}_{\omega}\mathcal{L}, \tag{2.43}$$

$$D_{\omega}(fg) = f D_{\omega}g + \mathcal{T}_{\omega}gD_{\omega}f = \mathcal{T}_{\omega}fD_{\omega}g + g D_{\omega}f, \qquad (2.44)$$

$$D_{\omega}(f\mathcal{L}) = f D_{\omega}\mathcal{L} + D_{\omega}f \mathcal{T}_{\omega}\mathcal{L} = \mathcal{T}_{\omega}f D_{\omega}\mathcal{L} + D_{\omega}f \mathcal{L}. \tag{2.45}$$

$$\omega D_{\omega} \mathcal{L} = (\mathcal{T}_{\omega} - \mathcal{I}_{d}) \mathcal{L}. \tag{2.46}$$

$$D_{\omega}(fg\mathcal{L}) = \mathcal{T}_{\omega} f D_{\omega}(g\mathcal{L}) - \mathcal{T}_{\omega} f D_{\omega} g\mathcal{L} + D_{\omega}(fg)\mathcal{L}. \ \forall f, g \in \mathbb{F}. \ \forall \mathcal{L} \in \mathbb{F}'.$$
 (2.47)

Notice that equation (2.42) means that:

$$\mathcal{T}_{-\omega}D_{\omega}\Phi = D_{\omega}\mathcal{T}_{-\omega}\Phi = D_{-\omega}\Phi, \ D_{\omega}D_{-\omega}\Phi = D_{-\omega}D_{\omega}\Phi, \ \forall \Phi \in \mathbb{P},$$

$$\mathcal{T}_{-\omega}D_{\omega}\bar{\Phi} = D_{\omega}\mathcal{T}_{-\omega}\bar{\Phi} = D_{-\omega}\bar{\Phi}, \ D_{\omega}D_{-\omega}\bar{\Phi} = D_{-\omega}D_{\omega}\bar{\Phi}, \ \forall \bar{\Phi} \in \mathbb{P}'.$$

This follows directly from Proposition 3.1.

The following lemma proves that the arithmetic shift of the associated orthogonal polynomials (resp. regular linear functional) are the associated shifted orthogonal polynomials and shifted regular linear functional, respectively.

**Lemma 2.5** Given a regular linear furctional  $\mathcal{L}$  and  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal polynomials, the rth associated  $P_n^{(r)}$  of  $P_n$  and  $\mathcal{L}^{(r)}$  of  $\mathcal{L}$  obey

$$\left(\mathcal{T}_{\omega}P_{n}\right)^{(r)} = \mathcal{T}_{\omega}P_{r}^{(r)}, \quad \mathcal{T}_{\omega}\mathcal{L}_{\perp}^{(r)} = \mathcal{T}_{\omega}\mathcal{L}^{(r)} \quad \forall r, n \in \mathbb{N}. \tag{2.48}$$

We shall give the proof by induction on r. It follows from Lemma 3.2 that  $\{T_\omega P_n\}_{n\in\mathbb{N}}$  are the monic orthogonal polynomials asso iated to  $\mathcal{T}_{\omega}|\mathcal{L}$ .

For 
$$r = 0$$
  $(\mathcal{T}_{\omega} P_n)^{(e)} = \mathcal{T}_{\omega} P_n^{(0)} = \mathcal{T}_{\omega} P_n$  and  $(\mathcal{T}_{\omega} \mathcal{L})^{(0)} = \mathcal{T}_{\omega} \mathcal{L}^{(0)} = \mathcal{T}_{\omega} \mathcal{L}$ .

For r = 0 ( $\mathcal{T}_{\omega} P_n$ )<sup>(c)</sup> =  $\mathcal{T}_{\omega} P_n^{(0)} = \mathcal{T}_{\omega} P_n$  and ( $\mathcal{T}_{\omega} \mathcal{L}$ )<sup>(e)</sup> =  $\mathcal{T}_{\omega} \mathcal{L}^{(e)} = \mathcal{T}_{\omega} \mathcal{L}$ . Suppose that (2.48) is satisfied up to a fixed r. Then using -2.22) and the fact that  $\mathcal{L}$  acts on the variable t, we get

$$(\mathcal{T}_{\omega} P_{n})^{(r+1)}(x) = \frac{1}{\gamma_{r}} ((\mathcal{T}_{\omega} \mathcal{L})^{(r)} \cdot \frac{(\mathcal{T}_{\omega} P_{n+1})^{(r)}(x - (\mathcal{T}_{\omega} P_{n+1})^{(r)})}{x - t})$$

$$= \frac{1}{\gamma_{r}} (\mathcal{T}_{\omega} \mathcal{L}^{(r)}, \frac{\mathcal{T}_{\omega} P_{n+1}^{(r)}(x) - \mathcal{T}_{\omega} P_{n+1}^{(r)}(t)}{x - t})$$

$$= \frac{1}{\gamma_{r}} (\mathcal{T}_{\omega} \mathcal{L}^{(r)}, \mathcal{T}_{\omega} \frac{\mathcal{T}_{\omega} P_{n+1}^{(r)}(x) - P_{n+1}^{(r)}(t)}{x - (t - \omega)})$$

$$= \frac{1}{\gamma_{r}} (\mathcal{L}^{(r)} \cdot \frac{P_{n-1}^{(r)}(x + \omega - P_{n+1}^{(r)}(t))}{x + \omega - t})$$

$$= \mathcal{T}_{\omega} P_{\omega}^{(r+1)}(x).$$

TLen.

$$\left(\mathcal{T}_{r}P_{n}\right)^{r} = \mathcal{T}_{r}P_{n}^{(r)}, \quad \forall n \in \mathbb{N}, \ \forall r \in \mathbb{N}. \tag{2.49}$$

 $\Box$ 

 $\Box$ 

We use remark 2.3 to get

$$\langle (\mathcal{T}_{\omega} \mathcal{L})^{(r+1)}, (\mathcal{T}_{\omega} P_n)^{(r+1)} \rangle = 0 = \langle \mathcal{T}_{\omega} \mathcal{L}^{(r+1)}, \mathcal{T}_{\omega} P_n^{(r+1)} \rangle, \ n \ge 1, \ r \ge 0.$$
 (2.50)

For n = 0 (see definition 2.13),

$$\langle (\mathcal{T}_{\omega} \mathcal{L})^{(r+1)}, 1 \rangle = \gamma_{r+1} = \langle \mathcal{T}_{\omega} \mathcal{L}^{(r+1)}, 1 \rangle. \tag{2.51}$$

We combine (2.49), (2.50) and (2.51) to get

$$\langle \left( \mathcal{T}_{\omega} \mathcal{L} \right)^{(r+1)}, \mathcal{T}_{\omega} P_{n}^{(r+1)} \rangle = \langle \mathcal{T}_{\omega} \mathcal{L}^{(r+1)}, \mathcal{T}_{\omega} P_{n}^{(r+1)} \rangle \quad \forall n \geq 0.$$

Hence  $(\mathcal{T}_{\omega} \mathcal{L})^{(r+1)} = \mathcal{T}_{\omega} \mathcal{L}^{(r+1)}$ , thanks to the fact that  $\{\mathcal{T}_{\omega} P_n^{(r+1)}\}_{n \in \mathbb{N}}$ , which is orthogonal with respect to  $\mathcal{T}_{\omega} \mathcal{L}^{(r+1)}$ , forms a basis of  $\mathbb{P}$ .

#### 2.3.5 Class of the $D_{\omega}$ -semi-classical linear functional

Let  $\mathcal{L}$  be a  $D_{\omega}$ -semi-classical linear functional satisfying

$$D_{\omega}(\phi \mathcal{L}) = \psi \mathcal{L},\tag{2.52}$$

where  $\phi$  is any non-zero polynomial and  $\psi$  a polynomial of degree at least one.  $\mathcal{L}$  satisfies  $D_{\omega}(f\phi\mathcal{L}) = (\phi D_{\omega}f + \psi \mathcal{T}_{\omega}f)\mathcal{L}$ , for any polynomial f.

**Definition 2.22** We define the class  $cl(\mathcal{L})$  of the  $D_{\omega}$ -semi-classical linear functional  $\mathcal{L}$  as

$$\operatorname{cl}(\mathcal{L}) = \min_{(f,g) \in \mathcal{R}_2} \{ \max(\deg(f) - 2, \deg(g) - 1) \},$$

where

$$\mathcal{R}_2 = \{(f,g) \in \mathbb{P}^2 \mid \deg(g) \ge 1 \text{ and } D_{\omega}(f\mathcal{L}) = g\mathcal{L}\}.$$

The following proposition give a characterisation of the class of a  $\mathcal{D}_q$ -semi-classical linear functional.

**Proposition 2.5 (Salto, 1995)** If  $\mathcal{L}$  is a  $D_{\omega}$ -semi-classical linear functional satisfying (2.52), then  $\mathcal{L}$  is of class  $s = \max(\deg(\phi) - 2, \deg(\psi) - 1)$  if and only if

$$\prod_{c \in Z_{\phi}} (|r_{\gamma,\omega}| + |\langle \mathcal{L}, \psi_{c,\omega} \rangle|) \neq 0.$$
(2.53)

where  $Z_{\phi}$  is the set of zeros of  $\phi$ . The complex number  $r_{c,\omega}$  and the polynomials  $\phi_c$ ,  $\psi_{c,\omega}$  are defined by

$$(x-c)\phi_c = \phi, \ \psi - \phi_c = (x+\omega-c)\psi_{c,\omega} + r_{c,\omega}. \tag{2.54}$$

*Proof:* This follows from Proposition 3.4.

More details about the class of a  $D_{\omega}$ -semi-classical linear functional can be found in [Salto, 1996] and [Godoy et al., 1997b].

**Remark 2.7** From the definition of the class of the semi-classical linear functional, we deduce that the  $D_{\omega}$ -classical linear functional is a  $D_{\omega}$ -semi-classical linear functional of class s=0.

**Lemma 2.6** The linear functional  $\mathcal{L}$  is regular if and only if  $\mathcal{T}_{\omega}\mathcal{L}$  is regular.

*Proof:* For a proof see Lemma 3.2.

Lemma 2.7 Let L be a regular linear functional.

i) If there exist two polynomials  $\psi \neq 0$  and  $\phi$  such that

$$D_{\omega}(\phi \mathcal{L}) = \psi \mathcal{L},\tag{2.55}$$

then  $\phi$  is a non-zero polynomial.

ii) Conversely, if there exist two polynomials  $\phi \neq 0$  and  $\psi$  such that (2.55) holds, then  $\psi$  is of degree at least one.

*Proof:* This follows from Lemma 3.1.

**Proposition 2.6 (Salto, 1995)** Let  $\mathcal{L}$  be a regular linear functional,  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family and  $\{\mathbf{P}_n\}_{n\in\mathbb{N}}$  the dual basis associated to  $\{P_n\}_{n\in\mathbb{N}}$ . If  $\{\tilde{\mathbf{Q}}_{n,1}\}_{n\in\mathbb{N}}$  is the dual basis associated to the monic family  $\{\tilde{Q}_{n,1}\}_{n\in\mathbb{N}}$  defined by

$$\tilde{Q}_{n,1} = \frac{D_{\omega} P_{n-1}}{n+1},$$

then we have

$$D_{-\omega}\tilde{\mathbf{Q}}_{n,1} = -(n+1)\mathbf{P}_{n+1}.$$

*Proof:* For a proof see Proposition 3.5.

#### 2.3.6 Characterisation of ∆-classical orthogonal polynomials

The following theorem which is a corollary of Theorem 3.1 gives a characterisation of the orthogonal polynomials of a discrete variable [Al-salam. 1990], [Nikiforov et al., 1991], [Garcia et al., 1995], [Salto, 1995].

**Theorem 2.8** Let  $\mathcal{L}$  be a regular linear functional,  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family and  $Q_{n,m}$  the monic polynomial of degree n defined by

$$B_{n,m} Q_{n,m} = \Delta^m P_{n+m}, \qquad 2.56$$

with

$$B_{n,m} = \frac{(n+m)!}{n!}, \ Q_{n,0} \equiv P_n.$$
 (2.57)

The following properties are equivalent:

i) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that

$$\Delta(\phi \mathcal{L}) = \psi \mathcal{L}.$$

ii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m,

$$\Delta(\phi \mathcal{L}_m) = \psi_m \mathcal{L}.$$
  
$$\langle \mathcal{L}_m . Q_{j,m} Q_{n,m} \rangle = k_n \delta_{j,n}, \quad (k_n \neq 0 \ \forall n \in \mathbb{N}),$$

with the linear functional  $\mathcal{L}_m$  and the polynomial  $\psi_m$  defined, recursively, by

$$\begin{split} \psi_{m+1} &= \Delta \phi \div \mathcal{T} \psi_m, \ \psi_0 \equiv \psi, \\ \mathcal{L}_{m+1} &= \mathcal{T} (\phi \ \mathcal{L}_m). \ \mathcal{L}_0 \equiv \mathcal{L} \end{split}$$

and given explicitly by

$$\psi_m(x) = \phi(x+m) - \phi(x) + \psi(x+m), \tag{2.58}$$

$$\mathcal{L}_m = \prod_{j=1}^m \phi(x+j) \, \mathcal{T}^m \mathcal{L}. \tag{2.59}$$

iii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m, the following second-order difference equation holds:

$$\phi\,\Delta\nabla\,Q_{n,m}+\psi_m\,\Delta\,Q_{n,m}+\lambda_{n,m}^*\,Q_{n,m}=0\quad\forall n\in\mathbb{N},$$

with the polynomial  $\psi_m$  given by (2.58) and the constant  $\lambda_{n,m}$  given by

$$\lambda_{n,m}^* = -n \left\{ \psi_m' + (n-1) \frac{\phi''}{2} \right\} = -n \left\{ \psi' + (2m+n-1) \frac{\phi''}{2} \right\}. \tag{2.60}$$

iv) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that, for any integer m, the following relation holds:

$$n\nabla\left[Q_{n-1,m+1}\mathcal{L}_{m+1}\right] = -\lambda_{n,m}^*Q_{n,m}\mathcal{L}_m \quad \forall n \in \mathbb{N},\tag{2.61}$$

with the polynomial  $\psi_m$ , the linear functional  $\mathcal{L}_{r}$ , and the constant  $\lambda_{n,m}^*$  given, respectively, by (2.58), (2.59) and (2.60).

v) There exist a polynomial  $\phi$  of degree at most two and three constants  $c_{n,n+1}$ ,  $c_{n,n}$ ,  $c_{n,n-1}$  with  $c_{n,n-1} \neq 0$  such that

$$\phi \nabla P_n = c_{n,n+1} P_{n-1} + c_{n,n} P_n + c_{n,n-1} P_{n-1}, \ n > 1.$$

vi) For any non-zero integer m, there exist sequence of complex numbers  $\{u_{n,m}\}_{n\in\mathbb{N}}$ ; such that

$$Q_{n,m-1} = Q_{n,m} + u_{n-1,m}Q_{n-1,m} + v_{n-2,m}Q_{n-2,m}, \quad \forall n \in \mathbb{N} - \{0,1\}.$$
(2.62)

**Remark 2.8** 1. For all  $m \in \mathbb{N}$ , the  $\Delta$ -derivative of order m,  $\{Q_{n,m}\}_{n \in \mathbb{N}}$ , of the family  $\{P_{n+m}\}_{n \in \mathbb{N}}$  is classical discrete and orthogonal with respect to the classical linear functional  $\mathcal{L}_m$ .

2. The analogue of the functional version of the generalised Rodrigues formula [Nikiforov et al., 1991], [Salto, 1995] given below, is obtained by iterating the relation (2.61)

$$Q_{n,m} \prod_{j=1}^m \phi(x+j) \mathcal{T}^m \mathcal{L} = \prod_{j=0}^{n-1} \frac{1}{\psi' + (2m+j+n-1)\frac{\phi''}{2}} \nabla^n (\prod_{j=1}^{n+m} \phi(x+j) \mathcal{T}^{n+m} \mathcal{L}).$$

3. If the linear functional  $\mathcal{L}$  is represented by the positive weight  $\rho$  on the interval I = [a, b[,

$$\langle \mathcal{L}. P = \sum_{x \in I} \rho(x) P(x) \quad \forall P \in \mathbb{F},$$
 (2.63)

with  $x^n \phi(x) \rho(x)|_a^b = 0$   $\forall n \in \mathbb{N}$ , then we have the equivalence

$$\Delta(\phi \mathcal{L}) = \psi \mathcal{L} \iff \Delta(\phi \rho) = \psi \rho. \tag{2.64}$$

#### 2.3.7 Operators $\mathcal{G}_q$ and $\mathcal{D}_q$

**Definition 2.23** The geometric shift operator  $G_2$  is defined by

$$\mathcal{G}_q: \mathbb{P} \longrightarrow \mathbb{P} 
P \longrightarrow \mathcal{G}_q P. \ \mathcal{G}_q P(x) = P(q x), \ q \neq 0.$$
(2.65)

**Definition 2.24 (Hahn, 1948)** The q-difference operator  $\mathcal{D}_q$ , called Hahn operator is defined by

$$\mathcal{D}_{q}: \mathbb{P} \longrightarrow \mathbb{P}$$

$$P \longrightarrow \mathcal{D}_{q}P, \ \mathcal{D}_{q}P(x) = \frac{P(qx) - P(x)}{(q-1)x}, \ q \in \mathbb{R}, \ q \neq 0, \ q \neq 1.$$

$$(2.66)$$

The applications  $P \longrightarrow \mathcal{G}_q P$  and  $P \longrightarrow \mathcal{D}_q P$  belong to  $L(\mathbb{P}, \mathbb{P})$ . We, therefore, use their transposes to define the action of the operators  $\mathcal{G}_q$  and  $\mathcal{D}_q$  on the linear functionals.

**Definition 2.25** The action of the geometric shift operator  $\mathcal{G}_q$  on the functional  $\mathcal{L}$  is defined by

$$\langle \mathcal{G}_q \mathcal{L}, P \rangle = \frac{1}{q} \langle \mathcal{L}, \mathcal{G}_{\frac{1}{q}} P \rangle \quad \forall P \in \mathbb{P}.$$
 (2.67)

**Definition 2.26** Given a linear functional  $\mathcal{L}$ , we define the  $\mathcal{D}_q$ -derivative of  $\mathcal{L}$ .  $\mathcal{D}_q\mathcal{L}$ , as

$$\mathcal{D}_{q}\mathcal{L} : \mathbb{F} \to \mathbb{C}$$

$$\langle \mathcal{D}_{q}\mathcal{L}, P \rangle = -\frac{1}{q} \langle \mathcal{L}, \mathcal{D}_{\frac{1}{q}}P \rangle \quad \forall P \in \mathbb{P}.$$
(2.68)

**Definition 2.27** Given a real number  $q \neq 1$  and an integer n, we define the real number  $[n]_q$  by

$$[n]_q = \frac{q^n - 1}{q - 1}, \ q \neq 1, \ n \ge 0.$$
 (2.69)

**Definition 2.28** The regular linear functional  $\mathcal{L}$  and the corresponding monic orthogonal polynomials are said to be  $\mathcal{D}_q$ -semi-classical if there exist two polynomials  $\psi$  of degree at least one, and  $\phi$  such that

$$\mathcal{D}_q(o\mathcal{L}) = \psi \mathcal{L}. \tag{2.70}$$

Moreover, if  $\phi$  is a polynomial of degree at most two and  $\psi$  a first-degree polynomial, then the linear functional and the corresponding orthogonal polynomials are called  $\mathcal{D}_q$ -classical or q-classical.

From the above definitions, we state the following corollary of Proposition 3.1 [Medem. 1996].

#### Proposition 2.7 (Medem, 1996)

$$\mathcal{G}_q \mathcal{D}_{\frac{1}{q}} = \mathcal{D}_q, \quad \mathcal{D}_q \mathcal{G}_t = t \mathcal{G}_t \mathcal{D}_q, \quad \mathcal{D}_{\frac{1}{2}} \mathcal{D}_q = q \mathcal{D}_q \mathcal{D}_{\frac{1}{2}}, \tag{2.71}$$

$$\mathcal{G}_{a}(f|q) = \mathcal{G}_{a}f\mathcal{G}_{a}q, \quad \mathcal{G}_{a}(f|\mathcal{L}) = \mathcal{G}_{a}f\mathcal{G}_{a}\mathcal{L}, \tag{2.72}$$

$$\mathcal{D}_q(fg) = f \mathcal{D}_q g + \mathcal{G}_q g \mathcal{D}_q f = \mathcal{G}_q f \mathcal{D}_q g + g \mathcal{D}_q f, \tag{2.73}$$

$$\mathcal{D}_q(f\mathcal{L}) = f\mathcal{D}_q\mathcal{L} + \mathcal{D}_qf\mathcal{G}_q\mathcal{L} = \mathcal{G}_qf\mathcal{D}_q\mathcal{L} + \mathcal{D}_qf\mathcal{L}, \tag{2.74}$$

$$(q-1)\mathcal{D}_q\mathcal{L} = x^{-1}(\mathcal{G}_q\mathcal{L} - \mathcal{L}). \tag{2.75}$$

$$\mathcal{D}_q(fg\mathcal{L}) = \mathcal{G}_q f \mathcal{D}_q(g\mathcal{L}) - \mathcal{G}_q f \mathcal{D}_q g \mathcal{L} + \mathcal{D}_q(fg) \mathcal{L}, \forall f, g \in \mathbb{F}, \forall \mathcal{L} \in \mathbb{P}'.$$
 (2.76)

Notice that the identities defined in (2.71) are valid when the operators  $\mathcal{G}_q$  and  $\mathcal{D}_q$  act on  $\mathbb{P}$  and also on  $\mathbb{P}^t$ 

#### 2.3.8 Class of the $\mathcal{D}_q$ -semi-classical linear functional

Let  $\mathcal{L}$  be a  $\mathcal{D}_q$ -semi-classical linear functional satisfying

$$\mathcal{D}_{q} \phi \mathcal{L}_{z} = i \mathcal{L}, \tag{2.77}$$

where  $\phi$  is any non-zero polynomial and  $\psi$  a polynomial of degree at least one.  $\mathcal{L}$  satisfies  $\mathcal{D}_q(f\phi\mathcal{L}) = (\phi\mathcal{D}_q f + \psi\mathcal{G}_q f)\mathcal{L}$ , for any polynomial f. We, therefore, define the class of the  $\mathcal{D}_q$ -semi-classical linear functional  $\mathcal{L}$  as:

**Definition 2.29** We define the class  $cl(\mathcal{L})$  of the  $\mathcal{D}_q$ -semi-classical linear functional  $\mathcal{L}$  as

$$\operatorname{cl}(\mathcal{L}) = \min_{(f|g) \in \mathcal{R}_3} \{ \max(\operatorname{deg}(f) - 2, \operatorname{deg}(g) - 1) \}.$$

where

$$\mathcal{R}_3 = \{ (f, g) \in \mathbb{P}^2 / \deg^r g \} \ge 1, \ \mathcal{D}_q(f\mathcal{L}) = g\mathcal{L} \}.$$

**Proposition 2.8 (Medem, 1996)** If  $\mathcal{L}$  is a  $\mathcal{D}_q$ -semi-classical linear functional satisfying (2.77), then  $\mathcal{L}$  is of class  $s = \max(\deg(\phi) - 2, \deg(\psi) - 1)$  if and only if

$$\prod_{c \in \mathcal{Z}_{\phi}} (|r_{z,q}| + |\langle \mathcal{L}, \psi_{c,q} \rangle|) \neq 0, \tag{2.78}$$

where  $Z_{\phi}$  is the set of zeros of  $\phi$ . The complex number  $r_{c,q}$  and the polynomials  $\phi_c$ ,  $\psi_{c,q}$  are defined by

$$(x-c)\phi_c = \phi$$
.  $\psi - \phi_c = (qx-c)\psi_{c,q} + r_{c,q}$ . (2.79)

*Proof:* For a proof see Proposition 3.4.

**Remark 2.9** It follows from the definition of the class of the linear functional that the  $\mathcal{D}_q$ -classical linear functional is a  $\mathcal{D}_q$ -semi-classical linear functional of class s=0.

**Lemma 2.8** The linear functional  $\mathcal{L}$  is regular if and only if  $\mathcal{G}_q\mathcal{L}$  (with  $q \neq 0$ ) is regular.

Proof: This follows from Lemma 3.2.

**Lemma 2.9** Let  $\mathcal{L}$  be a regular linear functional. we have:

i) If there exist two polynomials  $\psi \neq 0$ , and  $\phi$  such that

$$\mathcal{D}_q(\phi\mathcal{L}) = \psi\mathcal{L},\tag{2.80}$$

then  $\phi$  is a non-zero polynomial.

ii) Conversely, if there exist two polynomials  $\phi \neq 0$  and  $\psi$  such that (2.80) holds, then  $\psi$  is of degree at least one.

*Proof:* For a proof see Lemma 3.1.

**Proposition 2.9 (Medem, 1996)** Let  $\mathcal{L}$  be a regular linear functional,  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family and  $\{P_n\}_{n\in\mathbb{N}}$  the dual basis associated to  $\{P_n\}_{n\in\mathbb{N}}$ .

If  $\{\bar{\mathbf{Q}}_{n,1}\}_{n\in\mathbb{N}}$  is the dual basis associated to the monic family  $\{\bar{Q}_{n,1}\}_{n\in\mathbb{N}}$  defined by

$$\bar{Q}_{n,1} = \frac{\mathcal{D}_q P_{n+1}}{[n+1]_q}.$$

then we have

$$\mathcal{D}_{\frac{1}{2}}\bar{\mathbf{Q}}_{r,1} = -q[n+1]_q\mathbf{P}_{n+1}.$$

Proof: This follows from Proposition 3.5.

#### 2.3.9 Characterisation of $\mathcal{D}_q$ -classical orthogonal polynomials

We give some characterisations for  $\mathcal{D}_q$ -classical orthogonal polynomials. The following theorem is a corollary of Theorem 3.1 [Medem, 1996].

**Theorem 2.9** Let  $\mathcal{L}$  be a regular linear functional.  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family, and  $Q_{n,m}$  the monic polynomial of degree n defined by

$$B_{n,m} \neq Q_{n,m} = \mathcal{D}_a^m P_{n+m}$$
.

with

$$B_{n,m}(q) = \prod_{i=0}^{r_i-1} [n+m-j]_q, \ Q_{n,0} \equiv P_n.$$

The following properties are equivalent:

i) There exist two polynomials.  $\phi$  of degree at most two and  $\psi$  of degree one, such that

$$\mathcal{D}_{a}(\phi\mathcal{L})=\psi\mathcal{L}.$$

ii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m,

$$\mathcal{D}_{q}(\phi \mathcal{L}_{m}) = \psi_{m} \mathcal{L},$$

$$\langle \mathcal{L}_{m}, Q_{i,m} Q_{n,m} \rangle = k_{n} \delta_{i,n}, \ \forall j, n \in \mathbb{N}, \ \forall k_{n} \neq 0 \ \forall n \in \mathbb{N}),$$

with the linear functional  $\mathcal{L}_m$  and the polynomial  $\psi_m$  defined, recursively, by

$$\psi_{m+1} = \mathcal{D}_q \phi + q \mathcal{G}_q \psi_m, \quad \psi_0 \equiv \psi, 
\mathcal{L}_{m+1} = \mathcal{G}_q (\phi \mathcal{L}_m), \quad \mathcal{L}_0 \equiv \mathcal{L}$$

and given explicitly as

$$\psi_{m}(x) = \frac{\phi(q^{m}x) - \phi(x)}{(q - 1/x)} - q^{m} \psi(q^{m}x). \tag{2.81}$$

$$\mathcal{L}_m = \prod_{j=1}^m \phi(q^j x) \, \mathcal{G}_q^m \mathcal{L}. \tag{2.82}$$

iii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m, the following second-order q-difference equation holds:

$$\phi \, \mathcal{D}_q \mathcal{D}_{\frac{1}{q}} \, Q_{n,m} + \psi_m \, \mathcal{D}_q \, Q_{n,m} + \lambda_{n,m}^{**} \, Q_{n,m} = 0 \quad \forall n \in \mathbb{N},$$

with the polynomial  $\iota_m$  given by (2.81) and the constant  $\lambda_{n,m}^{**}$  given by

$$\lambda_{n,m}^{\bullet\bullet} = -[n]_q \{ \mathcal{D}_q \psi_m + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \} = -[n]_q q^{2m} \{ \mathcal{J}' + [2m+n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \}.$$
 (2.83)

iv) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that, for any integer m, the following relation holds:

$$[n]_q \mathcal{D}_{\frac{1}{q}} \left[ Q_{n-1,m+1} \mathcal{L}_{m+1} \right] = -q \, \lambda_{n,r}^{\bullet \bullet} \left[ Q_{n-1} \mathcal{L}_n \mid \forall n \in \mathbb{N}. \right]$$

$$(2.84)$$

with the polynomial  $\mathbb{C}_m$ , the linear functional  $\mathbb{L}_m$ , and the constant  $\lambda_{n,m}$  given respectively by (2.81). (2.82) and (2.83).

v) There exist a polynomial  $\phi$  of degree at most two and three constants  $c_{n,n+1}$ ,  $c_{n,n}$ ,  $c_{n,n-1}$  with  $c_{n,n-1} \neq 0$  such that

$$\phi \mathcal{D}_{\frac{1}{q}} P_n = c_{n,n+1} P_{n-1} + c_{n,n} P_n + c_{r_n,n-1} P_{n-1}.$$

vi) For any non-zero integer in, there exist a sequence of complex numbers  $\{u_{n,m}\}_{n\in\mathbb{N}}$  such that

$$Q_{n,m,-1} = Q_{n,m} + u_{n-1,m}Q_{n-1,n} + \iota_{n-2,m}Q_{n-2,m}, \quad \forall n \in \mathbb{N} - \{0,1\}.$$

- **Remark 2.10** 1. For all  $m \in \mathbb{N}$ , the  $\mathcal{D}_q$ -derivative of order m,  $\{Q_{n,m}\}_{n \in \mathbb{N}}$ , of the family  $\{P_{n+m}\}_{n \in \mathbb{N}}$  is q-classical and orthogonal with respect to the q-classical linear functional  $\mathcal{L}_m$ .
  - 2. The q-analogue of the functional version of the generalised Rodrigues formula [Medern, 1996] given below, is obtained by iterating the relation (2.8.):

$$Q_{n,m}\mathcal{L}_m = (-1)^{r}q^{-1} \prod_{j=0}^{n-1} \frac{[n-j]_q}{\lambda_{i,-j,n+j}^{*}} D_{\frac{1}{q}}^{n} \mathcal{L}_{n+m}.$$

#### 2.4 The q-integration

In this section, exploiting the thesis of Medem [Medem, 1996], we recall the definition of the concept of the q-integration with the assumption 0 < q < 1 and give some properties. More details can be found in [Jackson, 1919] and [Gasper et al., 1990] and [Medem, 1996].

#### 2.4.1 The q-integration on the interval [0, a], a > 0

Let f be a real function defined on the interval [0,a] and  $\mathcal{P}_q([0,a])$  the "q-partition" of the interval [0,a] defined by

$$\mathcal{P}_q([0,a]) = \{ \dots aq^{n+1} < aq^n < \dots < aq < a \}.$$

For any integer N, consider the "Riemann sum"

$$A_N(f) = \sum_{n=0}^{N} (aq^n - aq^{n+1})f(aq^n) = a(1-q)\sum_{n=0}^{N} q^n f(aq^n).$$

If the limit of  $A_N(f)$  when  $N \to \infty$  is finite, then f is said to be q-integrable and the q-integral of f on the interval [0,a], denoted  $\int_0^a f(s)d_qs$ , is given by

$$\int_0^a f(s)d_q s = \lim_{N \to \infty} A_N(f) = a(1-q) \sum_{n=0}^{\infty} q^n f(aq^n).$$
 (2.85)

#### 2.4.2 The q-integration on the interval [a, 0], a < 0

Let f be a real function defined on the interval [0,a] and  $\mathcal{P}_q([a,0])$  the "q-partition" of the interval [a,0] defined by

$$\mathcal{P}_q([a,0]) = \{ a < aq < \dots aq^n < aq^{n+1} < \dots \} = \{ aq^n, n \in \mathbb{N} \}.$$

For any integer N, consider the "Riemann sum"

$$A_N(f) = \sum_{n=0}^{N} (aq^{n+1} - aq^n) f(aq^n) = -a(1-q) \sum_{n=0}^{N} q^n f(aq^n).$$

If the limit of  $A_N(f)$  when  $N \to \infty$  is finite, then f is said to be q-integrable and the q-integral of f on the interval [a,0], denoted  $\int_a^0 f(s)d_qs$ , is given by

$$\int_{a}^{0} f(s)d_{q}s = \lim_{N \to \infty} A_{N}(f) = -a(1-q)\sum_{n=0}^{\infty} q^{n} f(aq^{n}).$$
 (2.86)

#### 2.4.3 The q-integration on the interval $[a, \infty[$ , a > 0]

Let f be a real function defined on the interval  $[a, \infty[$  and  $\mathcal{P}_q([a, \infty[$ . the "q-partition" of the interval  $[a, \infty[$  defined by

$$\mathcal{P}_{q}([a,\infty]) = \{a < aq^{-1} < \dots aq^{-n} < aq^{-n-1} < \dots\} = \{aq^{-n}, n \in \mathbb{N}\}.$$

For any integer N, consider the "Riemann sum"

$$A_N(f) = \sum_{n=0}^{N} (aq^{-n-1} - aq^{-n})f(aq^{-n-1}) = a(\frac{1}{q} - 1)\sum_{n=0}^{N} q^{-n}f(aq^{-n-1}).$$

If the limit of  $A_N(f)$  when  $N \to \infty$  is finite, then f is said to be q-integrable and the q-integral of f on the interval  $[a, \infty[$ , denoted  $\int_a^\infty f(s)d_qs$ , is given by

$$\int_{a}^{\infty} f(s)d_{q}s = \lim_{N \to \infty} A_{N}(f) = a(\frac{1}{q} - 1) \sum_{n=0}^{\infty} q^{-n} f(aq^{-n-1})$$
(2.87)

#### **2.4.4** The *q*-integration the interval $]-\infty,a],\ a<0$

Let f be a real function defined on the interval  $]-\infty,a]$  and  $\mathcal{P}_q(]-\infty,a]$ ) the "q-partition" of the interval  $]-\infty,a]$  defined by

$$\mathcal{P}_q([-\infty, a]) = \{a > aq^{-1} > \dots > aq^{-n-1} > \dots\} = \{aq^{-n}, n \in \mathbb{N}\}.$$

For any integer N, consider the "Riemann sum"

$$A_N(f) = \sum_{n=0}^{N} (aq^{-n} - aq^{-n-1})f(aq^{-n-1}) = -a(\frac{1}{q} - 1)\sum_{n=0}^{N} q^{-n}f(aq^{-n-1}).$$

If the limit of  $A_N(f)$  when  $N\to\infty$  is finite, then f is said to be q-integrable and the q-integral of f on the interval  $]-\infty,a]$ , denoted  $\int_{-\infty}^a f(s)d_qs$ , is given by

$$\int_{-\infty}^{a} f(s)d_{q}s = \lim_{N \to \infty} A_{N}(f) = -a(\frac{1}{q} - 1)\sum_{n=0}^{\infty} q^{-n}f(aq^{-n-1})$$
(2.88)

**Remark 2.11** The q-integration is extended to the whole real line by using relations (2.85)-(2.88) and the following rules

$$\int_{a}^{b} f(s)d_{q}s = \int_{a}^{0} f(s)d_{q}s + \int_{0}^{b} f(s)d_{q}s \quad \forall a, b \in \mathbb{R},$$

$$\int_{a}^{\infty} f(s)d_{q}s = \int_{a}^{b} f(s)d_{q}s + \int_{b}^{\infty} f(s)d_{q}s \quad \forall a, b \in \mathbb{R}, a < 0, b > 0$$

$$\int_{-\infty}^{b} f(s)d_{q}s = \int_{-\infty}^{a} f(s)d_{q}s + \int_{a}^{b} f(s)d_{q}s \quad \forall a, b \in \mathbb{R}. a < 0, b > 0$$

$$\int_{-\infty}^{\infty} f(s)d_{q}s = \int_{-\infty}^{a} f(s)d_{q}s + \int_{a}^{b} f(s)d_{q}s \quad \forall a, b \in \mathbb{R}. a < 0, b > 0$$

$$\int_{-\infty}^{\infty} f(s)d_{q}s = \int_{-\infty}^{a} f(s)d_{q}s + \int_{a}^{b} f(s)d_{q}s + \int_{b}^{\infty} f(s)d_{q}s \quad \forall a, b \in \mathbb{R}.$$
(2.89)

As the usual integration, the q-integration enjoys some properties. Here, we give some, which are proved using the definition of the concept of the q-integration.

**Lemma 2.10** 1. If f is a real function continuous at  $\theta$ , then we have

$$\int_0^a \mathcal{D}_q f(s) d_q s = f(a) - f(0).$$

2. For any function f integrable on [0,a], we have

$$\mathcal{D}_q \int_0^a f(s) d_q s = f(a),$$

assuming that the operator  $\mathcal{T}_q$  acts on the variable a.

3. If f is a real function continuous on the interval [0, a], then f is q-integrable on [0, a] and obeys

$$\lim_{q \to 1} \int_0^a f(s) d_q s = \int_0^a f(s) ds.$$

4. If f and g are two real functions, q-integrable on the interval [0,a], then we have

$$\int_{0}^{a} f(s) \mathcal{D}_{\beta} g(s) d_{q} s = f g_{0}^{a} - \int_{0}^{a} \mathcal{D}_{q} f(s) g(q s) d_{q} s = f(s/q) |g(s)|_{0}^{a} - \frac{1}{q} \int_{0}^{a} g(s) \mathcal{D}_{\frac{1}{q}} f(s) d_{q} s,$$
 with  $f g|_{0}^{a} = f(a)g(a) - f(0)g(b)$ .

Remark 2.12 The previous lemma can be extended to the whole real line by using (2.89).

## Chapter 3

## The $D_{q,\omega}$ -semi-classical orthogonal polynomials

#### 3.1 Introduction

We define the operators  $A_{q,\omega}$  and  $D_{q,\omega}$ . The first generalises the operators  $\mathcal{T}_{\omega}$  and  $\mathcal{G}_q$  and the second generalises the operators  $\mathcal{D}$ ,  $D_{\omega}$  and  $\mathcal{D}_q$ . We give some definitions related to these operators and then give the characterisation theorems for  $D_{q,\omega}$ -semi-classical orthogonal polynomials: and deduce by limit processes the characterisation theorems for  $\mathcal{D}$ ,  $D_{\omega}$  and  $\mathcal{D}_q$ -semi-classical orthogonal polynomials.

#### 3.1.1 Operators $A_{q,\omega}$ and $D_{q,\omega}$

**Definition 3.1** We combine the operators  $\mathcal{T}_{\omega}$  and  $\mathcal{G}_q$  to obtain a new operator denoted by  $A_{q,\omega}$  and defined by

$$A_{q,\omega}: \mathbb{P} \longrightarrow \mathbb{P}$$

$$P \longrightarrow A_{q,\omega}P, \ A_{q,\omega}P(x) = \mathcal{G}_q\mathcal{T}_\omega P(x) = P(qx+\omega), \ q \neq 0.$$
(3.1)

We denote

$$A_{q,\omega}^* = A_{\frac{1}{q}, -\frac{\omega}{q}}.\tag{3.2}$$

**Definition 3.2 (Hahn, 1948)** The difference operator  $D_{q,\omega}$  is defined by

$$D_{q,\omega}: \mathbb{P} \longrightarrow \mathbb{P}$$

$$P \longrightarrow D_{q,\omega}P, \ D_{q,\omega}P(x) = \frac{P(q\,x+\omega) - P(x)}{(q-1)x+\omega}, \ \omega \in \mathbb{R}, \ q \in \mathbb{R}, \ q \neq 0.$$
(3.3)

We denote

$$D_{q,\omega}^{\bullet} = D_{\frac{1}{q}, -\frac{\omega}{q}} \tag{3.4}$$

The applications  $P \longrightarrow A_{q,\omega}P$  and  $P \longrightarrow D_{q,\omega}P$  belong to  $L(\mathbb{P},\mathbb{P})$ . We, therefore, use their transposes to define the action of the operators  $A_{q,\omega}$  and  $D_{q,\omega}$  on the linear functionals.

**Definition 3.3** We define the action of the operator  $A_{q,\omega}$  on the functional  $\mathcal{L}$  as

$$\langle A_{q,\omega}\mathcal{L}, P \rangle = \frac{1}{q} \langle \mathcal{L} : A_{q,\omega}^* P \rangle \quad \forall P \in \mathbb{P}, \ (q \neq 0).$$
 (3.5)

**Definition 3.4** We define the  $D_{q,\omega}$ -derivative of a given linear functional  $\mathcal{L}$ ,  $D_{q,\omega}\mathcal{L}$ , as

$$D_{q,\omega}\mathcal{L} : \mathbb{P} \to \mathbb{C}$$

$$\langle D_{q,\omega}\mathcal{L}, P \rangle = -\frac{1}{q} \langle \mathcal{L}, D_{q,\omega}^{\star} P \rangle \quad \forall P \in \mathbb{P}, (q \neq 0).$$
(3.6)

**Definition 3.5** The regular linear functional  $\mathcal{L}$  and the corresponding monic orthogonal polynomials are said to be  $D_{a,\omega}$ -semi-classical if there exist two polynomials  $\psi$  of degree at least one, and  $\phi$  such that

$$D_{\gamma,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}. \tag{3.7}$$

Moreover, if  $\phi$  is a polynomial of degree at most two and  $\psi$  a first-degree polynomial, then the linear functional and the corresponding orthogonal polynomials are called  $D_{q,\omega}$ -classical.

**Remark 3.1** The operators  $D_{\omega}$  and  $D_q$  generalise the operator  $\frac{d}{dx}$  in the following way:

$$\lim_{\omega \to 0} D_{\omega} = \frac{d}{dx}. \lim_{q \to 1} \mathcal{D}_q = \frac{d}{dx},$$

while the operators  $\frac{d}{dx}$ ,  $D_{\omega}$  and  $D_{T}$  can be obtained from the operator  $D_{q,\omega}$  by the following limit processes:

$$\lim_{\omega \to 0} D_{q,\omega} = \mathcal{D}_q, \ \lim_{q \to 1} D_{q,\omega} = D_{\omega}, \ \lim_{q \to 1, \omega \to 0} D_{q,\omega} = \frac{d}{dx}.$$

**Lemma 3.1** Let  $\mathcal{L}$  be a regular linear functional.

i) If there exist two polynomials  $v \neq 0$ , and  $\phi$  such that

$$D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L},\tag{3.8}$$

then  $\phi$  is a nan-zero polynomial.

ii) Conversely, if there exist two polynomials  $\phi \neq 0$  and  $\psi$  such that (3.8) holds, then  $\psi$  is of degree at least one.

*Proof:* We give the proof for the operator  $D_{q,\omega}$  and extend it to the operators  $\frac{d}{dx}$ ,  $D_{\omega}$  and  $\mathcal{D}_{q}$  by limit processes (see Remark 3.1).

i) Suppose that

$$\psi(x) = \sum_{j=0}^{r} \psi_j x^j,$$

with  $\psi_r \neq 0$ ; and let  $\{P_n\}_{n \in \mathbb{N}}$  be the monic polynomial family orthogonal with respect to  $\mathcal{L}$ . If  $\phi = 0$ , we apply both sides of (3.8) to the polynomial  $\frac{1}{\psi_r}P_r$  and obtain

$$0 = \langle \mathcal{L}, \frac{1}{U_r} P_r \rangle = \langle \mathcal{L}, P_r P_r \rangle.$$

Then  $(\mathcal{L}, P_r P_r) = 0$ . This is a contradiction because  $\{P_n\}_{n \in \mathbb{N}}$  is the monic (OPS) associated to  $\mathcal{L}$  (see (2.5)). Thus  $\phi$  is a non-zero polynomial.

ii) Suppose that  $\psi$  is a constant denoted  $\psi_0$  and then apply both sides of (3.8) to the polynomial  $P_0P_0$   $(P_0,x)=1$  and get

$$\begin{split} \psi_0 \left\langle \mathcal{L}, P_0 P_0 \right\rangle &= \langle (D_q \cup \phi \mathcal{L}), P_0 P_0 \rangle \\ &= -\frac{1}{a} \left\langle \phi \mathcal{L}, D_{q, \cup}^* (P_0 P_0) \right\rangle = 0. \end{split}$$

Since  $\langle \mathcal{L}, P_0 P_0 \rangle \neq 0$ , we deduce that  $\psi_0 = 0$  and it results from (3.8) that

$$D_{q,\omega}(\phi \mathcal{L})=0.$$

The previous equation is equivalent to  $\phi \mathcal{L} = 0$ . In fact,

$$\begin{split} D_{q,\omega}(\phi\,\mathcal{L}) &= 0 &\iff \langle D_{I,\omega}(\phi\,\mathcal{L}), P \rangle = 0 \quad \forall P \in \mathbb{R} \\ &\iff \\ \langle \phi\,\mathcal{L}, D_{q,\omega}^{\bullet}P \rangle &= 0 \quad \forall P \in \mathbb{R} \\ &\iff \\ \phi\,\mathcal{L} &= 0. \end{split}$$

Since  $\phi \neq 0$ , we pose

$$\phi(x) = \sum_{j=0}^{t} \phi_j x^j.$$

with  $\phi_t \neq 0$ . Then applying both sides of (3.8) to the polynomial  $\frac{1}{\phi_t} P_t$ , we obtain

$$0 = \langle \phi \mathcal{L}, \frac{1}{\phi_t} P_t \rangle = \frac{1}{\phi_t} \langle \mathcal{L}, \phi P_t \rangle = \langle \mathcal{L}, P_t P_t \rangle.$$

The previous equation gives a contradiction since  $\{P_n\}_{n\in\mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}$ . We, therefore, conclude that the polynomial  $\psi$  is of degree at least one.

**Remark 3.2** The operators  $D_{\omega}$ ,  $\mathcal{D}_q$  or  $D_{q,\omega}$  transform any polynomial  $P_n$  of degree n in a polynomial of degree n-1.

**Lemma 3.2** If  $\mathcal{L}$  is a linear functional and  $\mathcal{Y}$  one of the difference operator  $\{\mathcal{T}_{\omega}, \mathcal{G}_{q}, A_{q,\omega}\}$ ,  $q \neq 0$ , then the linear functional  $\mathcal{Y}(\mathcal{L})$  is regular if and only if  $\mathcal{L}$  is regular.

*Proof:* We prove the lemma for the operator  $A_{q,\omega}$  and extend it to the operators  $\mathcal{T}_{\omega}$  and  $\mathcal{G}_q$ . If  $\{P_n\}_{n\in\mathbb{N}}$  is the monic polynomial family orthogonal with respect to  $\mathcal{L}$ , then  $\{q^{-n}A_{q,\omega}P_n\}_{n\in\mathbb{N}}$  is the monic polynomial family orthogonal with respect to  $A_{q,\omega}\mathcal{L}$ .

In fact,

$$\langle A_{q,\omega} \mathcal{L}, A_{q,\omega} P_n A_{q,\omega} P_m \rangle = \frac{1}{q} \delta_{n,m} \quad \forall n, m \in \mathbb{N}.$$

We prove the following proposition:

**Proposition 3.1** For all  $q, \omega \in \mathbb{R}$ ,  $q \neq 0$ ,  $q \neq 1$ ; for all  $f, g \in \mathbb{P}$ ; and for all  $\mathcal{L} \in \mathbb{P}'$  the following properties hold:

$$i)A_{q,\omega}A_{q,\omega}^* = A_{q,\omega}^*A_{q,\omega} = \mathcal{I}_d, \ D_{q,\omega}A_{q,\omega} = qA_{q,\omega}D_{q,\omega}, \ D_{q,\omega}A_{q,\omega}^* = \frac{1}{q}A_{q,\omega}^*D_{q,\omega} = \frac{1}{q}D_{q,\omega}^*,$$
 (3.9)

$$(3.10) D_{q,\omega}^* D_{q,\omega} = q D_{q,\omega} D_{q,\omega}^*, \quad A_{q,\omega} A_{q,\omega}^* = A_{q,\omega}^* A_{q,\omega} = \mathcal{I}_d.$$

$$iii)A_{q,\omega}(f|g) = A_{q,\omega}fA_{q,\omega}g, \quad A_{q,\omega}(f|\mathcal{L}) = A_{q,\omega}fA_{q,\omega}\mathcal{L}. \tag{3.11}$$

$$iv)D_{q,\omega}(fg) = fD_{q,\omega}g + A_{q,\omega}gD_{q,\omega}f = A_{q,\omega}fD_{q,\omega}g + gD_{q,\omega}f, \tag{3.12}$$

$$v)D_{q,\omega}(f\mathcal{L}) = fD_{q,\omega}\mathcal{L} + D_{q,\omega}fA_{q,\omega}\mathcal{L} = A_{q,\omega}fD_{q,\omega}\mathcal{L} + D_{q,\omega}f\mathcal{L}, \tag{3.13}$$

$$vi)(q-1)D_{q,\omega}\mathcal{L} = (x - \frac{\omega}{1-q})^{-1}(A_{q,\omega}\mathcal{L} - \mathcal{L}). \tag{3.14}$$

$$D_{q,\omega}(fg\mathcal{L}) = A_{q,\omega} f D_{q,\omega}(q\mathcal{L}) - A_{q,\omega} f D_{q,\omega} q \mathcal{L} + D_{q,\omega}(fq) \mathcal{L}, \ \forall f, q \in \mathbb{P}, \ \forall \mathcal{L} \in \mathbb{P}'.$$
 (3.15)

Proof: Properties i) and ii) are obtained directly from the definition.

It should be noted that the identities in relations (3.9) and (3.10) are valid when the operators  $A_{q,\omega}$  and  $D_{q,\omega}$  act on  $\mathbb{P}$  and also on  $\mathbb{P}'$ .

For iii), use of (3.5) gives

$$\langle A_{q,\omega}(f\mathcal{L}), P \rangle = \frac{1}{q} \langle f\mathcal{L}, A_{q,\omega}^{\star} P \rangle$$

$$= \frac{1}{q} \langle \mathcal{L}, f A_{q,\omega}^{\star} P \rangle$$

$$= \frac{1}{q} \langle \mathcal{L}, A_{q,\omega}^{\star} (A_{q,\omega} f P) \rangle$$

$$= \langle A_{q,\omega} \mathcal{L}, A_{q,\omega} f P \rangle$$

$$= \langle A_{q,\omega} f A_{q,\omega} \mathcal{L}, P \rangle,$$

36

thus

$$A_{q,\omega}(f\mathcal{L}) = A_{q,\omega}fA_{q,\omega}\mathcal{L}.$$

$$D_{q,\omega}(fg) = \frac{f(qx+\omega)g(qx+\omega) - f(x)g(x)}{(q-1)x+\omega}$$

$$= f(x)\frac{g(qx+\omega) - g(x)}{(q-1)x+\omega} + g(qx+\omega)\frac{f(qx+\omega) - f(x)}{(q-1)x+\omega}$$

$$= f(x)D_{q,\omega}g(x) + g(qx+\omega)D_{q,\omega}f(x).$$

then reversing the role of f and g, we deduce that

$$D_{q,\omega}(fg) = fD_{q,\omega}g + A_{q,\omega}gD_{q,\omega}f = A_{q,\omega}fD_{q,\omega}g + gD_{q,\omega}f.$$

We now use i), ii) and iii) to prove iv).

$$\begin{split} \langle D_{q,\omega}(f\mathcal{L}), P \rangle &= -\frac{1}{q} \langle f\mathcal{L}, D_{q,\omega}^* P \rangle \\ &= -\frac{1}{q} \langle \mathcal{L}, f D_{q,\omega}^* P \rangle \\ &= -\frac{1}{q} \langle \mathcal{L}, D_{q,\omega}^* (fP) - A_{q,\omega}^* P D_{q,\omega}^* f \\ &= \langle D_{q,\omega} \mathcal{L}, fP \rangle + \frac{1}{q} \langle \mathcal{L}, D_{q,\omega}^* f A_{q,\omega}^* P \rangle \\ &= \langle f D_{q,\omega} \mathcal{L}, P \rangle + \frac{1}{q} \langle D_{q,\omega}^* f \mathcal{L}, A_{q,\omega}^* P \rangle \\ &= \langle f D_{q,\omega} \mathcal{L}, P \rangle + \langle A_{q,\omega} (D_{q,\omega}^* f \mathcal{L}), P \rangle \\ &= \langle f D_{q,\omega} \mathcal{L}, P \rangle + \langle D_{q,\omega} f A_{q,\omega} \mathcal{L}, P \rangle. \end{split}$$

Then

$$D_{q,\omega}(f\mathcal{L}) = fD_{q,\omega}\mathcal{L} + D_{q,\omega}fA_{q,\omega}\mathcal{L} = A_{q,\omega}fD_{q,\omega}\mathcal{L} + D_{q,\omega}f\mathcal{L}.$$

For  $q \neq 1$  and  $q \neq 0$ , we have

$$\langle (x - \frac{\omega}{1 - q})^{-1} (A_{q,\omega} \mathcal{L} - \mathcal{L}), P \rangle = \langle A_{q,\omega} \mathcal{L} - \mathcal{L}, \theta_{\frac{\omega}{1 - q}} P \rangle$$

$$= \langle A_{q,\omega} \mathcal{L} - \mathcal{L}, \frac{P(x) - P(\frac{\omega}{1 - q})}{x - \frac{\omega}{1 - q}} \rangle$$

$$= \frac{1}{q} \langle \mathcal{L}, \frac{P(\frac{x - \omega}{q}) - P(\frac{z - \omega}{1 - q})}{\frac{x - \omega}{q} - \frac{\omega}{1 - q}} \rangle - \langle \mathcal{L}, \frac{P(x) - P(\frac{z - \omega}{1 - q})}{x - \frac{z - q}{1 - q}} \rangle$$

$$= -(q - 1) \frac{1}{q} \langle \mathcal{L}, \frac{P(\frac{x - \omega}{q} - P(x))}{(\frac{1}{q} - 1 | x - \frac{\omega}{q})}$$

$$= -(q - 1) \frac{1}{q} \langle \mathcal{L}, D_{q,\omega}^* \mathcal{L}, P \rangle$$

$$= (q - 1) \langle D_{q,\omega} \mathcal{L}, P \rangle.$$

Then.

$$(q-1)D_{q,\omega}\mathcal{L} = (x - \frac{\omega}{1-q})^{-1}(A_{q,\omega}\mathcal{L} - \mathcal{L}).$$

The relation (3.15) follows straightforwardly from (3.12) and (3.13).

Remark 3.3 The proof of Proposition 2.4 (resp. Proposition 2.7) is obtained in the same way just by replacing g by one and  $\omega$  by zero, respectively. In particular, to derive the relation (2.46) from (3.14), we first multiply both sides of (3.14) by  $(x - \frac{\omega}{1-a})$ , then use (2.14) to get

$$((q-1)x + \omega)D_{q,\omega}\mathcal{L} = (A_{q,\omega} - \mathcal{I}_d \mathcal{L}.$$

Therefore, (2.46) yields by taking q = 1 in the previous relation.

3.1. Introduction 37

**Proposition 3.2** If  $\mathcal{L}$  is a regular linear functional, and  $q \in \mathbb{R} - \{0\}$ , then we have

$$D_{q,\omega}(\phi \mathcal{L}) = \psi \mathcal{L} \iff D_{q,\omega}^*(\tilde{\phi}\mathcal{L}) = \psi \mathcal{L}, \tag{3.16}$$

with

$$\tilde{\phi} = \frac{1}{q}(\phi + [q-1]x + \omega]\psi). \tag{3.17}$$

Proof: Let  $\phi$  and  $\psi$  be two polynomials, then using Proposition 1.3 we have

$$\begin{split} D_{q,\omega}(\phi\mathcal{L}) &= \psi\mathcal{L} \Longleftrightarrow \langle D_{q,\omega}(\phi\mathcal{L}), A_{q,\omega}P \rangle = \langle \psi\mathcal{L}, A_{q,\omega}P \rangle \quad \forall P \in \mathbb{P} \\ &\iff -\frac{1}{q} \langle \phi\mathcal{L}, D_{q,\omega}^* A_{q,\omega}P \rangle = \langle \psi\mathcal{L}, A_{q,\omega}P \rangle \quad \forall P \in \mathbb{P} \\ &\iff -\langle \phi\mathcal{L}, A_{q,\omega}D_{q,\omega}^*P \rangle = \langle \psi\mathcal{L}, A_{q,\omega}P \rangle \quad \forall P \in \mathbb{P} \\ &\iff -\langle \phi\mathcal{L}, D_{q,\omega}P \rangle = \langle \psi\mathcal{L}, [-q-1)x + \omega]D_{q,\omega}P + P \rangle \quad \forall P \in \mathbb{P} \\ &\iff -\langle (\phi + [(q-1)x + \omega]\psi \cdot \mathcal{L}, D_{q,\omega}P \rangle = \langle \psi\mathcal{L}, P \rangle \quad \forall P \in \mathbb{P} \\ &\iff \frac{1}{q} \langle D_{q,\omega}^* [(\phi + [(q-1)x - \omega]\psi)\mathcal{L}] \cdot P \rangle = \langle \psi\mathcal{L}, P \rangle \quad \forall P \in \mathbb{P} \\ &\iff D_{q,\omega}^* (\tilde{\phi}\mathcal{L}) = \psi\mathcal{L}, \end{split}$$

with  $\tilde{\phi}$  given by (3.17).

Corollary 3.1 (Salto, 1995, Medem, 1996) From the above proposition, we deduce the following:

- i)  $\mathcal{L}$  is  $D_{q,\omega}$ -semi-classical  $\iff \mathcal{L}$  is  $D_{q,\omega}^*$ -semi-classical.
- ii)  $\mathcal{L}$  is  $\mathcal{D}_q$ -semi-classical  $\iff \mathcal{L}$  is  $\mathcal{D}_{\frac{1}{q}}$ -semi-classical.

Indeed,

$$\mathcal{D}_q(\phi \mathcal{L}) = \psi \mathcal{L} \iff \mathcal{D}_{\frac{1}{q}}(\tilde{\phi}\mathcal{L}) = \psi \mathcal{L}.$$

with

$$\tilde{\phi} = \frac{1}{q}(\phi + (q-1)x\psi).$$

iii)  $\mathcal{L}$  is  $D_{\omega}$ -semi-classical  $\iff \mathcal{L}$  is  $D_{--}$ -semi-classical. Moreover.

$$D_{\omega}(\phi \mathcal{L}) = \psi \mathcal{L} \iff D_{-\omega}(\tilde{\phi}\mathcal{L}) = \psi \mathcal{L},$$

where

$$\tilde{\phi} = \phi + \omega \psi.$$

**Remark 3.4** If  $\mathcal{Y}$  represents one of the operators:  $\mathcal{T}_{\omega}$ ,  $\mathcal{G}_q$ ,  $A_{q,\omega}$ ,  $\frac{d}{dx}$ ,  $D_{\omega}$ ,  $\mathcal{D}_{z}$ , and  $D_{q,\omega}$ , we define the power of  $\mathcal{Y}$ ,  $\mathcal{Y}^m$  as

$$\mathcal{Y}^m = \mathcal{Y}\mathcal{Y}^{m,-1}, \ m \geq 1 \ with \ \mathcal{Y}^0 \equiv \mathcal{I}_d,$$

where  $\mathcal{I}_d$  is the identity operator.

**Remark 3.5** One proves easily that  $\forall P \in \mathbb{P}$  and  $\forall n \in \mathbb{N}$ 

$$A_{q,\omega}^n P(x) = P(q^n x + \omega[n]_q), \ \mathcal{G}_q^n P(x) = P(q^n x), \ \mathcal{T}_\omega^n P(x) = P(x + n\omega). \tag{3.18}$$

# 3.1.2 Class of the $D_{q,\omega}$ -semi-classical linear functional

Let  $\mathcal{L}$  be a  $D_{q,\omega}$ -semi-classical linear functional satisfying

$$D_{g,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}. \tag{3.19}$$

where  $\phi$  is a non-zero polynomial and  $\psi$  a polynomial of degree at least one.  $\mathcal{L}$  satisfies  $D_{q,\omega}(f\phi\mathcal{L}) = (\phi D_{q,\omega}f + \psi A_{q,\omega}f)\mathcal{L}$ , for any polynomial f.

**Definition 3.6** We define the class  $cl(\mathcal{L})$  of the  $D_{q,\omega}$ -semi-classical linear functional  $\mathcal{L}$  as

$$\operatorname{cl}(\mathcal{L}) = \min_{f,g) \in \mathcal{R}} \{ \max(\operatorname{deg}(f) - 2, \operatorname{deg}(g) - 1) \}.$$
(3.20)

where

$$\mathcal{R} = \{ (f, g) \in \mathbb{P}^2 / \deg(g) > 1 \text{ and } D_{g,\omega}(f\mathcal{L} = g\mathcal{L}) \}. \tag{3.21}$$

We state the following lemmas and propostion which we shall use to prove the proposition characterising the class of the semi-classical linear functional.

**Lemma 3.3** Consider  $\mathcal{L}$  a regular linear functional,  $\psi$  a non-zero polynomial and  $\phi$  a polynomial of degree at least one. Then, for any zero, c. of  $\phi$ , we have

$$D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L} \iff D_{q,\omega}(\phi_c\mathcal{L}) = \psi_{c,q,\omega}\mathcal{L} - \langle \mathcal{L}, \psi_{c,q,\omega} \rangle \, \delta_{\frac{c-\omega}{q}} + \frac{1}{q} \, r_{c,q,\omega} \, (x - \frac{c - \omega}{q})^{-1} \mathcal{L}, \tag{3.22}$$

where

$$\phi = (x - c)\phi_c, \ \psi - \phi_c = (qx + \omega - c)\psi_{c,q,\omega} + r_{c,q,\omega}. \tag{3.23}$$

*Proof:* The proof is obtained straightforwardly by using (2.14), (3.13) and (3.23).

**Lemma 3.4** Let  $\mathcal{L}$  be a regular linear functional. If there exist four polynomials  $\phi$ ,  $\psi$ ,  $\tilde{\phi}$  and  $\tilde{\psi}$ , with  $\deg(\tilde{\phi} \geq 1)$ , such that

$$D_{J,\omega}(\phi \mathcal{L}) = \psi \mathcal{L}, \ D_{g,\omega}(\phi \tilde{\phi} \mathcal{L}) = \tilde{\psi} \mathcal{L}.$$
 (3.24)

then, for any zero, c. of  $\tilde{\phi}$ ,

$$r_{c,a,\omega} = \langle \mathcal{L}, \tilde{\psi}_{c,a,\omega} \rangle = 0.$$
 (3.25)

where,

$$\tilde{\phi} = (x - e^{-\tilde{\phi}_c}, \ \tilde{\psi} - \phi \tilde{\phi}_c = (qx + \omega - e)\tilde{\psi}_{c-l,z} + r_{\gamma,q,\omega}. \tag{3.26}$$

Proof: The second relation of 3.24) thanks to Lemma 3.3 is equivalent to

$$D_{q,\omega}(\phi\tilde{\phi}_{c}\mathcal{L}) = \tilde{\psi}_{c,q,\omega}\mathcal{L} - \langle \mathcal{L}, \tilde{\psi}_{c,q,\omega} \rangle \delta_{\frac{c-\omega}{q}} + \frac{1}{q}r_{c,q,\omega} \left( x - \frac{c-\omega}{q} \right)^{-1}\mathcal{L}, \tag{3.27}$$

where  $r_{c,q,\omega}$  and  $\psi_{c,q,\omega}$  are defined by (3.26). The previous relation, used together with (3.13) and the first relation of (3.24) gives

$$(\psi A_{q,\omega}\tilde{\phi}_c + \phi D_{q,\omega}\tilde{\phi}_c - \tilde{\psi}_{c,q,\omega})\mathcal{L} = -\langle \mathcal{L}, \tilde{\psi}_{c,q,\omega} \rangle \,\delta_{\frac{c_+}{4}} + \frac{1}{q} \,\tau_{c,q,-} \left(x - \frac{c - \omega}{q}\right)^{-1} \mathcal{L}. \tag{3.28}$$

The multiplication of the latter equation by  $(x - \frac{c - \omega}{q})$ , use of 2.14 and the relation  $(x - a)\delta_a = 0$ , gives

$$(x - \frac{c - \omega}{a}) \ \psi A_{q,\omega} \tilde{\phi}_c + \phi D_{q,\omega} \tilde{\phi}_c - \tilde{\psi}_{c,q,\omega}) \mathcal{L} = \frac{1}{a} r_{c,q,\omega} \mathcal{L}.$$

It follows from the previous equation and the fact that  $\mathcal{L}$  is regular that.

$$(x - \frac{c - \omega}{q})(\psi A_{q,\omega} \tilde{\phi}_c + \phi D_{q,\omega} \tilde{\phi}_c - \tilde{\psi}_{c,q\omega}) = \frac{1}{q} r_{c,q,\omega}.$$

Thus,  $r_{c,q,\omega} = 0$  and  $\psi A_{q,\omega} \tilde{\phi}_c + \phi D_{q,\omega} \tilde{\phi}_c - \tilde{\psi}_{c,q\omega} = 0$ . We, therefore, deduce that  $\langle \mathcal{L}, \tilde{\psi}_{c,q\omega} \rangle = 0$ . The following proposition, already known for the operator  $D_{\omega}$  [Salto, 1995], is also needed to characterise the class of the  $D_{q,\omega}$ -semi-classical linear functional.

3.1. Introduction

**Proposition 3.3** Consider  $\Phi_1$ .  $\Phi_2$ ,  $\Psi_1$  and  $\Psi_2$ , four polynomials such that:  $\Phi_1 \neq 0$ ,  $\Phi_2 \neq 0$ .  $\deg(\Phi_1) \leq \deg(\Phi_2)$ ,  $\deg(\Psi_1) \geq 1$  and  $\deg(\Psi_2) \geq 1$ . Let  $\mathcal{L}$  be a regular linear functional satisfying

$$D_{q,\omega}(\Phi_1 \mathcal{L}) = \Psi_1 \mathcal{L}, \quad D_{q,\omega}(\Phi_2 \mathcal{L}) = \Psi_2 \mathcal{L}, \quad q \neq 0.$$
(3.29)

If  $\Phi$  denotes the highest common factor of  $\Phi_1$  and  $\Phi_2$ :  $\Phi = hcf(\Phi_1, \Phi_2)$ , then, there exists a polynomial  $\Psi$  such that,

$$D_{q,\omega}(\Phi \mathcal{L}) = \Psi \mathcal{L} \tag{3.30}$$

and

$$\max(\deg(\Phi) - 2, \deg(\Psi) - 1) \le \max(\deg(\Phi_j) - 2, \deg(\Psi_j) - 1), \ j = 1, 2.$$
 (3.31)

Moreover, If  $\Phi_1$  is not a divisor of  $\Phi_2$  ( $\Phi_2 \neq f \Phi_1$ ,  $\forall f \in \mathbb{P}$ ), then the previous relation becomes

$$\max(\deg(\Phi) - 2, \deg(\Psi) - 1) < \max(\deg(\Phi_j) - 2, \deg(\Psi_j) - 1), j = 1, 2.$$
 (3.32)

Proof:

We shall give the proof mimicking the approach developed in [Salto, 1996] for the operator  $D_{\omega}$ . Since  $\Phi = \text{hcf}(\Phi_1, \Phi_2)$ , there exist two polynomials  $\tilde{\Phi}_1$  and  $\tilde{\Phi}_2$  satisfying

$$\Phi_1 = \Phi \,\tilde{\Phi}_1, \; \Phi_2 = \Phi \,\tilde{\Phi}_2, \tag{3.33}$$

with  $\bar{\Phi}_1$  and  $\bar{\Phi}_2$  having no common zero.

In the first step, we combine (3.12), (3.15) and (3.29) to get

$$D_{g,\omega}(\tilde{\Phi}_2\Phi_1\mathcal{L}) = A_{g,\omega}\tilde{\Phi}_2\Psi_1\mathcal{L} - A_{g,\omega}\tilde{\Phi}_2D_{g,\omega}\Phi_1\mathcal{L} + D_{g,\omega}(\Phi_1\tilde{\Phi}_2)\mathcal{L},\tag{3.34}$$

$$D_{q,\omega}(\tilde{\Phi}_1\Phi_2\mathcal{L}) = A_{q,\omega}\tilde{\Phi}_1\Psi_2\mathcal{L} - A_{q,\omega}\tilde{\Phi}_1D_{q,\omega}\Phi_2\mathcal{L} + D_{q,\omega}(\Phi_2\tilde{\Phi}_1)\mathcal{L}. \tag{3.35}$$

In the second step, we subtract the two previous equations taking care that  $\tilde{\Phi}_1\Phi_2=\tilde{\Phi}_2\Phi_1$ , to get

$$\left[A_{q,\omega}\tilde{\Phi}_2(\Psi_1 - \Phi D_{q,\omega}\tilde{\Phi}_1) - A_{q,\omega}\tilde{\Phi}_1(\Psi_2 - \Phi D_{q,\omega}\tilde{\Phi}_2)\right]\mathcal{L} = 0.$$

Since  $\mathcal{L}$  is regular, we deduce that

$$A_{q,\perp}\tilde{\Phi}_2(\Psi_1 - \Phi D_{q,\omega}\tilde{\Phi}_1) = A_{q,\omega}\tilde{\Phi}_1(\Psi_2 - \Phi D_{q,\omega}\tilde{\Phi}_2).$$

Using the previous relation and the fact that  $\tilde{\Phi}_1$  and  $\tilde{\Phi}_2$  have no common zero, it follows that there exists a polynomial  $\Psi$  verifying

$$A_{q,\omega}\tilde{\Phi}_1\Psi = \Psi_1 - \Phi D_{q,\omega}\tilde{\Phi}_1, \ A_{q,\omega}\tilde{\Phi}_2\Psi = \Psi_2 - \Phi D_{q,\omega}\tilde{\Phi}_2. \tag{3.36}$$

Use of (3.33) and (3.36) transforms (3.29) in

$$A_{q,\omega}\tilde{\Phi}_1 D_{q,\omega}(\Phi \mathcal{L}) = A_{q,\omega}\tilde{\Phi}_1 \Psi \mathcal{L},\tag{3.37}$$

$$A_{a,\omega}\tilde{\Phi}_2 D_{a,\omega}(\Phi \mathcal{L}) = A_{a,\omega}\tilde{\Phi}_2 \Psi \mathcal{L}. \tag{3.38}$$

Since  $\tilde{\Phi}_1$  and  $\tilde{\Phi}_2$  have no common zero, there exist two polynomials (Bezout identity)  $h_1$  and  $h_2$  such that  $\tilde{\Phi}_1 h_1 + \tilde{\Phi}_2 h_2 = 1$ . In the third step, we sum the two equations obtained by multiplying (3.37) and (3.38) by  $A_{q,\omega}h_1$  and  $A_{q,\omega}h_2$ , respectively, and get

$$D_{q,\omega}(\Phi \mathcal{L}) = \Psi \mathcal{L}.$$

The latter equation, used together with Lemma 3.1 gives  $deg(\Psi) \ge 1$ . In the fourth step, we use (3.33) and (3.36) to get

$$\deg(\Phi_j) = \deg(\Phi) + \deg(\tilde{\Phi}_j), \ \deg(\Psi) + \deg(\tilde{\Phi}_j) \le \max(\deg(\Psi_j), \deg(\Phi_j) - 1), \ j = 1, 2.$$
 (3.39)

We, therefore, deduce (3.31).

If we assume that polynomials  $\Phi_1$  and  $\Phi_2$  are such that  $\Phi_1 \neq f \Phi_2 \ \forall f \in \mathbb{P}$ , then,  $\deg(\Phi) < \deg(\Phi_j)$ . j = 1, 2. We finally use (3.39) to get (3.32).

The following proposition gives a characterisation for the class of semi-classical linear functional.

**Proposition 3.4** If  $\mathcal{L}$  is a  $D_{q,\omega}$ -semi-classical linear functional satisfying (3.19), then  $\mathcal{L}$  is of class  $\operatorname{cl}(\mathcal{L}) = \max(\operatorname{deg}(\phi) - 2, \operatorname{deg}(\psi) - 1)$  if and only if

$$\prod_{c \in \mathbf{Z}_c} (|r_{c,q,\omega}| + |\langle \mathcal{L}, \psi_{c,q,\omega} \rangle|) \neq 0.$$
(3.40)

where  $Z_{\phi}$  is the set of zeros of  $\phi$ . The complex number  $r_{c,q,\omega}$  and the polynomials  $\phi_c$ ,  $\psi_{c,q,\omega}$  are defined by

$$(x-c)\phi_c = \phi, \ \psi - \phi_c = (qx + \omega - c)\psi_{c,q,\omega} + r_{\varepsilon,q,\omega}. \tag{3.41}$$

*Proof:* We first recall the definition of the class  $cl(\mathcal{L})$  of  $\mathcal{L}$  (see (3.20) and (3.21)).

$$\operatorname{cl}(\mathcal{L}) = \min_{(f,g) \in \mathcal{R}} \{ \max(\deg f) - 2, \deg(g) - 1 \},$$

where

$$\mathcal{R} = \{ (f, g) \in \mathbb{P}^2 / \deg(g) \ge 1 \text{ and } D_{q, \omega}(f\mathcal{L}) = g\mathcal{L} \}.$$

Let  $(\phi, \psi) \in \mathcal{R}$  such that there exists a zero. c, of  $\phi$  verifying  $r_{c,q,\omega} = \mathcal{L}, \psi_{\varepsilon,q,\omega} \rangle = 0$ . We shall prove that,

$$\operatorname{cl}(\mathcal{L}) < \max(\operatorname{deg}(\phi) - 2, \operatorname{deg}(\psi) - 1).$$

Equation  $D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}$ , thanks to Lemma 3.3 is equivalent to

$$D_{q,\perp}(\phi_c \mathcal{L}) = \psi_{c,q,\perp} \mathcal{L},$$

therefore,  $(\phi_c, \psi_{c,q,\omega})$  belongs to  $\mathcal{R}$  (see Lemma 3.1). Moreover, the degree of  $\phi$ ,  $\psi$ ,  $\phi_c$  and  $\psi_{c,q,\omega}$  obey

$$\max(\deg \phi_c) - 2, \deg(\psi_{c,q,\perp}) - 1) = \max(\deg(\phi) - 2, \deg(\psi) - 1) - 1.$$

Thus,

$$\operatorname{cl}(\mathcal{L}) \leq \max(\operatorname{deg}(\phi_c) - 2, \operatorname{deg}(\psi_{c,q,\omega}) - 1) < \max(\operatorname{deg}(\phi) - 2, \operatorname{deg}(\psi) - 1).$$

We conclude that for any  $(\phi, \psi) \in \mathcal{R}$  such that  $\operatorname{cl}(\mathcal{L}) = \max(\operatorname{deg}(\phi) - 2, \operatorname{deg}(\psi) - 1)$ , then, for any zero, c, of  $\phi$ .

$$|r_{c,\eta,\omega}| + |\langle \mathcal{L}, \psi_{\gamma,q,\omega}| \neq 0.$$
 (3.42)

Conversely, we shall prove that for any  $(\phi, \psi) \in \mathcal{R}$  such that (3.42) holds for any zero, c, of  $\phi$ , then  $\operatorname{cl}(\mathcal{L}) = \max(\deg(\phi) - 2, \deg(\psi - 1))$ . Let  $(\phi_m, \psi_m) \in \mathcal{R}$  such that  $\operatorname{cl}(\mathcal{L}) = \max(\deg(\phi_m) - 2, \deg(\psi_m) - 1)$ . We assume without loss of generality that  $\deg(\phi_m) \leq \deg(\phi)$ . We write

$$o = \phi_m f + R$$
,  $R, f \in \mathbb{F}$ ,  $\deg(R) < \deg(\phi_{r_1})$ .

• If  $R \neq 0$ , then, from Proposition 3.3, there exists  $(\tilde{\varphi}, \tilde{\psi}) \in \mathcal{R}$ , with  $\tilde{\phi} = \text{hcf}(\phi, \phi_m)$  such that

$$\max(\deg(\tilde{e}) - 2, \deg(\tilde{\psi}) - 1) < \max(\deg(\phi_{n_k}) - 2, \deg(\psi_{n_k} - 1)) = \operatorname{cl}(\mathcal{L}).$$

This is a contradiction because  $(\tilde{\phi}, \tilde{\psi}) \in \mathcal{R}$ . Thus, R = 0.

• If  $deg(f) \ge 1$ , then, it yields from Lemma 3.4, that for any zero,  $\epsilon$ , of f (then of  $\phi$ ).

$$|r_{\cdot,q,\omega}| + |\langle \mathcal{L}, \psi_{c,q,\omega} \rangle| = 0.$$

The previous equation contradicts (3.42).

Finally, f is a constant and we have  $\phi = f\phi_r$ .  $\theta = f\psi_m$ . Thus,

$$\operatorname{cl}(\mathcal{L}) = \max(\operatorname{deg}(\varphi) - 2, \operatorname{deg}(\psi) - 1) = \max(\operatorname{deg}(\phi_{m_{\ell}} - 2, \operatorname{deg}(\psi_{m_{\ell}} - 1)).$$

The proof of the proposition is therefore complete. It should be noted that the proof of Propositions 2.3, 2.5 and 2.8 are deduced by limit processes (see Remark 3.1).

Remark 3.6 It follows from the definition of the class of the linear functional that the  $D_{q,\omega}$ -classical linear functional is a  $D_{q-\pi}$ -semi-classical linear functional of class s=0.

**Definition 3.7** The Pectron-type difference equation (3.19) is said to be irreducible on  $c \in Z_{\phi}$  if  $|r_{c,q,\omega}| + |\langle \mathcal{L}, \psi_{c,q,\omega} \rangle| \neq 0$ . Moreover, (3.19) is said to be irreducible if it is not reducible on any  $c \in Z_{\phi}$ .

**Proposition 3.5** Let  $\mathcal{L}$  be a regular linear functional.  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family and  $\{\mathbf{P}_n\}_{n\in\mathbb{N}}$  the dual basis associated to  $\{P_n\}_{n\in\mathbb{N}}$ . If  $\{\mathbf{Q}_{n,1}\}_{n\in\mathbb{N}}$  is the dual basis associated to the monic family  $\{Q_{n,1}\}_{n\in\mathbb{N}}$  defined by

$$Q_{n,1} = \frac{D_{q,\omega} P_{n+1}}{[n-1]_q},\tag{3.43}$$

then, we have

$$D_{q,\omega}^* \mathbf{Q}_{n,1} = -q [n+1]_q \mathbf{P}_{n+1}. \tag{3.44}$$

Proof:

$$\langle D_{q,\omega}^* \mathbf{Q}_{n,1}, P_{m+1} \rangle = -q \langle \mathbf{Q}_{n,1}, D_{q,\omega} P_{m+1} \rangle$$

$$= -q [m+1]_q \langle \mathbf{Q}_{n,1}, Q_{m,1} \rangle$$

$$= -q [n+1]_q \delta_{n,m}$$

$$= -q [n+1]_q \langle \mathbf{P}_{n+1}, P_{m+1} \rangle,$$

then

$$D_{q,\omega}^*\mathbf{Q}_{n,1} = -q[n+1]_q\mathbf{P}_{n+1}.$$

# 3.2 Characterisation theorems for $D_{q,\omega}$ -semi-classical orthogonal polynomials

#### 3.2.1 $D_{q,\omega}$ -classical orthogonal polynomials

**Theorem 3.1** Let  $\mathcal{L}$  be a regular linear functional,  $\{P_{-}\}_{n\in\mathbb{N}}$  the corresponding modified orthogonal family, and  $Q_{n,m}$  the monic polynomial of degree n defined by

$$B_{n,m}(q | Q_{n,m} = D_{q,\omega}^m P_{n+m}. (3.45)$$

with

$$B_{n,m}(q) = \prod_{j=0}^{m-1} [n+m-j]_q, \quad Q_{n,0} \equiv P_n \quad \forall n \in \mathbb{N}.$$
 (3.46)

The following properties are equivalent:

if There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that

$$D_{q,\omega}(\phi \mathcal{L}) = \psi \mathcal{L}. \tag{3.47}$$

ii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m,

$$D_{q,\omega}(\phi \mathcal{L}_m) = \psi_m \mathcal{L}. \tag{3.48}$$

$$\langle \mathcal{L}_m, Q_{j,m} Q_{n,m} \rangle = k_n \delta_{j,n}, \quad \forall j, n \in \mathbb{N}, \quad (k_n \neq 0 \ \forall n \in \mathbb{N}), \tag{3.49}$$

with the linear function of  $\mathcal{L}_m$  and the polynomial  $\psi_m$  defined, recursively, by

$$\psi_{m+1} = D_{q,\omega}\phi + qA_{q,\omega}\psi_m, \quad \psi_0 \equiv \psi, \tag{3.50}$$

$$\mathcal{L}_{m+1} = A_{q,\omega}(\mathcal{O}\mathcal{L}_{r_0}), \quad \mathcal{L}_0 \equiv \mathcal{L} \tag{3.51}$$

and given explicitly by

$$\psi_r(x) = \frac{\phi(q^m x + \omega[m]_q) - \phi(x)}{(q - 1)x + \omega} + q^m \psi(q^m x + \omega[m]_q). \tag{3.52}$$

$$\mathcal{L}_{r_{+}} = \prod_{j=1}^{m} \phi(q^{j}x + \omega[j]_{q}) A_{q,\omega}^{m} \mathcal{L}. \tag{3.53}$$

iii) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that for any integer m, the following second-order difference equation holds:

$$\phi D_{q,\omega} \mathcal{D}_{q,\omega}^* Q_{n,m} + \psi_m D_{q,\omega} Q_{n,m} + \lambda_{n,m} Q_{n,m} = 0 \quad \forall n \in \mathbb{N}.$$

$$(3.54)$$

with the polynomial  $U_m$  given by (3.52) and the constant  $\lambda_{n,m}$  given by

$$\lambda_{n,m} = -[n]_{\frac{1}{q}} \{ \mathcal{D}_q \psi_m + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \}.$$
 (3.55)

iv) There exist two polynomials,  $\phi$  of degree at most two and  $\psi$  of degree one, such that, for any integer m, the following relation holds:

$$[n]_q \mathcal{D}_{q,\omega}^{\star} \left[ Q_{n-1,m+1} \mathcal{L}_{m+1} \right] = -q \lambda_{n,m} Q_{n,m} \mathcal{L}_m \quad \forall n \in \mathbb{N}.$$

$$(3.56)$$

with the polynomial  $\psi_m$ , the linear functional  $\mathcal{L}_m$  and the constant  $\lambda_{n,m}$  given, respectively, by (3.52), (3.53) and (3.55).

v) There exist a polynomial  $\phi$  of degree at most two and three constants  $c_{n,n+1}$ ,  $c_{n,n}$ ,  $c_{n,n-1}$  with  $c_{n,n-1} \neq 0$  such that

$$\phi D_{1,\perp}^* P_n = c_{n,n-1} P_{n-1} + c_{n,n} P_n + c_{n,n-1} P_{n-1}, \ n > 1.$$
(3.57)

vi) For any non-zero integer m, there exist a sequence of complex numbers  $\{u_{n,m}\}_{n\in\mathbb{N}}$  such that

$$Q_{n,m+1} = \zeta_{n,m} + u_{n-1,m}Q_{n-1,m} + v_{n-2,m}Q_{n-2,m}, \quad \forall n \in \mathbb{N} - \{0,1\}.$$
(3.58)

Proof:  $i) \implies ii$ ). Suppose that the property i) is satisfied. We will show by induction on m that the relations (3.48) and (3.49) hold. From (3.47) and the orthogonality of the family  $\{P_n\}_{n\in\mathbb{N}}$ , it is obvious that the relations (4.48) and (3.49) are satisfied for m=0. Suppose that relations (3.48) and (3.49) are satisfied up to a fixed integer m. Using Proposition 3.1, we have

$$\begin{split} D_{q,\omega}\left(\phi\mathcal{L}_{m+1}\right) &= D_{q,\omega}\left(\phi A_{q,\omega}(\phi \mathcal{L}_{m}\right) \\ &= D_{q,\omega}\left(\phi A_{q,\omega}(\phi \mathcal{L}_{m}\right) + A_{q,\omega}\phi D_{q,\omega} A_{q,\omega}(\phi \mathcal{L}_{m}) \\ &= D_{q,\omega}\left(\phi A_{q,\omega}(\phi \mathcal{L}_{m}\right) + A_{q,\omega}\phi q A_{q,\omega} D_{q,\omega}\left(\phi \mathcal{L}_{m}\right) \\ &= D_{q,\omega}\left(\phi A_{q,\omega}(\phi \mathcal{L}_{m}\right) + q A_{q,\omega}\phi A_{q,\omega}\left(\psi_{m} \mathcal{L}_{m}\right) \\ &= D_{q,\omega}\left(\phi A_{q,\omega}(\phi \mathcal{L}_{m}\right) + q A_{q,\omega}\psi_{m} A_{q,\omega}\phi \mathcal{L}_{m}\right) \\ &= (D_{q,\omega}\left(\phi A_{q,\omega}(\phi \mathcal{L}_{m}\right) + q A_{q,\omega}\psi_{m}\right) A_{q,\omega}(\phi \mathcal{L}_{m}) \\ &= \psi_{m+1}\mathcal{L}_{m+1}. \end{split}$$

Thus, the relation (3.48) holds for all integers m.

Let j and n be two integers such that j < n. Using Proposition 3.1 and the fact that (3.48) and 3.49) hold up to a fixed integer m, we have

$$\begin{split} &[j+1]_q [n+1]_q {}^{\prime} \mathcal{L}_{m+1}, Q_{j,m+1} Q_{n,m+1} \rangle \\ &= \langle A_{q,\omega} (\phi \mathcal{L}_m , D_{q,\omega} Q_{j+1,m} D_{q,\omega} Q_{n+1,m} \rangle \\ &= \langle A_{q,\omega} (\phi \mathcal{L}_m , D_{q,\omega} [Q_{n+1,m} D_{q,\omega} Q_{j+1,m}] - A_{q,\omega} Q_{n+1,m} D_{q,\omega}^2 Q_{j+1,m} \rangle \\ &= -\frac{1}{q} \langle D_{q,\omega}^* A_{q,\omega} (\phi \mathcal{L}_m), Q_{n+1,m} D_{q,\omega} Q_{j+1,m} \rangle - \frac{1}{q} \langle \phi \mathcal{L}_m, Q_{n+1,m} A_{q,\omega}^* D_{q,\omega}^2 Q_{j+1,m} \rangle \end{split}$$

$$= -\langle D_{q,\omega}(\phi \mathcal{L}_m), Q_{n+1,m} D_{q,\omega} Q_{j+1,m} \rangle - \frac{1}{q} \langle \mathcal{L}_m, Q_{n+1,m} \phi A_{q,\omega}^* D_{q,\omega}^2 Q_{j+1,m} \rangle$$

$$= -\langle \mathcal{L}_m, \psi_n, Q_{n+1,m} D_{q,\omega} Q_{j+1,m} \rangle - \frac{1}{q} \langle \mathcal{L}_m, Q_{n+1,m} \phi A_{q,\omega}^* D_{q,\omega}^2 Q_{j+1,m} \rangle$$

$$= 0.$$

because  $\deg(\phi A_{q,\omega}^{\star} D_{q,\omega}^2 Q_{j+1,m}) \leq \deg(\psi_m D_{q,\omega} Q_{j+1,m}) = j+1 < n+1$  (see Lemma 3.5). Repeated use of the following relations, proved in Lemma 3.5,

$$[n+1]_q[n+1]_q \langle \mathcal{L}_{m+1}, Q_{n,m+1}Q_{n,m+1} \rangle = \lambda_{n+1,m} \langle \mathcal{L}_m, Q_{n+1,m}Q_{n+1-n} \rangle, \tag{3.59}$$

$$\lambda_{n+1,m} = \frac{[n+1]_q q^{2m}}{[n+1+2m]_q} \lambda_{n+1+2m,0} \neq 0 \quad \forall n, m \in \mathbb{N}$$
(3.60)

gives

$$\langle \mathcal{L}_m, Q_{n,m} Q_{n,m} \rangle = \prod_{i=0}^{m-1} \frac{\lambda_{n+1+j,m-1-j}}{[n+j+1]_q^2} \langle \mathcal{L}. P_{n+m} P_{n+m} \rangle.$$
 (3.61)

Thus,

$$\langle \mathcal{L}_m, Q_{n,m} Q_{n,m} \rangle \neq 0 \quad \forall n, m \in \mathbb{N}.$$

Iteration of relations (3.50) and taking into account (3.9) lead to

$$\psi_{m} = \sum_{j=0}^{m-1} q^{j} A_{q,\omega}^{j} D_{q,\omega} \phi(x) + q^{m} A_{q,\omega}^{m} \psi(x) 
= \sum_{j=0}^{m-1} q^{j} q^{-j} D_{q,\omega} A_{q,\omega}^{j} \phi(x) + q^{m} A_{q,\omega}^{m} \psi(x) 
= \sum_{j=0}^{m-1} \frac{A_{q,\omega}^{j+1} \phi(x) - A_{q,\omega}^{j} \phi(x)}{(q-1)x + \omega} + q^{m} A_{q,\omega}^{m} \psi(x).$$

Thus,

$$\psi_m(x) = \frac{\phi(q^m x + \omega[m]_q) - \phi(x)}{(q-1)x + \omega} + q^m \psi(q^m x + \omega[m]_q).$$

Taking into account Remark 3.5, relation (3.53) follows directly from the iteration of 3.51).

 $ii) \implies iii)$ . Assuming that the property ii) holds, it follows that for any integer m, the monic polynomial family  $\{Q_{n,m}\}_{n\in\mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}_m$ ; thus,  $\{Q_{n,m}\}_{n\in\mathbb{N}}$  forms a basis of  $\mathbb{P}$ . From the following expansion

$$\phi D_{q,\omega} D_{q,\omega}^* Q_{n,m} + \psi_m D_{q,\omega} Q_{n,m} = -\sum_{i=0}^n \lambda_{j,m} Q_{j,m}, \qquad (3.62)$$

we obtain

$$\lambda_{j,m}\langle \mathcal{L}_{m}, Q_{j,m}Q_{j,m}\rangle = -\langle \mathcal{L}_{m}, \phi Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m} + \psi_{m}Q_{j,m}D_{q,\omega}Q_{r,m}\rangle$$

$$= -\langle \mathcal{L}_{m}, \phi Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m}\rangle - \langle \psi_{m}\mathcal{L}_{m}, Q_{j,m}D_{q,\omega}\mathcal{L}_{n,m}\rangle$$

$$= -\langle \mathcal{L}_{m}, \phi Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m}\rangle - \langle D_{q,\omega}(\phi\mathcal{L}_{m}), Q_{j,m}D_{q,\omega}Q_{n,m}\rangle$$

$$= -\langle \mathcal{L}_{m}, \phi Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m}\rangle + \frac{1}{q}\langle \phi \mathcal{L}_{m}, D_{q,\omega}^{*}Q_{j,m}D_{q,\omega}Q_{n,m}\rangle$$

$$= -\langle \mathcal{L}_{m}, \phi Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{j,m}A_{q,\omega}^{*}D_{q,\omega}Q_{n,m}\rangle$$

$$+ \frac{1}{q}\langle \phi \mathcal{L}_{m}, D_{q,\omega}^{*}Q_{j,m}A_{q,\omega}^{*}D_{q,\omega}Q_{n,m}\rangle$$

$$= -\langle \mathcal{L}_{m}, \phi Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m}\rangle + \langle \phi \mathcal{L}_{m}, Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m}\rangle$$

$$= -\langle \mathcal{L}_{m}, \phi Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m}\rangle + \langle \phi \mathcal{L}_{m}, Q_{j,m}D_{q,\omega}D_{q,\omega}^{*}Q_{n,m}\rangle$$

$$\begin{split} & + \frac{1}{q} \langle \phi \, \mathcal{L}_m, D_{q,\omega}^* Q_{j,m} A_{q,\omega}^* D_{q,\omega} Q_{n,m} \rangle \\ = & + \frac{1}{q} \langle \phi \, \mathcal{L}_m, A_{q,\omega}^* D_{q,\omega} Q_{j,m} A_{q,\omega}^* D_{q,\omega} Q_{n,m} \rangle \\ = & + \frac{1}{q} \langle \phi \, \mathcal{L}_m, A_{q,\omega}^* (D_{q,\omega} Q_{j,m} D_{q,\omega} Q_{n,m}) \rangle \\ = & \langle A_{q,\omega} (\phi \, \mathcal{L}_m), D_{q,\omega} Q_{j,m} D_{q,\omega} Q_{n,m} \rangle \\ = & [j]_q [n]_q \langle \mathcal{L}_{m-1}, Q_{j-1,m+1} Q_{n-1,m+1} \rangle \\ = & 0 \text{ for } j < n, \end{split}$$

by orthogonality of  $\{Q_{n,m+1}\}_{n\in\mathbb{N}}$  with respect to  $\mathcal{L}_{m+1}$ . Thus,

$$\phi D_{q,\omega} D_{q,\omega}^* Q_{n,m} + \psi_m D_{q,\omega} Q_{n,m} + \lambda_{r,m} Q_{n,m} = 0 \qquad \forall n \in \mathbb{N}.$$

Identification of the coefficients of  $x^n$  in the previous equation gives

$$\frac{\phi''}{2}[n-1]_q[n]_{\frac{1}{q}} + \psi'_m[n]_q - \lambda_{n,m} = 0.$$

Then using the following relation

$$[n]_{\frac{1}{g}} = q^{1-n}[n]_q. \quad \forall n \in \mathbb{N},$$
 (3.63)

we obtain

$$\lambda_{n,m} = -[n]_q (\psi'_m + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q}).$$

 $iii \mapsto i$ ). Assuming that the property iii) holds, elementary computations using (3.12) and (3.54) for m = 1 give

$$[n+1]_{+}\langle D_{q,\omega}(\phi\mathcal{L}) - \psi\mathcal{L}, Q_{n,1}\rangle = -\frac{1}{q}\langle \phi\mathcal{L}, D_{q,\omega}^{*} D_{q,\omega} P_{n+1}\rangle - \langle \psi\mathcal{L}, D_{q,\omega} P_{n+1}\rangle$$

$$= -\langle \mathcal{L}, \phi D_{+,\omega} D_{q,\omega}^{*} P_{n+1} + \psi D_{q,\omega} P_{n+1}\rangle$$

$$= \langle \mathcal{L}, \lambda_{n+1,1} P_{n+1}\rangle$$

$$= 0 \quad \forall n \in \mathbb{N}.$$
(3.64)

Since the family  $\{C_{n,1}\}_{n\in\mathbb{N}}$  forms a basis of  $\mathbb{P}'$ , it is clear that  $D_{d,\omega}\left(\partial\mathcal{L}\right)=\psi\mathcal{L}.$ 

iii:  $\iff vi$ ). Computations using Proposition 3.1 shew straightforwardly that given an integer m, (3.56) is equivalent to

$$+ \ni D_{q,\omega} D_{n,m}^* + \psi_m D_{q,\omega} Q_{n,m} + \lambda_{n,\neg_i} Q_{r-m}) \mathcal{L}_m = 0 \qquad \forall n \in \mathbb{N}.$$

Since  $\mathcal{L}_m$  is regular (see property ii)) it is obvious that properties iii) and iv, are equivalent.

 $|i\rangle \Longrightarrow v$ . Expanding the polynomial  $\phi(x)D_{q,\omega}^*P_n$  in the basis  $\{P_n\}_{n\in\mathbb{N}}$  of  $\mathbb{R}$ ,

$$\phi(x)D_{q,\omega}^*P_n = \sum_{j=1}^{n-1} c_{n,j}P_j,$$

we obtain, using (3.10).

$$\begin{array}{lll} c_{n,j}\langle\mathcal{L},P_{j}P_{j}\rangle & = & \langle \phi\mathcal{L},P_{j}D_{q,\omega}^{*}P_{n}\rangle \\ & = & \langle \phi\mathcal{L},D_{q,\omega}^{*}(A_{q,\omega}P_{\parallel}P_{\perp}) - P_{n}D_{q,\omega}^{*}A_{q,\omega}P_{j} \rangle \\ & = & -q\langle D_{q,\omega}(\phi\mathcal{L}),A_{q,\omega}P_{j}P_{n}\rangle - q\langle \phi\mathcal{L},P_{\perp}D_{q,\omega}P_{j} \rangle \\ & = & -q\langle\mathcal{L},(\psi A_{q,\omega}P_{j} + \phi D_{q,\omega}P_{j},P_{n}) \\ & = & 0 \quad \text{for } j < n-1. \end{array}$$

Thus.

$$\phi(x)D_{q,\omega}^* P_n = c_{n,n+1}P_{n+1} + c_{n,n}P_n + c_{n,n-1}P_{n-1}.$$

When we set j = n - 1 in the above equations we obtain, taking into account (3.63),

$$c_{n,n-1}\langle \mathcal{L}. P_{n-1} P_{n-1} \rangle$$

$$= -q \langle \mathcal{L}, (\psi A_{q,\omega} P_{n-1} + \phi D_{q,\omega} P_{n-1}) P_n \rangle$$

$$= -q (q^{n-1} \psi' + [n-1]_{\frac{1}{q}} \frac{\phi''}{2}) \langle \mathcal{L}, x^n P_n \rangle$$

$$= -q^n (\psi' + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q}) \langle \mathcal{L}, P_n P_n \rangle$$

$$= q^n \frac{\lambda_{n,0}}{[n]_q} \langle \mathcal{L}, P_n P_n \rangle$$

$$\neq 0 \text{ for } n > 1$$

by relation (3.74).

 $v) \Longrightarrow i$ ). Expanding the linear functional  $D_{q,\omega}(\phi \mathcal{L})$  in the dual basis  $\{\mathbf{P}_n\}_{n\in\mathbb{N}}$  of the orthogonal family  $\{P_n\}_{n\in\mathbb{N}}$ .

$$D_{q,\omega}(\phi\mathcal{L}) = \sum_{n\geq 0} h_n \mathbf{P}_n,\tag{3.65}$$

we obtain, using (3.57),

$$h_n = \langle D_{q,\omega}(\phi \mathcal{L}), P_n \rangle$$

$$= -\frac{1}{q} \langle \mathcal{L}, \phi D_{q,\omega}^* P_n \rangle$$

$$= -\frac{1}{q} \langle \mathcal{L}, c_{n,n+1} P_{n+1} + c_{n,n} P_n + c_{n,n-1} P_{n-1} \rangle$$

$$= 0 \text{ for } n \geq 2,$$

then

$$h_n = 0 \text{ for } n \ge 2.$$
 (3.66)

On the other hand.

$$h_0 = \langle D_{q,\omega}(\phi \mathcal{L}), 1 \rangle = -\frac{1}{q} \langle \phi \mathcal{L}, D_{q,\omega}^* 1 \rangle = 0, \tag{3.67}$$

$$h_1 = -\frac{c_{1,0}}{q} \langle \mathcal{L}, P_0 P_0 \rangle \neq 0$$
 (3.68)

because by hypothesis,  $c_{n,n-1} \neq 0$  for  $n \geq 1$ .

Use of (3.65)-(3.68) and the fact that  $\mathbf{P}_1 = \frac{P_1}{(\mathcal{L}, P_1 P_1)} \mathcal{L}$  (see (2.17)) give

$$D_{q,\omega}(\phi\mathcal{L}) = h_1 \mathbf{P}_1 = -\frac{c_{1,0} \langle \mathcal{L}, P_0 P_0 \rangle P_1}{q \langle \mathcal{L}, P_1 P_1 \rangle} \mathcal{L}. \tag{3.69}$$

Then the regular linear functional  $\mathcal{L}$  satisfies  $D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}$ , where

$$\psi(x) = -\frac{c_{1,0}\langle \mathcal{L}, P_0 P_0 \rangle}{q\langle \mathcal{L}, P_1 P_1 \rangle} P_1(x),$$

with  $\deg(\phi) \leq 2$  and  $\deg(\psi) = \deg(P_1) = 1$ . Thus  $\mathcal{L}$  is  $D_{q,\omega}$ -classical, and therefore properties i) and v) are equivalent.

i)  $\Longrightarrow vi$ ). Since properties i) and ii) are equivalent, assuming that the property i) is satisfied, it yields that for any integer m the monic polynomial family  $\{Q_{n,m}\}_{n\in\mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}_m$ . Let m be a non-zero integer. We expand the polynomial  $Q_{n,m+1}$  in the basis  $\{Q_{n,m}\}_{n\in\mathbb{N}}$  of  $\mathbb{P}$ ,

$$Q_{n,m-1} = Q_{n,m} + \sum_{i=0}^{n-1} u_{j,m} Q_{j,m}$$

and obtain

$$\begin{aligned} u_{j,m}[j+1]_{q}\langle \mathcal{L}_{m}, Q_{j,m} \rangle &= [j+1]_{q}\langle \mathcal{L}_{m}, Q_{n,m-1} Q_{j,m} \rangle \\ &= \langle \mathcal{L}_{m}, D_{q,\omega} Q_{j+1,m-1} Q_{n,m-1} \rangle \\ &= \langle A_{q,\omega}(\phi \mathcal{L}_{m-1}), D_{q,\omega}(Q_{j+1,m-1} Q_{n,m-1}) \rangle \\ &- \langle A_{q,\omega}(\phi \mathcal{L}_{m-1}), A_{q,\omega} Q_{j+1,m-1} D_{q,\omega} Q_{n,m-1} \rangle \\ &= -\frac{1}{q}\langle D_{q,\omega}^{*} A_{q,\omega}(\phi \mathcal{L}_{m-1}), Q_{j+1,m-1} Q_{n,m-1} \rangle \\ &- [n]_{q}\langle \mathcal{L}_{m}, A_{q,\omega} Q_{j+1,m-1} Q_{n-1,m} \rangle \\ &= -\langle D_{q,\omega}(\phi \mathcal{L}_{m-1}), Q_{j+1,m-1} Q_{n,m-1} \rangle \\ &- [n]_{q}\langle \mathcal{L}_{m}, A_{q,\omega} Q_{j+1,m-1} Q_{n-1,m} \rangle \\ &= -\langle \mathcal{L}_{m-1}, \psi_{m-1} Q_{j+1,m-1} Q_{n,m-1} \rangle \\ &- [n]_{q}\langle \mathcal{L}_{m}, A_{q,\omega} Q_{j+1,m-1} Q_{n-1,m} \rangle \\ &= 0 \text{ for } j < n-2, \end{aligned}$$

by the orthogonality of  $\{Q_{n,m}\}_{n\in\mathbb{N}}$  and  $\{Q_{n,m-1}\}_{n\in\mathbb{N}}$  with respect to  $\mathcal{L}_m$  and  $\mathcal{L}_{m-1}$ , respectively. Therefore,

$$Q_{n,m+1} = Q_{n,m} - u_{n+1,m}Q_{n-1,m} + u_{n-2,m}Q_{n-2,m} \quad \forall n \in \mathbb{N} - \{0,1\} \quad \forall m \in \mathbb{N} - \{0\}.$$

 $iv) \Longrightarrow i$ ). Let  $\{\mathbf{P}_n\}_{n \in \mathbb{N}}$  and  $\{\mathbf{Q}_{n,m}\}_{n \in \mathbb{N}}$  be the dual basis associated to the monic families  $\{P_n\}_{n \in \mathbb{N}}$  and  $\{Q_{n,m}\}_{n \in \mathbb{N}}$ , respectively.

In the first step we expand  $\mathbf{Q}_{0,1}$  in the dual basis  $\{\mathbf{Q}_{n,0}\}_{n\in\mathbb{N}}$ ,

$$\mathbf{Q}_{0,1} = \sum_{j>0} \alpha_j \mathbf{Q}_{j,0}$$

and obtain, using. (3.58)

$$\begin{array}{rcl} \alpha_{.} & = & \langle \mathbf{Q}_{0,1}, Q_{j,0} \\ & = & \langle \mathbf{Q}_{0,1}, Q_{j,1} + u_{j-1,1}Q_{j-1,1} + u_{j-2,1}Q_{j-2,1} \rangle \\ & = & 0 \text{ for } > 3. \end{array}$$

Using (2.17), we, therefore, obtain,

$$\mathbf{Q}_{0,1} = \sum_{j=0}^{2} \alpha_j \mathbf{Q}_{j,0} = \sum_{j=0}^{2} \alpha_j \mathbf{P}_j = \phi \mathcal{L}, \tag{3.70}$$

where

$$\phi(x) = \sum_{j=0}^{2} \frac{\alpha_j P_j(x)}{\langle \mathcal{L}, P_j P_j \rangle}.$$
 (3.71)

In the second step, we compute  $D_{q,\omega}^* \mathbf{Q}_{0,1}$  using (2.17), (3.44) and obtain

$$D_{\gamma,\omega}^{\bullet}(\mathbf{Q}_{0,1}) = -q \, \mathbf{P}_1 = \psi \mathcal{L}. \tag{3.72}$$

where

$$\psi(x) = \frac{-qP_1(x)}{\langle \mathcal{L}. P_1 P_1 \rangle}.$$
(3.73)

Use of (3.70)-(3.72) permit us to conclude that

$$D_{q,\omega}^*(\phi \mathcal{L}) = \psi \mathcal{L}.$$

47

The previous equation, thanks to Proposition 3.2, is equivalent to

$$D_{q,\omega}(\tilde{\phi}\mathcal{L}) = \psi\mathcal{L},$$

with

$$\tilde{\phi} = q\phi - [(q-1)x + \omega]\psi.$$

We complete the proof of vi)  $\Longrightarrow$  i), by remarking that  $\deg(\phi) \le \deg(P_2 = 2)$  and  $\deg(\psi) = \deg(P_1) = 1$  (see (3.71) and (3.73)).

Summing up, we have proved that  $i) \Longrightarrow ii) \Longrightarrow iii) \Longrightarrow i)$ .  $iii \iff iv$ ).  $i) \iff v$ ) and  $i) \iff vi$ ; thus, the proof of the theorem is complete.

**Lemma 3.5** Let  $\mathcal{L}$  be a regular linear functional satisfying  $D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}$ , where  $\phi$  is a polynomial of degree at most two and  $\psi$  a first-degree polynomial. The following properties hold:

$$i) \lambda_{n+1,0} \neq 0 \forall n \in \mathbb{N}, (3.74)$$

*ii*) 
$$\lambda_{n+1,m} = \frac{[n+1]_q q^{2m}}{[n+1+2m]_q} \lambda_{n+1+2m,0}, \tag{3.75}$$

$$D_{q,\omega}\psi_m \neq 0 \quad \forall m \in \mathbb{N}. \tag{3.76}$$

$$[n+1]_q^2 \langle \mathcal{L}_{m+1}, Q_{n,m+1} Q_{n,m+1} \rangle = \lambda_{n-1,m} \langle \mathcal{L}_m, Q_{n+1,m} Q_{n+1,m} \rangle. \tag{3.77}$$

with  $\mathcal{L}_m$ ,  $\lambda_{n,m}$  and  $Q_{n,m}$  defined in Theorem 3.1.

Proof: i) From the relation

$$D_{q,\omega}^* x^n = [n]_{\frac{1}{q}} x^{n-1} + \sum_{j=2}^n a_{n,j}^*(q,\omega) \, z^{n-j}.$$

where  $a_{n,j}^*(q,\omega)$  are complex numbers given by

$$a_{n,j}^{\star}(q,\omega) = \left(\frac{\omega}{q}\right)^{j-1} \sum_{k=0}^{n-1} q^k \binom{k}{j-1},$$
 (3.78)

we obtain,

$$D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L} \iff \langle D_{q,\omega}(\phi\mathcal{L}), x^n \rangle = \psi\mathcal{L}, x^n \qquad \forall n \in \mathbb{N}$$

$$\iff -\frac{1}{q} \langle \phi\mathcal{L}, D_{l,\omega}^* x^n \rangle = \langle \psi\mathcal{L}, x^n \rangle \quad \forall n \in \mathbb{N}$$

$$\iff (\psi^l + [n]_{\frac{1}{q}} \frac{\phi^n}{2q}) M_{n+1} = \sum_{j=0}^n \tilde{f}_j M_j \quad \forall n \in \mathbb{N}$$

$$\iff -\frac{\lambda_{n+1,0}}{[n+1]_q} M_{n+1} = \sum_{j=0}^n \tilde{f}_j M_j \quad \forall n \in \mathbb{N}.$$

$$(3.79)$$

where  $M_j \equiv \langle \mathcal{L}, x^j \rangle$  is the moment of order j of the linear functional  $\mathcal{L}$  and  $\tilde{f}_j$  are complex numbers easily computed as function of coefficients  $a_{n,j}^*$  and those of the polynomials  $\phi$  and  $\psi$ . Since  $\mathcal{L}$  is regular, to have all its moments given in the unique way by the previous ones, it is necessary to have

$$\lambda_{n+1,0} \neq 0 \quad \forall n \in \mathbb{N}. \tag{3.80}$$

ii) The  $D_{q,\omega}$ -derivative of (3.52), taking into account (3.63), gives

$$\begin{split} D_{q,\omega}\psi_m &= \psi_m' &= [2m]_q \frac{\phi''}{2} + q^{2m} \psi' \\ &= q^{2m} (\psi' + q^{1-2r}) (2m)_q \frac{\phi''}{2q}) \\ &= q^{2m} (\psi' + [2m]_- \frac{\phi}{2}). \end{split}$$

then

$$D_{q,\omega}\psi_m = q^{2m}(\psi' + [2m]_{\frac{1}{q}}\frac{\phi''}{2q}) = \frac{-q^{2m}}{[2m+1]_q}\lambda_{2m+1,0} \quad \forall m \in \mathbb{N}.$$
(3.81)

We, therefore, conclude using (3.80) that for any integer m,  $\psi'_m \neq 0$  and  $\psi_m$  is a first-degree polynomial. iii) use of (3.55), (3.63) and (3.81) give

$$\begin{split} \lambda_{n,m} &= -[n]_q \{ \psi_m' + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \} \\ &= -[n]_q \{ q^{2m} (\psi' + [2m]_{\frac{1}{q}} \frac{\phi''}{2q}) + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \} \\ &= -[n]_q q^{2m} \{ \psi' + ([2m]_{\frac{1}{q}} + q^{-2m} [n-1]_{\frac{1}{q}}) \frac{\phi''}{2q} \} \\ &= -[n]_q q^{2m} \{ \psi' + [2m+n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \} \\ &= \frac{[n]_q q^{2m}}{[n+2m]_q} \lambda_{n+2m,0}. \end{split}$$

We derive the relation iverusing Proposition 3.1, the second property of Theorem 3.1 and the orthogonality of the family  $\{Q_{n,m}\}_{n\in\mathbb{N}}$  with respect to  $\mathcal{L}_m$ . In fact,

$$\begin{split} &[n+1]_q[n+1]_q \langle \mathcal{L}_{m+1}, Q_{n,m+1}Q_{n,m+1} \rangle \\ &= \langle A_{q,\omega} (\phi \mathcal{L}_m), D_{q,\omega}Q_{n+1,m} D_{q,\omega}Q_{n+1,m} \rangle \\ &= \langle A_{q,\omega} (\phi \mathcal{L}_m), D_{q,\omega} (Q_{n-1,m}D_{q,\omega}Q_{n+1,m}) - A_{q,\omega}Q_{n+1,m} D_{q,\omega}^2 Q_{n+1,m} \rangle \\ &= -\frac{1}{q} \langle D_{q,\omega}^* A_{q,\omega} (\phi \mathcal{L}_m), Q_{n+1,m}D_{q,\omega}Q_{n+1,m} \rangle \\ &= -\frac{1}{q} \langle \mathcal{O} \mathcal{L}_m - Q_{n+1,m} A_{q,\omega}^* D_{q,\omega}^2 Q_{n+1,m} \rangle \\ &= -\langle \mathcal{L}_{r_1}, \psi_{r_1}Q_{n-1,m}D_{q,\omega}Q_{n+1,m} \rangle - \frac{1}{q} \langle \mathcal{L}_m, \phi Q_{n+1,m}D_{q,\omega}^* D_{q,\omega}Q_{n+1,m} \rangle \\ &= -[n+1]_q \psi_m' \langle \mathcal{L}_m, Q_{n+1,m}Q_{n+1,m} \rangle - [n+1]_q [n]_{\frac{1}{q}} \frac{\phi''}{2q} \langle \mathcal{L}_m, Q_{n+1,m}Q_{n+1,m} \rangle \\ &= -[n+1]_q \{ \psi_m' + [n]_{\frac{1}{q}} \frac{\phi''}{2q} \} \langle \mathcal{L}_m, Q_{n+1,m}Q_{n+1,m} \rangle \\ &= \lambda_{n+1,m} \langle \mathcal{L}_m, Q_{n+1,m}Q_{n+1,m} \rangle. \end{split}$$

### 3.2.2 $D_{q,\omega}$ -semi-classical orthogonal polynomials

Let  $\mathcal{L}$  be a regular linear functional and  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family. When the linear functional  $\mathcal{L}$  is  $D_{q,\omega}$ -semi-classical of class s>0 satisfying (3.47), the characterisation theorem (see Theorem 3.1) is not valid anymore. In particular, the derivative  $D_{q,\omega}P_n$  of  $P_n$  is not orthogonal with respect to  $A_{q,\omega}(o\mathcal{L})$  but quasi-orthogonal of class s with respect to  $A_{q,\omega}(\phi\mathcal{L})$ . The following theorem, which generalise some results in [Salto, 1995] and [Medem, 1996], gives some characterisations for  $D_{q,\omega}$ -semi-classical orthogonal polynomials.

Theorem 3.2 Let  $\mathcal{L}$  be a regular linear functional and  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal family. The following properties are equivalent:

i) There exist two polynomials:  $\psi$  of degree at least one and  $\phi$  such that

$$D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}. \tag{3.82}$$

ii) There exists a polynomials  $\phi \neq 0$  and an integer s with deg  $\phi \leq s+2$  such that

$$\begin{cases}
\langle A_{q,\omega}(\phi\mathcal{L}), Q_{m,1}Q_{n,1} \rangle = 0, & |n-m| > s \\
\langle A_{q,\omega}(\phi\mathcal{L}), Q_{m,1}Q_{m+s,1} \rangle \neq 0, & \forall m \ge 1,
\end{cases}$$
(3.83)

where polynomials  $Q_{n,1}$  are defined in theorem 3.1.

iii) There exists a polynomial  $\phi \neq 0$  and an integer s with  $t = \deg \phi \leq s + 2$  such that

$$\phi D_{q,\omega}^* P_n = \sum_{j=n-s-1}^{n+t-1} \xi_{n,j} P_j \quad n > s+1, \tag{3.84}$$

with

$$\xi_{n,n-s-1} \neq 0, \ n > s+1.$$
 (3.85)

*Proof*:  $i) \Longrightarrow ii$ ). Suppose that (3.82) is satisfied. Then  $\phi \neq 0$  by Lemma 3.1 and (3.82). Let m and n be two integers such that n > m + s and pose  $s = \max\{\deg(\phi) - 2, \deg(\psi) - 1\}$ . Using (3.82) and Proposition 1.3, we get

$$\begin{split} &[m+1]_q \left[n+1\right]_q \langle A_{q,\omega}(\phi\mathcal{L}), Q_{n,1}Q_{m,1}\rangle \\ &= \langle A_{q,\omega}(\phi\mathcal{L}), D_{q,\omega} P_{m+1}D_{q,\omega} P_{n+1}\rangle \\ &= \langle A_{q,\omega}(\phi\mathcal{L}), D_{q,\omega} \left[P_{n+1}D_{q,\omega} P_{m+1}\right] - A_{q,\omega}P_{n+1}D_{q,\omega}^2 P_{m+1}\rangle \\ &= -\frac{1}{q} \langle D_{q,\omega}^* A_{q,\omega}(\phi\mathcal{L}), P_{n+1}D_{q,\omega} P_{m-1}\rangle - \frac{1}{q} \langle \phi\mathcal{L}, P_{n+1}A_{q,\omega}^* D_{q,\omega}^2 P_{m+1}\rangle \\ &= -\langle \mathcal{L}. P_{n+1} \psi D_{q,\omega} P_{m+1}\rangle - \frac{1}{q} \langle \mathcal{L}, P_{n+1}\phi A_{q,\omega}^* D_{q,\omega}^2 P_{m+1}\rangle \\ &= 0, \end{split}$$

because  $\deg(\frac{1}{q} \phi A_{q,\omega}^* D_{q,\omega}^2 P_{m+1} + \psi D_{q,\omega} P_{m+1}) \le m+s+1 < n+1$ . Given non-zero integer m, we have, from the previous computations,

$$\langle A_{q,\omega}(\phi \mathcal{L}), D_{q,\omega} P_{m+1} D_{q,\omega} P_{m-s+1} \rangle$$

$$= -\langle \mathcal{L}, P_{m+s+1} \psi D_{q,\omega} P_{m+1} \rangle - \frac{1}{q} \langle \mathcal{L}, P_{m+s+1} \phi A_{q,\omega}^* D_{q,\omega}^2 P_{m+1}$$

$$= -\{ [m+1]_q \psi_p \delta_{p,s+1} + [m]_q [m+1]_q q^{1-m} \frac{\phi_t}{q} \delta_{t,s+2} \} I_{0,m-s+1}$$

$$= -[m+1]_q \{ \psi_p \delta_{p,s+1} + \frac{\phi_t}{q} [m]_{\frac{1}{q}} \delta_{t,s+2} \} I_{0,m+s+1}, \qquad (3.86)$$

where  $I_{0,m}$  is defined by

$$I_{0,m} = \langle \mathcal{L}, P_m P_m \rangle, \ m \ge 0, \tag{3.87}$$

and the polynomials  $\phi$  and  $\psi$  are given by

$$\phi(x) = \sum_{j=0}^{t} \phi_j x^j, \ \psi(x) = \sum_{j=0}^{p} \psi_j x^j,$$
 (3.88)

with  $|\phi_t||\psi_p| \neq 0$ .

It results from (3.86) that  $\langle A_{q,\omega}(\phi \mathcal{L}), D_{q,\omega} P_{m+1} D_{q,\omega} P_{m+s+1} \rangle (I_{0,n+s+1})^{-1} = U(m,s)$ , takes one of the three values:

i): t < s + 2. p = s + 1.

$$U(m,s) = -[m+1]_q \psi_{s+1} \neq 0, \ m \geq 0,$$

ii): t = s + 2. p < s + 1.

$$U(m,s) = -[m]_{\frac{1}{q}} [m+1]_q \frac{\phi_{s+2}}{q} \neq 0, \ m \geq 1,$$

iii): t = s + 2, p = s + 1,

$$U(m,s) = -[m+1]_q \{ \psi_{s+1} + \frac{\phi_{s+2}}{q} [m]_{\frac{1}{q}} \}.$$

U(m,s) for the case iii) is not zero by the regularity of the linear functional  $\mathcal{L}$ .

In fact, mimicking the approach used in (3.79), we conclude that if  $\mathcal{L}$  is regular and satisfies (3.82), with  $\phi$  and  $\psi$  given by (3.88) and t = p + 1, then we have

$$\psi_p + \frac{\phi_{p+1}}{q} [m]_{\frac{1}{q}} \neq 0 \quad \forall m \in \mathbb{N}. \tag{3.89}$$

We deduce that  $\langle A_{q,\omega}(o\mathcal{L}), Q_{m,1}Q_{m+s,1} \rangle \neq 0 \quad \forall m \geq 1$  and therefore that the property ii) is fulfilled

 $(ii) \Longrightarrow (ii)$ . We assume that (ii) holds and expand  $\phi D_{q,\omega}^* P_n$  on the basis  $\{P_n\}_{n\in\mathbb{N}}$ 

$$\phi D_{q,\omega}^* P_n \sum_{j=0}^{n+t} \xi_{n,j} P_j,$$

where  $t = \deg(\phi)$ , and get

$$\begin{array}{lcl} \xi_{n,j}I_{0,j} & = & \langle \phi\mathcal{L}, P_j \ D_{q,\omega}^*P_n \rangle \\ & = & \langle \phi\mathcal{L}, A_{q,\omega}^*(A_{q,\omega}P_j \ D_{q,\omega}P_n) \rangle \\ & = & q[n]_q \langle A_{q,\omega}(o\mathcal{L}), A_{q,\omega}P_j \ Q_{n-1,1} \rangle \\ & = & 0 \text{ for } n > j+s+1, \end{array}$$

by (3.83).

Moreover, for n > s + 1,

$$\xi_{n,n-s-1}I_{0,n-s-1} = q[n]_q \langle A_{q,\omega}(\phi \mathcal{L}), A_{q,\omega}P_{n-s-1}Q_{n-1,1} \rangle \neq 0$$

also by (3.83).

 $iii) \Longrightarrow i$ ).

Let  $\{\mathbf{P}_n\}_{n\in\mathbb{N}}$  be the dual basis associated to the monic family  $\{P_n\}_{n\in\mathbb{N}}$  and t the degree of  $\varphi$ . We expand the linear functional  $D_{q,\omega}(\phi\mathcal{L})$  in the basis  $\{\mathbf{P}_n\}_{n\in\mathbb{N}}$ 

$$D_{q,\omega}(\phi\mathcal{L}) = \sum_{n\geq 0} \alpha_n \, \mathbf{P}_n$$

and get

$$\begin{array}{rcl} \alpha_n & = & \langle D_{q,\omega}(\phi\mathcal{L}), P_n \rangle \\ & = & -\frac{1}{q} \langle \mathcal{L}, \phi D_{q,\omega}^* P_n \rangle \\ \\ & = & -\frac{1}{q} \langle \mathcal{L}, \sum_{j=n-s-1}^{s+t-1} \xi_{n,j} P_j \rangle \\ \\ & = & 0 \text{ for } n > s+1. \end{array}$$

Then

$$D_{q,\omega}(\phi \mathcal{L}) = \sum_{j=0}^{s+1} \alpha_j \mathbf{P}_j$$

$$= \sum_{j=0}^{s+1} \alpha_j \frac{P_j}{\langle \mathcal{L}, P_j P_j \rangle} \mathcal{L}$$

$$= \psi \mathcal{L}$$

thanks to Proposition 2.1. We deduce from the previous equations, Lemma 3.1 and the fact that  $\phi \neq 0$ , that  $\psi$  is of degree at least one. The linear functional  $\mathcal{L}$  is, therefore,  $D_{q,\omega}$ -semi-classical.

# Chapter 4

# The formal Stieltjes function

# 4.1 The Stieltjes function and the Riccati difference equation

#### 4.1.1 Some definitions

**Definition 4.1** The formal Stieltjes function  $S(\mathcal{L})$  of a given linear functional  $\mathcal{L} \in \mathbb{P}'$  is defined by

$$S(\mathcal{L})(x) = -\sum_{k>0} \frac{(\mathcal{L})_k}{x^{k+1}},\tag{4.1}$$

where  $(\mathcal{L})_k = \langle \mathcal{L}, x^k \rangle$ , represents the moment of order k of the linear functional  $\mathcal{L}$  with respect to the sequence  $\{x^n\}_{n\geq 0}$ .

We define the action of the operators  $\mathcal{D}$ ,  $\mathcal{T}_{\omega}$ ,  $\mathcal{D}_{\omega}$ ,  $\mathcal{G}_{q}$ ,  $\mathcal{D}_{q}$ ,  $\mathcal{D}$ ,  $A_{q,\omega}$  and  $D_{q,\omega}$  on the Stieltjes function  $S(\mathcal{L})$  as is done in [Medem, 1996] (for more information see [Medem, 1996, p. 357]).

**Definition 4.2** (Medem, 1996) The operators  $\mathcal{T}_{\omega}$ ,  $D_{\omega}$ ,  $\mathcal{G}_{q}$ ,  $\mathcal{D}_{q}$ ,  $\mathcal{D}_{q}$ ,  $\mathcal{A}_{q,\omega}$  and  $D_{q,\omega}$  act on the Stieltjes function  $S(\mathcal{L})$  of the linear functional  $\mathcal{L}$  in the following ways:

$$\begin{split} \mathcal{T}_{\omega}S(\mathcal{L})(x) &= S(\mathcal{L})(x+\omega) = -\sum_{n\geq 0} \frac{(\mathcal{L})_n}{(x+\omega)^{n+1}}, \\ \mathcal{G}_qS(\mathcal{L})(x) &= S(\mathcal{L})(q\,x) = -\sum_{n\geq 0} \frac{(\mathcal{L})_n}{q^{n+1}\,x^{n+1}}, \, q \neq 0. \\ A_{q,\omega}S(\mathcal{L})(x) &= S(\mathcal{L})(q\,x+\omega) - \sum_{n\geq 0} \frac{(\mathcal{L})_n}{(qx+\omega)^{n+1}}, \\ \mathcal{D}S(\mathcal{L})(x) &= \sum_{n\geq 0} (n+1)\frac{(\mathcal{L})_n}{x^{n+2}}, \\ \mathcal{D}_qS(\mathcal{L})(x) &= \frac{S(\mathcal{L})(qx) - S(\mathcal{L})(x)}{(q-1)x} = \sum_{n\geq 0} \frac{[n+1]_q\,(\mathcal{L})_n}{q^{n+1}\,x^{n+2}}. \\ \mathcal{D}_{\omega}S(\mathcal{L})(x) &= \frac{S(\mathcal{L})(x+\omega) - S(\mathcal{L})(x)}{\omega} = -\sum_{n\geq 0} (\mathcal{L})_n \frac{1}{\omega} \left(\frac{1}{(x+\omega)^{n+1}} - \frac{1}{x^{n+1}}\right), \\ \mathcal{D}_{q,\omega}S(\mathcal{L})(x) &= \frac{S(\mathcal{L})(q\,x+\omega) - S(\mathcal{L})(x)}{(q-1)\,x+\omega} = -\sum_{n\geq 0} (\mathcal{L})_n \frac{1}{(q-1)\,x+\omega} \left(\frac{1}{(q\,x+\omega)^{n+1}} - \frac{1}{x^{n+1}}\right). \end{split}$$

Definition 4.3 The formal Stieltjes function  $S(\mathcal{L}) = S$  (see (4.1)) of the regular linear functional  $\mathcal{L}$  satisfies a Riccati differential equation if S satisfies an equation of type [Magnus. 1984], [Dzoumba, 1985]

$$\phi(x)DS(x) = A(x)S(x)^{2} + B(x)S(x) + C(x), \tag{4.2}$$

where  $\phi$  is a non-zero polynomial and A, B and C are polynomials.

When A = 0, the Riccati differential equation is called the affine Riccati differential equation.

**Definition 4.4** The formal Stieltjes function,  $S(\mathcal{L}) = S$ , of the regular linear functional  $\mathcal{L}$  satisfies a  $D_{\omega}$ -Riccati difference equation if S satisfies an equation of type

$$\phi(x)D_{\omega}S(x) = G(x;\omega)S(x)\mathcal{T}_{\omega}S(x) + E(x;\omega)S(x) + F(x;\omega)\mathcal{T}_{\omega}S(x) + H(x;\omega), \tag{4.3}$$

where  $\phi$  is a non-zero polynomial and E, F, G and H are polynomials in the variable x and depending on  $\omega$ .

When G=0, the  $D_{\omega}$ -Riccati difference equation is called the affine  $D_{\omega}$ -Riccati difference equation.

**Definition 4.5** The formal Stieltjes function  $S(\mathcal{L}) = S$  of the regular linear functional  $\mathcal{L}$  satisfies a  $\mathcal{D}_q$ -Riccati difference equation if S satisfies an equation of type

$$\phi(x)\mathcal{D}_q S(x) = G(x;q)S(x)\mathcal{G}_q S(x) + E(x;q)S(x) + F(x;q)\mathcal{G}_q S(x) + H(x;q)$$

$$(4.4)$$

where  $\phi$  is a non-zero polynomial and E, F, G and H are polynomials in the variable x and depending on q.

When G = 0, the  $\mathcal{D}_q$ -Riccati difference equation is called the affine  $\mathcal{D}_q$ -Riccati difference equation.

**Definition 4.6** The formal Stieltjes function,  $S(\mathcal{L}) \equiv S$ , of the regular linear functional  $\mathcal{L}$  satisfies a  $D_{g,\omega}$ -Riccati difference equation if S satisfies an equation of type

$$\phi(qx + \omega)D_{q,\omega}S(x) = G(x;q,\omega)S(x)A_{q,\omega}S(x) + E(x;,q,\omega)S(x) + F(x;,q,\omega)A_{q,\omega}S(x) + H(x;q,\omega),$$

$$(4.5)$$

where  $\phi$  is a non-zero polynomial and E, F, G and H are polynomials in the variable x and depending on q and  $\omega$ .

When G = 0, the  $D_{q,\omega}$ -Riccati difference equation is called the affine  $D_{q,\omega}$ -Riccati difference equation.

**Definition 4.7** Let  $\mathcal{Y}$  be any one of the four operators  $\{\frac{d}{dx}, \mathcal{D}, \mathcal{D}_q, \mathcal{D}_{q,\omega}\}$ . Then, the regular linear functional  $\mathcal{L}$  and the corresponding monic orthogonal polynomials belong to the  $\mathcal{Y}$ -Laguerre-Hahn class (resp. affine  $\mathcal{Y}$ -Laguerre-Hahn class) if the Stieltjes function of  $\mathcal{L}$  satisfies a  $\mathcal{Y}$ -Riccati difference equation (an affine  $\mathcal{Y}$ -Riccati difference equation). The regular linear functional and the corresponding orthogonal polynomials belonging to the  $\mathcal{Y}$ -Laguerre-Hahn class are called  $\mathcal{Y}$ -Laguerre-Hahn linear functional and  $\mathcal{Y}$ -Laguerre-Hahn orthogonal polynomials, respectively (see [Magnus, 1984], [Dzoumba, 1985]. [Guerfi. 1988]. [Medem, 1995], [Salto, 1996]. [Marcellan et al., 1998]).

#### 4.1.2 Some properties

**Proposition 4.1** The formal Stieltjes function,  $S(\mathcal{L})$ , of a given linear functional  $\mathcal{L} \in \mathbb{P}'$  obeys the relations

$$i) S(\alpha \mathcal{L} + \beta \mathcal{M}) = \alpha S(\mathcal{L}) + \beta S(\mathcal{M}) \quad \forall \alpha, \beta \in \mathbb{C}, \quad \forall \mathcal{M} \in \mathbb{P}'$$

$$(4.6)$$

$$ii) S(f\mathcal{L}) = fS(\mathcal{L}) + \mathcal{L}\theta_0 f, \quad \forall f \in \mathbb{P}.$$
 (4.7)

$$iii) A_{q,\omega} S(\mathcal{L}) = S(A_{q,\omega} \mathcal{L}), \tag{4.8}$$

$$iv) S(D_{q,\omega}\mathcal{L}) = D_{q,\omega}S(\mathcal{L}). \tag{4.9}$$

*Proof:* i) Let  $\alpha, \beta$  be two complex numbers and  $\mathcal{L}$ .  $\mathcal{M}$  two linear functionals. Then,

$$S(\alpha \mathcal{L} + \beta \mathcal{M}) = -\sum_{k \geq 0} \frac{\langle \alpha \mathcal{L} + \beta \mathcal{M}, x^k \rangle}{x^{k+1}}$$
$$= -\alpha \sum_{k \geq 0} \frac{\langle \mathcal{L}, x^k \rangle}{x^{k+1}} - \beta \sum_{k \geq 0} \frac{\langle \mathcal{M}, x^k \rangle}{x^{k+1}}$$
$$= \alpha S(\mathcal{L}) + \beta S(\mathcal{M}).$$

ii) Let k be an integer; we shall prove that

$$S(x^k \mathcal{L}) = x^k S(\mathcal{L}) + \mathcal{L}\theta_0 x^k \quad \forall k \in \mathbb{N}$$

and use property i) to deduce that (4.8) holds for any  $f \in \mathbb{F}$ . In fact,

$$S(x^{k}\mathcal{L})(x) = -\sum_{n=0}^{\infty} \frac{\langle x^{k}\mathcal{L}, x^{n} \rangle}{x^{n+1}}$$

$$= -\sum_{n=0}^{\infty} \frac{\langle \mathcal{L}, x^{n+k} \rangle}{x^{n-1}}$$

$$= -\sum_{m=k}^{\infty} \frac{\langle \mathcal{L}, x^{m} \rangle}{x^{m-k+1}} \text{ (taking } k + n = m)$$

$$= -x^{k} \left(\sum_{m=0}^{\infty} \frac{\langle \mathcal{L}, x^{m} \rangle}{x^{m+1}} - \sum_{m=0}^{k-1} \frac{\langle \mathcal{L}, x^{m} \rangle}{x^{m+1}}\right)$$

$$= x^{k} S(\mathcal{L})(x) + \sum_{m=0}^{k-1} \langle \mathcal{L}, x^{m} \rangle x^{k-1-m}$$

$$= x^{k} S(\mathcal{L})(x) + \sum_{j=1}^{k} \langle \mathcal{L}, x^{k-j} \rangle x^{j-1} \text{ (taking } k - m = j)$$

$$= x^{k} S(\mathcal{L})(x) + \mathcal{L}\theta_{0}x^{k}.$$

For property iii),

$$S(A_{q,\omega}\mathcal{L})(x) = -\sum_{n=0}^{\infty} \frac{\langle A_{q,\omega}\mathcal{L}, x^n \rangle}{x^{n-1}}$$

$$= -\frac{1}{q} \sum_{n=0}^{\infty} \frac{\langle \mathcal{L}, A_{j,\omega}^* x^n \rangle}{x^{r+1}}$$

$$= -\sum_{n=0}^{\infty} \frac{\langle \mathcal{L}, (x-\omega)^n \rangle}{(qx)^{n+1}}$$

$$= -\sum_{n=0}^{\infty} \frac{1}{(qx)^{n+1}} \langle \mathcal{L}, \sum_{j=0}^n x^j \binom{n}{j} (-\omega)^{n-j} \rangle$$

$$= -\sum_{n=0}^{\infty} \sum_{j=0}^n \binom{n}{j} (-\omega)^{n-j} \frac{\langle \mathcal{L}, x^j \rangle}{(qx)^{n+1}},$$

where

$$\binom{n}{j} = \begin{cases} \frac{n!}{j!(n-j)!} & \text{if } j \le n, \\ 0 & \text{if } j > n. \end{cases}$$
 (4.10)

Then,

$$S(A_{q,\omega}\mathcal{L})(x) = -\sum_{n=0}^{\infty} \sum_{j=0}^{n} \binom{n}{j} (-\omega)^{n-j} \frac{\langle \mathcal{L}, x^j \rangle}{(qx)^{n+1}}.$$
 (4.11)

On the other hand, using the series expansion of  $\frac{1}{(gx+\omega)^{n+1}}$ ,

$$\frac{1}{(qx+\omega)^{n-1}} = \sum_{p=0}^{\infty} \frac{\binom{n+p}{n}(-\omega)^p}{(qx)^{n+1+p}},\tag{4.12}$$

we obtain

$$\begin{split} A_{q,\omega}S(\mathcal{L})(x) &= -\sum_{n=0}^{\infty} \frac{\langle \mathcal{L}, x^n \rangle}{(qx+\omega)^{n+1}} \\ &= -\sum_{n=0}^{\infty} \sum_{p=0}^{\infty} \frac{\binom{n+p}{n} (-\omega)^p \langle \mathcal{L}, x^n \rangle}{(qx)^{n+p+1}}. \end{split}$$

Then changing the variable n + p = j, the previous equation gives

$$A_{q,\omega}S(\mathcal{L})(x) = -\sum_{j=0}^{\infty} \sum_{n=0}^{j} {j \choose n} (-\omega)^{j-n} \frac{\langle \mathcal{L}, x^n \rangle}{(qx)^{j+1}}$$
$$= S(A_{q,\omega}\mathcal{L})(x),$$

by (4.11) after reversing the role of n and j.

To derive relation iv), we compute both sides of (4.9) and remark that they are the same. In fact,

$$S(D_{q,\omega}\mathcal{L})(x) = -\sum_{n=0}^{\infty} \frac{\langle D_{q,\omega}\mathcal{L}, x^n \rangle}{x^{n+1}}$$

$$= \frac{1}{q} \sum_{n=1}^{\infty} \frac{\langle \mathcal{L}, D_{q,\omega}^* x^n \rangle}{x^{n+1}}$$

$$= \frac{1}{q} \sum_{n=1}^{\infty} \frac{1}{x^{n+1}} \langle \mathcal{L}, \frac{(\frac{x-\omega}{q})^n - x^n}{\frac{x-\omega}{q} - x} \rangle.$$

Then, using the relations (derived by induction on n),

$$(a+b)^n = \sum_{j=0}^n \binom{n}{j} a^j b^{n-j} \quad \forall a, b \in \mathbb{C} \quad \forall n \in \mathbb{N},$$

$$(4.13)$$

$$a^{n}-b^{n} = (a-b)\sum_{j=0}^{n-1} a^{j} b^{n-1-j} \quad \forall a,b \in \mathbb{C} \quad \forall n \in \mathbb{N},$$

$$(4.14)$$

we obtain

$$S(D_{q,\omega}\mathcal{L})(x) = \frac{1}{q} \sum_{n=1}^{\infty} \frac{1}{x^{n-1}} \sum_{k=0}^{n-1} \sum_{j=0}^{k} {k \choose j} (-\omega)^{k-j} q^{-k} \langle \mathcal{L}, x^{n+j-1-k} \rangle,$$

On replacing n by n+1, we obtain,

$$S(D_{q,\omega}\mathcal{L})(x) = \sum_{n=0}^{\infty} \frac{1}{x^{n+2}} \sum_{k=0}^{n} \sum_{j=0}^{k} {k \choose j} (-\omega)^{k-j} q^{-k-1} \langle \mathcal{L}, x^{r,+j-k} \rangle.$$
 (4.15)

Let us compute  $D_{q,\omega}S(\mathcal{L})(x)$ .

$$D_{q,\omega}S(\mathcal{L})(x) = -\sum_{n=0}^{\infty} \langle \mathcal{L}, x^n \rangle D_{q,\omega} \frac{1}{x^{n+1}}$$

$$= \sum_{n=0}^{\infty} \langle \mathcal{L}, x^n \rangle \frac{D_{q,\omega} x^{n+1}}{x^{n+1} (qx+\omega)^{n+1}}$$

$$= \sum_{n=0}^{\infty} \frac{(qx+\omega)^{n+1} - x^{n+1}}{qx+\omega - x} \frac{\langle \mathcal{L}, x^n \rangle}{x^{n+1} (qx+\omega)^{n+1}}.$$

Use of (4.14) transforms the previous relation into

$$D_{q,\omega}S(\mathcal{L})(x) = \sum_{n=0}^{\infty} \frac{\langle \mathcal{L}, x^n \rangle}{(qx+\omega)^{n+1} x^{n+1}} \sum_{j=0}^{n} (qx+\omega)^j x^{n-j}$$
$$= \sum_{n=0}^{\infty} \sum_{j=0}^{n} \frac{\langle \mathcal{L}, x^n \rangle}{(qx+\omega)^{n-j+1} x^{j+1}}.$$

Rewriting this equation, taking into account the series expansion of  $\frac{1}{(az+\omega)^{n-j+1}}$  (see (4.12)), leads to

$$D_{q,\omega}S(\mathcal{L})(x) = \sum_{n=0}^{\infty} \sum_{j=0}^{n} \frac{\langle \mathcal{L}, x^{n} \rangle}{x^{j+1}} \sum_{k=0}^{\infty} \frac{(n+k-j)! (-\omega)^{k}}{k! (n-j)! (qx)^{n+k-j+1}}$$
$$= \sum_{n=0}^{\infty} \sum_{j=0}^{n} \sum_{k=0}^{\infty} \frac{(n+k-j)! (-\omega)^{k} \langle \mathcal{L}, x^{n} \rangle}{k! (n-j)! x^{n+k+2} q^{n+k-j+1}}.$$

The change of variable, n + k = t. transforms the latter equation into

$$D_{q,\omega}S(\mathcal{L})(x) = \sum_{t=0}^{\infty} \sum_{k=0}^{t} \sum_{j=0}^{t-k} \frac{(t-j)! (-\omega)^{k} \langle \mathcal{L}, x^{t-k} \rangle}{k! (t-k-j)! x^{t-2} q^{t-j+1}}.$$

We obtain after replacing in the previous equation t - j by m that

$$\begin{split} D_{q} \, \, \cup \, S(\mathcal{L})(x) &= \sum_{t=0}^{\infty} \sum_{k=0}^{t} \sum_{m=k}^{t} \frac{m! \, (-\omega)^k \, \langle \mathcal{L}, x^{t-k} \rangle}{k! \, (m-k)! \, x^{t-2} \, q^{m+1}} \\ &= \sum_{t=0}^{\infty} \sum_{k=0}^{t} \sum_{m=k}^{t} \frac{\binom{m}{k} \, (-\omega)^k \, \langle \mathcal{L}, x^{t-k} \rangle}{x^{t+2} \, q^{m+1}} \\ &= \sum_{t=0}^{\infty} \sum_{k=0}^{t} \sum_{m=0}^{t} \frac{\binom{m}{k} \, (-\omega)^k \, \langle \mathcal{L}, x^{t-k} \rangle}{x^{t+2} \, q^{m+1}} \\ &= \sum_{t=0}^{\infty} \sum_{m=0}^{t} \sum_{k=0}^{m} \frac{\binom{m}{k} \, (-\omega)^k \, \langle \mathcal{L}, x^{t-k} \rangle}{x^{t+2} \, q^{m+1}}. \end{split}$$

Again, replacing m - k by l in the above equation, we obtain

$$\begin{split} D_{q,\omega}S(\mathcal{L})(x) &= \sum_{t=0}^{\infty} \sum_{m=0}^{t} \sum_{l=0}^{m} \frac{\binom{m}{m-l} (-\omega)^{m-l} \langle \mathcal{L}, x^{t+l-m} \rangle}{x^{t+2} \, q^{m+1}} \\ &= \sum_{t=0}^{\infty} \sum_{m=0}^{t} \sum_{l=0}^{m} \frac{\binom{m}{l} (-\omega)^{m-l} / \mathcal{L}. x^{t+l-m} \rangle}{x^{t+2} \, q^{m+1}} \\ &= S(D_{q,\omega} \mathcal{L})(x), \end{split}$$

**Corollary 4.1** The formal Stieltjes function  $S(\mathcal{L})$  of a given linear functional  $\mathcal{L} \in \mathbb{P}'$  obeys the relations:

$$\mathcal{T}_{\omega} S(\mathcal{L}) = S(\mathcal{T}_{\omega} \mathcal{L}), \ S(D_{\omega} \mathcal{L}) = D_{\omega} S(\mathcal{L}),$$
 
$$\mathcal{G}_{q} S(\mathcal{L}) = S(\mathcal{G}_{q} \mathcal{L}), \ S(\mathcal{D}_{q} \mathcal{L}) = \mathcal{D}_{q} S(\mathcal{L}).$$

We announce another corollary of Proposition 4.1. This result has been given for the operators  $\mathcal{D}$ ,  $\mathcal{D}_q$  and  $D_{\omega}$  (see [Dzoumba, 1985], [Guerfi, 1988], [Medem. 1996]).

**Theorem 4.1** Let  $\mathcal{L}$  be a regular linear functional.  $\mathcal{L}$  belongs to the affine  $D_{q,\omega}$ -Laguerre-Hahn class if and only if  $\mathcal{L}$  is  $D_{q,\omega}$ -semi-classical.

*Proof:* Suppose that  $\mathcal{L}$  is  $D_{q,\omega}$ -semi-classical and satisfies  $D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}$ , where  $\phi$  is a non-zero polynomial and  $\psi$  a polynomial of degree at least one. We first use Propositions 3.1 and 4.1 to compute  $S(D_{q,\omega}(\phi\mathcal{L}))$  and obtain

$$\begin{split} S(D_{q,\omega}(\phi\mathcal{L})) &= S(A_{q,\omega}\phi D_{q,\omega}\mathcal{L} + D_{q,\omega}\phi\mathcal{L}) \\ &= A_{q,\omega}\phi S(D_{q,\omega}\mathcal{L}) + (D_{q,\omega}\mathcal{L})\theta_0 A_{q,\omega}\phi + D_{q,\omega}\phi S(\mathcal{L}) + \mathcal{L}\theta_0 D_{q,\omega}\phi \\ &= A_{q,\omega}\phi D_{q,\omega}S(\mathcal{L}) + (D_{q,\omega}\mathcal{L})\theta_0 A_{q,\omega}\phi + D_{q,\omega}\phi S(\mathcal{L}) + \mathcal{L}\theta_0 D_{q,\omega}\phi. \end{split}$$

Secondly, we use again Proposition 4.1 to compute  $S(\iota \cdot \mathcal{L})$  and we obtain

$$S(\psi \mathcal{L}) = \psi S(\mathcal{L}) + \mathcal{L}\theta_0 \psi.$$

Since  $D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}$  and  $\phi$  is a non-zero polynomial, we deduce from the above computations that  $S(\mathcal{L})$  satisfies the affine  $D_{q,\omega}$ -Riccati difference equation

$$\phi(qx+\omega)D_{\pm,\omega}S(\mathcal{L})(x) = (\psi(x) - D_{q,\omega}\phi(x))S(\mathcal{L})(x) + \mathcal{L}\theta_0\psi(x) - (D_{q,\omega}\mathcal{L})\theta_0\phi(qx+\omega) - \mathcal{L}\theta_0D_{q,\omega}\phi(x).$$

Thus,  $\mathcal{L}$  belongs to the affine  $D_{q,\omega}$ -Laguerre-Hahn class.

Conversely, assume that the Stieltjes function  $S(\mathcal{L})$  of the regular linear functional  $\mathcal{L}$  satisfies an affine  $D_{q,\omega}$ -Riccati difference equation

$$A(x)D_{q,\omega}S(\mathcal{L})(x) = B(x)S(\mathcal{L})(x + C(x)).$$

where B and C are any polynomials and A is a non-zero polynomial. Using Propositions 3.1 and 4.1 we obtain

The right hand-side of the previous equation is a polynomial while the left hand-side is, by definition of the Stieltjes function of a given linear functional, an infinite (unless it vanishes) linear combination of  $\{\frac{1}{n+1}, n \in \mathbb{N}\}$ . Therefore, both sides of the previous equation vanish and we obtain

$$A(x)D_{q,\omega}\mathcal{L} - B(x)\mathcal{L} = 0 \tag{4.16}$$

and

$$(L_{1,\omega}\mathcal{L})\theta_0 A(x) - \mathcal{L}\theta_0 B(x) + C(x) = 0 \quad \forall x \in \mathbb{R}.$$
(4.17)

Again, we use Proposition 3.1 to deduce that (4.16) is equivalent to

$$D_{q,\omega}(A_{q,\omega}^{\bullet}A\mathcal{L}) = (B - \frac{1}{q}D_{q,\omega}^{\bullet}A)\mathcal{L}. \tag{4.18}$$

The previous equation, used together with Lemma 3.1 taking into account the fact that  $A \neq 0$ , allows us to conclude that the degree of  $B + \frac{1}{q} D_{q,\omega}^{\star} A$  is at least one. Then the regular linear functional  $\mathcal{L}$  is  $D_{q,\omega}$ -semi-classical.

# 4.2 $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials as $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials

In this section we prove that the  $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials can be deduced from  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials by a change of variable and then we give some consequences.

**Theorem 4.2** Let  $\mathcal{L}$  be any regular linear functional, then we have:

i)  $\mathcal{L}$  is a  $D_{q,\omega}$ -Laguerre-Hahn linear functional if and only if  $A_{a,\frac{\omega}{1-q}}\mathcal{L}$  is a  $\mathcal{D}_q$ -Laguerre-Hahn linear functional. This means that the Stieltjes function  $S(\mathcal{L})$  of  $\mathcal{L}$  satisfies

$$\phi(qx+\omega)D_{q,\omega}S(\mathcal{L})(x) = G(x;q,\omega)S(\mathcal{L})(x)A_{q,\omega}S(\mathcal{L})(x) + E(x;q,\omega)S(\mathcal{L})(x) 
+F(x;q,\omega)A_{q,\omega}S(\mathcal{L})(x) + H(x;q,\omega),$$
(4.19)

where a is any non-zero real number,  $\phi$  is a non-zero polynomial, E, F, G and H are polynomials in the variable x, if and only if the Stieltjes function  $S(A_a, \frac{\omega}{1-\sigma}\mathcal{L})$  of  $A_a, \frac{\omega}{1-\sigma}\mathcal{L}$  satisfies

$$\tilde{\phi}(qx)\mathcal{D}_{q}S(\tilde{\mathcal{L}})(x) = \tilde{G}(x;q,\omega)S(\tilde{\mathcal{L}})(x)\mathcal{G}_{q}S(\tilde{\mathcal{L}})(x) + \tilde{E}(x;q,\omega)S(\tilde{\mathcal{L}})(x) + \tilde{F}(x;q,\omega)\mathcal{G}_{q}S(\tilde{\mathcal{L}})(x) + \tilde{H}(x;q,\omega),$$
(4.20)

where  $\tilde{c} = a^{-1} A_{a, \frac{\omega}{1-a}} \phi$ .  $\tilde{\Phi} = A_{a, \frac{\omega}{1-a}} \Phi$  for  $\Phi \in \{\mathcal{L}, E, F, G, H\}$ .

ii) Let  $\mathcal{L}$  be a  $D_{q,\omega}$ -Laguerre-Hahn linear functional satisfying (4.19). If  $\{P_n\}_{n\in\mathbb{N}}$  and  $\{\tilde{P}_n\}_{n\in\mathbb{N}}$  represent the monic orthogonal families associated to  $\mathcal{L}$  and  $A_{a,\frac{\omega}{1-q}}\mathcal{L}$ , respectively, then we have the following results:

$$\tilde{P}_{n}(x) = a^{-n} P_{n}(ax + \frac{\omega}{1 - q}) \quad \forall x \in \mathbb{R},$$

$$\beta_{n}(q, \omega, \phi, E, F, G, H) = a \tilde{\beta}_{n}(q, 0, \tilde{\phi}, \tilde{E}, \tilde{F}, \tilde{G}, \tilde{H}) + \frac{\omega}{1 - q},$$

$$\gamma_{n}(q, \omega, \phi, E, F, G, H) = a^{2} \tilde{\gamma}_{n}(q, 0, \tilde{\phi}, \tilde{E}, \tilde{F}, \tilde{G}, \tilde{H}),$$

$$(4.21)$$

where  $\beta_n$ ,  $\gamma_n$ ,  $\tilde{\beta}_n$  and  $\tilde{\gamma}_n$  are coefficients of the three-term recurrence relation satisfied by  $\{P_n\}_{n\in\mathbb{N}}$  and  $\{\tilde{P}_n\}_{n\in\mathbb{N}}$ :

$$\begin{cases}
P_{n+1}(x) = (x - \beta_n(q, \omega, \zeta, E, F, G, H))P_n(x) - \gamma_n(q, \omega, \phi, E, F, G, H)P_{n-1}(x), & n \ge 0, \\
\tilde{P}_{n+1}(x) = (x - \tilde{\beta}_n(q, 0, \tilde{\phi}, \tilde{E}, \tilde{F}, \tilde{G}, \tilde{H})\tilde{P}_n(x) - \tilde{\gamma}_n(q, 0, \tilde{\phi}, \tilde{E}, \tilde{F}, \tilde{G}, \tilde{H})\tilde{P}_{n-1}(x), & n \ge 0, \\
P_{-1}(x) = 0, P_0(x) = 1, & \tilde{P}_{-1}(x) = 0, \tilde{P}_0(x) = 1.
\end{cases} (4.22)$$

Proof: i) We use the relation [Guerfi, 1988], [Medem. 1996]

$$A_{a,\frac{\omega}{1-q}} D_{q,\omega} = a^{-1} \mathcal{D}_q A_{a,\frac{\omega}{1-q}}, A_{a,\frac{\omega}{1-q}} A_{q,\omega} = \mathcal{G}_q A_{a,\frac{\omega}{1-q}}, q \neq 1. \ a \neq 0$$
 (4.23)

and get

$$\begin{split} & \phi(qx+\omega)D_{q,\omega}S(\mathcal{L})(x) = G(x;q,\omega)S(\mathcal{L})(x)A_{q,\omega}S(\mathcal{L})(x) + E(x;q,\omega)S(\mathcal{L})(x) \\ & + F(x;q,\omega)A_{q,\omega}S(\mathcal{L})(x) + H(x;q,\omega) \\ & \Longleftrightarrow \\ & A_{d,\frac{\pi}{1-q}}A_{q,\omega}\phi(x)A_{a,\frac{\pi}{1-q}}D_{q,\omega}S(\mathcal{L})(x) = \tilde{G}(x;q,\omega)A_{a,\frac{\pi}{1-q}}S(\mathcal{L})(x)A_{a,\frac{\pi}{1-q}}A_{q,\omega}S(\mathcal{L})(x) \\ & + \tilde{E}(x;q,\omega)A_{a,\frac{\pi}{1-q}}S(\mathcal{L})(x) + \tilde{F}(x;q,\omega)A_{a,\frac{\pi}{1-q}}A_{q,\omega}S(\mathcal{L})(x) + \tilde{H}(x;q,\omega) \\ & \Longleftrightarrow \\ & a^{-1}\mathcal{G}_{q}A_{a,\frac{\pi}{1-q}}\phi(x)\mathcal{D}_{q}A_{a,\frac{\pi}{1-q}}S(\mathcal{L})(x) = \tilde{G}(x;q,\omega)A_{a,\frac{\pi}{1-q}}S(\mathcal{L})(x)\mathcal{G}_{q}A_{a,\frac{\pi}{1-q}}S(\mathcal{L})(x) \\ & + \tilde{E}(x;q,\omega)A_{a,\frac{\pi}{1-q}}S(\mathcal{L})(x) + \tilde{F}(x;q,\omega)\mathcal{G}_{q}A_{a,\frac{\pi}{1-q}}S(\mathcal{L})(x) + \tilde{H}(x;q,\omega) \\ & \Longleftrightarrow \\ & \tilde{\phi}(qx)\mathcal{D}_{q}S(\tilde{\mathcal{L}})(x) = \tilde{G}(x;q,\omega)S(\tilde{\mathcal{L}})(x)\mathcal{G}_{q}S(\tilde{\mathcal{L}})(x) + \tilde{E}(x;q,\omega)S(\tilde{\mathcal{L}})(x) \\ & + \tilde{F}(x;q,\omega)\mathcal{G}_{q}S(\tilde{\mathcal{L}})(x) + \tilde{H}(x;q,\omega), \end{split}$$

by the relation (4.8):  $S(A_{q,\omega} \mathcal{L}) = A_{q,\omega} S(\mathcal{L})$ .

ii) Since the family  $\{A_{a,\frac{\omega}{1-q}}P_n\}_{n\in\mathbb{N}}$  is orthogonal with respect  $A_{a,\frac{\omega}{1-q}}\mathcal{L}$  (see Lemma 3.2), we deduce that  $\tilde{P}_n=a^{-n}A_{a,\frac{\omega}{1-q}}P_n$ , thanks to the uniqueness of the monic orthogonal polynomial family associated to a given regular linear functional.

Since  $\{P_n\}_{n\in\mathbb{N}}$  is orthogonal with respect to the linear functional  $\mathcal{L}$ , it satisfies

$$P_{n+1}(x) = (x - \beta_n(q, \omega, \phi, E, F, G, H))P_n(x) - \gamma_n(q, \omega, \phi, E, F, G, H)P_{n-1}(x),$$

where  $\beta_n(q, \omega, \phi, E, F, G, H)$ ,  $\gamma_n(q, \omega, \phi, E, F, G, H)$  are complex numbers depending on q,  $\omega$ ,  $\phi$ , E, F, G and H.

After applying the operator  $A_{a,\frac{\pi}{1-a}}$  to both sides of the previous equation, we obtain that

$$A_{a,\frac{\omega}{1-q}}P_{n+1}(x)=(a\,x+\frac{\omega}{1-q}-\beta_n(q,\omega,\phi,\psi))A_{a,\frac{\omega}{1-q}}P_n(x)-\gamma_n(q,\omega,\phi,\psi)A_{a,\frac{\omega}{1-q}}P_{n-1}(x).$$

This latter equation, used together with (4.21), gives

$$\tilde{P}_{n+1}(x) = \left(x + \frac{\omega}{1 - q^{\frac{1}{2}}a} - \frac{1}{a}\beta_n(q,\omega,\phi,\psi)\right)\tilde{P}_n(x) - \frac{1}{a^2}\gamma_n(q,\omega,\phi,\psi)\tilde{P}_{n-1}(x).$$

We complete the proof of the theorem by identifying the coefficients of the previous equation with the ones of the three-term recurrence relation satisfied by family  $\{\tilde{P}_n\}_{n\in\mathbb{N}}$ , orthogonal with respect to the  $\mathcal{D}_q$ -semi-classical linear functional  $A_{2,\frac{\omega}{1-\alpha}}\mathcal{L}$ 

$$\tilde{P}_{n+1}(x) = (x - \tilde{\beta}_n(q, 0, \tilde{\phi}, \tilde{\psi})) \tilde{P}_n(x) - \tilde{\gamma}_n(q, 0, \tilde{\phi}, \tilde{\psi}) \tilde{P}_{n-1}(x), \ n \ge 0,$$

with  $\tilde{\phi}$  and  $\tilde{\psi}$  defined by (4.24).

**Remark 4.1** Since the results stated in Theorem 4.2 are valid for any real number  $a \neq 0$ , without loss of generality, we choose a=1. In this case  $A_{1,\frac{\omega}{1-q}}=\mathcal{T}_{\frac{\omega}{1-q}}$  and we, therefore, get the following consequences:

Corollary 4.2 Let  $\mathcal{L}$  be any regular linear functional,  $\{P_n\}_{n\in\mathbb{N}}$  and  $\{\tilde{P}_n\}_{n\in\mathbb{N}}$  represent the orthogonal families associated to  $\mathcal{L}$  and  $\mathcal{T}_{\frac{n}{n-q}}\mathcal{L}$ , respectively. Then, we have the following results:

1.  $\mathcal{L}$  is  $D_{q,\omega}$ -semi-classical if and only if  $\mathcal{T}_{\frac{\omega}{1-\alpha}}\mathcal{L}$  is  $\mathcal{D}_q$ -semi-classical, i.e.,

$$D_{g,\omega}(\phi \mathcal{L}) = \psi \iff \mathcal{D}_g(\tilde{\phi}\,\tilde{\mathcal{L}}) = \tilde{\psi}\,\tilde{\mathcal{L}}.$$

where  $\phi$  is any polynomial and  $\psi$  a polynomial of degree at least one, with

$$\tilde{\mathcal{L}} = \mathcal{T}_{\frac{\omega}{1-q}} \mathcal{L}, \ \tilde{\phi}(x) = \phi(x + \frac{\omega}{1-q}). \ \tilde{\psi}(x) = \psi(x + \frac{\omega}{1-q}). \tag{4.24}$$

2. The coefficients of the TTRR satisfied by  $\{P_n\}_{n\in\mathbb{N}}$  and  $\{\tilde{P}_n\}_{n\in\mathbb{N}}$  are related by

$$\begin{split} &\beta_n(q,\omega,\phi,\psi) = \tilde{\beta}_n(q,0,\tilde{\phi},\tilde{\psi}) - \frac{\omega}{1-q}, \\ &\gamma_n(q,\omega,\phi,\psi) = \tilde{\gamma}_n(q,0,\tilde{\phi},\tilde{\psi}), \end{split}$$

where  $\beta_n$ ,  $\gamma_n$ ,  $\tilde{\beta}_n$  and  $\tilde{\gamma}_{-}$  are coefficients of the three-term recurrence relation

$$\left\{ \begin{array}{l} P_{n+1}(x) = (x - \beta_n(q, \omega, \phi, \psi)) P_n(x) - \gamma_n(q, \omega, \phi, \psi) P_{n-1}(x), \ n \geq 0, \\ \tilde{P}_{n+1}(x) = (x - \tilde{\beta}_n(q, 0, \tilde{\phi}, \tilde{\psi})) \tilde{P}_n(x) - \tilde{\gamma}_n(q, 0, \tilde{\phi}, \tilde{\psi}) \tilde{P}_{n-1}(x), \ n \geq 0, \\ P_{-1}(x) = 0, P_0(x) = 1, \ \tilde{P}_{-1}(x) = 0, \tilde{P}_0(x) = 1 \end{array} \right.$$

and  $\tilde{\phi}$ ,  $\tilde{\psi}$  given by (4.24).

*Proof:* The proof is similar to the one given for  $D_{q,\omega}$ -Laguerre-Hahn case. In particular, we have,

$$\begin{split} D_{q,\omega}(\phi\,\mathcal{L}) &= \psi \quad \iff \quad \mathcal{T}_{\frac{\omega}{1-q}}\,D_{q,\omega}(\phi\,\mathcal{L}) = \mathcal{T}_{\frac{\omega}{1-q}}\,\psi\,\mathcal{T}_{\frac{\omega}{1-q}}\,\mathcal{L} \\ &\iff \quad \mathcal{D}_q\,\mathcal{T}_{\frac{\omega}{1-q}}\,(\phi\,\mathcal{L}) = \mathcal{T}_{\frac{\omega}{1-q}}\,\psi\,\mathcal{T}_{\frac{\omega}{1-q}}\,\mathcal{L} \\ &\iff \quad \mathcal{D}_q(\mathcal{T}_{\frac{\omega}{1-q}}\,\phi\,\mathcal{T}_{\frac{\omega}{1-q}}\,\mathcal{L}) = \mathcal{T}_{\frac{\omega}{1-q}}\,\psi\,\mathcal{T}_{\frac{\omega}{1-q}}\,\mathcal{L}. \end{split}$$

# Chapter 5

# Difference equations for the first associated classical orthogonal polynomials

#### 5.1 Introduction

In this chapter we derive the single fourth order difference equation satisfied by the first associated of the q-classical orthogonal polynomials. We give this equation in the factored and simple form, we then use Theorem 4.2 to deduce the single fourth order difference (resp. differential) equation satisfied by the first associated of the classical orthogonal polynomials of a discrete variable and continuous variable, respectively.

Although the main result of this section is contained in the general theory given in the next chapter, this method is worth to be communicated because it uses the properties of the functions of a discrete variable of the second kind [Suslov, 1989] rather than the properties of the Stieltjes function which are used in the next chapter. It also allows us to have a factored and simple form for the fourth-order difference equation and to confirm the results obtained by the general theory.

# 5.2 q-classical weight

Let  $\rho(x)$  be a positive weight function defined on the interval I = ]a, b[ and let  $\mathcal{L}$  be a linear functional defined by

$$\langle \mathcal{L}, P \rangle = \int_{I} P(s) \, \rho(s) d_{q} s.$$
 (5.1)

The orthogonality weight  $\rho$  (defined in the interval I) is said to be q-classical if  $\rho$  satisfies:

i) There exists a monic polynomial family  $\{P_n\}_{n\in\mathbb{N}}$ , orthogonal with respect to  $\rho$ , i.e.,

$$\int_{I} P_{n}(s) P_{m}(s) \rho(s) d_{q}s = k_{n} \delta_{n,m} \quad \forall n, m \in \mathbb{N}. \quad (k_{n} \neq 0 \quad \forall n \in \mathbb{N}).$$
 (5.2)

ii) There exist two polynomials  $\phi$  of degree at most two and  $\tau$  of degree one such that

$$\mathcal{D}_{q}\left(\phi\,\rho\right) = \psi\,\rho,\tag{5.3}$$

with

$$|x^n \phi(x) \rho(x)|_a^b = 0 \quad \forall n \in \mathbb{N}.$$
 (5.4)

**Lemma 5.1** The linear functional  $\mathcal{L}$  represented by the q-classical weight  $\rho$  (see (5.1)) is  $\mathcal{D}_q$ -classical and satisfies

$$\mathcal{D}_q(\phi \,\mathcal{L}) = \psi \,\mathcal{L}.\tag{5.5}$$

*Proof:* If P is any element of  $\mathbb{P}$ , we use (5.1)-(5.4) and get.

$$\begin{split} \langle \mathcal{D}_{q}(\phi\,\mathcal{L}), P \rangle &= -\frac{1}{q} \left\langle \phi\,\mathcal{L}, \mathcal{D}_{\frac{1}{q}}\,P \right\rangle \\ &= -\frac{1}{q} \int_{I} \phi(s) \mathcal{D}_{\frac{1}{q}}\,P(s)\,\rho(s) d_{q}s \\ &= -\frac{1}{q} \int_{I} \left( \mathcal{D}_{\frac{1}{q}}(\phi | qs)\,\rho(qs)\,P(s) \right) - \mathcal{D}_{\frac{1}{q}}(\phi(qs)\rho(qs))\,P(s) \right) d_{q}s \\ &= -\int_{I} \left( \mathcal{D}_{q}(\phi(s | \rho(s)\,P(s/q)) + \mathcal{D}_{q}(\phi(s)\rho(s))\,P(s) \right) d_{q}s \\ &= -\phi(s)\,\rho(s)\,P(s/q)|_{a}^{b} + \int_{I} \psi(s)\rho(s)\,P(s) d_{q}s \\ &= \int_{I} \psi(s)\rho(s)\,P(s) d_{q}s \\ &= \langle \psi\mathcal{L}, P \rangle. \end{split}$$

Hence,  $\mathcal{D}_q(\mathcal{OL}) = \psi \mathcal{L}$ . We complete the proof by remarking that  $\{P_n\}_{n \in \mathbb{N}}$  is orthogonal with respect to  $\mathcal{L}$  (see (5.2)).

The monic polynomials  $\{P_n\}_{n\in\mathbb{N}}$ , orthogonal with respect to  $\mathcal{L}$ , satisfy the second order q-difference equation (see Theorem 3.1),

$$Q_{2,n}[y(x)] \equiv \left[\phi \mathcal{D}_q \,\mathcal{D}_{\frac{1}{q}} + \psi \,\mathcal{D}_q + \lambda_{n,0} \mathcal{I}_d\right] y(x) = 0, \tag{5.6}$$

an equation which can be written in the q-shifted form,

$$[(\phi_{(1)} + \psi_{(1)} t_1)\mathcal{G}_q^2 - ((1+q)\phi_{(1)} + \psi_{(1)} t_1 - \lambda_{n,0} t_1^2)\mathcal{G}_q + q\phi_{(1)} \mathcal{I}_d]y(x) = 0.$$
 (5.7)

with

$$\lambda_{n,0} = -[n]_q \{ \psi' + [n-1]_{\frac{1}{q}} \frac{\phi''}{2q} \},$$

$$\phi_{(i)} \equiv \phi(q^i x), \ \psi_{(i)} \equiv \psi(q^i x), \ t_i \equiv t(q^i x), \ t_i x) = (q-1)x.$$
(5.8)

# 5.3 Fourth-order q-difference equation for $P_{n-1}^{(1)}(x;q)$

The first associated  $P_{n-1}^{(1)}(x;q)$  of  $P_{n-1}(x;q)$  is a monic polynomial of degree n-1 defined by

$$P_{n-1}^{(1)}(x;q) = \frac{1}{\gamma_0} \langle \mathcal{L}, \frac{P_n(s;q) - P_n(x;q)}{s - x} \rangle = \frac{1}{\gamma_0} \int_I \frac{P_n(s;q) - P_n(x;q)}{s - x} \rho(s) d_q s, \tag{5.9}$$

where  $\gamma_0$  is given by  $\gamma_0 = (\mathcal{L}, 1) = \int_I \rho(s) d_q s$ .

Relation (5.9) can be rewritten as

$$P_{n-1}^{(1)}(x;q) = \rho(x) Q_n(x;q) - P_n(x;q)\rho(x) Q_0(x;q), \tag{5.10}$$

where

$$Q_n(x;q) = \frac{1}{\gamma_0 \rho(x)} \int_I \frac{P_n(s;q)}{s-x} \rho(s) d_s s.$$

It is well-known [Suslov, 1989] that  $Q_n(x;q)$  also satisfies Equation (5.6); hence, by (5.10)

$$Q_{2,n} \left[ \frac{P_{n-1}^{(1)}(x;q)}{\rho(x)} + P_n(x;q)Q_0(x;q) \right] = 0.$$
 (5.11)

In a first step, we eliminate  $\rho(x)$  and  $Q_0(x;q)$  in Equation (5.11) using Equation (5.3) and Equation (5.6) for  $P_n(x;q)$ . This can be easily carried out using a computer algebra system—we used Maple V Release 4 [Char et al., 1991]—and gives the relation

$$(\phi_{(1)} + \psi_{(1)} t_1) \mathcal{Q}_{2,n-1}^* \left[ P_{n-1}^{(1)}(x;q) \right] = [e \mathcal{G}_q + f \mathcal{I}_d] P_n(x;q), \tag{5.12}$$

with

$$Q_{2,n-1}^{\star} = \phi_{2} \mathcal{G}_{q}^{2} - ((1+q)\phi_{(1)} + \psi_{(1)} t_{1} - \lambda_{n,0} t_{1}^{2}) \mathcal{G}_{q} + q (\phi + \psi t) \mathcal{I}_{d}.$$

$$e = (\frac{\phi''}{2} - \psi') ((1+q)\phi_{(1)} + \psi_{(1)} t_{1} - \lambda_{n,0} t_{1}^{2}) t_{1},$$

$$f = -(\frac{\phi''}{2} - \psi') ((q+1)\phi_{(1)} + \psi_{(1)} t_{1}) t_{1}.$$

$$(5.13)$$

In a second step, we use Equations (5.12), (5.13) and the fact that the polynomials  $P_n(x;q)$  satisfy Equation (5.6), again. This gives—after some computations with Maple V.4—the operator  $\mathcal{Q}_{2,n-1}^{\star\star}$ annihilating the right-hand side of Equation (5.12),

$$Q_{2.n-1}^{*-} = (\phi_{(3)} + \psi_{(3)} t_3)[q^2 A_1 + (1+q) \phi_{(2)} + \psi_{(2)} t_2] \mathcal{G}_q^2 -[q^3 A_1 (\phi_{(2)} + \psi_{(2)} t_2) + A_3 (\phi_{(2)} + q A_1)] \mathcal{G}_q +q \phi_{(1)} [q^2 A_2 + (1+q) \phi_{(3)} + \psi_{(3)} t_3)] \mathcal{I}_d,$$
(5.14)

where  $A_j = (1+q)\phi_{(j)} + \psi_{(j)} t_j - \lambda_{n,0} t_j^2$ . We, therefore, obtain the factored form of the fourth-order q-difference equation satisfied by each  $P_{n-1}^{(1)}(x;q),$ 

$$Q_{2,n-1}^{\star\star} \frac{Q_{2,n-1}^{\star}}{q^2 (q-1)^2 x^2} \left[ P_{n-1}^{(1)}(x;q) \right] = 0.$$
 (5.15)

#### 5.4Applications

#### The first associated Little and Big q-Jacobi polynomials

For the Little q-Jacobi polynomials,  $p_n(x; a, b|q)$  [Area et al., 1998a], [Koekoek et. al., 1996]

$$\phi(x) = \frac{x(x-1)}{q}, \ \psi(x) = \frac{1-a\,q+(abq^2-1)\,x}{q(q-1)}.$$

and for the Big q-Jacobi polynomials,  $P_n(x; a, b, c; q)$  [Area et al., 1998a], [Koekoek et. al 1996]

$$\phi(x) = acq - a + c)x + \frac{x^2}{q}, \ \psi(x) = \frac{cq + aq(1 - (b + c)q) + (abq^2 - 1)x}{q(q - 1)},$$

the constant  $\phi'' - 2\psi'$  is equal to  $\frac{2(1-abq)}{q-1}$ . Therefore, the first associated of the Little q-Jacobi polynomials (resp. Big q-Jacobi polynomials) is still in the Little q-Jacobi (resp. Big q-Jacobi) family when abq = 1.

Let  $\beta_n(a,b|q)$  and  $\gamma_n(a,b|q)$  (resp.  $\beta_n(a,b,c;q)$  and  $\gamma_n(a,b,c;q)$ ) be the coefficients of the threeterm relation (see (2.18)) satisfied by the Little q-Jacobi polynomials  $p_n(x;a,b|q)$  and the Big q-Jacobi polynomials  $P_n(x; a, b, c; q)$ , respectively.

It follows immediately from Lemma 7.1 that they obey:

$$\beta_{n+1}(a, \frac{1}{q \, a} \, | q) = q \, a \, \beta_n(\frac{1}{a}, a \, q \, | q), \ \gamma_{n+1}(a, \frac{1}{q \, a} \, | q) = q^2 \, a^2 \, \gamma_n(\frac{1}{a}, a \, q \, q),$$

$$\beta_{n+1}(a, \frac{1}{q \, a}, c; q) = a \, \beta_n(\frac{1}{a}, a \, q, c \, q; q), \ \gamma_{n+1}(a, \frac{1}{q \, a}, c; q) = a^2 \, \gamma_n(\frac{1}{a}, a \, q, c \, q; q).$$

The previous equations used, together with (2.23), give:

**Theorem 5.1** The monic Little q-Jacobi (resp. monic Big q-Jacobi) polynomials and their respective first associated are related by

$$p_n^{(1)}(x; a, \frac{1}{q \, a} \, | q) = a^n \, q^n \, p_n(\frac{x}{a \, q}; \frac{1}{a}, a \, q \, | q), \tag{5.16}$$

$$P_n^{(1)}(x; a, \frac{1}{q a}, c; q) = a^n P_n(\frac{x}{a}; \frac{1}{a}, a q, c q; q).$$
 (5.17)

#### 5.4.2 The first associated $\mathcal{D}$ -classical orthogonal polynomials

Since  $\lim_{q\to 1} \mathcal{D}_q = \frac{d}{dx}$ , from Equations (5.13) and (5.14), we recover by a limit process the factored form of the fourth-order differential equation satisfied by the first associated  $P_{n-1}^{(1)}(x)$  of the (continuous) classical orthogonal polynomials  $P_{n-1}$  [Ronveaux. 1988],

$$Q_{2,n-1}^{\star\star c} Q_{2,n-1}^{\star c} \left[ P_{n-1}^{(1)}(x) \right] = 0, \tag{5.18}$$

with

$$\begin{aligned} \mathcal{Q}_{2,n-1}^{\star c} &=& \lim_{q \to 1} \frac{\mathcal{Q}_{2,n-1}^{\star}}{q^2 (q-1)^2 x^2} = \phi \frac{d^2}{dx^2} + (2\phi' - \psi) \frac{d}{dx} + (\phi'' - \psi' + \lambda_n) \mathcal{I}_d, \\ \mathcal{Q}_{2,n-1}^{\star \star c} &=& \frac{1}{4\phi(x)} \lim_{q \to 1} \frac{\mathcal{Q}_{2,n-1}^{\star \star}}{q^2 (q-1)^2 x^2} = \phi \frac{d^2}{dx^2} + (\phi' + \psi) \frac{d}{dx} + (\psi' + \lambda_n) \mathcal{I}_d, \end{aligned}$$

where

$$\lambda_n \equiv \lim_{q \to 1} \lambda_{n,0} = -n \left[ (n-1) \frac{\phi''}{2} + \psi' \right].$$

#### 5.4.3 The first associated $D_{q,\omega}$ -classical orthogonal polynomials

In this subsection we apply the result of Theorem 4.2 to deduce the fourth-order difference equation satisfied by the  $D_{q,\omega}$ -classical orthogonal polynomials and then deduce the difference equation for classical orthogonal polynomials of a discrete variable.

In the first step we replace in (5.15), the polynomials  $\phi$  (resp.  $\psi$  and  $P_{n-1}^{(1)}$ ) by  $\mathcal{T}_{\frac{\omega}{1-q}}\bar{\phi}$ ,  $\mathcal{T}_{\frac{\omega}{1-q}}\bar{\psi}$  and  $\mathcal{T}_{\frac{\omega}{1-q}}\bar{P}_{n-1}^{(1)}$ , respectively, i.e.,

$$\phi = \mathcal{T}_{\frac{\omega}{1-q}} \bar{\phi}, \ \psi = \mathcal{T}_{\frac{\omega}{1-q}} \bar{\psi}, \ P_{n-1}^{(1)}(x;q) = \mathcal{T}_{\frac{\omega}{1-q}} \bar{P}_{n-1}^{(1)}(x;q,\omega)$$
 (5.19)

and get an equation which multiplied by  $\mathcal{T}_{\underline{--}}$ , taking into account (4.23) and Proposition 3.5, gives

$$Q_{2,n-1}^{\star\star+} \frac{Q_{2,n-1}^{\star-}}{((q-1,x+\omega)^2} \left[ \bar{P}_{n-1}^{(1)}(x;q,\omega) \right] = 0, \tag{5.20}$$

where

$$\begin{split} \mathcal{Q}_{2,n-1}^{*+} &= \bar{\phi}_{[2]} \, A_{q,\omega}^2 - (\bar{-}1+q) \bar{\phi}_{[1]} + \bar{\psi}_{[1]} \, \bar{t}_1 - \lambda_{n,0} \, \bar{t}_1^2) A_{q,\omega} + q \, (\bar{\phi} + \bar{\psi} \, \bar{t}) \, \mathcal{I}_d, \\ \mathcal{Q}_{2,n-1}^{*++} &= (\bar{\phi}_{[3]} + \bar{\psi}_{[3]} \, \bar{t}_3) [q^2 \, \bar{A}_1 + (1+q) \, \bar{\phi}_{[2]} + \bar{\psi}_{[2]} \, \bar{t}_2] A_{q,\omega}^2 \\ &- [q^3 \, \bar{A}_1 \, (\bar{\phi}_{[2]} + \bar{\psi}_{[2]} \, \bar{t}_2) + \bar{A}_3 \, (\bar{\phi}_{[2]} + q \, \bar{A}_1)] \, A_{q,\omega} \\ &+ q \, \bar{\phi}_{[1]} \, [q^2 \, \bar{A}_2 + (1+q) \, \bar{\phi}_{[3]} + \bar{\psi}_{[3]} \, \bar{t}_3)] \, \mathcal{I}_d. \end{split}$$

with the notations

$$\bar{A}_{j}(x) \equiv \bar{A}_{j} = (1+q)\bar{\phi}_{[j]} + \bar{\psi}_{[j]}\bar{t}_{j} - \lambda_{n,0}\bar{t}_{j}^{2}, \ \bar{\phi}_{[j]} \equiv \bar{\phi}(q^{j} x + \omega[j]_{q}), \ \bar{\psi}_{[j]} \equiv \bar{\psi}(q^{j} x + \omega[j]_{q}), \ \bar{t}_{j} \equiv q^{j} \mathcal{T}_{\frac{-\omega}{1-q}} t(x) = q^{j} ((q-1)x + \omega_{+}, \ \bar{t}_{0} \equiv \bar{t}(x) = (q-1)x + \omega_{-}.$$

5.4. Applications 63

Since  $\{P_n\}_{n\in\mathbb{N}}$  is  $\mathcal{D}_q$ -classical with respect to  $\mathcal{L}$  (see (5.5)), it follows immediately from Theorem 4.2 that  $\{\bar{P}_n\}_{n\in\mathbb{N}}$  (with  $\bar{P}_n(x;q,\omega)=\mathcal{T}_{\frac{-\omega}{1-q}}P_n(x;q)$ ) is  $D_{q,\omega}$ -classical with respect to  $\bar{\mathcal{L}}=\mathcal{T}_{\frac{-\omega}{1-q}}\mathcal{L}$ , where the linear functional  $\bar{\mathcal{L}}$  satisfies  $D_{q,\omega}(\bar{\phi}\bar{\mathcal{L}})=\bar{\psi}\bar{\mathcal{L}}$ . Therefore (5.20) is the factored form of the fourth-order difference equation satisfied by the first associated  $\bar{P}_{n-1}^{(1)}(x;q,\omega)$  of the  $D_{q,\omega}$ -classical orthogonal polynomial  $\bar{P}_{n-1}(x;q,\omega)$ .

#### 5.4.4 The first associated $\Delta$ -classical orthogonal polynomials

We obtain the difference equation satisfied by the first associated  $P_{n-1}^{(1)}$  of the polynomial of a discrete variable,  $P_{n-1}$ , orthogonal with respect to the classical linear functional  $\tilde{\mathcal{L}}$  (with  $\mathcal{L}$  satisfying  $\Delta(\tilde{\phi}\tilde{\mathcal{L}}) = \tilde{\psi}\tilde{\mathcal{L}}$ ) [Atakishiyev et al., 1988], [Ronveaux et al. 1998a], [Foupouagnigni et al., 1998b] by limit processes  $\begin{pmatrix} \lim_{q \to 1} D_{q,\omega} = \Delta \end{pmatrix}$ :

$$Q_{2,n-1}^{**d}Q_{2,n-1}^{*d}\left[P_{n-1}^{(1)}(x)\right] = 0, (5.21)$$

where

$$Q_{2,n-1}^{\star d} = \lim_{\omega \to 1, \ q \to 1} Q_{2,n-1}^{\star +}$$

$$= \tilde{\phi}_{(2)} \mathcal{T}^2 - (2\tilde{\phi}_{(1)} + \tilde{\phi}_{(1)} - \lambda_n) \mathcal{T} + (\tilde{\phi}_{(0)} + \tilde{\phi}_{(0)}) \mathcal{I}_d$$

and

$$\begin{array}{lll} \mathcal{Q}_{2,n-1}^{\star\star d} & = & \lim_{\omega \to 1, \ q \to 1} \mathcal{Q}_{2,n-1}^{\star\star +} \\ & = & (\tilde{\phi}_{(3)} + \tilde{\psi}_{(3)})(\lambda_n - 2\tilde{\phi}_{(1)} - \tilde{\psi}_{(1)} - 2\tilde{\phi}_{(2)} - \tilde{\psi}_{(2)}) \, \mathcal{T}^2 \\ & & + [\tilde{\phi}_{(3)} \, (2\,\tilde{\phi}_{(2)} + 4\,\tilde{\phi}_{(1)} + 2\,\tilde{\psi}_{(1)} - 2\,\lambda_n) + \tilde{\phi}_{(2)} \, (2\,\tilde{\phi}_{(1)} + \tilde{\psi}_{(1)} + \tilde{\psi}_{(3)} - 2\,\lambda_n) \\ & & + 2\,\tilde{\phi}_{(1)} \, (\tilde{\psi}_{(2)} + \tilde{\psi}_{(3)} - \lambda_n) + (\tilde{\psi}_{(2)} + \tilde{\psi}_{(3)})(\tilde{\psi}_{(1)} - \lambda_n) + \lambda_n \, (\lambda_n - \tilde{\psi}_{(1)})]\mathcal{T} \\ & & + \tilde{\phi}_{(1)} (\lambda_n - 2\,\tilde{\phi}_{(2)} - \tilde{\psi}_{(2)} - 2\,\tilde{\phi}_{(3)} - \tilde{\psi}_{(3)})\mathcal{I}_d, \end{array}$$

with the notations

$$\tilde{\phi}_{(j)} \equiv \tilde{\phi}(x+j), \ \tilde{\psi}_{(j)} \equiv \tilde{\psi}(x+j), \ \lambda_n = \lim_{q \to 1} \lambda_{n,0} = -n \left( \tilde{\psi}' + (n-1) \frac{\tilde{\phi}''}{2} \right).$$

The results given in this chapter (see Equations (5.12) and (5.14)), which agree with the ones obtained using the Stieltjes properties of the associated linear functional [Foupouagnigni et al., 98e19], can be used for connection problems (see [Askey, 1965, 1975], [Askey et al., 1984], [Lewanowicz, 1995, 1996], [Godoy et al., 1997a] [Area et al., 1998b]), expanding the first associated  $P_{n-1}^{(1)}$  in terms of  $P_n$ , in the same spirit as in [Lewanowicz, 1995]; and also in order to represent finite modifications inside the Jacobi matrices of the q-classical starting family [Ronveaux et al., 1996]. We have also computed the coefficients of the fourth order q-difference equation satisfied by the first associated q-classical orthogonal polynomials appearing in the q-Hahn tableau. In particular, from the Big q-Jacobi polynomials, we derive by limit processes [Koekoek et al., 1996] the fourth-order differential (resp. q-difference) equation satisfied by the first associated classical (resp. q-classical) orthogonal polynomials.

For the Little q-Jacobi polynomials, for example, the operators  $\mathcal{Q}_{2,n-1}^{\star}$  and  $\mathcal{Q}_{2,n-1}^{\star\star}$  are given below, with the notation:  $\nu=q^n$ .

$$\begin{array}{rcl} \mathcal{Q}_{2,n-1}^{\star} & = & q\,x\, \big[ (q^2\,x - 1)\,\mathcal{G}_q^2 - \nu^{-1}\, (-\nu - c\,\nu + q^2\,x\,a\,b\,\nu^2 + q\,x)\,\mathcal{G}_q \\ & & + a\, (-1 + b\,q\,x)\,\mathcal{I}_d \big]\,, \\ \\ \mathcal{Q}_{2,n-1}^{\star\star} & = & \nu^{-1}\,q^4x^2\, \big[ qa\, (-1 + bq^4x)\,\,\times \\ & & (q^3xab\nu + q^3xab\nu^2 + q^2x\nu + q^2x - q\nu - qa\nu - \nu - a\nu)\mathcal{G}_q^2 \\ & - & \nu^{-1}(q^5x^2 + a\nu^2 + q\nu^2 - q^2x\nu^2 - q^3xab\nu^3 + q^7x^2a^2b^2\nu^3 \\ & - & q^3xa^2b\nu^3 - q^5xab\nu^3 + q^2a^2\nu^2 - q^5xab\nu^2 - q^5xa^2b\nu^2 + q^2a\nu^2 \end{array}$$

$$\begin{array}{lll} - & q^5xa^2b\nu^3 - q^2xa\nu - q^4xa\nu - q^2x\nu - q^4x\nu - q^3xa\nu + q^5x^2\nu \\ - & q^3x\nu + q^7x^2a^2b^2\nu^4 + q^6x^2ab\nu - q^4xa^2b\nu^3 + qa^2\nu^2 - q^2xa\nu^2 \\ + & 2q^6x^2ab\nu^2 + q^6x^2ab\nu^3 + 2qa\nu^2 + \nu^2 - q^4xab\nu^3)\mathcal{T}_q \\ + & (-1+qx)(q^4xab\nu + q^4xab\nu^2 + q^3x\nu + q^3x - q\nu \\ - & qa\nu - \nu - a\nu)\mathcal{I}_d \right]. \end{array}$$

# Chapter 6

# Difference equations for the rth associated Laguerre-Hahn orthogonal polynomials

# 6.1 Introduction

Using the properties of the Stieltjes function of a given Laguerre-Hahn linear functional, we derive the single fourth-order difference equation satisfied by the rth associated  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials [Foupouagnigni et al., 1998d, 1998e]. We deduce by the limit process,  $\lim_{q\to 1} \mathcal{D}_q = \frac{d}{dx}$ , the fourth-order differential equation satisfied by the rth associated  $\mathcal{D}$ -Laguerre-Hahn orthogonal polynomials [Belmehdi et al., 1991].

Moreover, we use Theorem 4.2 to give the fourth-order difference equation satisfied by the rth associated  $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials. Then follows, immediately, the fourth-order difference equation satisfied by the rth associated  $\Delta$ -Laguerre-Hahn orthogonal polynomials [Letessier et al., 1996], [Foupouagnigni et al., 1998b, 1998c].

# 6.2 The associated $\mathcal{D}_q$ -Laguerre-Hahn linear functional

# 6.2.1 The associated $\mathcal{D}_q$ -Laguerre-Hahn linear functional is a $\mathcal{D}_q$ -Laguerre-Hahn linear functional

**Theorem 6.1 (Foupouagnigni** et al., 1998e) The associated of any integer order of the regular linear functional belonging to the  $\mathcal{D}_q$ -Laguerre-Hahn class belongs to the  $\mathcal{D}_q$ -Laguerre-Hahn class.

The proof of the above theorem is given by induction on the order of association using the following proposition.

Proposition 6.1 (Foupouagnigni et al., 1998e) Let  $\mathcal{L}$  be a given regular linear functional;  $\mathcal{L}^{(r)}$  the associated of order r of  $\mathcal{L}$  and  $S_r (\equiv S(\mathcal{L}^{(r)}))$  the Stieltjes function of  $\mathcal{L}^{(r)}$ .

If  $S_r$  satisfies the  $\mathcal{D}_q$ -Riccati difference equation,

$$\phi(qx)\mathcal{D}_{q}S_{r}(x) = G_{r}(x;q)S_{r}(x)G_{q}S_{r}(x) + E_{r}(x;q)S_{r}(x) 
+ F_{r}(x;q)G_{q}S_{r}(x) + H_{r}(x;q), \quad r \ge 0,$$
(6.1)

where  $\phi$  is a non-zero polynomial and  $E_r$ ,  $F_r$ ,  $G_r$  and  $H_r$  are polynomials in the variable x depending on q, then the same property holds for  $S_{r+1}$ :

$$\phi(qx)D_{q,\omega}S_{r+1}(x) = G_{r+1}(x;q)S_{r+1}(x)G_qS_{r+1}(x)$$

$$+E_{r+1}(x;q)S_{r+1}(x) +F_{r+1}(x;q)\mathcal{G}_q S_{r+1}(x) + H_{r+1}(x;q),$$
(6.2)

with

$$G_{r+1} = \frac{H_r}{\gamma_r}, (6.3)$$

$$E_{r+1} = (qx - \beta_r) \frac{H_r}{\gamma_r} - F_r, \qquad (6.4)$$

$$F_{r+1} = (x - \beta_r) \frac{H_r}{\gamma_r} - E_r. \tag{6.5}$$

$$H_{r+1} = -\phi(qx) + \gamma_r G_r - (qx - \beta_r) E_r - (x - \beta_r) F_r + (x - \beta_r) (qx - \beta_r) \frac{H_r}{\gamma_r}.$$
(6.6)

*Proof:* Application of the  $\mathcal{D}_q$  -derivative rule

$$\mathcal{D}_q\left(\frac{f}{g}\right)(x) \equiv \frac{\frac{f(qx)}{g(qx)} - \frac{f(x)}{g(x)}}{(q-1)x} = \frac{g(x)\mathcal{D}_q f(x) - f(x)\mathcal{D}_q g(x)}{g(x)g(qx)},\tag{6.7}$$

to (6.21) gives

$$\mathcal{D}_q S_r(x) = \frac{\gamma_r \left[ 1 + \mathcal{D}_q S_{r+1}(x) \right]}{(qx - \beta_r + \mathcal{G}_q S_{r+1}(x)) (x - \beta_r + S_{r+1}(x))}.$$
 (6.8)

Using (6.21), (6.1) and (6.8), we obtain the  $\mathcal{D}_q$ -Riccati difference equation for  $S_{r+1}$ 

$$\begin{split} \phi(qx)\mathcal{D}_{q}S_{r+1} &= \frac{H_{r}}{\gamma_{r}}S_{r+1}\mathcal{G}_{q}S_{r+1} \\ &+ [(qx - \beta_{r})\frac{H_{r}}{\gamma_{r}} - F_{r}]S_{r+1} - [(x - \beta_{r})\frac{H_{r}}{\gamma_{r}} - E_{r}]\mathcal{G}_{q}S_{r+1} \\ &- \phi(qx) + \gamma_{r}G_{r} - (qx - \beta_{r})E_{r} - (x - \beta_{r})F_{r}. \end{split}$$

Identification of the previous difference equation with (6.2) completes the proof.

**Remark 6.1** Use of (6.4)-(6.6) gives the following properties:

i)

$$E_{r+1} - F_{r+1} - E_r + F_r = (q-1)x \frac{H_r}{\gamma_r}.$$
 (6.9)

$$E_{r+1} + F_{r+1} + E_r + F_r = \frac{H_r}{\gamma_r} \left( (1-q)x - 2\beta_r \right),$$
 (6.10)

$$E_{r+1}F_{r+1} - E_rF_r = \phi(qx)\frac{H_r}{\gamma_r} + \frac{H_rH_{r+1}}{\gamma_r} - H_rG_r.$$
 (6.11)

ii) Knowing polynomials  $\phi$ ,  $E_0$ ,  $F_0$ ,  $G_0$ ,  $H_0$ ,

Note that the coefficients  $\beta_n$  and  $\gamma_n$  of the three-term recurrence relation (see (2.18)), for  $\mathcal{D}_q$ -semi-classical orthogonal polynomials of class one are given by Theorem 8.1.

Let  $\mathcal{L}$  be a regular  $\mathcal{D}_q$ -Laguerre-Hahn linear functional. By Theorem 6.1, the rth associated of  $\mathcal{L}$ ,  $\mathcal{L}^{(r)}$ , belongs to the  $\mathcal{D}_q$ -Laguerre-Hahn class and its Stieltjes function  $S_r$  satisfies the following  $\mathcal{D}_q$ -Riccati difference equation

$$\phi(qx)\mathcal{D}_q S_r(x) = G_r(x;q)S_r(x)\mathcal{G}_q S_r(x) + E_r(x;q)S_r(x) + F_r(x;q)\mathcal{G}_q S_r(x) + H_r(x;q), \quad r \ge 0,$$

where  $\phi$  is a non-zero polynomial and  $E_r, F_r, G_r$  and  $H_r$  are polynomials in the variable x depending eventually on q. The following proposition proves that the degrees of the polynomials  $E_r, F_r, G_r$  and  $H_r$  are bounded.

Proposition 6.2 (Foupouagnigni et al., 1998e) The polynomial coefficients  $E_r, F_r, G_r$  and  $H_r$  satisfy:

$$\deg(H_r) \le m - 1. \ \deg(E_r) \le m \ and \ \deg(F_r) \le m, \ r \ge 0, \tag{6.12}$$

where m is given by  $m = \max\{\deg(E_0), \deg(F_0), \deg(H_0) + 1\}.$ 

*Proof:* For r = 0, (6.12) holds by hypothesis. Suppose that (6.12) holds up to a fixed integer r. Then using (6.4), we obtain

$$\deg(E_{r+1}) = \deg((qx - \beta_r) \frac{H_r}{\gamma_r} - F_r) \le m, \tag{6.13}$$

by the above hypothesis. Likewise, using (6.5), we have  $\deg(F_{r+1}) \leq m$ . Finally use of (6.4) and the fact that the last two inequalities of (6.12) hold for any integer r, give

$$\deg(H_{r+1}) + 1 = \deg(F_{r+1} + E_{r+2}) \le m.$$

Corollary 6.1 Let  $\mathcal{L}$  be a  $\mathcal{D}_q$ -semi-classical linear functional satisfying

$$\mathcal{D}_q(\phi \mathcal{L}) = \psi \mathcal{L},\tag{6.14}$$

where  $\phi$  is any non-zero polynomial,  $\psi$  a polynomial of degree at least one, and  $E_r, F_r, G_r$  and  $H_r$  are defined by (6.1). Then the following properties hold:

$$\begin{aligned}
\deg(H_r) &\leq \max\{\deg(\psi), \deg(\phi) - 1\} - 1 \quad \forall r \in \mathbb{N}. \\
\deg(E_r) &\leq \max\{\deg(\psi), \deg(\phi) - 1\} \quad \forall r \in \mathbb{N}, \\
\deg(F_r) &\leq \max\{\deg(\psi), \deg(\phi) - 1\} \quad \forall r \in \mathbb{N}.
\end{aligned} \tag{6.15}$$

*Proof:* We shall give the proof by showing that

$$m = \max\{\deg(E_0), \deg(F_0), \deg(H_0) + 1\} \le \max\{\deg(\psi), \deg(\phi) - 1\},$$

then use Proposition 6.2.

In fact, since  $\mathcal{L}$  is  $\mathcal{D}_q$ -semi-classical satisfying (6.14), we deduce from Theorem 4.1 that  $\mathcal{L}$  is a  $\mathcal{D}_q$ -Laguerre-Hahn linear functional and its Stieltjes function  $S_0$  satisfies

$$\phi(qx)\mathcal{D}_q S_0 = G_0 S_0 \mathcal{G}_q S_0 + E_0 S_0 + F_0 \mathcal{G}_q S_0 + H_0,$$

where

$$E_{0}(x;q) = \psi(x) - \mathcal{D}_{q}\phi(x),$$

$$F_{0}(x;q) = G_{0}(x;q) = 0,$$

$$H_{0}(x;q) = \mathcal{L}\theta_{0}\psi(x) - (\mathcal{D}_{q}\mathcal{L})\theta_{0}\phi(qx) - \mathcal{L}\theta_{0}\mathcal{D}_{q}\phi(x).$$
(6.16)

From (6.16) results immediately

$$\deg(F_0) \le \deg(E_0) \le \max\{\deg(\psi), \deg(\phi) - 1\}. \tag{6.17}$$

It follows from (2.11) and (2.12) that

$$\deg(\mathcal{L}\theta_0\psi) \le \deg(\psi) - 1, \ \deg(\mathcal{L}\theta_0\mathcal{D}_q\phi) \le \deg(\phi) - 2. \tag{6.18}$$

To show that

$$\deg((\mathcal{D}_q \mathcal{L})\theta_0 \phi(qx) \le \deg(\phi) - 2, \tag{6.19}$$

we assume

$$\phi(qx) = \sum_{j=0}^{n} \phi_j x^j$$

and deduce that

$$\theta_0 \phi(qx) = \sum_{j=0}^{n-1} \phi_{j+1} x^j,$$

$$(\mathcal{D}_q \mathcal{L}) \theta_0 \phi(qx) = \sum_{j=0}^{n-1} \tilde{\phi}_j x^j,$$

with

$$\tilde{\phi}_j = \sum_{k=j}^{n-1} \phi_{k+1} \langle \mathcal{D}_q \mathcal{L}, x^{k-j} \rangle.$$

It turns out that

$$\tilde{\phi}_{n-1} = \phi_n \langle \mathcal{D}_q \mathcal{L}. 1 \rangle = -\frac{1}{q} \phi_n \langle \mathcal{L}, \mathcal{D}_q^* 1 \rangle = 0,$$

then  $\deg((\mathcal{D}_q \mathcal{L})\theta_0 \phi(qx)) \leq \deg(\phi) - 2$ .

Using (6.18) and (6.19), we deduce that

$$\deg(H_0) \le \max\{\deg(\psi), \deg(\phi) - 1\} - 1.$$
 (6.20)

It results from (6.17) and (6.20) that

$$m = \max\{\deg(E_0), \deg(F_0), \deg(H_0) + 1\} < \max\{\deg(\psi), \deg(\phi) - 1\}.$$

The previous equation, combined with Proposition 6.2, completes the proof of the corollary.

# 6.3 Fourth-order difference equation

Through the following steps, we will show that the rth associated Laguerre-Hahn orthogonal polynomials are solution of a single fourth-order linear difference equation with polynomial coefficients. To do this, we shall need the following identities giving relation between  $S_r$  and the associated orthogonal polynomials.

**Lemma 6.1** (Sherman, 1933, Maroni, 1986a) Let  $\mathcal{L}$  be a given regular linear functional;  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal polynomials satisfying (2.18);  $\mathcal{L}^{(r)}$  the rth associated of  $\mathcal{L}$  and  $S_r (\equiv S(\mathcal{L}^{(r)}))$  the Stieltjes function of  $\mathcal{L}^{(r)}$ ; then, we have

$$S_r(x) = \frac{-\gamma_r}{x - \beta_r + S_{r-1}(x)}, \quad \forall r \in \mathbb{N}, \tag{6.21}$$

where  $\beta_n$  and  $\gamma_n$  are defined in (2.18).

**Lemma 6.2 (Dzoumba, 1985)** Let  $\mathcal{L}$  be a given regular linear functional;  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding monic orthogonal polynomials satisfying (2.18);  $\mathcal{L}^{(r)}$  the associated of order r of  $\mathcal{L}$ ;  $\{P_n^{(r)}\}_{n\in\mathbb{N}}$  the orthogonal polynomials associated to  $\mathcal{L}^{(r)}$  and  $S_r = S(\mathcal{L}^{(r)})$  the Stieltjes function of  $\mathcal{L}^{(r)}$ . Then, the following identity holds:

$$S_r = -\gamma_r \frac{P_n^{(r+1)} + S_{n+r+1} P_{n-1}^{(r+1)}}{P_{n+1}^{(r)} + S_{n+r+1} P_n^{(r)}},$$
(6.22)

where  $\beta_n$  and  $\gamma_n$  are defined in (2.18).

We suppose that  $\mathcal{L}$  is a regular linear functional belonging to the  $\mathcal{D}_q$ -Laguerre-Hahn class, that  $\mathcal{L}^{(r)}$  is the rth associated of  $\mathcal{L}$ , and that  $\{P_n^{(r)}\}_{n\in\mathbb{N}}$  is the family of monic polynomials, orthogonal with respect to  $\mathcal{L}^{(r)}$ . If  $S_r$  represents the Stieltjes function of  $\mathcal{L}^{(r)}$ , then by Theorem 6.1, for any integer r,  $S_r$  satisfies a  $\mathcal{D}_q$ -Riccati difference equation (see (6.1)). We first apply the difference operator  $\mathcal{G}_q$  to (6.22) and obtain

$$\mathcal{G}_q S_r = -\gamma_r \frac{\mathcal{G}_q P_n^{(r+1)} + \mathcal{G}_q S_{n+r+1} \mathcal{G}_q P_{n-1}^{(r+1)}}{\mathcal{G}_q P_{n+1}^{(r)} + \mathcal{G}_q S_{n+r+1} \mathcal{G}_q P_n^{(r)}}.$$
(6.23)

Secondly, we apply the quotient rule (see (6.7)) to (6.22) and obtain

$$\left(P_{n+1}^{(r)} + S_{n+r+1}P_{n}^{(r)}\right) \left(\mathcal{G}_{q}P_{n+1}^{(r)} + \mathcal{G}_{q}S_{n+r+1}\mathcal{G}_{q}P_{n}^{(r)}\right) \frac{\mathcal{D}_{q}S_{r}}{\gamma_{r}}$$

$$= \left(\mathcal{G}_{q}P_{n}^{(r+1)}\mathcal{D}_{q}P_{n}^{(r)} - \mathcal{G}_{q}P_{n+1}^{(r)}\mathcal{D}_{q}P_{n-1}^{(r+1)}\right) S_{n+r+1}$$

$$+ \left(\mathcal{G}_{q}P_{n-1}^{(r+1)}\mathcal{D}_{q}P_{n+1}^{(r)} - \mathcal{G}_{q}P_{n}^{(r)}\mathcal{D}_{q}P_{n}^{(r+1)}\right) \mathcal{G}_{q}S_{n+r+1}$$

$$- \left(\mathcal{G}_{q}P_{n}^{(r)}\mathcal{D}_{q}P_{n-1}^{(r+1)} - \mathcal{G}_{q}P_{n-1}^{(r+1)}\mathcal{D}_{q}P_{n}^{(r)}\right) S_{n+r+1}\mathcal{G}_{q}S_{n+r+1}$$

$$- \mathcal{G}_{q}P_{n+1}^{(r)}\mathcal{D}_{q}P_{n}^{(r+1)} + \mathcal{G}_{q}P_{n}^{(r+1)}\mathcal{D}_{q}P_{n+1}^{(r)}$$

$$+ \left(\mathcal{G}_{q}P_{n}^{(r+1)}\mathcal{G}_{q}P_{n}^{(r)} - \mathcal{G}_{q}P_{n-1}^{(r+1)}\mathcal{G}_{q}P_{n+1}^{(r)}\right) \mathcal{D}_{q}S_{n+r+1}.$$
(6.24)

Further, we replace  $S_r$ ,  $\mathcal{G}_q S_r$  and  $\mathcal{D}_q S_r$ , given by (6.22), (6.23) and (6.24), respectively, in (6.1) and obtain after taking into account (2.24), the  $\mathcal{D}_q$ -Riccati difference equation for  $S_{n+r+1}$ ; an equation which when compared with

$$\phi(qx)\mathcal{D}_{q} S_{n+r+1}(x) = G_{n+r+1}(x;q)S_{n+r+1}(x)\mathcal{G}_{q} S_{n+r+1}(x) + E_{n+r+1}(x;q)S_{n+r+1}(x) + F_{n+r+1}(x;q)\mathcal{G}_{q} S_{n+r+1}(x) + H_{n+r+1}(x;q),$$

gives the following proposition:

Proposition 6.3 (Foupouagnigni et al., 1998e) The coefficients of the  $\mathcal{D}_q$ -Riccati difference equation for  $S_{n+r+1}$  are given by

$$\pi_{n,r} E_{n+r+1} = -\phi(qx) \left( \mathcal{G}_q P_n^{(r+1)} \mathcal{D}_q P_n^{(r)} - \mathcal{G}_q P_{n+1}^{(r)} \mathcal{D}_q P_{n-1}^{(r+1)} \right)$$

$$- E_r P_{n-1}^{(r+1)} \mathcal{G}_q P_{n+1}^{(r)} - F_r P_n^{(r)} \mathcal{G}_q P_n^{(r+1)}$$

$$+ \frac{H_r}{\gamma_r} P_n^{(r)} \mathcal{G}_q P_{n+1}^{(r)} + \frac{\gamma_r H_{r-1}}{\gamma_{r-1}} P_{n-1}^{(r+1)} \mathcal{G}_q P_n^{(r+1)},$$

$$(6.25)$$

$$\pi_{n,r} F_{n+r+1} = \phi(qx) \left( \mathcal{G}_q P_n^{(r)} \mathcal{D}_q P_n^{(r+1)} - \mathcal{G}_q P_{n-1}^{(r+1)} \mathcal{D}_q P_{n+1}^{(r)} \right) \\
- E_r P_n^{(r+1)} \mathcal{G}_q P_n^{(r)} - F_r P_{n+1}^{(r)} \mathcal{G}_q P_{n-1}^{(r+1)} \\
+ \frac{H_r}{\gamma_r} P_{n+1}^{(r)} \mathcal{G}_q P_n^{(r)} + \frac{\gamma_r H_{r-1}}{\gamma_{r-1}} P_n^{(r+1)} \mathcal{G}_q P_{n-1}^{(r+1)}, \tag{6.26}$$

$$\pi_{n,r} H_{n+r+1} = -\phi(qx) \left( \mathcal{G}_q P_n^{(r+1)} \mathcal{D}_q P_{n+1}^{(r)} - \mathcal{G}_q P_{n+1}^{(r)} \mathcal{D}_q P_n^{(r+1)} \right)$$

$$-E_r P_n^{(r+1)} \mathcal{G}_q P_{n+1}^{(r)} - F_r P_{n+1}^{(r)} \mathcal{G}_q P_n^{(r+1)}$$

$$+ \frac{H_r}{\gamma_r} P_{n+1}^{(r)} \mathcal{G}_q P_{n+1}^{(r)} + \frac{\gamma_r H_{r-1}}{\gamma_{r-1}} P_n^{(r+1)} \mathcal{G}_q P_n^{(r+1)},$$

$$(6.27)$$

$$\pi_{n-1,r} H_{n+r} = -\phi(qx) \left( \mathcal{G}_q P_{n-1}^{(r+1)} \mathcal{D}_q P_n^{(r)} - \mathcal{G}_q P_n^{(r)} \mathcal{D}_q P_{n-1}^{(r+1)} \right)$$

$$- E_r P_{n-1}^{(r+1)} \mathcal{G}_q P_n^{(r)} - F_r P_n^{(r)} \mathcal{G}_q P_{n-1}^{(r+1)}$$

$$+ \frac{H_r}{\gamma_r} P_n^{(r)} \mathcal{G}_q P_n^{(r)} + \frac{\gamma_r H_{r-1}}{\gamma_{r-1}} P_{n-1}^{(r+1)} \mathcal{G}_q P_{n-1}^{(r+1)},$$

$$(6.28)$$

where  $\pi_{n,r}$  is given by (2.24).

We combine (2.24) and (6.25)-(6.28) to obtain:

Theorem 6.2 (Foupouagnigni et al., 1998e) The associated polynomials obey:

$$\phi(qx)\mathcal{D}_{q} P_{n}^{(r)} = -E_{n+r+1}\mathcal{G}_{q} P_{n}^{(r)} - F_{r} P_{n}^{(r)} + \frac{H_{n+r}}{\gamma_{n+r}} \mathcal{G}_{q} P_{n+1}^{(r)} + \frac{\gamma_{r} H_{r-1}}{\gamma_{r-1}} P_{n-1}^{(r+1)}.$$

$$(6.29)$$

$$\phi(qx)\mathcal{D}_{q} P_{n-1}^{(r+1)} = -E_{n+r+1}\mathcal{G}_{q} P_{n-1}^{(r+1)} + E_{r} P_{n-1}^{(r+1)} + \frac{H_{n+r}}{\gamma_{n+r}} \mathcal{G}_{q} P_{n}^{(r+1)} - \frac{H_{r}}{\gamma_{r}} P_{n}^{(r)},$$

$$(6.30)$$

$$\phi(qx)\mathcal{D}_{q} P_{n+1}^{(r)} = F_{n+r+1}\mathcal{G}_{q} P_{n+1}^{(r)} - F_{r} P_{n+1}^{(r)} - H_{n+r+1}\mathcal{G}_{q} P_{n}^{(r)} + \frac{\gamma_{r} H_{r-1}}{\gamma_{r-1}} P_{n}^{(r+1)},$$

$$(6.31)$$

$$\phi(qx)\mathcal{D}_{q} P_{n}^{(r+1)} = F_{n+r+1}\mathcal{G}_{q} P_{n}^{(r+1)} + E_{r} P_{n}^{(r+1)} - H_{n+r+1}\mathcal{G}_{q} P_{n-1}^{(r+1)} - \frac{H_{r}}{\gamma_{r}} P_{n+1}^{(r)}.$$

$$(6.32)$$

Proof: We subtract the two equations obtained after multiplying (6.25) (and (6.28), respectively) by  $\mathcal{G}_q P_n^{(r)}$ ,  $\mathcal{G}_q P_{n+1}^{(r+1)}$  and obtain

$$\pi_{n,r} E_{n+r+1} \mathcal{G}_{q} P_{n}^{(r)} - \pi_{n-1,r} H_{n+r} \mathcal{G}_{q} P_{n+1}^{(r)} \\
= -\phi(qx) \left( \mathcal{G}_{q} P_{n}^{(r)} \mathcal{G}_{q} P_{n}^{(r+1)} - \mathcal{G}_{q} P_{n+1}^{(r)} \mathcal{G}_{q} P_{n-1}^{(r+1)} \right) \mathcal{D}_{q} P_{n}^{(r)} \\
- \left( \mathcal{G}_{q} P_{n}^{(r)} \mathcal{G}_{q} P_{n}^{(r+1)} - \mathcal{G}_{q} P_{n+1}^{(r)} \mathcal{G}_{q} P_{n-1}^{(r+1)} \right) F_{r} P_{n}^{(r)} \\
+ \left( \mathcal{G}_{q} P_{n}^{(r)} \mathcal{G}_{q} P_{n}^{(r+1)} - \mathcal{G}_{q} P_{n+1}^{(r)} \mathcal{G}_{q} P_{n-1}^{(r+1)} \right) \frac{\gamma_{r} H_{r-1}}{\gamma_{r-1}} P_{n-1}^{(r+1)}.$$
(6.33)

Then use of the relation obtained from (2.24)

$$\mathcal{G}_q P_n^{(r)} \mathcal{G}_q P_n^{(r+1)} - \mathcal{G}_q P_{n+1}^{(r)} \mathcal{G}_q P_{n-1}^{(r+1)} = \pi_{n,r}, \tag{6.34}$$

and the fact that

$$\pi_{n-1,r} = \frac{\pi_{n,r}}{\gamma_{n+r}},$$

transform (6.33) in (6.29).

Again, we multiply both sides of (6.25) (and(6.28), respectively) by  $\mathcal{G}_q P_{n-1}^{(r+1)}$ .  $\mathcal{G}_q P_n^{(r+1)}$  and obtain two equations which subtracted give

$$\pi_{n,r} E_{n+r+1} \mathcal{G}_{q} P_{n-1}^{(r+1)} - \pi_{n-1,r} H_{n+r} \mathcal{G}_{q} P_{n}^{(r+1)} - \dots \\
= -\phi(qx) \left( \mathcal{G}_{q} P_{n}^{(r)} \mathcal{G}_{q} P_{n}^{(r+1)} - \mathcal{G}_{q} P_{n+1}^{(r)} \mathcal{G}_{q} P_{n-1}^{(r+1)} \right) \mathcal{D}_{q} P_{n-1}^{(r+1)} \\
+ \left( \mathcal{G}_{q} P_{n}^{(r)} \mathcal{G}_{q} P_{n}^{(r+1)} - \mathcal{G}_{q} P_{n+1}^{(r)} \mathcal{G}_{q} P_{n-1}^{(r+1)} \right) E_{r} P_{n-1}^{(r+1)} \\
+ \left( \mathcal{G}_{q} P_{n}^{(r)} \mathcal{G}_{q} P_{n}^{(r+1)} - \mathcal{G}_{q} P_{n+1}^{(r)} \mathcal{G}_{q} P_{n-1}^{(r+1)} \right) \frac{H_{r}}{\gamma_{r}} P_{n}^{(r)}. \tag{6.35}$$

Then use of (6.34) transforms (6.35) in (6.30).

Equations (6.31) and (6.32) are obtained in the same way by combining (6.26), (6.27) and (6.34).  $\square$  For the sake of simplicity and uniformity we shall present difference equations in terms of powers of the operator  $A_{q,\omega}$  instead of  $D_{q,\omega}$ . This is possible because for  $q \neq 1$  or  $\omega \neq 0$ , all powers of the operator  $D_{q,\omega}$  can be expressed in terms of the powers of  $A_{q,\omega}$  and conversely. To do this we present the following lemma (proved by solving system of equations).

**Lemma 6.3** The powers of the operators  $D_{q,\omega}$  and  $A_{q,\omega}$  are linked by the following relations:

$$\begin{array}{rcl} D_{q,\omega}^{0} & = & A_{q,\omega}^{0} = \mathcal{I}_{d}. \\ ((q-1)x+\omega)\,D_{q,\omega} & = & A_{q,\omega} - \mathcal{I}_{d}, \\ ((q-1)x+\omega)^{2}\,D_{q,\omega}^{2} & = & q^{-1}\,A_{q,\omega}^{2} - [2]_{q}\,q^{-1}\,A_{q,\omega} + \mathcal{I}_{d}, \\ ((q-1)x+\omega)^{3}\,D_{q,\omega}^{3} & = & q^{-3}\,A_{q,\omega}^{3} - [3]_{q}\,q^{-3}\,A_{q,\omega}^{2} + [3]_{q}\,A_{q,\omega} - \mathcal{I}_{d}, \\ ((q-1)x+\omega)^{4}\,D_{q,\omega}^{4} & = & q^{-6}\,A_{q,\omega}^{4} - (q-1)\,[4]_{q}\,q^{-6}\,A_{q,\omega}^{3} + (1+q^{2})[3]_{q}\,q^{-5}\,A_{q,\omega}^{2} \\ & - [4]_{q}\,q^{-3}\,A_{q,\omega} + \mathcal{I}_{d}. \\ ((q-1)x+\omega)^{5}\,D_{q,\omega}^{5} & = & q^{-10}\,A_{q,\omega}^{5} - [5]_{q}\,q^{-10}\,A_{q,\omega}^{4} + (1+q^{2})\,[5]_{q}\,q^{-9}\,A_{q,\omega}^{3} \\ & - (1+q^{2})\,[5]_{q}\,q^{-7}\,A_{q,\omega}^{2} + [5]_{q}\,q^{-4}\,A_{q,\omega} - \mathcal{I}_{d}, \end{array}$$

$$\begin{array}{lll} A_{q,\omega}^{0} & = & D_{q,\omega}^{0} = \mathcal{I}_{d}, \\ A_{q,\omega} & = & ((q-1)x+\omega)D_{q,\omega} + \mathcal{I}_{d}, \\ A_{q,\omega}^{2} & = & q\left((q-1)x+\omega\right)^{2}D_{q,\omega}^{2} + (1+q)\left((q-1)x+\omega\right)D_{q,\omega} + \mathcal{I}_{d}, \\ A_{q,\omega}^{3} & = & q^{3}\left((q-1)x+\omega\right)^{3}D_{q,\omega}^{3} + q[3]_{q}\left((q-1)x+\omega\right)^{2}D_{q,\omega}^{2} + [3]_{q}\left((q-1)x+\omega\right)D_{q,\omega} + \mathcal{I}_{d}, \\ A_{q,\omega}^{4} & = & q^{6}\left((q-1)x+\omega\right)^{4}D_{q,\omega}^{4} + q^{3}\left(q-1\right)[4]_{q}\left((q-1)x+\omega\right)^{3}D_{q,\omega}^{3}, \\ & + q\left(1+q^{2}\right)[3]_{q}\left((q-1)x+\omega\right)^{2}D_{q,\omega}^{2} + (q-1)\left[4\right]_{q}\left((q-1)x+\omega\right)D_{q,\omega} + \mathcal{I}_{d}, \\ A_{q,\omega}^{5} & = & q^{10}\left((q-1)x+\omega\right)^{5}D_{q,\omega}^{5} + q^{6}\left[5\right]_{q}\left((q-1)x+\omega\right)^{4}D_{q,\omega}^{4} \\ & + q^{3}\left(1+q^{2}\right)\left[5\right]_{q}\left((q-1)x+\omega\right)^{3}D_{q,\omega}^{3} + q\left(1+q^{2}\right)\left[5\right]_{q}\left((q-1)x-\omega\right)^{2}D_{q,\omega}^{2} \\ & + \left[5\right]_{q}\left((q-1)x+\omega\right)D_{q,\omega} + \mathcal{I}_{d}. \end{array}$$

**Remark 6.2** If we take  $\omega = 0$ ,  $q \neq 1$  (resp.  $\omega \neq 0$ , q = 1) in the previous lemma, we find the link between the powers of the operators  $\mathcal{D}_q$  and  $\mathcal{G}_q$  (resp.  $D_{\omega}$  and  $\mathcal{T}_{\omega}$ ).

Theorem 6.3 (Foupouagnigni et al., 1998e) Let  $\mathcal{L}$  be a regular linear functional belonging to the  $\mathcal{D}_q$ -Laguerre-Hahn class,  $\mathcal{L}^{(r)}$  the rth associated of  $\mathcal{L}$  and  $\{P_n^{(r)}\}_{n\in\mathbb{N}}$  the family of monic polynomials, orthogonal with respect to  $\mathcal{L}^{(r)}$ . If  $S_r$  represents the Stieltjes function of  $\mathcal{L}^{(r)}$ , by Theorem 6.1, for any integer r,  $S_r$  satisfies a  $\mathcal{D}_q$ -Riccati difference equation (see (6.1)). The associated polynomials  $P_n^{(r)}$  satisfy

$$\mathcal{D}_{r,n} \left[ P_n^{(r)} \right] = \mathcal{N}_{r+1,n-1} \left[ P_{n-1}^{(r+1)} \right], \tag{6.36}$$

$$\bar{\mathcal{D}}_{r+1,n-1}\left[P_{n-1}^{(r+1)}\right] = \bar{\mathcal{N}}_{r,n}\left[P_{n}^{(r)}\right],\tag{6.37}$$

where the operators  $\mathcal{D}_{r,n}$ ,  $\mathcal{N}_{r+1,n-1}$ ,  $\bar{\mathcal{D}}_{r+1,n-1}$  and  $\bar{\mathcal{N}}_{r,n}$  are given by

$$\mathcal{D}_{r,n} = a_2 \, \mathcal{G}_q^2 + a_1 \, \mathcal{G}_q + a_0 \, \mathcal{I}_d, \quad \mathcal{N}_{r+1,n-1} = \tilde{a}_1 \, \mathcal{G}_q + \tilde{a}_0 \, \mathcal{I}_d, \tag{6.38}$$

$$\bar{\mathcal{D}}_{r+1,n-1} = b_2 \, \mathcal{G}_q^2 + b_1 \, \mathcal{G}_q + b_0 \, \mathcal{I}_d, \ \bar{\mathcal{N}}_{r,n} = \tilde{b}_1 \, \mathcal{G}_q + \tilde{b}_0 \, \mathcal{I}_d. \tag{6.39}$$

The coefficients  $a_i$ ,  $b_i$ ,  $\tilde{a}_i$  and  $\tilde{b}_i$  are defined as

$$a_{2} = K_{3,0}(K_{1,1} K_{7,1} - K_{3,1} K_{8,1}), \qquad b_{2} = K_{3,0}(K_{1,1} K_{7,1} - K_{3,1} K_{8,1})$$

$$a_{1} = -K_{2,1}(K_{3,0} K_{7,1} + K_{1,0} K_{3,1}), \qquad b_{1} = -K_{5,1}(K_{3,0} K_{7,1} + K_{1,0} K_{3,1})$$

$$a_{0} = K_{3,1}(K_{2,0} K_{2,1} + K_{4,1} K_{6,0}), \qquad b_{0} = K_{3,1}(K_{5,0} K_{5,1} + K_{4,0} K_{6,1})$$

$$\tilde{a}_{1} = K_{4,1}(K_{3,0} K_{7,1} + K_{1,0} K_{3,1}), \qquad \tilde{b}_{1} = K_{6,1}(K_{3,0} K_{7,1} + K_{1,0} K_{3,1})$$

$$\tilde{a}_{0} = -K_{3,1}(K_{2,1} K_{4,0} + K_{4,1} K_{5,0}), \qquad \tilde{b}_{0} = -K_{3,1}(K_{5,1} K_{6,0} + K_{6,1} K_{2,0}),$$

$$(6.40)$$

where the coefficients  $K_{i,j}$  are given below with the notations:

$$\begin{cases}
K_i \equiv K_{i,0}(x;r,n,q) = K_i(x;r,n,q), \\
K_{i,j} \equiv K_{i,j}(x;r,n,q) = \mathcal{G}_q^j K_i(x;r,n,q) = K_i(q^j x;r,n,q).
\end{cases} (6.41)$$

$$K_{1} = \frac{\phi(qx)}{(q-1)x} + E_{n+r+1}(x;q), \quad K_{2} = \frac{\phi(qx)}{(q-1)x} - F_{r}(x;q),$$

$$K_{3} = \frac{H_{n+r}(x;q)}{\gamma_{n+r}}, \quad K_{4} = \begin{cases} \gamma_{r} \frac{H_{r-1}(x;q)}{\gamma_{r-1}} & \text{if } r \geq 1\\ \gamma_{0} G_{0} & \text{if } r = 0 \end{cases},$$

$$K_{5} = \frac{\phi(qx)}{(q-1)x} + E_{r}(x;q), \quad K_{6} = -\frac{H_{r}(x;q)}{\gamma_{r}},$$

$$K_{7} = \frac{\phi(qx)}{(q-1)x} - F_{n+r+1}(x;q), \quad K_{8} = -\gamma_{r,+r+1} \frac{H_{n-r+1}(x;q)}{\gamma_{n+r+1}}.$$

$$(6.42)$$

Proof: Use of the relation

$$(q-1)x\mathcal{D}_q P(x) = \mathcal{G}_q P(x) - P_{\geq} x$$
  $\forall P \in \mathbb{P}$ .

transforms relations (6.29)-(6.32) in

$$K_1 \mathcal{G}_q P_n^{(r)} = K_2 P_n^{(r)} + K_3 \mathcal{G}_q P_{n+1}^{(r)} + K_4 P_{n-1}^{(r-1)},$$
 (6.43)

$$K_1 \mathcal{G}_q P_{n-1}^{(r+1)} = K_5 P_{n-1}^{(r+1)} + K_3 \mathcal{G}_q P_n^{(r-1)} + K_6 P_n^{(r)},$$
 (6.44)

$$K_7 \mathcal{G}_q P_{n+1}^{(r)} = K_2 P_{n+1}^{(r)} + K_8 \mathcal{G}_q P_n^{(r)} + K_4 P_n^{(r+1)},$$
 (6.45)

$$K_7 \mathcal{G}_q P_n^{(r+1)} = K_5 P_n^{(r+1)} + K_8 \mathcal{G}_q P_{n-1}^{(r+1)} + K_6 P_{n+1}^{(r)},$$
 (6.46)

where  $K_j$  are given by (6.42).

In the first step, we solve equations (6.43) and (6.44) in terms of  $\mathcal{G}_q \, P_{n+1}^{\,\,r)}$  and  $\mathcal{G}_q \, P_n^{\,\,(r+1)}$  and obtain

$$\mathcal{G}_q P_{n+1}^{(r)} = \frac{K_1 \mathcal{G}_q P_n^{(r)} - K_2 P_n^{(r)} - K_4 P_{n-1}^{(r+1)}}{K_3}.$$
 (6.47)

$$\mathcal{G}_q P_n^{(r+1)} = \frac{K_1 \mathcal{G}_q P_{n-1}^{(r+1)} - K_5 P_{n-1}^{(r+1)} + K_6 P_n^{(r)}}{K_3}.$$
 (6.48)

In the second step we apply the operator  $\mathcal{G}_q$  to both sides of (6.45) and (6.46) and get

$$K_{7,1} \mathcal{G}_{q}^{2} P_{n+1}^{(r)} = K_{2,1} \mathcal{G}_{q} P_{n-1}^{(r)} + K_{8,1} \mathcal{G}_{q}^{2} P_{n}^{(r)} + K_{4,1} \mathcal{G}_{q} P_{n}^{(r+1)},$$
  

$$K_{7,1} \mathcal{G}_{q}^{2} P_{n}^{(r+1)} = K_{5,1} \mathcal{G}_{q} P_{n}^{(r+1)} + K_{8,1} \mathcal{G}_{q}^{2} P_{n-1}^{(r+1)} + K_{6,1} \mathcal{G}_{q} P_{n+1}^{(r)}.$$

Then, we replace  $\mathcal{G}_q P_{n+1}^{(r)}$  and  $\mathcal{G}_q P_n^{(r+1)}$  given by (6.47) and (6.48) respectively, in the two previous equations and obtain

$$\mathcal{G}_{q}^{2} P_{n+1}^{(r)} = \frac{K_{8,1}}{K_{7,1}} \mathcal{G}_{q}^{2} P_{n}^{(r)} + \frac{K_{1} K_{2,1}}{K_{3} K_{7,1}} \mathcal{G}_{r} P_{n}^{(r)} - \frac{(K_{2} K_{2,1} + K_{6} K_{4,1})}{K_{3} K_{7,1}} P_{n}^{(r)} + \frac{K_{1} K_{4,1}}{K_{3} K_{7,1}} \mathcal{G}_{q} P_{n-1}^{(r+1)} - \frac{(K_{1} K_{2,1} + K_{5} K_{4,1})}{K_{3} K_{7,1}} P_{n-1}^{(r+1)},$$
(6.49)

$$\mathcal{G}_{q}^{2} P_{n}^{(r+1)} = \frac{K_{8,1}}{K_{7,1}} \mathcal{G}_{q}^{2} P_{n-1}^{(r+1)} + \frac{K_{1} K_{5,1}}{K_{3} K_{7,1}} \mathcal{G}_{q} P_{n-1}^{(r+1)} - \frac{(K_{5} K_{5,1} + K_{4} K_{6,1})}{K_{3} K_{7,1}} P_{n-1}^{(r+1)} + \frac{K_{1} K_{6,1}}{K_{3} K_{7,1}} \mathcal{G}_{q} P_{n}^{(r)} - \frac{(K_{6} K_{5,1} + K_{2} K_{6,1})}{K_{3} K_{7,1}} P_{n}^{(r)}.$$
(6.50)

In the third step we apply the operator  $\mathcal{G}_q$  to both sides of (6.43) and (6.44) and obtain

$$K_{1,1} \mathcal{G}_{z}^{2} P_{n}^{(r)} = K_{2,1} \mathcal{G}_{q} P_{n}^{(r)} + K_{3,1} \mathcal{G}_{q}^{2} P_{n+1}^{(r)} + K_{4,1} \mathcal{G}_{q} P_{n-1}^{(r+1)}.$$

$$K_{1,1} \mathcal{G}_{q}^{2} P_{n-1}^{(r+1)} = K_{5,1} \mathcal{G}_{q} P_{n-1}^{(r+1)} + K_{3,1} \mathcal{G}_{q}^{2} P_{n}^{(r+1)} + K_{6,1} \mathcal{G}_{q} P_{n}^{(r)}.$$

Finally, use of (6.49) and (6.50) transforms the two previous equations in

$$\begin{split} &K_{3,0}(K_{1,1}\ K_{7,1}-K_{3,1}\ K_{8,1})\mathcal{G}_q^2\ P_n^{(r)}-K_{2,1}(K_{3,0}\ K_{7,1}+K_{1,0}\ K_{3,1})\mathcal{G}_q\ P_n^{(r)}\\ &+K_{3,1}(K_{2,0}\ K_{2,1}+K_{4,1}\ K_{6,0})P_n^{(r)}=\\ &K_{4,1}(K_{3,0}\ K_{7,1}+K_{1,0}\ K_{3,1})\mathcal{G}_q\ P_{n-1}^{(r+1)}-K_{3,1}(K_{2,1}\ K_{4,0}+K_{4,1}\ K_{5,0})P_{n-1}^{(r+1)},\\ &K_{3,0}(K_{1,1}\ K_{7,1}-K_{3,1}\ K_{8,1})\mathcal{G}_q^2\ P_{n-1}^{(r+1)}-K_{5,1}(K_{3,0}\ K_{7,1}+K_{1,0}\ K_{3,1})\mathcal{G}_q\ P_{n-1}^{(r+1)}\\ &+K_{3,1}(K_{5,0}\ K_{5,1}+K_{4,0}\ K_{6,1})P_{n-1}^{(r+1)}=\\ &K_{6,1}(K_{3,0}\ K_{7,1}+K_{1,0}\ K_{3,1})\mathcal{G}_q\ P_n^{(r)}-K_{3,1}(K_{5,1}\ K_{6,0}+K_{6,1}\ K_{2,0})P_{n-1}^{(r+1)}. \end{split}$$

thus the proof of Theorem 6.3 is complete.

After proving Theorem 6.3, we have now all ingredients to derive the single fourth-order difference equation satisfied by  $P_n^{(r)}$ .

In fact, we apply the operator  $\mathcal{G}_q$  to both sides of (6.36) and eliminate  $\mathcal{G}_q^{-2}P_{n-1}^{(r+1)}$  in the equation obtained, by using (6.37) and obtain

$$c_{3} \mathcal{G}_{q}^{3} P_{n}^{(r)} + c_{2} \mathcal{G}_{q}^{2} P_{n}^{(r)} + c_{1} \mathcal{G}_{q} P_{n}^{(r)} + c_{0} P_{n}^{(r)}$$

$$= \tilde{c}_{1} \mathcal{G}_{q} P_{n-1}^{(r+1)} + \tilde{c}_{0} P_{n-1}^{(r+1)}, \qquad (6.51)$$

with polynomials  $c_j$  and  $\tilde{c}_j$  given by

$$c_3 = b_2 a_{2,1}, c_2 = b_2 a_{1,1}, c_1 = b_2 a_{0,1} - \tilde{b}_1 \tilde{a}_{1,1}$$
  

$$c_3 = -\tilde{b}_0 \tilde{a}_{1,1}, \tilde{c}_1 = b_2 \tilde{a}_{0,1} - b_1 \tilde{a}_{1,1}, \tilde{c}_0 = -b_0 \tilde{a}_{1,1},$$

where  $\chi_{i,j} = \mathcal{G}_a^j \chi_i$  for  $\chi_i \in \{a_i, b_i, \tilde{a}_i, \tilde{b}_i\}$ .

By the same process, we apply the operator  $\mathcal{G}_q$  to both sides of (6.51) and eliminate  $\mathcal{G}_q^{-2}P_{n-1}^{(r+1)}$  in the equation obtained, by using (6.37) and get

$$d_4 \mathcal{G}_q \stackrel{:}{\cdot} P_n^{(r)} + d_3 \mathcal{G}_q \stackrel{3}{\cdot} P_n^{(r)} + d_2 \mathcal{G}_q \stackrel{2}{\cdot} P_n^{(r)} + d_1 \mathcal{G}_q P_n^{(r)} + d_0 P_n^{(r)}$$

$$= \tilde{d}_1 \mathcal{G}_q P_{n-1}^{(r+1)} + \tilde{d}_0 P_{n-1}^{(r+1)}, \qquad (6.52)$$

with

$$\begin{array}{lll} d_4 & = & b_2\,b_{2,1}a_{2,1},\ d_3 = b_2\,b_{2,1}a_{1,1},\ d_2 = b_2(a_{0,1}b_{2,1} - \tilde{a}_{1,1}\tilde{b}_{1,1}),\\ d_1 & = & (\tilde{b}_1\,b_{1,1}\,\tilde{a}_{1,1} - \tilde{b}_1\,\tilde{a}_{0,1}\tilde{b}_{2,1}) - \tilde{b}_2\,\tilde{b}_{0,1}\tilde{a}_{1,1}),\ d_0 = (\tilde{a}_{1,1}\,b_{1,1} - \tilde{a}_{0,1}\,b_{2,1})\tilde{b}_0,\\ \tilde{d}_1 & = & (b_1\,b_{1,1}\,\tilde{a}_{1,1} - b_1\,\tilde{a}_{0,1}b_{2,1}) - b_2\,b_{0,1}\tilde{a}_{1,1}),\ \tilde{d}_0 = (\tilde{a}_{1,1}\,b_{1,1} - \tilde{a}_{0,1}\,b_{2,1})b_0. \end{array}$$

We, therefore, deduce from (6.36), (6.51) and (6.52) the following result:

Theorem 6.4 (Foupouagnigni et al., 1998e) The associated polynomials  $P_n^{(r)}$ , for any integer n and for any integer r, satisfies the single fourth-order difference equation

$$\begin{vmatrix} a_2 \mathcal{G}_q^2 P_n^{(r)} + a_1 \mathcal{G}_q P_n^{(r)} + a_0 P_n^{(r)} & \tilde{a}_1 & \tilde{a}_0 \\ c_3 \mathcal{G}_q^3 P_n^{(r)} + c_2 \mathcal{G}_q^2 P_n^{(r)} + c_1 \mathcal{G}_q P_n^{(r)} + c_0 P_n^{(r)} & \tilde{c}_1 & \tilde{c}_0 \\ d_4 \mathcal{G}_q^4 P_n^{(r)} + d_3 \mathcal{G}_q^3 P_n^{(r)} + d_2 \mathcal{G}_q^2 P_n^{(r)} + d_1 \mathcal{G}_q P_n^{(r)} + d_0 P_n^{(r)} & \tilde{d}_1 & \tilde{d}_0 \end{vmatrix} = 0,$$
 (6.53)

which by Lemma 6.3 can be written in the two different forms:

$$\sum_{j=0}^{4} I_j(r, n, q; x) \mathcal{G}_q^j P_n^{(r)}(x) = 0, \tag{6.54}$$

$$\sum_{j=0}^{4} I_{j}^{*}(r, n, q; x) \mathcal{D}_{q}^{j} P_{n}^{(r)}(x) = 0.$$
 (6.55)

where  $I_i(r,n,q;x)$ ,  $I_i^*(r,n,q;x)$  are polynomials in the variable x and depending on r,n and q.

#### 6.3.1 Fourth-order differential equation for $P_n^{(r)}$

We deduce from the previous results and by the limit process,  $\lim_{q\to 1} \mathcal{D}_q = \frac{d}{dx}$ , the fourth-order differential equation satisfied by the rth associated orthogonal polynomial of the  $\mathcal{D}$ -Laguerre-Hahn class [Belmehdi et al., 1991]. Moreover, we recover relations used in [Belmehdi et al., 1991] to derive the fourth-order differential equation satisfied by the rth associated  $\mathcal{D}$ -Laguerre-Hahn orthogonal polynomials.

From (6.55) and by the limit process we get

$$\sum_{i=0}^{4} I_{j}^{*}(r, n, 1; x) \frac{d}{dx}^{j} P_{n}^{(r)}(x) = 0,$$
(6.56)

where  $I_{j}^{*}(r, n, 1; x) = \lim_{q \to 1} I_{j}^{*}(r, n, q; x)$ .

To compare more easily the equations obtained from (6.29)-(6.32) by this limit process with those given in [Belmehdi et al., 1991], we state the following lemma:

**Lemma 6.4** If  $E_r(x;1)$ ,  $F_r(x;1)$  are the limit when  $q \to 1$  of  $E_r(x;q)$  and  $F_r(x;q)$  respectively, we have

$$E_{n+r+1}(x;1) - E_r(x;1) = F_{n+r+1}(x;1) - F_r(x;1) \quad \forall n \in \mathbb{N}.$$
(6.57)

*Proof:* We shall prove the lemma using the relation

$$E_{r+1} x; 1 - E_r(x; 1) = F_{r+1}(x; 1) - F_r(x; 1),$$

easily derived by limit process from (6.9).

In fact, use of the previous relation gives:

$$E_{n+r+1}(x;1) - E_r(x;1) = \sum_{j=0}^n E_{j+r+1}(x;1) - E_{j+r}(x;1)$$
$$= \sum_{j=0}^n F_{j+r+1}(x;1) - F_{j+r}(x;1)$$
$$= F_{n+r+1}(x;1) - F_r(x;1).$$

When we take the limit of equations (6.29)-(6.32) as  $q \to 1$ , we obtain, taking into account the previous lemma [Magnus, 1984], [Dzoumba, 1985] [Belmehdi et al., 1994],

$$\begin{split} \phi \frac{d}{dx} P_{n}^{(r)} &= -\frac{C_{n-r-1} + C_{r}}{2} P_{n}^{(r)} + \frac{D_{n+r}}{\gamma_{n+r}} P_{n+1}^{(r)} + \frac{\gamma_{r}}{\gamma_{r-1}} D_{r-1} P_{n-1}^{(r+1)}, \\ \phi \frac{d}{dx} P_{n-1}^{(r+1)} &= -\frac{C_{n-r+1} - C_{r}}{2} P_{n-1}^{(r+1)} + \frac{D_{n+r}}{\gamma_{n+r}} P_{n}^{(r+1)} - \frac{D_{r}}{\gamma_{r}} P_{n}^{(r)}, \\ \phi \frac{d}{dx} P_{n-1}^{(r)} &= \frac{C_{n+r-1} - C_{r}}{2} P_{n-1}^{(r)} - D_{n+r+1} P_{n}^{(r)} + \frac{\gamma_{r}}{\gamma_{r-1}} D_{r-1} P_{n}^{(r+1)}, \\ \phi \frac{d}{dx} P_{n}^{(r+1)} &= \frac{C_{n+r+1} + C_{r}}{2} P_{n}^{(r+1)} - D_{n+r-1} P_{n-1}^{(r+1)} - \frac{D_{r}}{\gamma_{r}} P_{n+1}^{(r)}, \end{split}$$

where the polynomial coefficients  $C_r$  and  $D_r$  are given by

$$C_r \equiv C_r(x) = E_r(x;1) + F_r(x;1), \ D_r \equiv D_r(x) = \lim_{q \to 1} H_r(x;q).$$
 (6.58)

Notice that the previous four differential equations, already known earlier [Magnus, 1984]. [Dzoumba, 1985], are exactly those which allow Belmehdi et al. (1991) to derive the fourth-order differential equation satisfied by the associated orthogonal polynomial of the  $\mathcal{D}$ -Laguerre-Hahn class. The coefficients  $C_r$  and  $D_r$ , for the associated  $\mathcal{D}$ -classical orthogonal polynomials, are given by

$$C_r = (x - \beta_r)(2r\phi_2 + \psi_1) - \phi', \ \frac{D_r}{\gamma_r} = (2r - 1)\phi_2 + \psi_1,$$

where  $\phi$  and  $\psi$  are the polynomials appearing in the Pearson differential equation satisfied by the regular linear functional  $\mathcal{L}$ :  $\mathcal{D}(\phi\mathcal{L}) = \psi\mathcal{L}$ , with

$$\phi(x) = \phi_2 x^2 + \phi_1 x + \phi_0, \ \psi(x) = \psi_1 x + \psi_0.$$

# 6.3.2 Fourth-order difference equation for the rth associated $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials

We deduce the difference equation satisfied by the associated  $D_{q,\omega}$ -Laguerre-Hahn class from Theorem 6.4.

Consider  $\mathcal{L}$  a  $\mathcal{D}_q$ -Laguerre-Hahn linear functional and  $\{P_n\}_{n\in\mathbb{N}}$  the corresponding family of monic orthogonal polynomials. Let  $P_n^{(r)}$  and  $\mathcal{L}^{(r)}$  be the rth associated of  $P_n$  and  $\mathcal{L}$ , respectively. The Stieltjes function  $S_r$  of  $\mathcal{L}^{(r)}$  satisfies (6.1):

$$\phi(qx)\mathcal{D}_q S_r(x) = G_r(x;q)S_r(x)\mathcal{G}_q S_r(x) + E_r(x;q)S_r(x) + F_r(x;q)\mathcal{G}_q S_r(x) + H_r(x;q), \quad r \ge 0,$$

where  $\phi$ ,  $E_r$ ,  $F_r$ ,  $G_r$  and  $H_r$  are polynomials in x and depending on q. It follows from Theorem 6.4 that  $P_n^{(r)}$  satisfies the fourth-order q-difference equation (6.54) where the polynomials  $I_j(r,n,q;x)$  depend on the polynomial coefficients  $\phi$ ,  $E_r$ ,  $F_r$ ,  $G_r$  and  $H_r$ . To be more explicit, we denote  $I_j(r,n,q;x) = I_j(r,n,q;x;\phi,E_r,F_r,G_r,H_r)$ .

It results from Theorem 4.2 and Lemma 2.5 that the polynomials  $\{\tilde{P}_n\}_{n\in\mathbb{N}}$ , with  $\tilde{P}_n = \mathcal{T}_{\frac{-\omega}{1-q}}P_n$ , are orthogonal with respect to  $\tilde{\mathcal{L}} = \mathcal{T}_{\frac{-\omega}{1-q}}\mathcal{L}$  and that the Stieltjes function  $\tilde{S}_r$  of  $\tilde{\mathcal{L}}^{(r)}$  satisfies

$$\tilde{\phi}(qx+\omega)D_{q,\omega}\tilde{S}_r(x) = \tilde{G}_r(x;q,\omega)\tilde{S}_r(x)A_{q,\omega}\tilde{S}_r(x) + \tilde{E}_r(x;q,\omega)\tilde{S}_r(x) + \tilde{F}_r(x;q,\omega)A_{q,\omega}\tilde{S}_r(x) + \tilde{H}_r(x;q,\omega), \quad r > 0,$$

where  $\tilde{\phi} = \mathcal{T}_{\frac{-\omega}{1-q}} \phi$ , and  $\tilde{\Phi}(x;q,\omega) = \mathcal{T}_{\frac{-\omega}{1-q}} \Phi(x;q), \ \Phi \in \{E_r,F_r,G_r,H_r\}$ . We state the following

Theorem 6.5 The rth associated  $\tilde{P}_n^{(r)}$  of the polynomial  $\tilde{P}_n$  satisfies the fourth-order difference equation

$$\sum_{i=0}^{4} I_{j}^{g}(r, n, q, \omega; x) A_{q,\omega}^{j} \tilde{P}_{n}^{(r)}(x) = 0.$$
 (6.59)

where the polynomial coefficient  $I_j^g(r,n,q,\omega;x)$  depending on  $\tilde{\phi},\ \tilde{E}_r,\ \tilde{F}_r,\ \tilde{G}_r,\ \tilde{H}_r$  and denoted  $I_j^g(r,n,q,\omega;x)=I_j^g(r,n,q,\omega;x;\tilde{\phi},\tilde{E}_r,\tilde{F}_r,\tilde{G}_r,\tilde{H}_r)$ , are given by

$$I_j^g(r,n,q,\omega;x;\tilde{\phi},\tilde{E}_r,\tilde{F}_r,\tilde{G}_r,\tilde{H}_r) = I_j(r,n,q;x - \frac{\omega}{1-q};\phi,E_r,F_r,G_r,H_r).$$

*Proof:* We replace in (6.54)  $P_n^{(r)}$  by  $\mathcal{T}_{\frac{n}{1-2}}\tilde{P}_n^{(r)}$ , i.e.,

$$P_n^{(r)} = \mathcal{T}_{\frac{\omega}{1-q}} \tilde{P}_n^{(r)},$$

and obtain an equation which multiplied by the operator  $\mathcal{T}_{\frac{-\omega}{1-\alpha}}$  gives

$$\sum_{j=0}^{4} \mathcal{T}_{\frac{-\omega}{1-q}} I_{j}(r, n, q; x) \mathcal{T}_{\frac{-\omega}{1-q}} \mathcal{G}_{q}^{j} \mathcal{T}_{\frac{\omega}{1-q}} \tilde{P}_{n}^{(r)}(x) = 0.$$

We therefore use the relation (4.23):  $\mathcal{T}_{\frac{-\omega}{1-q}} \mathcal{G}_q^j \mathcal{T}_{\frac{\omega}{1-q}} = A_{q,\omega}^j$  to transform the previous equation in

$$\sum_{j=0}^{4} I_{j}(r, n, q; x - \frac{\omega}{1-q}) A_{q,\omega}^{j} \tilde{P}_{n}^{(r)}(x) = 0.$$

We complete the proof by identifying the coefficients of  $A_{q,\omega}^{j} \tilde{P}_{n}^{(r)}(x)$  in the previous equation with the ones of (6.59).

# 6.3.3 Fourth-order difference equation for the rth associated $\Delta$ -Laguerre-Hahn orthogonal polynomials

From the fourth-order difference equation satisfied by the rth associated orthogonal polynomial of the  $D_{q,\omega}$ -Laguerre-Hahn class, we deduce, again, by the limit process the fourth-order difference equation satisfied by the rth associated orthogonal polynomial of the  $\Delta$ -Laguerre-Hahn class [Foupouagnigni et al., 1998b]

$$\sum_{j=0}^4 I_j^{\Delta}(r,n;x) \, \mathcal{T}^j \, \tilde{P}_n^{(r)}(x) = 0,$$

with

$$I_{\boldsymbol{j}}^{\Delta}(\boldsymbol{r},\boldsymbol{n};\boldsymbol{x}) = \lim_{\boldsymbol{c} \to 1} \lim_{\boldsymbol{a} \to 1} I_{\boldsymbol{j}}^{\boldsymbol{q}}(\boldsymbol{r},\boldsymbol{n},\boldsymbol{q},\boldsymbol{\omega};\boldsymbol{x};\tilde{\boldsymbol{o}},\tilde{E}_{r},\tilde{F}_{r},\tilde{G}_{r},\tilde{H}_{r}).$$

#### 6.4 Application of difference equations to classical situations

#### 6.4.1 Coefficients $E_r$ , $F_r$ and $H_r$ for classical situations

Here we suppose that the regular linear functional  $\mathcal{L}$  satisfies the  $\mathcal{D}_q$ -Pearson linear functional equation,  $\mathcal{D}_q(\phi \mathcal{L}) = \psi \mathcal{L}$ , where  $\phi$  is a polynomial of degree at most two, and  $\psi$  is a first-degree polynomial given by

$$\phi(x) = \phi_2 x^2 + \phi_1 x + \phi_0, \ \psi(x) = \psi_1 x + \psi_0, \ |\psi_1|(|\phi_3| + |\phi_2| + |\phi_1| + |\phi_0|) \neq 0.$$

It follows from Proposition 6.2 that  $H_r$  is constant and  $E_r$  and  $F_r$  are polynomials of degree at most one. Let us compute first polynomials  $E_r$ .  $F_r$ , and  $\frac{H_r}{r}$  in terms of  $\phi$  and  $\psi$ . The first  $\mathcal{D}_q$ -derivative of (6.4). (6.5) and the first and second  $\mathcal{D}_q$ -derivative of (6.6) give, respectively,

$$\mathcal{D}_q E_{r+1} = q \frac{H_r}{\gamma_r} - \mathcal{D}_q F_r, \quad r \ge 0, \tag{6.60}$$

$$\mathcal{D}_q F_{r+1} = \frac{H_r}{\gamma_r} - \mathcal{D}_q E_r, \quad r \ge 0, \tag{6.61}$$

$$qE_r + F_r = -q\mathcal{G}_q \mathcal{D}_q \phi - (q^2x - \beta_r) \mathcal{D}_q E_r - (qx - \beta_r) \mathcal{D}_q F_r + (1+q)(qx - \beta_r) \frac{H_r}{\gamma_r}, \ r \ge 0,$$

$$(6.62)$$

$$q\mathcal{D}_q E_r + \mathcal{D}_q F_r = q \frac{H_r}{\gamma_r} - q^2 \phi_2, \ r \ge 0.$$
 (6.63)

In the first step, we solve equations (6.60), (6.61) and (6.63), taking into account the initial conditions (6.16)

$$H_{0}(x,q) = \mathcal{L}\theta_{0}\psi(x) - (\mathcal{D}_{q}\mathcal{L})\theta_{0}\phi(qx) - \mathcal{L}\theta_{0}\mathcal{D}_{q}\psi = (\psi_{1} - \phi_{2})\gamma_{0},$$

$$\mathcal{D}_{q} E_{0}(x,q,w) = \psi' - \mathcal{D}_{q}^{2}\phi(x) = \psi_{1} - (1+q)\phi_{2}.$$

$$F_{0}(x,q) = 0.$$

and obtain [Foupouagnigni et al., 1998e]

$$\mathcal{D}_{q} E_{r} = q^{r} \psi_{1} + ([r]_{q} - [2]_{q}) \phi_{2}, 
\mathcal{D}_{q} F_{r} = q^{2-r} [r]_{q} \phi_{2}, 
\frac{H_{r}}{\gamma_{r}} = q^{r} \psi_{1} + q^{-r} ([2r]_{q} - 1) \phi_{2}.$$
(6.64)

In a second step, we compute the coefficients  $E_r$  and  $F_r$  using (6.62), (6.64) and the equation obtained after iterating (6.9):

$$E_r - F_r = ((q-1)x) \sum_{k=0}^{r-1} \frac{H_k}{\gamma_k} + \psi - \mathcal{D}_q \phi$$

and we get huge expressions for  $E_r$  and  $F_r$ . Finally, use of Maple V.4 and the simplification procedures for q-hypergeometric terms developed in [Böing et al., 1998] allow us to have readable expressions for  $E_r$  and  $F_r$  [Foupouagnigni et al., 1998e],

$$E_{r}(x;q) = ((q^{r} - q^{2}) \phi_{2} + q^{r} \psi_{1} (q - 1))((q^{r} - q) (q^{r} x q - q^{r} x + x q^{2} - x q) \phi_{2}$$

$$+ (q - 1) (q^{r} - q) \phi_{1} + q^{r} (q - 1) (q^{r} x q \psi_{1} + q \psi_{0} - q^{r} x \psi_{1} - \psi_{0})) /$$

$$(q - 1)^{2} (q^{r} - q) (q^{r} + q) \phi_{2} + (q^{r})^{2} \psi_{1} (q - 1)),$$

$$(6.65)$$

$$F_{r}(x;q) = (-1+q^{r}) (q^{r}-q) (q^{r} x q^{2}-q^{r} x q + x q^{3}-x q^{2}) \phi_{2}^{2} + (q^{r} (q-1) (q^{r}-q) \phi_{1}+q^{r} (q-1) (q^{r} x q^{2} \psi_{1}-\psi_{0} q^{2}-q^{r} x q \psi_{1}+q \psi_{0})) \phi_{2} + \psi_{1} (q^{r})^{2} (q-1)^{2} \phi_{1}) q /(q^{r} (q-1)^{2} ((q^{r}-q) (q^{r}+q) \phi_{2}+(q^{r})^{2} \psi_{1} (q-1))).$$

$$(6.66)$$

Remark 6.3 1. For q-classical situations, coefficients  $K_2$ .  $K_4$ ,  $K_6$  and  $K_8$  (see (6.42)) are constant with respect to the variable x.

- 2. For r = 0,  $K_4 = 0$ , then it follows from (6.40). (6.42) and (6.64)-(6.66) that (6.36) and (6.37) (for r = 0 and for  $\mathcal{D}_q$ -classical situations) are respectively, equivalent to equations (5.7) and (5.12).
- 3. When the regular functional  $\mathcal{L}$  is  $\mathcal{D}_q$ -semi-classical,  $K_4 = \gamma_0 G_0 = 0$  (for r = 0). This allows us to obtain the factored form of the fourth-order difference equation for the first associated  $\mathcal{D}_q$ -semi-classical orthogonal polynomials.

#### 6.4.2 Results on general associated $\mathcal{D}_q$ -classical orthogonal polynomials

The coefficients  $I_j(r,n,q;x)$  (see (6.53)) can be computed using the algorithm described in (6.21)-(6.53). But this involves heavy computations due to huge expressions containing powers of q which need to be factored. To avoid these difficulties, we again used Maple V.4 to compute symbolically the coefficients  $I_j(r,n,q;x)$  and to simplify common factors as was done for the associated classical discrete orthogonal polynomial [Foupouagnigni et al., 1997c] to obtain

Theorem 6.6 (Foupouagnigni et al., 1998c, 1998e) The coefficients  $I_j(r,n,q;x)$  of the fourth-order q-difference equation satisfied by the rth associated  $\mathcal{D}_q$ -classical orthogonal polynomials are

given by

$$I_{4} = K_{9,2}(K_{10,0}K_{10,1} - K_{12,0}K_{12,1}),$$

$$I_{3} = K_{10,2}(K_{12,0}(k_{2,3}K_{12,1} + K_{13,1}) - K_{10,0}K_{10,1}(K_{2,3} + K_{5,2})) + K_{9,1}K_{10,0}K_{12,2},$$

$$I_{2} = K_{10,1}(K_{10,2}(K_{10,0}K_{10,1} + K_{13,0} - K_{5,1}K_{12,0}) - K_{9,1}K_{10,0}) - K_{12,1}(K_{12,2}K_{13,0} + k_{11,2}K_{12,0}),$$

$$I_{1} = K_{10,0}K_{12,2}(k_{2,2}K_{12,0} + K_{13,0}) + K_{10,2}K_{12,0}(K_{9,0} - K_{10,0}K_{10,1}),$$

$$I_{0} = K_{9,-1}(K_{10,1}K_{10,2} - K_{12,1}K_{12,2}),$$

$$(6.67)$$

where the coefficients  $K_{i,j}$  are obtained from (6.41), (6.42) and

$$\begin{array}{lcl} K_{9}(x) & = & K_{7}(qx)K_{1}(qx) - K_{3}(x)K_{8}(x), & K_{10}(x) = K_{7}(qx) + K_{1}(x). \\ K_{11}(x) & = & K_{2}(qx)K_{2}(x) + K_{4}(x)K_{6}(x), & K_{12}(x) = K_{2}(qx) + K_{5}(x), \\ K_{13}(x) & = & K_{5}(qx)K_{5}(x) + K_{4}(x)K_{6}(x), & K_{14}(x) = K_{5}(qx) + K_{2}(x), \end{array}$$

with coefficients  $E_n$ ,  $F_n$ , and  $\frac{H_n}{\gamma_n}$  given by (6.64)-(6.66).

Notice that coefficients  $I_j(r, n, q; x)$ , are given in appendix III, for some  $\mathcal{D}_q$ -classical orthogonal polynomials.

# 6.4.3 Fourth-order differential equation for the rth associated $\mathcal{D}$ -classical orthogonal polynomials

From the relation  $\lim_{q\to 1} \mathcal{D}_q = \frac{d}{dx}$  and by the limit process, we recover using Maple V Release 4 the fourth-order differential equation satisfied by the rth associated classical continuous orthogonal polynomials (see [Belmehdi et al., 1991], [Zarzo et al., 1993]) [Lewanowicz, 1995], [Foupouagnigni et al., 1998e]). This equation is given in terms of the factored form of the fourth-order differential equation satisfied by the first associated classical continuous orthogonal polynomials as already done earlier [Lewanowicz, 1995].

$$\mathcal{O}^{c}(r,n;x)P_{n}^{(r)}(x)=0,$$
 (6.68)

where

$$\mathcal{O}(r,n,q;x) = \sum_{j=0}^4 I_j(r,n,q;x) \mathcal{G}_q^j$$

and

$$\mathcal{O}^{c}(r, n; x) = \frac{1}{2\phi(x)} \lim_{\eta(r, n)} \lim_{q \to 1} \frac{\mathcal{O}(r, n, q; x)}{q^{2}(q - 1)^{2}x^{2}}$$

$$= \mathcal{Q}_{2,n}^{\star\star c} \mathcal{Q}_{2,n}^{\star c} + (1 - r)\zeta(n, r)\mathcal{Q}_{2,n}^{c}, \qquad (6.69)$$

with.

$$Q_{2n}^{c} = 2\phi \frac{d^{2}}{dx^{2}} + 3\phi' \frac{d}{dx} - n(n+2)\phi'' \mathcal{I}_{d}.$$

$$\zeta(r,n) = ((n+r-1)\phi'' + 2\psi'),$$

$$\eta(r,n) = (n+1)((n-2r-2)\phi'' + 2\psi').$$

 $Q_{2,r}^{**}$  and  $Q_{2,n}^{**c}$  are given by (5.13, and (5.14).

# 6.4.4 Fourth-order difference equation for the rth associated $\Delta$ -classical orthogonal polynomials

We first deduce the fourth-order difference equation for the rth associated  $D_{q,\omega}$ -classical orthogonal polynomials using Theorems 4.2 and 6.6, then deduce the difference equation for the rth associated  $\Delta$ -classical orthogonal polynomial by the limit process:  $\lim_{\omega \to 1, q \to 1} D_{q,\omega} = \Delta$ .

We assume that  $\{P_n\}_{n\in\mathbb{N}}$ , orthogonal with respect to  $\mathcal{L}$ , is  $\mathcal{D}_q$ -classical with  $\mathcal{L}$  satisfying  $\mathcal{D}_q(\phi\mathcal{L})=\psi\mathcal{L}$  where  $\phi$  and  $\psi$  are polynomials of degree at most two and degree one, respectively. The rth associated  $P_n^{(r)}$  satisfies Theorem 6.6. It yields from Theorem 4.2 and 6.6 that  $\{\bar{P}_n\}_{n\in\mathbb{N}}$ , with  $\bar{P}_n=\mathcal{T}_{\frac{-\omega}{1-q}}P_n$ , is orthogonal with respect to  $\bar{\mathcal{L}}=\mathcal{T}_{\frac{-\omega}{1-q}}\mathcal{L}$  and  $\bar{\mathcal{L}}$  is  $D_{q,\omega}$ -classical satisfying  $D_{q,\omega}(\bar{\phi}\bar{\mathcal{L}})=\bar{\psi}\bar{\mathcal{L}}$ , where  $\bar{\phi}=\mathcal{T}_{\frac{-\omega}{1-q}}\phi$  and  $\bar{\psi}=\mathcal{T}_{\frac{-\omega}{1-q}}\psi$ . Again, we use Theorem 4.2 and 6.6 to conclude that the rth associated  $\bar{P}_n^{(r)}$  of the  $D_{q,\omega}$ -classical orthogonal polynomial  $\bar{P}_n$  satisfies the fourth-order difference equation

$$\sum_{j=0}^{4} I_{j}(r, n, q, \omega; x; \bar{\phi}, \bar{\psi}) A_{q,\omega}{}^{j} \bar{P}_{n}^{(r)}(x) = 0.$$

We, therefore, use the limit process to state the following:

Theorem 6.7 (Foupouagnigni et al., 1998c) Let  $\tilde{P}_n$  be the classical orthogonal polynomials of a discrete variable associated with the linear functional  $\tilde{\mathcal{L}}$  satisfying  $\Delta(\tilde{\phi}) = \tilde{\psi}\tilde{\mathcal{L}}$ . Then, the rth associated  $\tilde{P}_n^{(r)}$  of  $\tilde{P}_n$  satisfies the fourth-order difference equation

$$\sum_{j=0}^4 I_j^{\Delta}(r,n;x) \, \mathcal{T}^j \, \tilde{P}_n^{(r)}(x) = 0,$$

where the coefficients  $I_i^{\Delta}$  are given by

$$I_{4}^{\Delta} = K_{9,2}(K_{10,0}K_{10,1} - K_{12,0}K_{12,1}),$$

$$I_{3}^{\Delta} = K_{10,2}(K_{12,0}(k_{2,3}K_{12,1} + K_{13,1}) - K_{10,0}K_{10,1}(K_{2,3} + K_{5,2})) + K_{9,1}K_{10,0}K_{12,2},$$

$$I_{2}^{\Delta} = K_{10,1}(K_{10,2}(K_{10,0}K_{10,1} + K_{13,0} - K_{5,1}K_{12,0}) - K_{9,1}K_{10,0}) - K_{12,1}(K_{12,2}K_{13,0} + k_{11,2}K_{12,0}),$$

$$I_{1}^{\Delta} = K_{10,0}K_{12,2}(k_{2,2}K_{12,0} + K_{13,0}) + K_{10,2}K_{12,0}(K_{9,0} - K_{10,0}K_{10,1}),$$

$$I_{0}^{\Delta} = K_{9,-1}(K_{10,1}K_{10,2} - K_{12,1}K_{12,2}).$$

$$(6.70)$$

with the notation:  $K_{i,j} \equiv k_i(x+j)$ . Coefficients  $k_i$  read as:

$$\begin{array}{lll} k_1(x) & = & \tilde{\phi}(x+1) + \tilde{E}_{n+r+1}(x), \ k_2(x) = \tilde{\phi}(x+1) - \tilde{F}_r(x), \ k_3(x) = \frac{\tilde{H}_{n+r}}{\tilde{\gamma}_{n+r}}, \\ k_4(x) & = & \begin{cases} & \tilde{\gamma}_r \frac{\tilde{H}_{r-1}}{\tilde{\gamma}_{r-1}} & \text{if} \quad r \geq 1 \\ & 0 \quad \text{if} \quad r = 0 \end{cases}, \ k_5(x) = \tilde{\phi}(x+1) + \tilde{E}_r(x), \ k_6(x) = -\frac{\tilde{H}_r}{\tilde{\gamma}_r}, \\ k_7(x) & = & \tilde{\phi}(x+1) - \tilde{F}_{n+r+1}(x), \ k_8(x) = -\tilde{H}_{n+r+1}, \ k_9(x) = k_7(x+1)k_1(x+1) - k_3(x)k_8(x), \\ k_{10}(x) & = & k_7(x+1) + k_1(x), \ k_{11}(x) = k_2(x+1)k_2(x) + k_4(x)k_6(x), \ k_{12}(x) = k_2(x+1) + k_5(x), \\ k_{13}(x) & = & k_5(x+1)k_5(x) + k_4(x)k_6(x), \ k_{14}(x) = k_5(x+1) + k_2(x), \end{array}$$

with

$$\begin{split} \tilde{E}_r(x,\tilde{\phi},\tilde{\psi}) &= \lim_{\omega \to 1, q \to 1} E_r(x - \frac{\omega}{1 - q}, \mathcal{T}_{\frac{\omega}{1 - q}}\tilde{\phi}, \mathcal{T}_{\frac{\omega}{1 - q}}\tilde{\psi}) \\ &= (\tilde{\phi}_2 \, n - 2\,\tilde{\phi}_2 + \psi_1) \, x \\ &\quad + \frac{(\tilde{\phi}_2 \, n - 2\,\tilde{\phi}_2 + \psi_1) \, (\tilde{\phi}_2 \, n^2 - \tilde{\phi}_2 + \tilde{\phi}_1 \, n + \psi_1 \, n - \tilde{\phi}_1 + \psi_0)}{2 \, (n - 1) \, \tilde{\phi}_2 + \psi_1}, \\ \tilde{F}_r(x,\tilde{\phi},\tilde{\psi}) &= \lim_{\omega \to 1, q \to 1} F_r(x - \frac{\omega}{1 - q}, \mathcal{T}_{\frac{\omega}{1 - q}}\tilde{\phi}, \mathcal{T}_{\frac{\omega}{1 - q}}\tilde{\psi}) \\ &= \tilde{\phi}_2 \, x \, n \\ &\quad - \frac{n \, (\tilde{\phi}_2 \, \tilde{\phi}_1 + 3 \, \tilde{\phi}_2^2 + \tilde{\phi}_2^2 \, n^2 - 4 \, \tilde{\phi}_2^2 \, n - 2 \, \tilde{\phi}_2 \, \psi_1 - \tilde{\phi}_2 \, \tilde{\phi}_1 \, n + \psi_1 \, n \, \tilde{\phi}_2 + \tilde{\phi}_2 \, \psi_0 - \psi_1 \, \tilde{\phi}_1)}{2 \, (n - 1) \, \tilde{\phi}_2 + \psi_1}, \\ \tilde{H}_r &= \lim_{\omega \to 1, q \to 1} H_r(x - \frac{\omega}{1 - q}, \mathcal{T}_{\frac{\omega}{1 - q}}\tilde{\phi}, \mathcal{T}_{\frac{\omega}{1 - q}}\tilde{\phi}, \mathcal{T}_{\frac{\omega}{1 - q}}\tilde{\psi}) = ((2r - 1)\tilde{\phi}_2 + \tilde{\psi}_1)\tilde{\gamma}_r. \end{split}$$

It should be mentioned that  $E_r$ ,  $F_r$  and  $H_r$  are given by (6.64). (6.65) and (6.66), respectively. The coefficients  $\beta_n$ ,  $\gamma_n$  are given in Lemma 7.1.

$$\begin{split} \tilde{\beta}_n & \equiv & \tilde{\beta}_n(\tilde{\phi},\tilde{\psi}) = \lim_{\omega \to 1, q \to 1} (\beta_n(\mathcal{T}_{\frac{\omega}{1-q}}\tilde{\phi},\mathcal{T}_{\frac{\omega}{1-q}}\tilde{\psi}) + \frac{\omega}{1-q}) \\ & = & -\frac{\tilde{\phi}_2(\tilde{\psi}_1 + 2\tilde{\phi}_1)n^2 - (\tilde{\psi}_1 + 2\tilde{\phi}_1)(-\tilde{\psi}_1 + \tilde{\phi}_2)n - \tilde{\psi}_0(-\tilde{\psi}_1 + 2\tilde{\phi}_2)}{(\tilde{\psi}_1 + 2\tilde{\phi}_2n)(2(n-1)\tilde{\phi}_2 + \tilde{\psi}_1)}, \end{split}$$

$$\begin{split} \tilde{\gamma}_{n} & \equiv & \tilde{\gamma}_{n}(\tilde{\phi},\tilde{\psi}) = \lim_{\omega \to 1, q \to 1} \gamma_{n}(\mathcal{T}_{\frac{\omega}{1-q}}\tilde{\phi},\mathcal{T}_{\frac{\omega}{1-z}}\tilde{\psi}) \\ & = & -((n-2)\,(n-1)^{4}\,\tilde{\phi}_{2}^{4} \\ & + (4\,(n-2)\,(n-1)^{2}\,\tilde{\phi}_{0} + (n-1)^{2}\,\,3\,\tilde{\psi}_{1}\,n^{2} + 2\,n\,\tilde{\psi}_{0} - 8\,\tilde{\psi}_{1}\,n - 4\,\tilde{\psi}_{0} + 5\,\tilde{\psi}_{1}))\,\tilde{\phi}_{2}^{3} + \\ & -(n-2)\,(n-1)^{2}\,\tilde{\phi}_{1}^{2} - \tilde{\psi}_{1}\,(n-2)\,(n-1)^{2}\,\tilde{\phi}_{1} + 4\,\tilde{\psi}_{1}\,(n-1)\,(-3+2\,n)\,\tilde{\phi}_{0} \\ & + (\tilde{\psi}_{0} + \tilde{\psi}_{1}\,n - \tilde{\psi}_{1})\,(n\,\tilde{\psi}_{0} - 2\,\tilde{\psi}_{0} + 4\,\tilde{\psi}_{1} - 7\,\tilde{\psi}_{1}\,n + 3\,\tilde{\psi}_{1}\,n^{2}))\tilde{\phi}_{2}^{2} \\ & + (-\tilde{\psi}_{1}\,(n-1)\,(-3+2\,n)\,\tilde{\phi}_{1}^{2} - \tilde{\psi}_{1}\,-5\,\tilde{\psi}_{1}\,n + 3\,\tilde{\psi}_{1} - 2\,\tilde{\psi}_{0} + n\,\tilde{\psi}_{0} + 2\,\tilde{\psi}_{1}\,n^{2})\,\tilde{\phi}_{1} \\ & + \tilde{\psi}_{1}^{2}\,(-6+5\,n)\,\tilde{\phi}_{0} + \tilde{\psi}_{1}\,(\tilde{\psi}_{0} - \tilde{\psi}_{1}\,n - \tilde{\psi}_{1})^{2})\tilde{\phi}_{2} - \tilde{\psi}_{1}^{2}\,(n-1)\,\tilde{\phi}_{1}^{2} - \tilde{\psi}_{1}^{2}\,(\tilde{\psi}_{0} + \tilde{\psi}_{1}\,n - \tilde{\psi}_{1})\,\tilde{\phi}_{1} \\ & + \tilde{\phi}_{0}\,\tilde{\psi}_{1}^{3})n\,\Big/(((2\,n-1)\,\tilde{\phi}_{2} + \tilde{\psi}_{1})\,((-2+2\,n)\,\tilde{\phi}_{2} + \tilde{\psi}_{1}^{-2}\,((-3+2\,n)\,\tilde{\phi}_{2} + \tilde{\psi}_{1})). \end{split}$$

**Remark 6.4** The coefficients  $I_0^{\Delta}(r,n,x)$ , as well as operators  $\mathcal{D}_{r,n}$ ,  $\mathcal{N}_{r,n}$ .  $\tilde{\mathcal{D}}_{r,n}$  and  $\tilde{\mathcal{N}}_{r,n}$  (see Theorem 6.3), are given in Appendix II for all classical orthogonal polynomial of a discrete variable. They are obviously deduced from those of q-classical case by Theorem 4.2.

## Chapter 7

# Three-term recurrence relation coefficients for classical situations

#### 7.1 Introduction

We describe the method used to compute the coefficients  $\beta_n$  and  $\gamma_n$  for the  $\mathcal{D}_q$ -classical case.

This method, already used in [Koepf et al., 1996] but for classical continuous and classical discrete cases, consists to derive from the second order difference equation satisfied by  $\{P_n\}_{n\in\mathbb{N}}$  (3.54) a system of equations satisfied by  $T_{n,1}$ ,  $T_{n,2}$  and  $\lambda_{n,0}$  [Foupouagnigni et al., 1998a], then solve these equations and deduce coefficients  $\beta_n$  and  $\gamma_n$ .

# 7.2 Three-term recurrence relation coefficients for $\mathcal{D}_q$ -classical situations

#### 7.2.1 Coefficients $T_{n,1}$ and $T_{n,2}$

Let  $\mathcal{L}$  be a  $\mathcal{D}_q$ -classical linear functional satisfying

 $\mathcal{D}_q(\phi\mathcal{L}) = \psi\mathcal{L}$ , where  $\phi$  is of degree at most two and  $\psi$  a first-degree polynomial i.e.,

$$\phi(x) = \phi_2 x^2 + \phi_1 x + \phi_0 \cdot \psi(x) = \psi_1 x + \psi_0, \ |\psi_1|(|\phi_2| + |\phi_1| + |\phi_0|) \neq 0.$$
 (7.1)

It follows from Theorem 3.1 that  $\{P_n\}_{n\in\mathbb{N}}$  satisfies

$$\phi \mathcal{D}_q \mathcal{D}_{\perp} P_n + \psi \mathcal{D}_q P_n + \lambda_{n,0} P_n = 0 \qquad \forall n \in \mathbb{N},$$
(7.2)

with  $\lambda_{n,0}$  given by (3.55).

Use of the expansions (see [Foupouagnigni et al., 1998a], see also (8.8)),

$$P_n(x) = \sum_{j=0}^n T_{n,j} x^{n-j}, \ \mathcal{D}_q x^n = [n]_q x^{n-1}$$

allows us to write (7.2) as

$$\sum_{j=0}^{n} d_{n,j} x^{n-j} = 0. (7.3)$$

We compute the first three coefficients  $d_{n,j}$  and obtain, with  $\rho = q^n$ ,

$$d_{n,0} = q^4 \rho (q-1)^2 \lambda_{n,0} - q^4 (\rho - 1) (-\rho + q) \phi_2 + q^4 \rho (\rho - 1) (q-1) \psi_1,$$

$$d_{n,1} = (2 q^{3} (-\rho + q) (-\rho + q^{2}) \phi_{1} - 2 q^{3} \rho (q - 1) (-\rho + q) \psi_{0}) A(n)$$

$$+ (2 q^{4} \rho (q - 1)^{2} \lambda_{n,0} + 2 q^{2} (-\rho + q^{2}) (q^{3} - \rho) \phi_{2} - 2 q^{2} \rho (q - 1) (-\rho + q^{2}) \psi_{1}) B(n)$$

$$- 2 q^{4} (\rho - 1) (-\rho + q) \phi_{0}$$

$$d_{n,2} = (q^{4} \rho (q - 1)^{2} \lambda_{n,0} + q^{3} (-\rho + q) (-\rho + q^{2}) \phi_{2} - q^{3} \rho (q - 1) (-\rho + q) \psi_{1}) A(n)$$

$$- q^{4} (\rho - 1) (-\rho + q) \phi_{1} + q^{4} \rho (\rho - 1) (q - 1) \psi_{0}.$$

We solve the equations  $d_{n,0}=0$ ,  $d_{n,1}=0$  and  $d_{n,2}=0$  in terms of  $\lambda_{n,0}$ ,  $T_{n,1}$  and  $T_{n,2}$  and get

$$\begin{split} \lambda_{n,0} &= \frac{(\rho-1)\left(-\phi_{2}\,\rho + \rho\,\psi_{1} - q\,\rho\,\psi_{1} + \phi_{2}\,q\right)}{\rho\,(q-1)^{2}} = -[n]_{q}\,(\psi_{1} + [n-1]_{\frac{1}{q}}\,\frac{\phi_{2}}{q}), \\ T_{n,1} &= \frac{(-1+\rho)\,q\,(q-1)\,(q-\rho)\,\phi_{1} - \rho\,q\,(q-1)^{2}\,(-1+\rho)\,\psi_{0}}{(q-1)^{2}\,(q^{2}\,\phi_{2} - \psi_{1}\,q\,\rho^{2} + \psi_{1}\,\rho^{2} - \phi_{2}\,\rho^{2})}. \\ T_{n,2} &= \frac{1}{2}\,(2\,q^{2}\,(-1+\rho)\,(q-1)^{3}\,(q+\rho)\,(q-\rho)^{2}\,\phi_{0}\,\phi_{2} \\ &\quad - 2\,q^{2}\,(-1+\rho)\,(q-1)^{2}\,(-\rho+q^{2})\,(q-\rho)^{2}\,\phi_{1}^{2} \\ &\quad + 2\,q^{2}\,\rho\,(-1+\rho)\,(q-1)^{3}\,(q-\rho)\,(-2\,\rho+q^{2} + q)\,\psi_{0}\,\phi_{1} \\ &\quad - 2\,\rho^{2}\,q^{2}\,(-1+\rho)\,(q-1)^{4}\,(q-\rho)\,\psi_{1}\,\phi_{0} - 2\,\rho^{2}\,q^{2}\,(-1+\rho)\,(q-1)^{4}\,(q-\rho)\,\psi_{0}^{2})\,\Big/ \\ &\quad ((q+1)\,(q-1)^{4}\,(-\psi_{1}\,q\,\rho^{2} + \psi_{1}\,\rho^{2} + \phi_{2}\,q^{3} - \phi_{2}\,\rho^{2})\,(q^{2}\,\phi_{2} - \psi_{1}\,q\,\rho^{2} + \psi_{1}\,\rho^{2} - \phi_{2}\,\rho^{2})). \end{split}$$

#### 7.2.2 Coefficients $\beta_n$ and $\gamma_n$ for $\mathcal{D}_q$ -classical orthogonal polynomials

We use the following identities already given in [Foupouagnigni, 1998a] (see also (8.8))

$$\beta_n = T_{n,1} - T_{n+1,1}, \ \gamma_n = T_{n,2} - T_{n+1,2} - \beta_n T_{n,1}$$

$$(7.4)$$

to compute the coefficients  $\beta_n$ ,  $\gamma_n$  and get:

**Lemma 7.1** (Medem, 1996) The coefficients  $\beta_n$  and  $\gamma_n$  of TTRR satisfied by the polynomials  $\{P_n\}_{n\in\mathbb{N}}$  (see (2.18)), orthogonal with respect to the  $\mathcal{D}_q$ -classical linear functional  $\mathcal{L}$ , satisfying  $\mathcal{D}_q(\phi\mathcal{L}) = \psi\mathcal{L}$ , where  $\phi$  and  $\psi$  are defined in (7.1), are given by:

$$\begin{split} \beta_{n}(q,\phi,\psi) &= -\rho((-(q+1)\,(-1+\rho)\,(-q+\rho)\,\phi_{1} - (q-1)\,(-\rho\,q^{2} + q - q\,\rho + \rho^{2})\,\psi_{0},\phi_{2} \\ &- \rho\,(q-1)\,(q+1)\,(-1+\rho)\,\psi_{1}\,\phi_{1} - \rho^{2}\,(q-1)^{2}\,\psi_{0}\,\psi_{1}) \,\Big/ \\ &- ((-1+\rho)\,(\rho+1)\,\phi_{2} + \rho^{2}\,(q-1)\,\psi_{1})\,(-(-q+\rho)\,(q+\rho)\,\phi_{2} - \rho^{2}\,(q-1)\,\psi_{1}),\\ \gamma_{n}(q,\phi,\psi) &= -(-1+\rho)\,((-\rho+q^{2})\,\phi_{2} - (q-1)\,\rho\,\psi_{1})((-q+\rho)^{2}\,(q+\rho)^{2}\,\phi_{0}\,\phi_{2}^{2} \\ &+ (-q\,\rho\,(-q+\rho)^{2}\,\phi_{1}^{2} - q\,\rho\,(q-1)\,(-q+\rho^{2}\,\psi_{0}\,\phi_{1} \\ &+ 2\,\rho^{2}\,(q-1)\,(-q+\rho)\,(q+\rho)\,\psi_{1}\,\phi_{0} + q^{2}\,\rho^{2}\,(q-1)^{2}\,\psi_{0}^{2})\phi_{2} \\ &- \rho^{2}\,q\,(q-1)(-q+\rho)\,\psi_{1}\,\phi_{1}^{2} - q\,\rho^{3}\,(q-1)^{2}\,\psi_{0}\,\psi_{1}\,\phi_{1} \\ &+ \rho^{4}\,(q-1)^{2}\,\psi_{1}^{2}\,\phi_{0})\rho\,q\,\Big/(((-q+\rho^{2})\,\phi_{2} + \rho^{2}\,(q-1)\,\psi_{1})). \end{split}$$

# 7.3 Three-term recurrence relation coefficients for $\mathcal{D}$ -classical situations

#### 7.3.1 Coefficients $\tilde{T}_{n,1}$ and $\tilde{T}_{n,2}$

If we denote by  $\tilde{T}_{n,1}$  and  $\tilde{T}_{n,2}$  the coefficients  $T_{n,1}$  and  $T_{n,2}$  when the linear functional  $\mathcal{L}$  is  $\mathcal{D}$ -classical satisfying  $\mathcal{D}(\phi\mathcal{L}) = \psi\mathcal{L}$ , we obtain  $\tilde{T}_{n,1}$  and  $\tilde{T}_{n,2}$  by limit process [Koepf et al., 1996]:

$$\tilde{T}_{n,1} = \lim_{q \to 1} T_{n,1} = \frac{n((n-1)\phi_1 + \phi_2)}{2(n-1)\phi_2 + \psi_1},$$

$$\tilde{T}_{n,2} = \lim_{q \to 1} T_{n,2} 
= \frac{1}{2} n (2 (n-1)^2 \phi_0 \phi_2 + (n-2) (n-1)^2 \phi_1^2 + \psi_0 (n-1) (2 n-3) \phi_1 
+ \psi_1 (n-1) \phi_0 + \psi_0^2 (n-1))) / (((2 n-3) \phi_2 + \psi_1) ((2 n-2) \phi_2 + \psi_1)).$$

We, therefore, use (7.4) to deduce coefficients  $\tilde{\beta}_n$  and  $\tilde{\gamma}_n$  and get [Lesky, 1985], [Koepf et al., 1996]....

$$\tilde{\beta}_n = \lim_{q \to 1} \beta_n = -\frac{(2 n (n-1) \phi_1 - 2 \psi_0) \phi_2 + 2 \phi_1 \psi_1 n + \psi_0 \psi_1}{(\psi_1 + 2 \phi_2 n) ((2 n-2) \phi_2 + \psi_1)},$$

$$\begin{split} \tilde{\gamma}_{n} &= \lim_{q \to 1} \gamma_{n} = -n(4 \phi_{0} (n-2) (n-1)^{2} \phi_{2}^{3} + (-(n-2) (n-1)^{2} \phi_{1}^{2} \\ &+ 4 (n-1) (2 n-3) \psi_{1} \phi_{0} + (n-2) \psi_{0}^{2}) \phi_{2}^{2} + (-\psi_{1} (n-1) (2 n-3) \phi_{1}^{2} \\ &+ (2-n) \psi_{0} \psi_{1} \phi_{1} + (-6+5 n) \psi_{1}^{2} \phi_{0} + \psi_{0}^{2} \psi_{1}) \phi_{2} \\ &+ \psi_{1}^{2} (-n+1) \phi_{1}^{2} - \psi_{0} \phi_{1} \psi_{1}^{2} + \phi_{0} \psi_{1}^{3}) \Big/ \\ &(((2 n-1) \phi_{2} + \psi_{1}) ((2 n-2) \phi_{2} + \psi_{1})^{2} ((2 n-3) \phi_{2} + \psi_{1})). \end{split}$$

# 7.4 Three-term recurrence relation coefficients for $\Delta$ -classical situations

We state the following:

Lemma 7.2 (Salto, 1995, Koepf et al., 1996) Let  $\{P_n\}_{n\in\mathbb{N}}$  be a family of monic polynomials, orthogonal with respect to the  $\Delta$ -classical linear functional  $\tilde{\mathcal{L}}$  satisfying  $\Delta(\tilde{\phi}\tilde{\mathcal{L}}) = \tilde{\psi}\tilde{\mathcal{L}}$ . If  $\tilde{\beta}_n$  and  $\tilde{\gamma}_n$  are the coefficients of TTRR satisfied by  $\{P_n\}_{n\in\mathbb{N}}$ , then, they are given by

$$\begin{split} \tilde{\beta}_{n} &= -\frac{\tilde{\phi}_{2} \left(\tilde{\psi}_{1} + 2\,\tilde{\phi}_{1}\right)n^{2} - \left(\tilde{\psi}_{1} + 2\,\tilde{\phi}_{1}\right)\left(-\tilde{\psi}_{1} + \tilde{\phi}_{2}\right)n - \tilde{\psi}_{0}\left(-\tilde{\psi}_{1} + 2\,\tilde{\phi}_{2}\right)}{\left(\tilde{\psi}_{1} + 2\,\tilde{\phi}_{2}\,n\right)\left(2\,(n-1)\,\tilde{\phi}_{2} + \tilde{\psi}_{1}\right)}. \\ \tilde{\gamma}_{n} &= -\left(\left(n-2\right)\left(n-1\right)^{4}\,\tilde{\phi}_{2}^{4} \times \right. \\ &\left. + \left(4\,(n-2)\,(n-1)^{2}\,\tilde{\phi}_{0} + \left(n-1\right)^{2}\left(3\,\tilde{\psi}_{1}\,n^{2} + 2\,n\,\tilde{\psi}_{0} - 8\,\tilde{\psi}_{1}\,n - 4\,\tilde{\psi}_{0} + 5\,\tilde{\psi}_{1}\right)\right)\,\tilde{\phi}_{2}^{3} \\ &\left. + \left(-\left(n-2\right)\left(n-1\right)^{2}\,\tilde{\phi}_{1}^{2} - \tilde{\psi}_{1}\,\left(n-2\right)\left(n-1\right)^{2}\,\tilde{\phi}_{1} + 4\,\tilde{\psi}_{1}\,\left(n-1\right)\left(-3 + 2\,n\right)\,\tilde{\phi}_{0} \right. \\ &\left. + \left(\tilde{\psi}_{0} + \tilde{\psi}_{1}\,n - \tilde{\psi}_{1}\right)\left(n\,\tilde{\psi}_{0} - 2\,\tilde{\psi}_{0} + 4\,\tilde{\psi}_{1} - 7\,\tilde{\psi}_{1}\,n + 3\,\tilde{\psi}_{1}\,n^{2}\right)\!\tilde{\phi}_{2}^{2} \right. \\ &\left. + \left(-\tilde{\psi}_{1}\,\left(n-1\right)\left(-3 + 2\,n\right)\,\tilde{\phi}_{1}^{2} - \tilde{\psi}_{1}\,\left(-5\,\tilde{\psi}_{1}\,n + 3\,\tilde{\psi}_{1} - 2\,\tilde{\psi}_{0} + n\,\tilde{\psi}_{0} + 2\,\tilde{\psi}_{1}\,n^{2}\right)\tilde{\phi}_{1} \right. \\ &\left. + \tilde{\psi}_{1}^{2}\left(-6 + 5\,n\right)\,\tilde{\phi}_{0} + \tilde{\psi}_{1}\left(\tilde{\psi}_{0} + \tilde{\psi}_{1}\,n - \tilde{\psi}_{1}\right)^{2}\tilde{\phi}_{2} - \tilde{\psi}_{1}^{2}\left(n-1\right)\tilde{\phi}_{1}^{2} - \tilde{\psi}_{1}^{2}\left(\tilde{\psi}_{0} + \tilde{\psi}_{1}\,n - \tilde{\psi}_{1}\right)\tilde{\phi}_{1} \right. \\ &\left. + \tilde{\phi}_{0}\,\tilde{\psi}_{1}^{3}\right)n\left/\left(\left(\left(2\,n - 1\right)\,\tilde{\phi}_{2} + \tilde{\psi}_{1}\right)\left(\left(-2 + 2\,n\right)\,\tilde{\phi}_{2} + \tilde{\psi}_{1}\right)^{2}\left(\left(-3 + 2\,n\right)\,\tilde{\phi}_{2} + \tilde{\psi}_{1}\right)\right). \end{split}$$

The corresponding coefficients  $\tilde{T}_{n,1}$  and  $\tilde{T}_{n,2}$  are deduced by the same way [Koepf et al., 1996].

### Chapter 8

# Laguerre-Freud equations for the recurrence coefficient of the semi-classical orthogonal polynomials of class one

#### 8.1 Introduction

We assume that  $\mathcal{L}$  is a regular linear functional satisfying

$$\mathcal{D}_q(\phi \mathcal{L}) = \psi \mathcal{L},\tag{8.1}$$

with polynomials  $\phi$  and  $\psi$  given by

$$\phi(x) = \sum_{i=0}^{t} \phi_{i} x^{j}, \ \psi(x) = \sum_{j=0}^{p} \psi_{j} x^{j}, \ p \ge 1, \ |\phi_{t}| |\psi_{p}| \ne 0.$$
 (8.2)

We suppose that (8.1) is not reducible and that the class of the linear functional  $\mathcal{L}$ ,  $\operatorname{cl}(\mathcal{L})$  is  $\operatorname{cl}(\mathcal{L}) \equiv s = \max\{\deg(\phi) - 2, \deg(\psi) - 1\}$ .  $\{P_n\}_{n \in \mathbb{N}}$ , which is a family of monic polynomials orthogonal with respect to  $\mathcal{L}$ , satisfies the TTRR:

$$\begin{cases}
P_{n+1}(x) = (x + \beta_n)P_n(x) - \gamma_n P_{n-1}(x), & n \ge 1, \\
P_0(x) = 1, P_1(x) = x - \beta_0,
\end{cases}$$
(8.3)

where  $\beta_n$  and  $\gamma_n$  are complex numbers with  $\gamma_n \neq 0 \quad \forall n \in \mathbb{N}$ .

When  $\mathcal{L}$  is  $\mathcal{D}$ ,  $\mathcal{D}_q$  or  $\Delta$ -classical, the coefficients  $\beta_n$  and  $\gamma_n$  can be given explicitly in terms of polynomials c and  $\psi$  appearing in (8.1) (see the previous chapter).

But if  $\mathcal{L}$  is  $\mathcal{D}$ ,  $\mathcal{D}_q$  or  $\Delta$ -semi-classical of class s > 0, it is very difficult to give the coefficients  $\beta_n$  and  $\gamma_n$  explicitly in terms of the polynomials  $\phi$  and  $\psi$ .

We propose a method which enables us to compute them recursively when the linear functional  $\mathcal{L}$  is  $\mathcal{D}_q$ -semi-classical of class s=1. Then, we use Theorem 4.2 to extend this result to the  $\Delta$ -classical orthogonal polynomials.

This method consists to derive two non-linear equations satisfied by  $\beta_n$  and  $\gamma_n$ , called Laguerre-Freud equations.

#### 8.2 Starting the Laguerre-Freud equations

The initial form of the Laguerre-Freud equations is obtained by applying both sides of (8.1) to the polynomials  $P_nP_n$  and  $P_nP_{n+1}$ , respectively

$$\langle D_{q,\omega}(\phi\mathcal{L}), P_n P_n \rangle = \langle \psi\mathcal{L}, P_n P_n \rangle,$$
  
$$\langle D_{q,\omega}(\phi\mathcal{L}), P_n P_{n+1} \rangle = \langle \psi\mathcal{L}, P_n P_{n+1} \rangle.$$

Then, we apply the rules (3.6) and (3.12) to the previous equations and obtain

$$\langle \mathcal{L}, \mathcal{O}_{\frac{1}{q}} P_n \mathcal{G}_{\frac{1}{q}} P_n \rangle + \langle \mathcal{L}, \phi \mathcal{D}_{\frac{1}{q}} P_n P_n \rangle = -q \langle \psi \mathcal{L}, P_n P_n \rangle, \tag{8.4}$$

$$\langle \mathcal{L}, \mathcal{O}D_{\frac{1}{2}}P_{n+1}\mathcal{G}_{\frac{1}{2}}P_n\rangle + \langle \mathcal{L}, \phi \mathcal{D}_{\frac{1}{2}}P_n P_{n+1}\rangle = -q \langle \psi \mathcal{L}, P_n P_{n+1}\rangle. \tag{8.5}$$

The respective right-hand sides of the previous equations are given by

#### Lemma 8.1 (Belmehdi et al., 1994)

$$\begin{cases}
\langle \psi \mathcal{L}, P_n P_n \rangle = [\psi(\beta_n) + \psi_2(\gamma_n + \gamma_{n+1})] I_{0,n}, \\
\langle \psi \mathcal{L}, P_n P_{n+1} \rangle = [\psi_1 + \psi_2(\beta_n + \beta_{n+1})] \gamma_{n+1} I_{0,n}.
\end{cases}$$
(8.6)

*Proof:* Using the three-term recurrence relation (8.3), we first derive the relation

$$I_{0,n+1} = \gamma_{n+1} I_{0,n} \quad \forall n \in \mathbb{N}$$

$$\tag{8.7}$$

and then use it together with (8.3) to prove the lemma. In fact, use of (8.3) and (8.7) give

$$\begin{split} I_{0,n+1} &= \langle \mathcal{L}.P_{n+1}P_{n+1} \rangle \\ &= \langle \mathcal{L}.P_{n+1}((x-\beta_n)P_n - \gamma_n P_{n-1}) \rangle \\ &= \langle \mathcal{L}.P_{n+1}xP_n \rangle \\ &= \langle \mathcal{L}.P_n(P_{n+2} + \beta_{n+1}P_{n+1} + \gamma_{n+1}P_n) \rangle \\ &= \gamma_{n+1} \langle \mathcal{L}.P_n P_n \rangle \\ &= \gamma_{n+1} I_{0,n}. \end{split}$$

Using (8.7) we obtain

$$\begin{split} \langle \psi \mathcal{L}, P_{n} P_{n} \rangle &= \psi_{0} \langle \mathcal{L}, P_{n} P_{n} \rangle + \psi_{1} \langle \mathcal{L}, x P_{n} P_{n} \rangle + \psi_{2} \langle \mathcal{L}, x^{2} P_{n} P_{n} \rangle \\ &= \psi_{0} I_{0,n} + \psi_{1} \langle \mathcal{L}, (P_{n+1} + \beta_{n} P_{n} + \gamma_{n} P_{n-1}) P_{n} \rangle \\ &+ \psi_{2} \langle \mathcal{L}, (P_{n+1} + \beta_{n} P_{n} + \gamma_{n} P_{n-1})^{2} \rangle \\ &= \psi_{0} I_{0,n} + \psi_{1} \beta_{n} I_{0,n} + \psi_{2} (I_{0,n+1} + \beta_{n}^{2} I_{0,n} + \gamma_{n}^{2} I_{0,r,-1}) \\ &= [\psi(\beta_{n}) + \psi_{2} (\gamma_{n} + \gamma_{n+1})] I_{0,n}. \end{split}$$

$$\begin{split} \langle \psi \mathcal{L}, P_{n} P_{n+1} \rangle &= \psi_{0} \langle \mathcal{L}, P_{n} P_{n+1} \rangle + \psi_{1} \langle \mathcal{L}, x P_{n} P_{n+1} \rangle + \psi_{2} \langle \mathcal{L}, x^{2} P_{n+1} P_{n} \rangle \\ &= \psi_{1} \langle \mathcal{L}, (P_{n+1} + \beta_{n} P_{n} + \gamma_{n} P_{n-1}) P_{n+1} \rangle \\ &+ \psi_{2} \langle \mathcal{L}, (P_{n+2} + \beta_{n+1} P_{n+1} + \gamma_{n+1} P_{n}) (P_{n+1} + \beta_{n} P_{n} + \gamma_{n} P_{n-1}) \rangle \\ &= \psi_{1} I_{0,n+1} + \psi_{2} (\beta_{n+1} I_{0,n+1} + \gamma_{n+1} \beta_{n}) I_{0,n} \\ &= [\psi_{1} + \psi_{2} (\beta_{n} + \beta_{n+1})] \gamma_{n+1} I_{0,n} \end{split}$$

In order to express all terms of (8.4) and (8.5) in terms of  $\beta_n$  and  $\gamma_n$ , we need to expand the polynomials  $P_n$  in the basis  $\{x^n\}_{n\in\mathbb{N}}$  with coefficients depending on  $\beta_n$  and  $\gamma_n$ .

#### 8.3 Intermediate coefficients

#### 8.3.1 Coefficients $T_{n,i}$

Lemma 8.2 (Foupouagnigni et al., 1998a) All basic coefficients  $T_{n,i}$  in the expansion of

$$P_n(x) = \sum_{i=0}^{n} T_{n,i} x^{n-i}$$
(8.8)

can be computed recursively from the relations:

$$T_{1,1} = -\beta_0,$$
  
 $T_{n,0} = 1, \quad n \ge 0,$  (8.9)

$$T_{n+1,1} = T_{n,1} - \beta_n, \quad n \ge 1,$$
 (8.10)

$$T_{n+1,j} = T_{n,j} - \beta_n T_{n,j-1} - \gamma_n T_{n-1,j-2}, \quad 2 \le j \le n, \tag{8.11}$$

$$T_{n+1,n+1} = -\beta_n T_{n,n} - \gamma_n T_{n-1,n-1}, \quad n \ge 1.$$
 (8.12)

*Proof:* We use the relation (8.8) and the three-term recurrence relation (8.3) to obtain

$$\sum_{i=0}^{n} T_{n,i} x^{n+1-i} = \sum_{i=0}^{n+1} T_{n+1,i} x^{n+1-i} + \beta_n \sum_{j=0}^{n} T_{n,j} x^{n-j} + \gamma_n \sum_{k=0}^{n-1} T_{n-1,k} x^{n-1-k}.$$

We replace the variable j and k in the previous equation by j-1 and k-2, respectively, to obtain

$$\sum_{i=0}^{n} T_{n,i} x^{n+1-i} = \sum_{i=0}^{n+1} T_{n+1,i} x^{n+1-i} + \beta_n \sum_{j=1}^{n+1} T_{n,j-1} x^{n+1-j} + \gamma_n \sum_{k=2}^{n+1} T_{n-1,k-2} x^{n+1-k},$$

an equation which is equivalent to

$$\begin{split} &(T_{n+1,0}-T_{n,0})x^{n+1}+(T_{n+1,1}-T_{n,1}+\beta_nT_{n,0})x^n\\ &+\sum_{k=2}^n(T_{n+1,k}-T_{n,k}+\beta_nT_{n,k-1}+\gamma_nT_{n-1,k-2})x^{n+1-k}+T_{n+1,n+1}+\beta_nT_{n,n}\\ &+\gamma_nT_{n-1,n-1}=0. \end{split}$$

From the relation  $P_1 = T_{1,0} x + T_{1,1} = x - \beta_0$ , it follows that  $T_{1,0} = 1$  and  $T_{1,1} = -\beta_0$ . We complete the proof by identifying to zero all coefficients of the polynomial on the left hand-side of the previous equation.

**Corollary 8.1** Using Lemma 8.2, we compute the coefficients  $T_{n,j}$  j = 0.3 as:

$$T_{n+1,1} = -\sum_{i=0}^{n} \beta_{i}, \quad n \ge 0,$$

$$T_{n+1,2} = \sum_{0 \le i \le j \le n} \beta_{i} \beta_{j} - \sum_{i=1}^{n} \gamma_{i}, \quad n \ge 1,$$

$$T_{n+1,3} = -\sum_{0 \le i \le j \le k \le n} \beta_{i} \beta_{j} \beta_{k} + \sum_{1 \le i \le j \le n} (\gamma_{i} \beta_{j} + \beta_{i} \gamma_{j}) + \beta_{0} \sum_{i=1}^{n} \gamma_{i}$$

$$-\sum_{i=1}^{n} \beta_{i-1} \gamma_{i}, \quad n \ge 2.$$

$$8.13)$$

All other terms can be computed in the same way, but for class s = 1, only these 3 terms will be used.

Let us emphasise that the two terms  $T_{n,1}$  and  $T_{n,2}$  are already given in [Chihara, 1978]; the computation of the higher order coefficients allows to generate Laguerre-Freud equations for any arbitrary class s > 1. These coefficients play the role (but in a simpler way) of the Turán determinants introduced in [Belmehdi et al., 1994] showing the interest of Laguerre-Freud equations.

87

#### 8.3.2 Coefficients $B_n^k$

The coefficients  $B_n^k$  appear from the action of the linear functional  $\mathcal L$  on the polynomial  $x^{n+k}\,P_n$ 

$$B_n^k = \langle \mathcal{L}. x^{n+k} P_n \rangle, \tag{8.14}$$

with the initial condition

$$B_n^0 = \langle \mathcal{L}, x^n P_n \rangle = \langle \mathcal{L}, P_n P_n \rangle = I_{0,n}.$$

From the relation  $0 = \langle \mathcal{L}, P_{n+k} P_n. \rangle; k \geq 1$  and (8.8) we deduce that

$$B_n^k = -\sum_{i=1}^k T_{n+k,i} B_n^{k-i}.$$

We use the previous equation to compute, recursively, the coefficient  $B_n^k$ . In particular, we have:

$$B_{n}^{1} = -T_{n+1,1}I_{0,n},$$

$$B_{n}^{2} = (T_{n+1,1}T_{n+2,1} - T_{n+2,2})I_{0,n},$$

$$B_{n}^{3} = [T_{n+1,1}(T_{n+3,2} - T_{n+2,1}T_{n+3,1}) + T_{n+3,1}T_{n+2,2} - T_{n+3,3}]I_{0,n}.$$
(8.15)

Notice that the connection between  $B_n^k$  and the coefficients  $C_{j,k}^n$  introduced in [Belmehdi et al., 1994],

$$x^{n+k}P_n(x) = \sum_{i=0}^{2n+k} C_{j,n}^{n+k} P_j(x),$$

is obviously

$$B_n^k = C_{0,n}^{n+k} I_{0,0}.$$

#### 8.3.3 Structure relations

We first recall the structure relation (3.84):

$$\phi \, \mathcal{D}_{\frac{1}{q}} P_n = \sum_{i=n-s-1}^{n+t-1} \xi_{n,j} \, P_j, \ , \ n > s.$$

with  $t = \deg(\phi)$ .  $\xi_{n,n-s-1} \neq 0$ , n > s+1 and then apply the linear functional  $\mathcal{L}$  to both sides of the equation obtained when multiplying the previous one by  $P_j$  and get

$$\xi_{n,j}I_{0,j} = \langle \phi \mathcal{L}, P_j \mathcal{D}_{\frac{1}{q}} P_n \rangle, \ n-s-1 \le j \le n+t-1.$$

$$(8.16)$$

Then, using (8.2), (8.14) and the previous equation we get

$$\begin{split} \xi_{n,j}I_{0,j} &= \langle \phi \mathcal{L}, P_j \mathcal{D}_{\frac{1}{2}}P_n \rangle \\ &= \langle \mathcal{L}, P_j \sum_{i=0}^t \phi_i x^i \rangle \sum_{k=1}^n [n+1-k]_{\frac{1}{q}} T_{n,k-1} x^{n-k} \rangle \\ &= \sum_{k=0}^{n+t-1} c_k(q) \langle \mathcal{L}, x^k P_j \rangle \\ &= \sum_{k=j}^{n+t-1} c_k(q) \langle \mathcal{L}, x^k P_j \rangle \\ &= \sum_{k=j}^{n+t-1} c_k(q) B_j^{k-j}. \end{split}$$

with

$$c_{k}(q) = \sum_{i+j=k, i \le t, j \le n-1}^{n+t-1} \phi_{i} [j+1]_{\frac{1}{q}} T_{n,n-j-1}.$$
(8.17)

Thus,

$$\xi_{n,j}I_{0,j} = \sum_{k=j}^{n+t-1} c_k(q)B_j^{k-j}.$$
(8.18)

Once (8.18) is derived, we are now able to compute the coefficients  $\xi_{n,i}$  in terms of  $\beta_n$ ,  $\gamma_n$  and the polynomials  $\phi$  and  $\psi$ , by using (8.2) (8.15) and (8.18). To be more precise, we assume that the linear functional  $\mathcal{L}$  is of class at most one, this implies that

$$\phi(x) = \sum_{j=0}^{3} \phi_j x^j, \ \psi(x) = \sum_{j=0}^{2} \psi_j x^j,$$

with

$$(|\phi_0| + |\phi_1| + |\phi_2| + \phi_3)(|\psi_1| + |\psi_2|) \neq 0.$$

We use the method described above to compute the coefficients  $\xi_{n,j}$ ,  $n-1 \leq j \leq n+2$  and get

$$\xi_{n,n+2} = [n]_{\frac{1}{q}}\phi_3,$$

$$\xi_{n,n+1} = q^{1-n} \{ [[n]_n(\beta_n + \beta_{n+1}) - T_{n,1}]\phi_3 + [n]_q\phi_2 \},$$

$$\xi_{n,n} = q^{1-n} \{ [n]_q\phi_1 + [[n]_q\beta_n - T_{n,1}]\phi_2 + [[n]_q(\gamma_n + \gamma_{n+1} + \beta_n^2) + T_{n,1}^2 - \beta_n T_{n,1} - (1+q) T_{n,2}]\phi_3 \},$$

$$\xi_{n,n-1} = q^{1-n} \{ [n]_q \phi_0 - T_{n,1} \phi_1 - [T_{n,1}^2 - (1+q)T_{n,2} + [n]_q \gamma_n] \phi_2$$
$$- [-T_{n,1}^3 + (1+q)T_{n,2} - \gamma_n T_{n,1} - [3]_q T_{n,3} + [n]_q \gamma_n (\beta_{n-1} + \beta_n)] \phi_3 \}.$$

The search for  $\xi_{n,n-2}$  requires the constant  $T_{n,4}$  which is huge and needs heavy computation. To get rid of this difficulty we, again, use (3.12), (8.1) and (8.16) to get

$$\begin{split} \xi_{n,j}I_{0,j} &= \langle \delta\mathcal{L}, P_j \, \mathcal{D}_{\frac{1}{q}}P_n \rangle \\ &= \langle \mathcal{OL}, \mathcal{D}_{\frac{1}{q}}[A_{q,\omega}P_jP_n] - \mathcal{D}_{\frac{1}{q}}A_{q,\omega}P_jP_n \rangle \\ &= -q \langle \mathcal{D}_q(\mathcal{OL}), \mathcal{G}_qP_jP_n \rangle - q \langle \mathcal{OL}, \mathcal{D}_qP_jP_n \rangle \\ &= -q \langle \mathcal{L}, \mathcal{L} \, \mathcal{G}_qP_jP_n \rangle - q \langle \mathcal{OL}, \mathcal{D}_qP_jP_n \rangle \\ &= -q \langle \mathcal{L}, (\mathcal{L} \, \mathcal{G}_qP_j + \mathcal{OD}_qP_j)P_n \rangle, \end{split}$$

hence

$$\xi_{n,i}I_{0,i} = -q\langle \mathcal{L}_{+}(\psi A_{q,\omega}P_{i} + \phi \mathcal{D}_{q}P_{i})P_{n}\rangle. \tag{3.19}$$

It follows immediately that

$$\xi_{n,n-2}I_{0,n-2} = -q(\mathcal{L}, q^{n-2}\psi_2 + [n-2]_g\phi_3)x^nP_n\rangle.$$

We use (3.63) and (8.7) to simplify the expression of  $\xi_{n,n-2}$  and get

$$\xi_{n,n-2} = -q^{n-1} + (2 + [n-2]_{\frac{1}{q}} \frac{\phi_3}{q}) \gamma_n \gamma_{n-1}. \tag{8.20}$$

In the same way, we compute another expression for  $\xi_{n,n-1}$  which we denote by  $\xi_{n,n-1}^+$ 

$$\begin{split} \xi_{n,n+1}^{+} I_{0,n+1} &= -q \langle \mathcal{L}, (\psi A_{q-1} P_{n+1} + \phi \mathcal{D}_{q} P_{n+1}) P_{n} \rangle \\ &= -q \langle \mathcal{L}, [(\psi_{0} + \psi_{1} x + \psi_{2} x^{2}) | q^{n+1} | x^{n+1} + q^{n+2} T_{n+1,1} | x^{n+2} \\ &+ (\phi_{0} + \phi_{1} x + \phi_{2} x^{2} + \phi_{3} x^{-}) [n+1]_{q} | x^{n+2} + [n-2]_{q} | T_{n+1,1} | x^{n+3}) ] P_{n} \rangle \end{split}$$

and get, after simplifications,

$$\xi_{n,n-1}^{+} = q[n-1]_q \gamma_n T_{n,1} \phi_3 - q([n-2]_q T_{n-1,1} + [n-1]_q \beta_n) \gamma_n \phi_3 -q[n-1]_q \gamma_n \phi_2 + q^n \gamma_n T_{n,1} \psi_2 - (q\psi_1 + q\psi_2 \beta_2 + T_{n-1,1} \psi_2) \gamma_n q^{n-1}.$$

#### 8.4 Final form of the Laguerre-Freud equations

We prove the following theorem which is the main result of this chapter.

**Theorem 8.1** The coefficients  $\beta_n$  and  $\gamma_n$  of the three-term recurrence relation

$$P_{n+1}(x) = (x - \beta_n)P_n(x) - \gamma_n P_{n-1}(x), n \ge 1, P_0(x) = 1, P_1(x) = x - \beta_0$$

satisfied by the  $\mathcal{D}_q$ -semi-classical orthogonal polynomials of class at most one,  $\{P_n\}_{n\in\mathbb{N}}$ , can be computed recursively from the two non-linear equations

$$\begin{cases}
(\psi_2 + [2n]_{\frac{1}{q}} \frac{\phi_3}{q})(\gamma_n + \gamma_{n+1}) = F_1(q, ; \beta_0, \dots, \beta_n; \gamma_1, \dots, \gamma_n), \\
(\psi_2 + [2n+1]_{\frac{1}{q}} \frac{\phi_3}{q})\beta_{n+1}\gamma_{n+1} = F_2(q, ; \beta_0, \dots, \beta_n; \gamma_1, \dots, \gamma_{n+1}).
\end{cases}$$
(8.21)

 $\phi_j$  and  $\psi_j$  are the coefficients of the polynomials  $\phi$  and  $\psi$  appearing in the Pearson equation,  $\mathcal{D}_q(\phi\mathcal{L}) = \psi\mathcal{L}$ , satisfied by the regular linear functional  $\mathcal{L}$ .  $F_1$  is a polynomial of 2n+1 variables and of degree 2 and  $F_2$  a polynomial of 2n+2 variables and of degree 3, with the initial conditions

$$\beta_0 = \frac{\langle \mathcal{L}, x \rangle}{\langle \mathcal{L}, 1 \rangle}, \ \psi_2 \, \gamma_1 = -\psi(\beta_0). \tag{8.22}$$

*Proof:* In the first step we use the structure relation (3.84) to transform Equations (8.4) and (8.5) as

$$\xi_{n,n-2}\langle \mathcal{L}, \mathcal{G}_{\frac{1}{q}} P_n P_{n-2} \rangle + \xi_{n,n-1}^+ \langle \mathcal{L}, \mathcal{G}_{\frac{1}{q}} P_n P_{n-1} \rangle + (1 + q^{-n}) \xi_{n,n} I_{0,n} 
= -q \langle \mathcal{L}, \psi P_n P_n \rangle, 
\xi_{n+1,n-1} \langle \mathcal{L}, \mathcal{G}_{\frac{1}{q}} P_n P_{n-1} \rangle + \xi_{n+1,n} q^{-n} I_{0,n} + \xi_{n,n-1} I_{0,n+1} 
(8.23)$$

$$\begin{array}{ll}
\varsigma_{n+1,n-1}\langle \mathcal{L}, \mathcal{G}_{\frac{1}{q}}^{\perp} \stackrel{n}{}_{1} \stackrel{n-1}{}_{1} \stackrel{+}{}_{1} \varsigma_{n+1,n} \stackrel{q}{}_{1} \stackrel{1}{}_{0,n} + \varsigma_{n,n-1} \stackrel{1}{}_{10,n+1} \\
&= -q \langle \mathcal{L}, \psi P_{n} P_{n+1} \rangle.
\end{array}$$
(8.24)

In the second step we compute  $\langle \mathcal{L}, \mathcal{G}_{\frac{1}{2}} P_n P_{n-1} \rangle$  and  $\langle \mathcal{L}, \mathcal{G}_{\frac{1}{2}} P_n P_{n-2} \rangle$  using (8.8) and get

$$\begin{split} \langle \mathcal{L}, \mathcal{G}_{\frac{1}{q}} \, P_{n} \, P_{n-1} \rangle &= \langle \mathcal{L}, (q^{-n} \, x^{n} + q^{1-n} \, T_{n,1} \, x^{n-1}) \, P_{n-1} \rangle \\ &= q^{-n} B_{n-1}^{1} + q^{1-n} \, T_{n,1} \, I_{0,n-1} \\ &= q^{-n} (q-1) T_{n,1} \, I_{0,n-1}, \\ \langle \mathcal{L}, \mathcal{G}_{\frac{1}{q}} \, P_{n} \, P_{n-2} \rangle &= \langle \mathcal{L}, (q^{-n} \, x^{n} + q^{1-n} \, T_{n,1} \, x^{n-1} + q^{2-n} \, T_{n,2} \, x^{n-2}) \, P_{n-2} \rangle \\ &= q^{-n} B_{n-2}^{2} + q^{1-n} \, T_{n,1} B_{n-2}^{1} + q^{2-n} \, T_{n,2} I_{0,n-2}. \end{split}$$

In the third step we use (8.6) and the previous equations to simplify (8.23) and (8.24) and obtain:

$$-q^{2n+2} (\psi_{2} + [2n]_{\frac{1}{q}} \frac{\phi_{3}}{q}) (\gamma_{n} + \gamma_{n+1}) = [q^{2} [2n]_{q} \beta_{n}^{2} + q(q+q^{2n})(T_{n,1}^{2} - \beta_{n}T_{n,1})$$

$$-(q+1)(q^{2} + q^{2n})T_{n,2}]\phi_{3} + [q^{2} [2n]_{q}\beta_{n} - q(q+q^{2n})T_{n,1}]\phi_{2} + q^{2} [2n]_{q}\phi_{1} + q^{2n+2}\beta_{n}^{2}$$

$$+q^{2n} (q-1)\psi_{2} [qT_{n,1}^{2} - q\beta_{n}T_{n,1} - (1+q)T_{n,2}] + q^{2n+1}[q\beta_{n} - (q-1)T_{n,1}]\psi_{1} + q^{2n+2}\psi_{0},$$
(8.25)

$$-q^{2n+1} \left(\psi_{2} + [2n+1]_{\frac{1}{q}} \frac{\phi_{3}}{q}\right) \gamma_{n+1} \beta_{n+1} = \left\{\beta_{n}^{3} + (q-1)\beta_{n}^{2} T_{n,1} + [(q+2)\gamma_{n} - (q-1)T_{n,1}^{2} + (q^{2}-1)T_{n,2} + (1+[2n+1]_{q})\gamma_{n+1}]\beta_{n}\right\}$$

$$+ \left[(q^{2}-1)T_{n,1} + [3]_{q} \beta_{n+1}\right] \gamma_{n} - T_{n,1}^{3} + \left[(q+2)T_{n,2} - (1+q^{2\gamma_{n}+1})\gamma_{n+1}\right] T_{n,1} - [3]_{q} T_{n,3}\right\} \phi_{3}$$

$$+ \left\{\beta_{n}^{2} + (q-1)\beta_{n} T_{n,1} + (q+1)\gamma_{n} + T_{n,1}^{2} - (q+1)T_{n,2} + [2n+1]_{q}\gamma_{n+1}\right\} \phi_{2} + (\beta_{n} - T_{n,1})\phi_{1}$$

$$+ [n+1]_{q} \phi_{0} + q^{2n+1}\beta_{n}\gamma_{n+1} \psi_{2} - q^{2n}(q-1)\gamma_{n+1}T_{n,1}\psi_{2} + q^{2n+1}\gamma_{n+1}\psi_{1}.$$

$$(8.26)$$

The first initial condition is obtained by applying the linear functional  $\mathcal{L}$  to  $P_1 = x - \beta_0$  while the second comes from the application of both sides of (8.1) to the polynomial  $P_0P_0$ .

In fact, it follows from (8.1) that

$$\langle \psi \mathcal{L}, P_0 P_0 \rangle = \langle \mathcal{D}_q(\phi \mathcal{L}), P_0 P_0 \rangle = -\frac{1}{q} \langle \phi \mathcal{L}, \mathcal{D}_{\frac{1}{q}} P_0 P_0 \rangle = 0.$$

The previous equation used together with (8.6) gives  $\psi(\beta_0) + \gamma_1 \psi_2 = 0$ .

We complete the proof of the theorem by saying that:

- 1. For any non-zero integer n, the coefficients  $\psi_2 + [j]_{\frac{1}{q}} \frac{\phi_3}{q}$ , j = 2n, 2n + 1 of the right-hand sides of the two previous equations, thanks to the fact that the  $\mathcal{D}_q$ -semi-classical linear functional  $\mathcal{L}$  is regular (see (3.89)), are non-zero (except if  $\phi_3 = \psi_2 = 0$ ).
- 2. The left-hand sides of the previous equations contain only constants, sums and products of coefficients  $\beta_j$  and  $\gamma_j$ . Polynomials  $F_1$  and  $F_2$  are obtained by replacing  $T_{n,j}$ , j=1,2,3 in equations (8.25) and (8.26) by (8.13).

Notice that we can also obtain the second Laguerre-Freud Equation (8.26) by identification of the two expressions  $\xi_{n,n-1}$  and  $\xi_{n,n-1}^+$ .

Equation (8.25) gives, linearly,  $\gamma_{n+1}$  in terms of  $\beta_j$ , j=0,n and  $\gamma_j$ , j=1,n; when (8.26) gives  $\beta_{n+1}$  in terms of  $\beta_j$ , j=0,n,  $\gamma_j$ , j=1,n and the previous  $\gamma_{n+1}$  via the non-linear term  $(\psi_2 + [2n+1]_{\frac{1}{q}}, \frac{\phi_3}{\sigma})\beta_{n+1}\gamma_{n+1}$ .

The fact that  $\beta_{n+1}$  is not obtained linearly (except for the classical case) in terms of the previous  $\beta_j$  and  $\gamma_j$  exemplify the fundamental barrier between semi-classical of class s > 0 and classical situation in which both  $\phi_3$  and  $\psi_2$  are zero. For  $\mathcal{D}_q$ -semi-classical of class s > 0, both relations (8.25) and (8.26) must be used simultaneously, starting with the initial values given by (8.22). In the classical situation Equations (8.25) and (8.26) can be decoupled.

#### 8.4.1 Laguerre-Freud equations for $\mathcal{D}_q$ -classical orthogonal polynomials

When we take  $\phi_3 = \psi_2 = 0$  in Equations (8.25) and (8.26), we obtain the Laguerre-Freud equations for  $\mathcal{D}_q$ -classical orthogonal polynomials:

$$q^{2n}(\psi_{1} + [2n]_{\frac{1}{q}} \frac{\phi_{2}}{q}) \beta_{n} + [(1+q^{2n-1})\phi_{2} + (q-1)q^{2n-1}\psi_{1}] \sum_{j=0}^{n-1} \beta_{j}$$

$$+ [2n]_{q} \phi_{1} + q^{2n-1}\psi_{0} = 0, \qquad (8.27)$$

$$q^{2n+1}(\psi_{1} + [2n+1]_{\frac{1}{q}} \frac{\phi_{2}}{q}) \gamma_{n+1} + (1+q)\phi_{2} \sum_{j=0}^{n} \gamma_{j} \qquad (8.28)$$

$$= -(\sum_{j=0}^{n-1} \beta_{j})^{2} \phi_{2} + [(q-1)\phi_{2}\beta_{n} - \phi_{1}] \sum_{j=0}^{n-1} \beta_{j}$$

$$+ (q-1)\phi_{2} \sum_{0 \le i \le j \le n-1}^{n-1} \beta_{i}\beta_{j} - \phi_{2} \beta_{n}^{2} - \phi_{1} \beta_{n} - [n+1]_{q} \phi_{0}.$$

Remark 8.1 Using Maple V.4 and the simplification procedures for q-hypergeometric terms developed in [Böing et al., 1998], we have solved (8.27) with the initial condition  $\beta_0 = -\frac{\psi_0}{\psi_1}$  to get  $\beta_n$ .

Taking into account the  $\beta_n$  obtained above, we have solved (8.28) with the initial condition  $\gamma_1 = -\frac{\phi(\beta_0)}{\phi_2 + q\psi_1}$  to get  $\gamma_n$ . Obviously the coefficients  $\beta_n$  and  $\gamma_n$  obtained coincide with the ones given in Lemma 7.1.

# 8.5 Applications to $\mathcal{D}$ , $D_{\omega}$ and $D_{q,\omega}$ -semi-classical orthogonal polynomials of class one

# 8.5.1 Laguerre-Freud equations for $\mathcal{D}$ -semi-classical orthogonal polynomials of class one

We obtain these equations by computing the limits of (8.25) and (8.26) as  $q \to 1$  to obtain [Belmehdi et al., 1994], [Foupouagnigni et al., 1998a]

$$\psi(\beta_{n}) + 4\phi_{3} \sum_{i=1}^{n-1} \gamma_{i} + 2 \sum_{i=0}^{n-1} \theta_{\beta_{n}} \phi(\beta_{i}) = -(\psi_{2} + 2n\phi_{3})(\gamma_{n} + \gamma_{n+1}),$$

$$\sum_{i=0}^{n} \phi(\beta_{i}) + 3\phi_{3} \sum_{i=1}^{n} \gamma_{i}(\beta_{i-1} + \beta_{i}) + \left[ (2n+1)\gamma_{n+1} + 2\sum_{i=1}^{n} \gamma_{i} \right] \phi_{2} + 2\gamma_{n+1}(n\beta_{n} + \sum_{i=0}^{n} \beta_{i})\phi_{3} + [\psi_{1} + \psi_{2}\beta_{n}]\gamma_{n+1} = -[\psi_{2} + (2n+1)\phi_{3}]\beta_{n+1}\gamma_{n+1}.$$
(8.29)

where

$$\theta_a \phi(x) = \frac{\phi(x) - \phi(a)}{x - a},$$

with the initial conditions

$$\beta_0 = \frac{\langle \mathcal{L}, x \rangle}{\langle \mathcal{L}, 1 \rangle}, \ \psi_2 \gamma_1 = -\psi(\beta_0).$$

# 8.5.2 Laguerre-Freud equations for $D_{\omega}$ -semi-classical orthogonal polynomials of class one

It follows from Theorem 4.2 that the Laguerre-Freud equations for  $\mathcal{D}_q$  semi-classical linear functional of class one is obtained just by replacing  $\beta_j$  (resp.  $\phi$  and  $\psi$ ) in (8.25) and (8.26) by  $\beta_j - \frac{\omega}{1-q}$  ( $\mathcal{T}_{\frac{\omega}{1-q}}\phi$  and  $\mathcal{T}_{\frac{\omega}{1-q}}\psi$  respectively). For this reason, we need to control the behaviour of  $T_{n,1}$ .  $T_{n,2}$  and  $T_{n,3}$  when  $\beta_j$  is replaced by  $\beta_j - \frac{\omega}{1-q}$ .

**Lemma 8.3** If the coefficients  $\bar{T}_{n,j}$ , j=1,2,3 represent the coefficients  $T_{n,j}$ , j=1,2,3 in which  $\beta_j$  is replaced by  $\beta_j - \frac{\omega}{1-g}$ , then, they are related by

$$\bar{T}_{n,1} = T_{n,1} + \frac{n\omega}{1-q}, 
\bar{T}_{n,2} = T_{n,2} + \frac{(n-1)\omega}{1-q} T_{n,1} + \frac{\omega^2}{(1-q)^2} \binom{n}{2}, 
\bar{T}_{n,3} = T_{n,3} + \frac{(n-2)\omega}{1-q} T_{n,2} + \frac{\omega^2}{(1-q)^2} \binom{n-1}{2} T_{n,1} + \frac{\omega^3}{(1-q)^3} \binom{n}{3}.$$
(8.30)

Proof:

The proof follows immediately from (8.13).

We replace  $\phi$  and  $\psi$  in (8.25) and (8.26) by  $\mathcal{T}_{\frac{\omega}{1-q}}\phi$  and  $\mathcal{T}_{\frac{\omega}{1-q}}\psi$ , respectively (and implicitly  $\beta_j$  by  $\beta_j - \frac{\omega}{1-q}$ ), taking into account the previous lemma and obtain the Laguerre-Freud equations for the recurrence coefficients of the  $D_{q,\omega}$ -semi-classical [Azatas-on et al, 1998] orthogonal polynomials  $\{P_n\}_{n\in\mathbb{N}}$ . These polynomials are orthogonal with respect to the linear functional  $\mathcal{L}$ , of class at most one, satisfying  $D_{q,\omega}(\phi\mathcal{L}) = \psi\mathcal{L}$ . Hence we take the limit of these two equations as  $q \to 1$  and obtain:

Theorem 8.2 (Foupouagnigni et al., 1998a) The coefficients  $\beta_j$  and  $\gamma_j$  of the three-term recurrence relation.

$$P_{n+1} = (x - \beta_n) P_n - \gamma_n P_{n-1}, \ n \ge 0, \ P_{-1} = 0, \ P_0 = 1,$$

satisfied by the monic polynomials  $\{P_n\}_{n\in\mathbb{N}}$ , orthogonal with respect to the  $D_{\omega}$ -semi-classical linear functional  $\mathcal{L}$ , of class at most one, satisfying  $D_{\omega}(\phi\mathcal{L}) = \psi\mathcal{L}$ , are given by

$$\psi(\beta_n) + 4\phi_3 \sum_{i=1}^{n-1} \gamma_i + 2 \sum_{i=0}^{n-1} \theta_{\beta_n} \phi(\beta_i) + \omega \sum_{i=0}^{n-1} \theta_{\beta_n} \psi(\beta_i) + 2 \binom{n}{3} \omega^2 \phi_3 + \binom{n}{2} \omega^2 \psi_2 = -(\psi_2 + 2n\phi_3)(\gamma_n + \gamma_{n+1}),$$
(8.31)

where

$$\theta_a \phi(x) = \frac{\phi(x) - \phi(a)}{x - a}$$

and

$$\sum_{i=0}^{n} \phi(\beta_{i}) + \left[ (2n+1)\gamma_{n+1} + 2\sum_{i=1}^{n} \gamma_{i} \right] \phi_{2} + 
+ 3\phi_{3} \sum_{i=1}^{n} \gamma_{i} (\beta_{i-1} + \beta_{i}) + 2\gamma_{n+1} (n\beta_{n} + \sum_{i=0}^{n} \beta_{i}) \phi_{3} 
+ n\omega\psi_{2}\gamma_{n+1} - \binom{n+1}{2} \omega\phi_{1} + \left[ -n\omega \sum_{i=0}^{n} \beta_{i} + \binom{n+1}{3} \omega^{2} \right] \phi_{2} 
- \omega \left[ \sum_{0 \leq i < j \leq n} \beta_{i}\beta_{j} + n \sum_{i=0}^{n} \beta_{i}\beta_{i} + (2n-1) \sum_{i=1}^{n} \gamma_{i} + n\gamma_{n+1} \right] \phi_{3} 
+ \left[ \binom{n}{2} \omega^{2} \sum_{i=0}^{n} \beta_{i} - \binom{n+1}{4} \omega^{3} \right] \phi_{3} + [\psi_{1} + \psi_{2}\beta_{n}]\gamma_{n+1} 
= -[\psi_{2} + (2n+1)\phi_{3}]\beta_{n+1}\gamma_{n+1},$$
(8.32)

with the initial conditions

$$\beta_0 = \frac{\langle \mathcal{L}, x \rangle}{\langle \mathcal{L}, 1 \rangle}, \ \psi_2 \, \gamma_1 = -\psi(\beta_0).$$

#### 8.5.3 Laguerre-Freud equations for $D_{\omega}$ -classical orthogonal polynomials

The Laguerre-Freud equations obtained in (8.31) and (8.32) contain, obviously, the classical cases when  $\psi_2 = \phi_3 = 0$ . We use the notation of [Salto, 1995] so that we can compare more easily with the results therein.

$$\phi(x) = ax^2 + bx + c$$
 and  $\psi(x) = px + q$ .

Equations (8.31) and (8.32) reduce to:

$$\psi(\beta_{n}) + 2a \sum_{i=0}^{n-1} \beta_{i} + 2nb + 2na\beta_{n} = -n\omega p,$$

$$\sum_{i=0}^{n} \phi(\beta_{i}) + \left[ (2n+1)\gamma_{n+1} + 2\sum_{i=1}^{n} \gamma_{i} \right] a - \binom{n+1}{2} \omega b$$

$$+ \left[ -n\omega \sum_{i=0}^{n} \beta_{i} + \binom{n+1}{3} \omega^{2} \right] a = -p\gamma_{n+1}.$$
(8.34)

Rewriting the second equation with  $n \to n-1$  and subtracting we get:

$$\phi(\beta_n) + [p + (2n+1)a]\gamma_{n+1} - [p + (2n-3)a]\gamma_n$$

$$-n\omega b - an\omega \beta_n - a\omega \sum_{i=0}^{n-1} \beta_i + a\omega^2 \binom{n}{2} = 0.$$
(8.35)

Using symbolic computation with Maple V.4 we have checked positively that for the classical discrete orthogonal polynomials, the coefficients  $\beta_n$  and  $\gamma_n$ , given explicitly in terms of polynomials  $\phi$  and  $\psi$  (see Lemma 7.2), are solutions of Equations (8.33) and (8.34) (with  $\omega = 1$ ).

Equations (8.33) and (8.35) are exactly the ones derived in the thesis [Salto, 1995] taking into account the  $D_{\omega}$  derivative of the linear functional given by definition 2.20 and the one used in [Salto, 1995]. Let us remark, however, that in [Salto, 1995] the  $\gamma_n$  equation is obtained using the so-called  $D_{\omega}$  representation, expanding a classical orthogonal polynomial  $P_n$  as a sum of (see (2.62)) (maximum three)  $D_{\omega}P_i(i=n+1,n,n-1)$ . This technique cannot be extended to the class 1, because of the non-existence of such a representation for semi-classical orthogonal polynomials of class s>0.

# 8.6 Applications to generalised Charlier and generalised Meixner polynomials of class one

## 8.6.1 Laguerre-Freud equations for the generalised Meixner polynomial of class one

These polynomials with  $\ell$  parameters were introduced in [Ronveaux, 1986] in order to show the quasiorthogonality character of the  $D_{\omega}$  derivative (with  $\omega = 1$ ). The weight  $\rho$  is given by:

$$\rho(i) = \frac{\mu^i}{(i!)^{\ell}} \prod_{j=1}^{\ell} \Gamma(i + \alpha_j), \quad (0 < \mu < 1, \alpha_j > 0), i = 0, 1, 2, \dots$$
(8.36)

Generalised Meixner polynomials are denoted by  $m_n^{(\vec{\alpha},\mu)}$ , where  $\vec{\alpha} = (\alpha_1, \ldots, \alpha_\ell)$ , which reduce of course, to the well-known classical Meixner polynomials when  $\vec{\alpha}$  is the scalar  $\alpha$  ( $\ell = 1$ ).

If  $\ell = 2$ ,  $\alpha_1 \neq 1$  and  $\alpha_2 \neq 1$ , the weight  $\rho$  obeys

$$\Delta(\phi\rho) = \psi\rho.$$

with

$$\phi(x) = x^2 \text{ and } \psi(x) = (\mu - 1)x^2 + (\alpha_1 + \alpha_2)\mu x + \mu \alpha_1 \alpha_2.$$
 (8.37)

The family is, therefore, discrete semi-classical of class one.

In fact, we have

$$\phi(x) = x \phi_0(x). \ \psi(x) - \phi_0(x) = (x+1)\psi_{0,1}(x) + r_{0,1},$$

with

$$\phi_0(x) = x, \ \psi_{0,1}(x) = (\mu - 1)x + \mu \left(\alpha_1 + \alpha_2 - 1\right), \ r_{0,1} = \mu \left(\alpha_1 - 1\right)(\alpha_2 - 1). \tag{8.38}$$

Since the only root of  $\phi$  is zero, it follows from Proposition 2.5, and the fact that  $r_{0,1} \neq 0$  (for  $(\alpha_1 - 1)(\alpha_2 - 1) \neq 0$ ), that for  $\ell = 2$  and for  $\alpha_1 - 1)(\alpha_2 - 1) \neq 0$ , the generalised Meixner polynomial is of class one.

Of course, when  $\alpha_1 = 1$  (or  $\alpha_2 = 1$ ), the class reduce to 0 and we obtain the classical Meixner polynomials  $m_n^{(\alpha_2,\mu)}$  [Nikiforov et al., 1991]. In particular for  $\alpha_1 = \alpha_2 = 1$ , the generalised Meixner polynomials of class 1 reduces to the particular case of the Meixner polynomials, called discrete Laguerre polynomials and denoted [Chihara, 1978]

$$la.(x) = m_n^{(1,\mu)}(x).$$
 (8.39)

We have checked, positively, the Laguerre-Freud equations when  $\omega \to 1$  with the known  $\beta_n, \gamma_n$  of the classical Meixner polynomials and the discrete Laguerre polynomials.

It should be noted that for  $\ell = 2$  and for arbitrary positive  $\alpha_1$  and  $\alpha_2$ , the weight given by Equation (8.36), is not a polynomial modification of the Meixner weight, except when  $\alpha_1$  or  $\alpha_2$  is an integer.

Replacing in Equations (8.31) and (8.32)  $\omega$  by one and polynomials  $\phi$  and  $\psi$  given by Equation (8.38), we obtain the Laguerre-Freud equations for the generalised Meixner polynomial of class s = 1:

$$(1 - \mu)(\gamma_n + \gamma_{n+1}) = (\mu - 1)(\binom{n}{2} + \beta_n^2) + ((1 + \mu)n$$

$$+ \mu(\alpha_1 + \alpha_2))\beta_n + (1 + \mu)\sum_{i=0}^{n-1}\beta_i$$

$$+ \mu(\alpha_1 + \alpha_2)n + \mu\alpha_1\alpha_2,$$

$$(1 - \mu)(\beta_n + \beta_{n+1})\gamma_{n+1} = -n\sum_{i=0}^n \beta_i + ((1 + \mu)n + \mu(\alpha_1 + \alpha_2) + 1)\gamma_{n+1}$$

$$+ \binom{n+1}{3} + \sum_{i=0}^n \beta_i^2 + 2\sum_{i=1}^n \gamma_i,$$

$$(8.41)$$

with initial values

$$\beta_0 = \frac{M_1}{M_0} = \frac{\mu \alpha_1 \ \alpha_2 \ _2 F_1(1 + \alpha_1, 1 + \alpha_2; 2; \mu)}{_2 F_1(\alpha_1, \alpha_2; 1; \mu)}, \quad \gamma_1 = \frac{\psi(\beta_0)}{1 - \mu}.$$
 (8.42)

## 8.6.2 Laguerre-Freud equations for generalised Charlier polynomial of class one

The generalised Charlier polynomials introduced in [Hounkonnou et al., 1998] are discrete semi-classical orthogonal polynomials associated with the weight

$$\rho(x) = \frac{\mu^x}{(x!)^N}, \ N \ge 1, \ (\mu > 0), x = 0, 1, 2, \dots$$
(8.43)

The generalised Charlier weight  $\rho$  is semi-classical and satisfies the Pearson equation

$$\Delta(\phi\rho) = \psi\rho$$

with

$$\phi(x) = x^N \text{ and } \psi(x) = \mu - x^N.$$
 (8.44)

If N=2, the orthogonal polynomial family associated to the weight  $\rho(x)=\frac{\mu^x}{(x!)^2}$  is discrete semi-classical of class one (and called generalised Charlier polynomials of class one).

Replacing in Equations (8.31) and (8.32)  $\omega$  by one and the polynomials  $\phi$  and  $\psi$  given by Equation (8.44) (but with N=2), we obtain the Laguerre-Freud equations for the generalised Charlier polynomials of class one:

$$\gamma_n + \gamma_{n+1} = -\binom{n}{2} - \beta_n^2 + n\beta_n + \sum_{i=0}^{n-1} \beta_i + \mu, \tag{8.45}$$

$$(\beta_n + \beta_{n+1})\gamma_{n+1} = -n \sum_{i=0}^n \beta_i + n \gamma_{n+1} + \binom{n+1}{3} + \sum_{i=0}^n \beta_i^2 + 2 \sum_{i=1}^n \gamma_i + \gamma_{n+1}, \tag{8.46}$$

with initial values

$$\beta_0 = \frac{M_1}{M_0} = \frac{\sqrt{\mu}I_1(2\sqrt{\mu})}{I_0(2\sqrt{\mu})} = \frac{\mu_0 F_1(2;\mu)}{{}_0F_1(1;\mu)},$$

$$\gamma_1 = \mu - \beta_0^2,$$
(8.47)

where  $I_0(x)$  and  $I_1(x)$  are the modified Bessel functions of order 0 and 1, respectively.

Remark 8.2 The polynomials  $P_n$  have been computed for the generalised Meixner and Charlier polynomials of class one, up to n=10 from  $\beta_n, \gamma_n$  generated by the Laguerre-Freud equations given above and also from the Hankel representation of polynomials (see (2.6)) which requires the computation of the moments  $M_1$  up to j=19. These moments were computed from the moment recurrence relation for the generalised Meixner and Charlier polynomials of class one, respectively:

$$(1-\mu)M_{k+2} = \alpha_1\alpha_2\mu M_k + (\alpha_1 + \alpha_2)\mu M_{k+1}$$
$$-\sum_{j=1}^k (-1)^j \binom{k}{j} M_{k+2-j},$$
$$M_{k+2} = \mu M_k - \sum_{j=1}^k (-1)^j \binom{k}{j} M_{k+2-j}.$$

The polynomial coefficients in both approaches are written in terms of  $M_0$  and  $M_1$  using the initial values of the Laquerre-Freud recurrence given by Equations (8.42) and (8.47). The polynomials obtained in these two ways coincide, of course, and the Laquerre-Freud approach is obviously more efficient.

#### 8.6.3 Asymptotic behaviour

In the first step we compute numerically, up to n = 100000, the coefficients  $\beta_n$  and  $\gamma_n$  Using (8.40) and (8.41), for several values of the coefficients  $\alpha_1$ .  $\alpha_2$  and  $\mu$  and the result of the plot for all cases indicates that the sequences  $\frac{\gamma_n}{n^2}$  and  $\frac{\beta_n}{n}$  are convergent. Assuming that they converge, their limits,  $a(\mu)$  and  $b(\mu)$ 

$$a(\mu) = \frac{\mu}{(1-\mu)^2}, \ b(\mu) = \frac{1+\mu}{1-\mu},$$
 (8.48)

are obtained using Maple V.4 and Equations (8.40), (8.41) with the approximations:  $\gamma_n \cong a(\mu) n^2$  and  $\beta_n \cong b(\mu) n$ , for n large.

In the same way, but with  $\beta_n$  and  $\gamma_n$  replaced by  $\beta_n - b(\mu)n$  and  $\gamma_n - a(\mu)n^2$ , respectively, using numerical and symbolic computation with Maple V.4 and analysis of Equations (8.40) and (8.41) Foundamigni et al., 1998f, we observe the asymptotic behaviour for the coefficients  $\beta_n$  and  $\gamma_n$ .

The same process, applied to (8.45) and (8.46), allows to observe the asymptotic behaviour for the generalised Charlier polynomials of class one. We, therefore, give the following conjecture about the asymptotic behaviour for the generalised Charlier and Meixner polynomials of class one. These results, obtained by the Laguerre-Freud equations with Maple V.4, are under investigation [Foundamigni et al., 1998f] in order to give a suitable proof.

Conjecture 8.1 The coefficients  $\beta_n$  and  $\gamma_n$  of the three-term recurrence relation satisfied by the monic generalised Meixner polynomials of class one obey:

$$\lim_{n\to\infty}\left(\beta_n-\frac{1+\mu}{1-\mu}n-\frac{\mu\left(\alpha_1+\alpha_2-1\right)}{1-\mu}\right)=0,\ \lim_{n\to\infty}\left(\gamma_n-\frac{\mu(n+\alpha_1-1)(n+\alpha_2-1)}{(1-\mu)^2}\right)=0,$$

and those of the three-term recurrence relation satisfied by the monic generalised Charlier polynomials of class one obey.

$$\lim_{n\to\infty}(\beta_n-n)=0.\ \lim_{n\to\infty}(\gamma_n-\mu)=0.$$

It should be mentioned that the coefficients  $\beta$  and  $\gamma_n$  of the generalised Meixner polynomials of class 1, are known when  $\alpha_1$  or  $\alpha_2$  is an integer [Ronveaux et al., 1998b]. They obviously confirm the asymptotic behaviour of the coefficients eta and  $\gamma_n$  stated in the previous conjecture.

## Chapter 9

## Conclusion and perspectives

We first list our main contributions to the theory of orthogonal polynomials, then give some open problems which can be investigated as the continuation of this work.

#### 9.1 Conclusion

Chapter 1 introduces the work while Chapter 2 recalls some known materials on orthogonal polynomials.

The main results of Chapter 3 are theorems 3.1 and 3.2. Theorem 3.1 gives a general caracterisation of classical orthogonal polynomials. This result gives a more general caracterisation of classical orthogonal polynomials, and is valid for classical orthogonal polynomials of a continuous variable, classical orthogonal polynomials of a discrete variable and also for q-classical polynomials. It constitutes a unified theory for classical orthogonal polynomials.

Theorem 3.2 caracterises the semi-classical orthogonal polynomials. It gives some links between the semi-classical aspect of the orthogonal polynomials, the quasi-orthogonal aspect of the derivative of these orthogonal polynomials and the structure relations satisfied by these polynomials.

In Chapter 4, we study the properties of the formal Stieltjes function. We mention two results. The first is the theorem 4.1, stating that the affine  $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials and the  $D_{q,\omega}$ -semi-classical orthogonal polynomials are the same. This result is used to obtain the coefficients of the affine  $D_{q,\omega}$ -Riccati difference equation and the coefficients of the fourth-order difference equation satisfied by the associated Laguerre-Hahn orthogonal polynomials.

The second result is theorem 4.2. It proves that the  $D_{q,\omega}$ -Laguerre-Hahn orthogonal polynomials can be deduced, using a suitable change of variable, from the  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials. This result is very interesting and could have lot of applications. We have used it to deduce the coefficients of the fourth-order difference equation satisfied by the rth associated  $\Delta$ -Laguerre-Hahn orthogonal polynomials from the coefficients of the fourth-order difference equation satisfied by the rth associated  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials. The Laguerre-Freud equations for the recurrence coefficients of the  $\mathcal{D}_{\omega}$ -semi-classical orthogonal polynomials of class 1 have also been obtained using theorem 4.2.

In Chapter 5 we use a result by Suslov [Suslov, 1989] to obtain the factored form of the fourth-order difference equation satisfied by the first associated  $\mathcal{D}_q$ -classical orthogonal polynomials. We have again used theorem 4.2 to deduce the factored form of the fourth-order difference equation satisfied by the first associated  $\Delta$ -classical orthogonal polynomials.

We mention that equation (5.12) can be used to obtain some families of classical orthogonal polynomials for which the first associated is still classical. These families:

1- For classical continuous orthogonal polynomials we note the Grosjean polynomials [Ronveaux et al., 1996] of the first kind  $G_n^{\alpha}$  for which the first associated is a Grosjean polynomial of the second kind  $g_n^{\alpha}$  [Grosjean, 1985, 1986], i.e.,

$$(G_n^{\alpha})^{(1)} = g_n^{-\alpha}, -1 < \alpha < 0,$$

9.2. Perspectives 97

where  $G_n^{\alpha}(x) = P_n^{(\alpha,-1-\alpha)}(x)$ ,  $-1 < \alpha < 0$  and  $g_n^{\alpha}(x) = P_n^{(\alpha,1-\alpha)}(x)$ ,  $-1 < \alpha < 2$ .  $P_n^{(\alpha,\beta)}$  represents the monic Jacobi polynomials with the parameters  $\alpha$  and  $\beta$ .

2- For the classical orthogonal polynomials of a discrete variable, we note that the first associated of the monic Hahn polynomial  $H_n(x, \alpha, \beta, N)$  with  $\alpha + \beta + 1 = 0$  is classical and is related to the Hahn family by [Area et al., 1996]

$$H_n(x, \alpha, \beta, N)^{(1)} = H_n(x - \alpha - 1, -\alpha, 1 + \alpha, N - 1).$$

3- For the q-classical polynomials, we have already pointed out the situations for which the first associated little q-Jacobi polynomials  $p_n(x; a, b|q)$  and big q-Jacobi polynomials  $P_n(x; a, b, c; q)$  are still classical.

The monic little q-Jacobi (resp. monic big q-Jacobi) polynomials and their respective first associated are related by

$$p_n^{(1)}(x; a, \frac{1}{q a} | q) = a^n q^n p_n(\frac{x}{a q}; \frac{1}{a}, a q | q),$$

$$P_n^{(1)}(x; a, \frac{1}{q a}, c; q) = a^n P_n(\frac{x}{a}; \frac{1}{a}, a q, c q; q).$$

In Chapter 6 we have proved (see theorem 6.1 and proposition 6.1) that the associated of any integer order of the Laguerre-Hahn linear functional is a Laguerre-Hahn linear functional. We also gave upper bounds for the degrees of coefficients  $E_r$ ,  $F_r$ ,  $G_r$  and  $H_r$  of the  $\mathcal{D}_q$ -Riccati difference equation satisfied by the Stieltjes function  $S_r$  of the rth associated  $\mathcal{L}^{(r)}$  of  $\mathcal{L}$  (see proposition 6.2).

Theorem 6.3 gives fondamental relations which lead to the fourth-order difference equation for the rth associated  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials

$$\sum_{j=0}^{4} I_j(r, n, q; x) \mathcal{G}_q^j P_n^{(r)} = 0.$$

given in theorem 6.4. Theorem 6.3 and 6.4 are valid for  $\mathcal{D}$ -Laguerre-Hahn orthogonal polynomials (by limit process) and for  $\Delta$ -Laguerre-Hahn orthogonal polynomials (via theorem 4.2). We have also given explicitly coefficients  $E_r$ ,  $F_r$ ,  $G_r$ ,  $H_r$  and  $I_j(r, n, q; x)$  for classical situations.

Chapter 7 contains known materials needed for this work.

The main result of Chapter 8 is theorem 8.1 which shows that it is possible to compute recursively via two non-linear equations, coefficients  $\beta_n$  and  $\gamma_n$  of the three-term recurrence relation satisfied by the  $\mathcal{D}_q$ -semi-classical orthogonal polynomials of class one. This new result (theorem 8.1) is used, together with theorem 4.2 and lemma 8.3, to obtain Theorem 8.2 giving the Laguerre-Freud equations for the recurrence coefficients of the  $D_{\omega}$ -semi-classical orthogonal polynomials of class one.

Using theorem 8.2, we have given a conjecture about the asymptotic behaviour of the coefficients  $\beta_n$  and  $\gamma_n$  of the three-term recurrence relation satisfied by the generalised Charlier and generalised Meixner polynomials of class 1.

#### 9.2 Perspectives

As the continuation of this work, many investigations can be done:

- 1. Theorem 4.2 proves that the  $\Delta$ -Laguerre-Hahn orthogonal polynomials can be obtained from the  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials. In principle, this result means that any result obtained for the  $\mathcal{D}_q$ -Laguerre-Hahn orthogonal polynomials can be extended to the  $\Delta$ -Laguerre-Hahn orthogonal polynomials. It will be interesting to see how these results are extended and see their consequences in the applications of orthogonal polynomials.
- 2. It might be possible to simplify and writte the fourth-order difference equation for the  $\mathcal{D}_q$ -classical orthogonal polynomials in the compact form as was done for  $\mathcal{D}$ -classical orthogonal polynomials (see (6.69)) [Lewanowicz, 1995].

3. One can use the fourth-order difference equation satisfied by the rth associated  $\mathcal{D}_q$ -classical orthogonal polynomials  $P_n^{(r)}$ 

$$\mathbf{M}(r, n, q; x) P_n^{(r)} = \sum_{j=0}^{4} I_j(r, n, q; x) \mathcal{G}_q^j P_n^{(r)} = 0,$$

to expand the rth associated  $P_n^{(r)}$  in the basis  $\{P_n\}_{n\in\mathbb{N}}$ 

$$P_n^{(r)}(x) = \sum_{j=0}^n C(n,j) P_j,$$

as was done for  $\mathcal{D}$ -classical orthogonal polynomials and  $\Delta$ -classical orthogonal polynomials (see [Lewanowicz, 1996,1997], [Area et al. ,1998a, 1998b], [Godoy et al., 1996], [Askey 1965,1975], [Askey et al., 1984] . . . ).

- 4. The fourth-order difference equation can be established for the general Laguerre-Hahn orthogonal polynomials. We mention for example that Bangerezako [Bangerezako, 1998], had derived the fourth-order difference equation for the Laguerre-Hahn polynomials orthogonal on special non-uniform lattices (snul).
- 5. The Laguerre-Freud equations for class s > 1 can be obtained by mimicking the approach developed in this thesis. This generalisation is already under investigation [Azatassou et al., 1998].
- 6. The conjecture obtained using the Laguerre-Freud equations need to be proved and extended to the semi-classical orthogonal polynomials of class s > 1. For this purpose, It may be hepful to have a look at the papers giving the proof of Freud's conjecture (see the Introduction).

## Chapter 10

# Appendices

#### 10.1 Appendix I

#### 10.1.1 About $\mathcal{D}$ -classical orthogonal polynomials

We give the polynomials  $\phi$  and  $\psi$  appearing in the Pearson equation satisfied by the weight  $\rho$   $(\mathcal{D}(\phi\rho) = \psi\rho)$  defining the classical orthogonal polynomials of a continuous variable [Chihara, 1978], [Nikiforov et al., 1983, 1991], [Szegő, 1939].

1. Jacobi 
$$P_n^{(\alpha,\beta)}(x)$$

$$\phi(x) = 1 - x^2$$
,  $\psi(x) = -(\alpha + \beta + 2) x + \beta - \alpha$ 

2. Laguerre 
$$L_n^{\alpha}(x)$$

$$\phi(x) = x, \ \psi(x) = -x + \alpha + 1$$

3. Hermite 
$$H_n(x)$$

$$\phi(x) = 1, \ \psi(x) = -2x$$

4. Bessel B(x)

$$\phi(x) = x^2, \ \psi(x) = -2(x+1)$$

#### 10.1.2 About $\Delta$ -classical orthogonal polynomials

We give the polynomials  $\phi$  and  $\psi$  appearing in the Pearson equation satisfied by the weight  $\rho$  ( $\Delta(\phi\rho)=\psi\rho$ ) defining the classical orthogonal polynomials of a discrete variable [Chihara, 1978], [Nikiforov et al., 1983, 1991], [Szegő, 1939].

1. Hahn 
$$h_n^{(\alpha,\beta)}(x)$$

$$\phi(x) = x (N + \alpha - x), \ \psi(x) = -(\alpha + \beta + 2) x + (\beta + 1)(N - 1)$$

2. Meixner 
$$m_n^{(\nu,\mu)}(x)$$

$$\phi(x) = x, \ \psi(x) = \mu\nu + (\mu - 1) x$$

3. Krawtchouk 
$$k_n^{(p)}(x)$$

$$\phi(x) = x, \ \psi(x) = \frac{(Np - x)}{1 - p}$$

4. Charlier 
$$c_n^{(\mu)}(x)$$

$$\phi(x) = x, \ \psi(x) = \mu - x$$

#### 10.1.3 About q polynomials

We give the polynomials  $\phi$  and  $\psi$  appearing in the Pearson equation satisfied by the weight  $\rho$  ( $\mathcal{D}_q(\phi\rho) = \psi\rho$ ) defining the polynomials appearing in the q-Hahn tableau. [Kockock et al, 1996], [Koornwinder, 1994]. Notice that these polynomials  $\phi$  and  $\psi$  were already given case by case in [Medem, 1996] and [Ivan et al, 1998].

1. Big q-Jacobi  $P_n(x; a, b, c; q)$ 

$$\phi(x) = a c q - (a + c) x + \frac{x^2}{2}, \ \psi(x) = \frac{c q - x + a q (1 - (b + c) q + b q x)}{(-1 + q) q}$$

2. Little q-Jacobi  $p_n(x; a, b|q)$ 

$$\phi(x) = \frac{x(x-1)}{q}, \ \psi(x) = \frac{1 - x + aq(bqx - 1)}{(-1+q)q}$$

3. Stieltjes-Wigert  $S_n(x;q)$ 

$$\phi(x) = \frac{x}{q}, \ \psi(x) = \frac{q x - 1}{(-1 + q) q}$$

4. q-Meixner  $m_n(x; b, c; q)$ 

$$\phi(x) = -bc + \frac{cx}{q}, \ \psi(x) = \frac{c + q - bcq - qx}{(1 - q)q}$$

5. Alternative q-Charlier  $K_n(x; a; b)$ 

$$\phi(x) = \frac{x(1-x)}{q}, \ \psi(x) = \frac{-1+x(1+aq)}{(-1+q)q}$$

6. Little q-Laguerre/Wall  $L_n^{(a)}(x;q)$ 

$$\phi(x) = \frac{x(1-x)}{q}, \ \psi(x) = \frac{-1+a\,q+x}{(-1+q)\,q}$$

7. q-Charlier  $U_n^{(a)}(x;q)$ 

$$\phi(x) = a - (1+a)x + x^2, \ \psi(x) = \frac{1+a-x}{a-1}$$

8. Discrete q-Hermite  $h_n(q;x)$ 

$$\phi(x) = x^2 - 1, \ \psi(x) = \frac{x}{1 - q}$$

#### 10.2 Appendix II

#### 10.2.1 Results on general associated classical discrete polynomials

We use Theorem 4.2 and 6.3 to obtain operators  $\mathcal{D}_{r,n}$ ,  $\mathcal{N}_{r+1,n-1}$ ,  $\bar{\mathcal{D}}_{r+1,n-1}$  and  $\bar{\mathcal{N}}_{r,n}$  for the classical orthogonal polynomials of a discrete variable (see Foupouagnigni et al. 1998b]. These basic operators (see (6.38) and (6.39)) and the coefficients of the fourth order difference equation for associated classical discrete orthogonal polynomials (see (6.7)) are written down in each case (for notations see [Nikiforov et al.,1991]).

10.2. Appendix II

Charlier case  $C_n^{\mu}(x), \ \mu > 0$ 

$$\phi(x) = x, \quad \psi(x) = -x + \mu,$$

$$\mathcal{D}_{r,n} = \mu(2+x)\mathcal{T}^2 - (2+x-r)(x-n-r+1+\mu)\mathcal{T}$$

$$-(-3x-2+3r-x^2+2xr-r^2+r\mu)\mathcal{I}_d,$$

$$\mathcal{N}_{r+1,n-1} = -r\mu(x-n-r+1+\mu)\mathcal{T} + r\mu(\mu+2+x-r)\mathcal{I}_d,$$

$$\bar{\mathcal{D}}_{r+1,n-1} = \mu(2+x)\mathcal{T}^2 - \mu(x-n-r+1+\mu)\mathcal{T} + \mu(-r+\mu)\mathcal{I}_d,$$

$$\bar{\mathcal{N}}_{r,n} = -(-x+n+r-1-\mu)\mathcal{T} - (\mu+x+1-r)\mathcal{I}_d,$$

$$I_0(r,n,x) = \mu(1+x)(-2+n+2R),$$

$$I_1(r,n,x) = (2x\mu+2R+4\mu-2R^3+nR-3nR^2-n^2R),$$

$$I_2(r,n,x) = (2x\mu+2R+4\mu-5\mu n-2x\mu n-4\mu xR+4R^3-10\mu R-6R^2,$$

$$-4nR+6nR^2+4n^2R-n^2+n^3),$$

$$I_3(r,n,x) = +(-2x\mu-4R-6\mu-2n-2R^3+6R^2+7nR-3nR^2-n^2R+2n^2),$$

where R is given by  $R = r - x - \mu - 2$ .

Meixner case  $M_n^{(\nu,\mu)}(x), \ \nu > 0, \ 0 < \mu < 1$ 

 $I_{A}(r,n,x) = \mu (4+x) (n+2R),$ 

$$\phi(x) = x; \ \psi(x) = (\mu - 1)x + \mu \nu,$$

$$\mathcal{D}_{r,n} = \mu (x+2) (x+1+\nu) (\mu - 1) \mathcal{T}^{2} + (-2-x+r) (1+x-r-n+\mu \nu + x \mu + r \mu + n \mu + \mu) (\mu - 1) \mathcal{T} - (-r \mu + r \mu \nu + r^{2} \mu - 3 x - r^{2} + 2 x r - x^{2} - 2 + 3 r) (\mu - 1) \mathcal{I}_{d},$$

$$\mathcal{N}_{r+1,n-1} = +r(\nu+r-1)(1+x-r-n+\mu\nu+x\mu+r\mu+n\mu+\mu)\mu\mathcal{T} -r(\nu+r-1)(\mu\nu+x\mu+x-r+r\mu+2)\mu\mathcal{I}_d,$$

$$\bar{\mathcal{D}}_{r+1,n-1} = \mu (x + 2) (x + 1 + \nu) \mathcal{T}^{2}$$

$$-(x + 1 + \nu + r) (1 + x - r - n + \mu \nu + x \mu + r \mu + n \mu + \mu) \mu \mathcal{T}$$

$$+(r - r \nu + x \mu + r \mu + r^{2} \mu + \mu \nu^{2} + x^{2} \mu + 2 x \mu \nu + 2 r \mu \nu$$

$$+2 x \mu r + \mu \nu - r^{2}) \mu \mathcal{I}_{d},$$

$$\bar{\mathcal{N}}_{r,n} = -(\mu - 1) (1 + x - r - n + \mu \nu + x \mu - r \mu + n \mu + \mu) \mathcal{T} + (\mu - 1) (x \mu + \mu + \mu \nu + r \mu - r + x + 1) \mathcal{I}_d,$$

$$I_0(r, n, x) = -\mu (-3 \mu + M + 2 R - 3) (x + \nu) (x + 1),$$

$$I_{1}(r,n,x) = -6 \mu^{2} x - 2 x^{2} \mu^{2} - 4 \mu^{2} - 4 \mu^{2} \nu - 2 x \mu^{2} \nu - 3 \mu R^{2} - 3 \mu M R - 6 x \mu$$
$$-2 x^{2} \mu - 4 \mu - 4 \mu \nu - 2 x \mu \nu - 3 M R - 3 R^{2} + M^{2} R + 3 R^{2} M + 2 R^{3},$$

$$\begin{split} I_2(r,n,x) & = & -4\,R^3 - 6\,\mu^2 - 4\,M^2\,R - 9\,\mu^2\,\nu - 14\,\mu^2\,x - 4\,\mu\,R - 2\,\mu\,M - 5\,M - M^3 \\ & - 6\,\mu - 4\,x^2\,\mu^2 - 4\,x^2\,\mu - 4\,x\,\mu\,\nu - 4\,x\,\mu^2\,\nu - 14\,x\,\mu - 9\,\mu\,\nu + 2 + 16\,R\,x\,\mu + 10\,R\,\mu\,\nu \\ & + 12\,\mu\,R^2 + 4\,R\,x^2\,\mu + 4\,R\,x\,\mu\,\nu + 8\,\mu\,M\,x + 12\,\mu\,M\,R + 12\,M\,R + 2\,\mu^3 + 4\,M^2 - 10\,\mu^2\,R \\ & + 5\,M\,\mu\,\nu - 6\,R^2\,M + 2\,x^2\,\mu\,M - 5\,\mu^2\,M + 2\,x\,\mu\,\nu\,M - 10\,R + 12\,R^2 + 4\,\mu\,M^2 \,, \end{split}$$

$$I_{3}(r,n,x) = 2R^{3} + M^{2}R + 6\mu^{2}\nu + 10\mu^{2}x + 24\mu R + 12\mu M + 6M + 2x^{2}\mu^{2} + 2x^{2}\mu + 2x\mu\mu + 2x\mu^{2}\nu + 10x\mu + 6\mu\nu - 4 - 9\mu R^{2} - 9\mu M R - 9NR - 4\mu^{3} - 2M^{2} + 12\mu^{2}R + 3R^{2}M + 6\mu^{2}M + 12R - 9R^{2} - 2\mu M^{2},$$

$$I_4(r,n,x) = -(x+4)(x+3+\nu)(-\mu+M+2R-1)\mu,$$
 where  $R=r-x-2-\mu(r+x+\nu)$ , and  $M=(n+1)(1-\mu)$ .

Krawtchouk case  $k_n^{(p)}(x)$ , p>0, q>0, p+q=1

$$\phi(x) = x, \ \psi(x) = \frac{1}{q}((1-q)N - x),$$

$$\mathcal{D}_{r,n} = (q-1)(x+2)(-x-1+N)\mathcal{T}^{2} + (-2-x+r)(-2q-2xq+qN-N+x+r+1+n)\mathcal{T} + (-3xq-2q-Nrq+r-r^{2}-x^{2}q+rN+2xqr+2rq)\mathcal{I}_{d},$$

$$\mathcal{N}_{r+1,n-1} = (q-1)(N-r+1)(-2q-2xq+qN-N+x+r+1+n)r\mathcal{T} + (q-1)(N-r+1)(qN-N-2q-2xq+r+x)r\mathcal{I}_d,$$

$$\begin{split} \bar{\mathcal{D}}_{r+1,n-1} &= q \, (q-1)(x+2) \, (-x-1+N) \, \mathcal{T}^2 \\ &+ (q-1)(N-x-r-1) \, (-2 \, q-2 \, x \, q+q \, N-N+x+r+1+n) \, \mathcal{T} \\ &- (q-1)(-2 \, x \, q \, N+q \, N^2-q \, N+2 \, r \, q+2 \, x \, q \, r+x^2 \, q+x \, q-N \, r \, q+2 \, r \, N \\ &-r^2-r-2 \, x \, r-N^2+N+2 \, N \, x-x^2-x) \mathcal{I}_d, \end{split}$$

$$\tilde{\mathcal{N}}_{r,n} = -(-x-r-qN+2q+2xq-n-1+N)\mathcal{T} - (-2xq-2q+qN+r+x-N+1)\mathcal{I}_d,$$

$$\begin{split} I_0(r,n,x) &= q \, (1+x) \, (x-N) \, (q-1) \, (2\,R+n), \\ I_1(r,n,x) &= (6\,x\,q+9\,n\,q-4\,N\,q-12\,q^2 \\ &-4\,x\,q^3\,N+2\,n^2-2\,x\,N\,q+2\,x^2\,q-8\,q^3\,N+12\,x\,q^3+4\,x^2\,q^3+8\,q^3 \\ &-9\,n\,q^2+4\,q-2\,n-4\,R-3\,n^2\,q+12\,q^2\,N+6\,x\,q^2\,N-18\,x\,q^2-6\,x^2\,q^2 \\ &-12\,n\,q\,R-3\,n\,R^2-2\,R^3-18\,R\,q^2-12\,q\,R^2-n^2\,R+7\,n\,R+18\,R\,q+6\,R^2), \end{split}$$

$$\begin{split} I_2(r,n,x) & = & -(10\,x\,q - 8\,n\,q - 6\,N\,q - 42\,q^2 \\ & - 5\,n\,N\,q^2 + 8\,x\,q^2\,n - 4\,x\,q^3\,N + 2\,x^2\,n\,q^2 + 5\,n\,N\,q + 2\,x\,N\,q\,n + n^2 \\ & - 2\,x\,N\,q + 2\,x^2\,q - 12\,q^3\,N + 20\,x\,q^3 + 4\,x^2\,q^3 - 2\,x\,N\,q^2\,n + 28\,q^3 + 6\,n\,q^2 \\ & + 14\,q - 2\,R - 8\,x\,q\,n - n^3 - 4\,n^2\,q + 18\,q^2\,N + 6\,x\,q^2\,N - 30\,x\,q^2 - 2\,x^2\,n\,q \\ & - 6\,x^2\,q^2 - 12\,n\,q\,R - 4\,x^2\,q\,R - 6\,n\,R^2 - 4\,R^3 + 12\,R\,q^2 - 12\,q\,R^2 - 4\,n^2\,R \\ & + 4\,n\,R - 12\,R\,q + 6\,R^2 - 10\,q^2\,N\,R - 16\,x\,q\,R + 16\,x\,q^2\,R - 4\,x\,q^2\,N\,R \\ & + 4\,x\,N\,q\,R + 10\,N\,q\,R + 4\,x^2\,q^2\,R), \end{split}$$

10.2. Appendix II

$$I_{3}(r,n,x) = -(10 x q + n q - 6 N q - 42 q^{2} + 4 x q^{3} N - 2 x N q + 2 x^{2} q - 12 q^{3} N +20 x q^{3} + 4 x^{2} q^{3} + 28 q^{3} - 3 n q^{2} + 14 q - 2 R - n^{2} q + 18 q^{2} N + 6 x q^{2} N -30 x q^{2} - 6 x^{2} q^{2} + 3 n R^{2} + 2 R^{3} - 6 R q^{2} + n^{2} R - n R + 6 R q),$$

$$I_{4}(r,n,x) = q (4 + x) (x + 3 - N) (q - 1) (n - 2 + 4 q + 2 R).$$

where R is given by R = r + x - 2xq + qN - 5q - N + 2.

Hahn case  $h_n^{(\alpha,\beta)}(x,N)$   $\alpha > -1$ ,  $\beta > -1$ ,

$$\phi(x) = x(N + \alpha - x), \quad \psi(x) = (\beta + 1)(N - 1) - (\alpha + \beta + 2)x$$

The rth associated  $P_n^{(r)}$  of the Hahn polynomials, with  $n + r \leq N$ , is annihilated by the following difference operator, by a decomposition already used in the r associated Meixner case (see [Lewanowicz, 1997]).

$$M_n^{(r)} \equiv \sum_{j=0}^4 I_j(r, n, x) \, \mathcal{T}^j = \bar{\mathcal{D}}_{1,n}^{\star\star} \, \bar{\mathcal{D}}_{1,n}^{\star} + (r-1) \sum_{j=0}^4 \bar{I}_j(r, n, x) \, \mathcal{T}^j, \tag{10.1}$$

where from (5.21),

$$\bar{\mathcal{D}}_{1,n}^{*} = (x+2)(\alpha+N-x-2)\mathcal{T}^{2} 
+(7+n(3+n+\alpha)-3N+6x-(\alpha+2N)x+2x^{2}+\beta(3+n+x-N))\mathcal{T} 
+(x+\beta+1)(N-x-1)\mathcal{I}_{d},$$

$$\bar{\mathcal{D}}_{1,n}^{**} = (x+4+\beta) (N-x-4) (20+3n+n^2-8N-4(N-4)x+4x^2 \\ +\beta (6+n-2N+2x) +\alpha (n-2x-2)) \mathcal{T}^2 \\ + (360+141n+56n^2+6n^3+n^4-260N-45Nn-15Nn^2+44N^2 \\ -2 (52+3n(3+n)-20N)(N-5)x \\ +2 (3n(3+n)+152+4(-15+N)N)x^2-16(N-5)x^3+8x^4 \\ +\alpha^2 (n-2-x) (n-2x-2) +\beta^2 (3+n+x-N)(n+8-2N+2x) +\alpha (2n^3+2n^2-3n^2x-2(x+1)(38-12N+23x-4Nx+4x^2) \\ +n (35-15N+21x-6Nx+6x^2) \\ +\beta (2n^2+7n-3Nn+4(N-x-4)(x+1))) +\beta (2n^3 \\ +n^2 (17-3N+3x) +n (80+39x+6x^2-24N-6Nx) \\ +2 (N-x-4) (-23+9N+4Nx-17x-4x^2))) \mathcal{T} \\ +(x+1) (x+1-N-\alpha) (-n (3+n)+\alpha (4-n+2x)-\beta (n+8-2N+2x) \\ -40+12N-24x+4Nx-4x^2) \mathcal{I}_d ,$$

$$\bar{I}_0(r, n, x) = -2(x+1)(x+1-N)(x+1-N-\alpha)(x+\beta+1)(r+\beta+n+1+\alpha),$$

$$\bar{I}_{1}(r,n,x) = (r+\beta+n+1+\alpha)(840 x - 24 r N \alpha + 18 \alpha N \beta + 4 \alpha \beta r + 74 n \beta - 234 N \beta + 58 \beta r + 58 \alpha r + 50 n \alpha - 30 N \beta^{2} - 10 \alpha r^{2} + 14 r \beta^{2} - 24 N r^{2} + 20 n^{2} + 58 r^{2} + 36 \beta^{2} + 90 N^{2} + 2 r^{4} + 270 \beta - 420 N - 150 \alpha + 40 n + 600 x^{2} + 192 x^{3} + 24 x^{4} + 58 r n - 42 \alpha \beta + 66 N \alpha + 14 \beta r^{2} + 12 \alpha^{2} - 24 r N \beta + 56 \beta x n + 48 \beta N^{2} + 6 N^{2} \beta^{2} + 4 \alpha r^{3} + 2 r^{2} \beta^{2} + 4 \beta r^{3} + 6 x^{2} \beta^{2} + 348 \beta x - 252 x \alpha - 132 x^{2} \alpha + 156 x^{2} \beta + 30 \beta^{2} x + 48 x r^{2} + 24 \beta x^{3} - 24 x^{3} \alpha + 12 x^{2} r^{2} + 6 x^{2} \alpha^{2} + 18 x \alpha^{2} - 10 r \alpha^{2} + 2 \alpha^{2} r^{2}$$

```
-600 N x - 6 r N \alpha \beta + 4 \alpha r^2 \beta - 6 r^2 N \beta - 6 N \beta^2 r + 24 \beta N^2 x
-12 x^2 \alpha \beta + 12 x^2 \beta r + 12 x^2 \alpha r - 12 \beta^2 N x - 48 x \alpha \beta + 84 \alpha N x
.204 \beta N x + 24 N x^2 \alpha + 48 x \alpha r + 48 x \beta r - 48 \beta N x^2 - 6 x \alpha r^2
-12 x N r^2 + 6 x \beta r^2 + 6 \beta^2 x r - 12 \beta N x r - 12 \alpha N x r + 12 \alpha N x \beta
-6 x \alpha^2 r + 24 n^2 x + 48 n x + 24 N^2 x^2 - 12 N n^2 + 16 n x^2 + 8 n^2 x^2
-48 N x^3 - 24 N n - 288 N x^2 + 14 n \beta^2 + r^2 n + 4 r^3 n + 3 r^2 n^2 + r n^2
+9 \beta n^2 - 10 n \alpha^2 - 3 \alpha n^2 + \alpha^2 n^2 + n^2 \beta^2 + n^3 \beta + n^3 \alpha + r n^3 + 96 x N^2
-8 N n^2 x - 16 N n x - 6 N \beta^2 n + 3 \beta^2 r n + 7 r^2 n \beta + 15 r n \beta + 4 \alpha n \beta
-4 N \beta n^2 + 4 \beta r n^2 + 3 \alpha^2 r n + 7 \alpha r^2 n + 4 \alpha r n^2 + 2 \alpha \beta n^2 + 6 \alpha r n \beta
-6 N \alpha n \beta - 6 N r n \beta - 32 N \beta n - 9 \alpha r n - 24 N \alpha n - 24 N r n
+6 \beta^2 x n + 4 \beta x n^2 - 6 x \alpha^2 n - 4 x \alpha n^2 + 40 x \alpha n + 12 x^2 \alpha n
+12 x^2 \beta n + 12 x^2 r n + 48 x r n - 6 x \alpha r n + 6 x r n \beta - 12 \alpha N x n
-12 \beta N x n - 12 x N r n + 450),
```

 $\bar{I}_2(r, n, x) = -2(r + \beta + n + 1 + \alpha)(1540x - 30rN\alpha + 19\alpha N\beta + 4\alpha\beta r)$  $+ 140 \, n \, \beta - 326 \, N \, \beta + 88 \, \beta \, r + 88 \, \alpha \, r + 60 \, n \, \alpha - 36 \, N \, \beta^2 - 13 \, \alpha \, r^2$  $+17 r \beta^2 -30 N r^2 +131 n^2 +88 r^2 +54 \beta^2 +133 N^2 +2 r^4 +475 \beta$  $-770 N - 295 \alpha + 238 n + 858 x^{2} + 220 x^{3} + 22 x^{4} + 88 r n - 55 \alpha \beta$  $+103 N \alpha + 17 \beta r^{2} + 24 \alpha^{2} - 30 r N \beta + 76 \beta x n + 55 \beta N^{2} + 6 N^{2} \beta^{2}$  $+4 \alpha r^{3} + 2 r^{2} \beta^{2} + 4 \beta r^{3} + 6 x^{2} \beta^{2} + 489 \beta x - 369 x \alpha - 153 x^{2} \alpha$  $+ 177 x^{2} \beta + 36 \beta^{2} x + 60 x r^{2} + 22 \beta x^{3} - 22 x^{3} \alpha + 12 x^{2} r^{2} + 6 x^{2} \alpha^{2}$  $+24 x \alpha^{2} - 13 r \alpha^{2} + 2 \alpha^{2} r^{2} - 858 N x - 6 r N \alpha \beta + 4 \alpha r^{2} \beta - 6 r^{2} N \beta$  $-6N\beta^{2}r + 22\beta N^{2}x - 10x^{2}\alpha\beta + 12x^{2}\beta r + 12x^{2}\alpha r - 12\beta^{2}Nx$  $-50 x \alpha \beta + 98 \alpha N x - 232 \beta N x + 22 N x^{2} \alpha + 60 x \alpha r + 60 x \beta r$  $-44 \beta N x^2 - 6 x \alpha r^2 - 12 x N r^2 + 6 x \beta r^2 + 6 \beta^2 x r - 12 \beta N x r$  $-12 \alpha N x r + 10 \alpha N x \beta - 6 x \alpha^{2} r + 12 n^{3} + 80 n^{2} x + 3 n^{4} + 160 n x$  $+22 N^2 x^2 - 40 N n^2 + 32 n x^2 + 16 n^2 x^2 - 44 N x^3 - 80 N n - 330 N x^2$  $+ 19 n \beta^{2} + 9 r^{2} n + 4 r^{3} n + 7 r^{2} n^{2} + 9 r n^{2} + 35 \beta n^{2} - 11 n \alpha^{2} - 5 \alpha n^{2}$  $+2\alpha^{2}n^{2}+2n^{2}\beta^{2}+5n^{3}\beta+5n^{3}\alpha+5rn^{3}+110xN^{2}-16Nn^{2}x$  $-32 N n x - 6 N \beta^2 n + 3 \beta^2 r n + 7 r^2 n \beta + 26 r n \beta + 8 \alpha n \beta - 8 N \beta n^2$  $+8\beta r n^{2} + 3\alpha^{2}r n + 7\alpha r^{2}n + 8\alpha r n^{2} + 4\alpha\beta n^{2} + 6\alpha r n\beta - 6N\alpha n\beta$  $-6Nrn\beta - 46N\beta n - 4\alpha rn - 30N\alpha n - 30Nrn + 6\beta^2xn + 8\beta xn^2$  $-6 x \alpha^{2} n - 8 x \alpha n^{2} + 44 x \alpha n + 12 x^{2} \alpha n + 12 x^{2} \beta n + 12 x^{2} r n$  $+60 x r n - 6 x \alpha r n + 6 x r n \beta - 12 \alpha N x n - 12 \beta N x n - 12 x N r n$ +1093).

$$\begin{split} I_3(r,n,x) &= (r+\beta+n+1+\alpha)(2760\,x-36\,r\,N\,\alpha+30\,\alpha\,N\,\beta+4\,\alpha\,\beta\,r \\ &+ 150\,n\,\beta-486\,N\,\beta+118\,\beta\,r+118\,\alpha\,r+94\,n\,\alpha-42\,N\,\beta^2-16\,\alpha\,r^2 \\ &+ 20\,r\,\beta^2-36\,N\,r^2+100\,n^2+118\,r^2+72\,\beta^2+210\,N^2+2\,r^4+810\,\beta \\ &- 1380\,N-570\,\alpha+200\,n+1320\,x^2+288\,x^3+24\,x^4+118\,r\,n-102\,\alpha\,\beta \\ &+ 174\,N\,\alpha+20\,\beta\,r^2+36\,\alpha^2-36\,r\,N\,\beta+80\,\beta\,x\,n+72\,\beta\,N^2+6\,N^2\,\beta^2 \\ &+ 4\,\alpha\,r^3+2\,r^2\,\beta^2+4\,\beta\,r^3+6\,x^2\,\beta^2+732\,\beta\,x-588\,x\,\alpha-204\,x^2\,\alpha \\ &+ 228\,x^2\,\beta+42\,\beta^2\,x+72\,x\,r^2+24\,\beta\,x^3-24\,x^3\,\alpha+12\,x^2\,r^2+6\,x^2\,\alpha^2 \\ &+ 30\,x\,\alpha^2-16\,r\,\alpha^2+2\,\alpha^2\,r^2-1320\,N\,x-6\,r\,N\,\alpha\,\beta+4\,\alpha\,r^2\,\beta-6\,r^2\,N\,\beta \end{split}$$

10.3. Appendix III

$$-6N\beta^{2}r + 24\beta N^{2}x - 12x^{2}\alpha\beta + 12x^{2}\beta r + 12x^{2}\alpha r - 12\beta^{2}Nx$$

$$-72x\alpha\beta + 132\alpha Nx - 300\beta Nx + 24Nx^{2}\alpha + 72x\alpha r + 72x\beta r$$

$$-48\beta Nx^{2} - 6x\alpha r^{2} - 12xNr^{2} + 6x\beta r^{2} + 6\beta^{2}xr - 12\beta Nxr$$

$$-12\alpha Nxr + 12\alpha Nx\beta - 6x\alpha^{2}r + 2250 + 56n^{2}x + 112nx + 24N^{2}x^{2}$$

$$-28Nn^{2} + 16nx^{2} + 8n^{2}x^{2} - 48Nx^{3} - 56Nn - 432Nx^{2} + 20n\beta^{2} + r^{2}n$$

$$+4r^{3}n + 3r^{2}n^{2} + rn^{2} + 17\beta n^{2} - 16n\alpha^{2} - 11\alpha n^{2} + \alpha^{2}n^{2} + n^{2}\beta^{2} + n^{3}\beta$$

$$+n^{3}\alpha + rn^{3} + 144xN^{2} - 8Nn^{2}x - 16Nnx - 6N\beta^{2}n + 3\beta^{2}rn$$

$$+7r^{2}n\beta + 21rn\beta + 4\alpha n\beta - 4N\beta n^{2} + 4\beta rn^{2} + 3\alpha^{2}rn + 7\alpha r^{2}n$$

$$+4\alpha rn^{2} + 2\alpha\beta n^{2} + 6\alpha rn\beta - 6N\alpha n\beta - 6Nrn\beta - 44N\beta n$$

$$-15\alpha rn - 36N\alpha n - 36Nrn + 6\beta^{2}xn + 4\beta xn^{2} - 6x\alpha^{2}n - 4x\alpha n^{2}$$

$$+64x\alpha n + 12x^{2}\alpha n + 12x^{2}\beta n + 12x^{2}rn + 72xrn - 6x\alpha rn$$

$$+6xrn\beta - 12\alpha Nxn - 12\beta Nxn - 12xNrn),$$

105

$$\bar{I}_4(r,n,x) = -2(x+4)(x+4-N)(x+4-N-\alpha)(x+4+\beta)(r+\beta+n+1+\alpha).$$

#### 10.3 Appendix III

We give the coefficients  $I_j(r, n, q; x)$  of the fourth-order difference equation satisfied by the rth associated  $\mathcal{D}_q$ -classical orthogonal polynomials.  $\phi_j$  and  $\psi_j$  are the coefficients of the polynomials  $\phi$  and  $\psi$ , both related to the q-Pearson equation:  $\mathcal{D}_q(\phi \mathcal{L}) = \psi \mathcal{L}$ .

#### Coefficients $I_i(r, n, q; x)$ for some q-classical orthogonal polynomials

For the discrete q-Hermite and Stieltjes-Wigert cases [see Koekoek et al., 1996], we compute the coefficients  $I_j(r,n,q;x)$  using the results given in Theorem 6.6 and obtain after cancelling common factors the following results with the notations:  $\nu = q^r$  and  $\rho = q^n$ .

1. Discrete q-Hermite case  $(\phi(x) = x^2 - 1, \ \psi(x) = \frac{x}{1-a})$ .

$$\begin{split} I_0(r,n,q;x) &= & (x^2 \, \rho \, q^8 + q^7 \, x^2 - q^3 \, \nu \, \rho - q^2 \, \nu \, \rho - \nu \, \rho \, q - \nu \, \rho) \\ & (q \, x - 1) \, (q \, x + 1), \\ I_1(r,n,q;x) &= & (q^{15} \, x^6 - \nu^3 \, \rho^2 + 2 \, \nu^2 \, \rho^2 \, q^{10} \, x^2 + 2 \, \nu^2 \, \rho \, q^9 \, x^2 - q^8 \, \nu \, \rho \, x^4 \\ & + q^4 \, \nu^2 \, \rho \, x^2 + q^4 \, \rho^2 \, x^2 \, \nu^3 + 2 \, q^6 \, x^2 \, \nu^2 \, \rho^2 + 2 \, \nu^2 \, \rho^2 \, q^8 \, x^2 - 2 \, \nu \, \rho \, q^{12} \, x^4 \\ & + 2 \, \nu^2 \, \rho^2 \, x^2 \, q^7 - 2 \, \nu \, \rho \, q^{13} \, x^4 + 2 \, \nu^2 \, \rho \, q^8 \, x^2 + 2 \, \nu^2 \, \rho^2 \, q^8 \, x^2 - \nu \, \rho^2 \, q^{13} \, x^4 \\ & - q^{11} \, x^4 \, \nu \, \rho - q^{10} \, x^4 \, \nu \, \rho + 2 \, \nu^2 \, \rho \, q^6 \, x^2 - q^9 \, \nu \, \rho \, x^4 + 2 \, x^2 \, \rho \, \nu^2 \, q^5 \\ & + \nu^2 \, \rho^2 \, q^{11} \, x^2 + \nu^2 \, \rho \, q^{10} \, x^2 - \nu^3 \, \rho^2 \, q^9 \, x^2 + \nu^2 \, \rho^2 \, q^5 \, x^2 - \nu^3 \, \rho^2 \, q^8 \, x^2 \\ & + \nu^3 \, \rho^2 \, x^2 \, q^5 - \nu \, \rho^2 \, q^{14} \, x^4 + 2 \, \nu^2 \, \rho \, x^2 \, q^7 - \nu \, q^{12} \, x^4 - q^{11} \, x^4 \, \nu + x^6 \, \rho \, q^{16} \\ & - \nu^3 \, \rho^2 \, q^6 - 3 \, \nu^3 \, \rho^2 \, q^4 - 4 \, \nu^3 \, \rho^2 \, q^3 - 3 \, \nu^3 \, \rho^2 \, q^2 - 2 \, \nu^3 \, \rho^2 \, q - 2 \, \nu^3 \, \rho^2 \, q^5) \\ & \rho^{-1} \nu^{-2} q^{-3}, \\ I_2(r,n,q;x) &= & -(q^{17} \, x^6 + 3 \, \nu^2 \, \rho^2 \, q^{10} \, x^2 + 3 \, q^6 \, x^2 \, \nu^2 \, \rho^2 + 4 \, \nu^2 \, \rho^2 \, q^8 \, x^2 \\ & - 2 \, \nu \, \rho \, q^{12} \, x^4 + 4 \, \nu^2 \, \rho^2 \, x^2 \, q^7 - 2 \, \nu \, \rho \, q^{13} \, x^4 + 4 \, \nu^2 \, \rho^2 \, q^8 \, x^2 \\ & - 2 \, \nu \, \rho \, q^{12} \, x^4 + 4 \, \nu^2 \, \rho^2 \, x^2 \, q^7 - 2 \, \nu \, \rho \, q^{13} \, x^4 + 4 \, \nu^2 \, \rho^2 \, q^8 \, x^2 \\ & + 2 \, \nu^2 \, \rho^3 \, q^{15} \, x^4 - \nu \, q^{16} \, \rho^3 \, x^4 + \nu^2 \, \rho^3 \, q^8 \, x^2 + 4 \, \nu^2 \, \rho^3 \, q^9 \, x^2 + 2 \, \nu^2 \, \rho^2 \, q^{11} \, x^4 \\ & + \nu^2 \, \rho^3 \, q^{15} \, x^4 - \nu \, q^{16} \, \rho^3 \, x^4 + \nu^2 \, \rho^2 \, q^{14} \, x^4 - \nu \, \rho^2 \, q^{12} \, x^4 - \nu \, \rho \, q^{14} \, x^4 \\ & + \nu^2 \, \rho^3 \, q^{15} \, x^4 - \nu \, q^{16} \, \rho^3 \, x^4 + \nu^2 \, \rho^2 \, q^{14} \, x^4 - \nu \, \rho^2 \, q^{12} \, x^4 - \nu \, \rho \, q^{14} \, x^4 \\ & + 2 \, \nu^2 \, \rho^3 \, q^{14} \, x^4 + \nu^2 \, q^{13} \, \rho^3 \, x^4 - 2 \, \nu^3 \, \rho^3 \, q^{10} \, x^2 + 4 \, \nu^2 \, \rho^3 \, q^{15} \, x^4 - \nu \, \rho \, q^{14} \, x^4 \\ & + 2 \, \nu^2 \, \rho^3 \, q^{12} \, x^2 - 2 \, \nu^3 \, \rho^3 \, q^9 \, x^2 + 3 \, \nu^2 \, \rho^3$$

$$\begin{array}{lll} & -\nu\,\rho^2\,q^{11}\,x^4-\nu\,\rho^3\,q^{12}\,x^4-\nu\,\rho^2\,q^{16}\,x^4-\nu\,q^{13}\,\rho^3\,x^4+q^{18}\,x^6\,\rho+q^{19}\,x^6\,\rho^2\\ & -5\,\nu^3\,\rho^3\,q^3-4\,\nu^3\,\rho^3\,q^2-4\,\nu^3\,\rho^3\,q^5-5\,\nu^3\,\rho^3\,q^4-\nu^3\,\rho^3\,q^7-2\,\nu^3\,\rho^3\,q^6\\ & -2\,\nu^3\,\rho^3\,q+q^{20}\,x^6\,\rho^3)\nu^{-2}\,\rho^{-2}\,q^{-5}\,,\\ I_3(r,n,q;x) & = & (-\nu^3\,\rho^2+2\,\nu^2\,\rho^2\,q^{10}\,x^2+2\,\nu^2\,\rho\,q^9\,x^2+q^6\,x^2\,\nu^2\,\rho^2\\ & +2\,\nu^2\,\rho^2\,q^8\,x^2+2\,\nu^2\,\rho^2\,x^2\,q^7-2\,\nu\,\rho\,q^{13}\,x^4+2\,\nu^2\,\rho\,q^8\,x^2+2\,\nu^2\,\rho^2\,q^9\,x^2\\ & +2\,\nu^2\,\rho\,q^6\,x^2+x^2\,\rho\,\nu^2\,q^5+2\,\nu^2\,\rho^2\,q^{11}\,x^2+2\,\nu^2\,\rho\,q^{10}\,x^2-\nu^3\,\rho^2\,q^8\,x^2\\ & -\nu\,\rho^2\,q^{14}\,x^4+2\,\nu^2\,\rho\,x^2\,q^7+q^{20}\,x^6-\nu\,x^4\,q^{15}\,\rho^2-2\,\nu\,\rho\,q^{14}\,x^4-\nu\,\rho\,x^4\,q^{15}\\ & -\nu\,q^{12}\,x^4-\nu^3\,\rho^2\,q^6-3\,\nu^3\,\rho^2\,q^4-4\,\nu^3\,\rho^2\,q^3-3\,\nu^3\,\rho^2\,q^2-2\,\nu^3\,\rho^2\,q\\ & -2\,\nu^3\,\rho^2\,q^5-\nu\,q^{13}\,x^4-\nu\,\rho\,q^{16}\,x^4-\nu\,\rho\,q^{18}\,x^4-\nu\,\rho\,q^{17}\,x^4-\nu^3\,\rho^2\,x^2\,q^7\\ & +\nu^2\,\rho\,q^{11}\,x^2+\nu^3\,\rho^2\,q^{11}\,x^2+\nu^3\,\rho^2\,q^{12}\,x^2+\nu^2\,\rho^2\,q^{12}\,x^2+q^{21}\,x^6\,\rho)\\ & \nu^{-2}\,\rho^{-1}\,q^{-6}\,,\\ I_4(r,n,q;x) & = & (x^2\,\rho\,q^6+q^5\,x^2-q^3\,\nu\,\rho-q^2\,\nu\,\rho-\nu\,\rho\,q-\nu\,\rho)\\ & (q^4\,x-1)\,(q^4\,x+1)q^{-6}\,. \end{array}$$

2. Stieltjes-Wigert case ( $\phi(x) = \frac{x}{q}$ ,  $\psi(x) = \frac{qx-1}{q(q-1)}$ )

$$I_{0}(r,n,q;x) = (q^{3}x\nu\rho + x\nuq^{2} + q + 1) q x,$$

$$I_{1}(r,n,q;x) = -(q^{4}\rho^{2}x^{3}\nu^{3} + q^{4}x^{2}\nu^{2}\rho^{2} + q^{3}x^{3}\nu^{3}\rho + 2q^{3}\nu^{2}\rho x^{2} + q^{3}x\nu\rho - q^{2}x + x\nuq^{2} + q^{2}\nu^{2}x^{2} + q^{2}x^{2}\nu^{2}\rho + \nu\rho q^{2}x + q + qx\nu + qx^{2}\nu^{2}\rho + \nu\rho qx + 1 + x\nu + x) q,$$

$$I_{2}(r,n,q;x) = (1 + 2q + 2q^{4}\nu^{2}\rho x^{2} + q^{6}x^{2}\nu^{2}\rho^{2} + q^{2}\nu^{2}x^{2} + 2qx\nu - 2q^{2}x + x^{2}\rho\nu^{2}q^{5} + 2\nu^{2}\rho^{2}q^{5}x^{2} - 2q^{3}x + q^{3}\nu^{2}\rho x^{2} + q^{4}x^{2}\nu^{2}\rho^{2} + 2\nu\rho q^{2}x + q^{2} + 2q^{3}x\nu\rho + 2x\nuq^{2} + 2q^{3}x^{2}\nu^{2} + 2q^{3}x\nu + q^{4}x^{2}\nu^{2} - \nu q^{3}x^{2} - \nu q^{4}x^{2} + q^{4}x^{3}\nu^{3} + \nu^{3}\rho^{2}q^{6}x^{3} + \nu^{3}\rho^{3}q^{7}x^{3} + \nu^{3}\rho q^{5}x^{3} - \nu\rho q^{4}x^{2} - \nu q^{5}x^{2}\rho + 2q^{4}x\nu\rho),$$

$$I_{3}(r,n,q;x) = -(\nu^{3}\rho^{2}x^{3}q^{7} + \nu^{3}\rho q^{6}x^{3} + \nu^{2}\rho q^{6}x^{2} + \nu^{2}\rho^{2}q^{5}x^{2} + x^{2}\rho\nu^{2}q^{5} + q^{4}x\nu\rho + q^{4}x + 2q^{4}\nu^{2}\rho x^{2} + q^{3}x^{2}\nu^{2} + q^{3}x\nu + q^{3}x\nu\rho - q^{2}x + x\nu q^{2} + \nu\rho q^{2}x + q + qx\nu + 1),$$

$$I_{4}(r,n,q;x) = (\nu\rho q^{2}x + q + qx\nu + 1)q^{3}x.$$

#### Bibliography

- [1] Al-Salam, W.A. (1990). Characterisation theorems for orthogonal polynomials, in Orthogonal Polynomials: Theory and practice, (P. Nevai, ed.), NATO ASI Series C: Math. and Phys. Sciences 294, Kluwer, Dordrecht, 1-24
- [2] Area, I., Godoy, E., Ronveaux, A., Zarzo, A. (1996) Discrete Grosjean orthogonal polynomials, SIAM, Orthogonal Polynomials and Special Functions-Newsletter (February, 1996).
- [3] Askey, R.(1965). Orthogonal expansions with positive coefficients, Proc. Amer. Math. Soc. 26, 1191-1194.
- [4] Askey, R.(1975). Orthogonal Polynomials and Special Functions, Regional Conf. Ser. Appl. Math. 21, (SIAM, Philadelphia, Pennsylvania).
- [5] Askey, R., Winp, J.(1984). Associated Laguerre and Hermitte Polynomials, Proc. Roy. Soc. Edinburgh, 96A, 15-37.
- [6] Area, I. Godoy, E., Ronveaux, R. and Zarzo, A.(1998a). Inversion problems in the q-Hahn tableau, submitted.

[7] Area, I., Godoy, E., Ronveaux, A., Zarzo, A.(1998b). Minimal recurrence relations for connection coefficients between classical orthogonal polynomials: Discrete case, J. Comput. Appl. Math.87(2), 321-337.

- [8] Atakishiyev, N.M., Ronveaux, A., Wolf, K.B. (1996). Difference equation for the associated polynomials on the linear lattice, Zt. Teoret. Mat. Fiz. 106 76-83.
- [9] Azatassou, E., Hounkonnou, M. N., and Ronveaux, A. (1998) Les équations de Laguerre-Freud pour l'opérateur  $D_{q,\omega}$  de classe s. Journées Scientifiques de l'Université du Bénin, Lomé 98; TOGO.
- [10] Bangerezako, G. (1998). The fourth-order difference equation for the Laguerre-Hahn polynomials orthogonal on non-uniform lattices, submitted.
- [11] Belmehdi, S., Ronveaux, A. (1989). Polynômes associés des polynômes orthogonaux classiques. Construction via "REDUCE", in . Arias et al. Eds., Orthogonal Polynomials and their Applications, Univ. Oviedo, 72-83.
- [12] Belmehdi, S. (1990a) "Formes Linéaires et Polynômes Orthogonaux Semi-Classiques de classe s=1. Description et Classification", Thèse d'Etat. Université P. et M. Curie. Paris VI, 1990.
- [13] Belmehdi, S. (1990b). On the associated orthogonal polynomials. J. Comput. Appl. Math. 32, 311-319.
- [14] Belmehdi, S., Ronveaux, A. (1991). Fourth-order differential equation satisfied by the associated orthogonal polynomials. *Rend. Mat. Appl.* (7) 11, 313-326.
- [15] Belmehdi, S., Renveaux, A. (1994). Laguerre-Freud Equations for the Recurrence Coefficients of Semi-Classical O thogonal Polynomials, *J. Approx. Theory* **76**, **351**–**368**.
- [16] Böing, H., Koel f, w.(1998) Algorithms for q-hypergeometric Summation in Computer Algebra Konrad-Zuse-Zentrum Berlin, Preprint SC 98-02, 1998.
- [17] Bonan, S. S. (1984). Application of G. Freud theory, I, in "Approximation Theory, IV" (C. K. Chui et al., Eds), pp. 447-351, Academic Press, New york.
- [18] Char, B., W. et al. (1991).: Maple V Language Reference Manual. Springer, New York.
- [19] Chihara, T.S. (1978). An Introduction to Orthogonal Polynomials. Gordon and Breach, New York.
- [20] Dini, J. (1988). Sur les formes linéaires et polynômes orthogonaux de Laguerre-Hahn. Thèse de Doctorat, Univ. P. et M. Curic, Paris.
- [21] Dzoumba, J. (1985).: Sur les Polynômes de Laguerre-Hahn, Thèse de 3e cycle, Université P. et M. Curie, Paris VI.
- [22] Favard, J. (1935). Sur les polynômes de Tchebicheff, C.R. Acad. Sci. Paris, 200, 2052-2053.
- [23] Foupouagnigni, M. (1995). Equations de Laguerre-Freud: Cas des Polynomes Orthogonaux semiclassiques de cla se 2. Memoire de DEA. IMSP, Benin
- [24] Foupouagnigni, M., Hounkonnou, M. N., Ronveaux, A. (1998a). Laguerre-Freud Equations for the Recurrence Coefficients of  $D_{\omega}$ -Semi-classical Orthogonal polynomials of class one, J. Comp. App. Math., 99/1-2 pp. 143-154.
- [25] Foupouagnigni, M., Ronveaux, A., Hounkonnou, M. N. (1998b). The fourth-order difference equation satisfied by the associated orthogonal polynomials of the Δ-Laguerre-Hahn Class. Konrad-Zuse-Zentrum Berlin, Preprint SC 97-71 (accepted for publication in *Journal of Symbolic Computation*).
- [26] Foupouagnigni, `L., Koepf, W., Ronveaux, A. (1998c). The fourth-order difference equation satisfied by the associated classical discrete orthogonal polynomials. J. Comput. Appl. Math., 92 pp. 103-108.
- [27] Foupouagnigni, J., Ronveaux, A., Koepf, W., (1998d). The fourth-order q-difference equation satisfied by the first associated of the q-classical orthogonal polynomials Konrad-Zuse-Zentrum Berlin, Preprint SC 98–96. (accepted for publication in J. Comput. Appl. Math.).

[28] Foupouagnigni, M., Ronveaux, A., Hounkonnou, M. N. (1998e) The fourth-order difference equation satisfied by the associated orthogonal polynomials of the  $\mathcal{D}_q$ -Laguerre-Hahn Class (submitted).

- [29] Foupouagnigni, M., and Van Assche, W. (1998f) Analysis of a system of two non-linear recurrence relations for recurrence coefficients of generalised Meixner and Charlier polynomials (in progress).
- [30] Freud, G. (1976). On the coefficients in the recursion formulae of the orthogonal polynomials, Proc. Roy. Irish Acad. Sect. Λ (1) 76, 1-6.
- [31] Freud, G. (1977). On the zeros of orthogonal polynomials with respect to measures with non-compact support, Anal. Numér. Thér. Approx. 6, 125-131.
- [32] Freud, G. (1986) On the greatest zero of an orthogonal polynomials., J. Approx. Theory 46, 15-23.
- [33] Garcia, A.G., Marcellàn, F., Salto, L. (1995). A distributional study of discrete classical orthogonal polynomials, J. Comput. Appl. Math. 57(2), 147-162.
- [34] Gasper, G. and Rahman, R. (1990) Basic Hypergeometric Series. Encyclopedia of Mathematics and its Applications 35 Cambridge University Press, Cambridge.
- [35] Godoy, E., Ronveaux, A., Zarzo, A., Area, I., (1997a). Minimal recurrence relations for connection coefficients between classical orthogonal polynomials: Continuous case, *J. Comput. Appl. Math.*84(2), 257-275.
- [36] Godoy, E., Marcellàn, F., Salto, L., Zarzo, A. (1997b) Pertubations of discrete semi-classical functionals by Dirac masses, Integral Transforms and Special functions, Vol 5, No 1-2, pp. 19-46.
- [37] Godoy, E., Ronveaux, A., Zarzo, A. and Area, I. (1998). Connection problems for polynomial solutions of non-homogeneous differential and difference equations. NAVIMA preprint No 5107.
- [38] Grosjean, C.C. (1986). The weight functions, generating functions and miscellaneous properties of the sequences of orthogonal polynomials of the second kind associated with the Jacobi and Gegenbauer polynomials, *J. Comput. Appl. Math.* 16 259-307.
- [39] Grossjean, C.C. (1985). Theory of recursive generation of systems of orthogonal polynomials: an illustrative example, J. Comput. Appl. Math. 12,13 299-318.
- [40] Guerfi, M. (1988). Les polynômes de Laguerre-Hahn affines discrets. Thèse. Université P. et M. Curie. Paris VI.
- [41] Hahn, W. (1949) Über Orthogonalpolynome, die q-Differenzengleichungen genügen, Math. Nachr. **2**, 4- 34.
- [42] Hahn, W. (1983) Über Differenzengleichungen für Orthogonalpolynome, Monat. Math. 95, p.269-274
- [43] Hounkonnou, M. N., Hounga, C, and Ronveaux, A. (1998). Discrete semi-classical orthogonal polynomials: Generalized Charlier. Submitted.
- [44] Koekoek, R. and Swarttouw, R. S. (1996) The Askey-scheme of hypergeometric orthogonal polynomials and its q-analogue Report Fac. of technical Math. and Informatics 94-05 T.U. Delft, revised version from February 1996, available at: http://www.can.nl/~renes.
- [45] Koepf, K., Schmersau, D. (1996). Algorithms for classical orthogonal polynomials. Konrad-Zuse-Zentrum Berlin, Preprint SC 96-23.
- [46] Koornwinder, T. (1994). Compact quantum groups and q-special functions. In: Representations of Lie groups and quantum groups, (V. Baldoni and M.A. Picardello eds.), Pitman Research Notes in Mathematics Series 311, Longman Scientific and technical, 46-128.
- [47] Lewanowicz, S. (1995). Results on the associated classical orthogonal polynomials. J. Comput. Appl. Math. 65 215-231.
- [48] Lewanowicz, S. (1996). Recurrence relations for the connection coefficients of orthogonal polynomials of a discrete variable. *J. Comput. Appl. Math.* **76** 213-229.

[49] Lewanowicz, S. (1997). On the fourth order difference equation for the associated Meixner polynomials. J. Comput. Appl. Math. 80, 351-358.

- [50] Lesky, P.(1985): Über Polynomlösungen von Differentialgleichungen und Differenzengleichungen zweiter Ordnung. Anzeiger der Österreichischen Akademie der Wissenschaften, math.-naturwiss. Klasse 121, 29 33.
- [51] Letessier, J., Ronveaux, A., Valent, G. (1996). Fourth order difference equation for the associated Meixner and Charlier polynomials, J. Comput. Appl. Math. 71 331-341.
- [52] Lubinsky, D., S. (1984) A weighted polynomial inequality, Proc. Amer. Math. Soc. 92, 263-267
- [53] Lubinsky, D., S. (1985a) Estimates of Freud-Christoffel functions for some weights with the whole real line as support, J. Approx. Theory 44, 343-379.
- [54] Lubinsky, D., S. (1985b) On Nevai's bounds for orthogonal polynomials associated with exponential weights, J. Approx. Theory 44, 86-91.
- [55] Lubinsky, D. S., Mhaskar, H.N., and Saff, E. B. (1986), Freud conjecture for exponential weights, Bull. Amer. Math. Soc., 15, 217-281
- [56] Lubinsky, D. S., Mhaskar, H.N., and Saff, E. B. (1988). A proof of Freud, conjecture for exponential weights, *Constructive Approximations*, 4, 65-83.
- [57] Magnus, A. (1984). Riccati acceleration of Jacobi continued fractions and Laguerre-Hahn orthogonal polynomials. Lect. Notes Math., 1071, Springer Verlag.
- [58] Magnus, A. (1985a). Asymptotic behaviour of continued fractions coefficients related to singularities of the weight function, in "The Recursion Method and it's Applications," (D. G. Pettifor and D. L. Weaires, Eds.) Solid-State Sciences, Vol. 58, pp.22-45, Springer-Verlag, New york.
- [59] Magnus, A. (1985b). A proof of Freud conjecture about orthogonal polynomials related to  $|x|^{\rho} \exp(-|x|^{2m})$ , in "Orthogonal Polynomials and their applications, Laguerre Symposium, Barle-Duc, 1984" (C. Brezinski et al. Eds.), Lecture Notes in Mathematics, vol 1171, Springer-Verlag, Berlin.
- [60] Magnus, A. (1986). On Freud equations for exponential weights J. Approx. Theory 46, 65-99.
- [61] Magnus, A. (1991). Freud equations for the simplest generalised Jacobi orthogonal polynomials. Internal report, Universite Catholique de Louvain (Belgium).
- [62] Marcellán, F.(1988). Polinomios orthogonales semiclásicos, Una Aproximación Constructiva. Acta V Simposium Polinomios orthogonales. A. Cachafero, E. Godoy. Eds. Vigo (Spain), 100-123.
- [63] Marcellán, F., Branquinho, A., Petronilho (1994). Classical Orthogonal Polynomials: A Functional Approach, Acta Applicandae Mathematicae 34 p. 283-303.
- [64] Marcellán, F., Prianes E. (1996). Orthogonal polynomials and Stieltjes function: The Laguerre-Halm case. Rend. di Matematica. Series VII, 16, 117-141.
- [65] Marcellàn, F., Salto, L. (1998). Discrete semi-classical orthogonal polynomials. Journal of Difference Equations, 4 463-496.
- [66] Maroni, P. (1985a). Sur quelques espaces de distributions qui sont des formes linéaires sur l'espace vectoriel des polynômes. In "Polynômes orthogonaux et applications", C. Brezinski et al. eds., LNM 1171, Springer-Verlag, Berlin 1985, pp.184-194.
- [67] Maroni, P. (1985b). Une caractérisation des polynômes orthogonaux semi-classiques, C. R. Acad. Sci. Paris 301, Série I ,nº 6, P. 269-272
- [68] Maroni, P. (1986a). Sur la suite associée à une suite de polynômes, Publi. Labo. Analyse Numérique, Paris VI, CNRS 86012.
- [69] Maroni, P. (1986b). Le calcul des formes linéaires et les polynômes orthogonaux semi-classiques, Pub. du Labo. Analyse Numérique CNRS, n° 86023, Paris VI (a paraître dans Ségovia 1988).

[70] Maroni, P. (1987). Prolégomènes à l'étude des polynômes orthogonaux semi-classiques. Ann. Mat. Pura. et Appl. (4) 149 (1987), pp. 165-184.

- [71] Màté, A., Nevai, P. and T. Zaslavsky (1985). Asymptotic Expansion of Ratios of Coefficients of Orthogonal Polynomials with Exponential Weight. Trans. Amer. Math. Soc. 287, 495-505
- [72] Medem, J. C. (1996) Polinomios ortogonales q-semi-clásicos, Ph.D. Dissertation, Universidad Politécnica de Madrid.
- [73] Mhaskar, H., N. and Saff, E., B. (1984a). Extremal problems for polynomials with exponential weights, *Trans. Amer. Math. Soc.* **285**, 203-234.
- [74] Mhaskar, H., N. and Saff, E., B. (1984b). Weighted polynomials on finite and infinite intervals: A unified approach, Bull. Amer. Math. Soc. 11, 351-354.
- [75] Morton, R.D., Krall, A.M.(1978) Distributional weight functions for orthogonal polynomials. SIAM, J. Math. Anal. 9 p.604-626.
- [76] Nevai, P. (1973) Orthogonal polynomials on the real line associated with the weight  $|x|^{\alpha} \exp(-|x|^{\beta})$ , I, Acta. Math. Sci. Math. Hungar., 24, 335-342.
- [77] Nevai, P. (1983) Orthogonal polynomials associated with  $\exp(-|x|^4)$ , in "Second Edmonton Conference on Approximation Theory", pp. 263-285, Canadian Math. Soc. Conference Proceedings, Vol. 3.
- [78] Nevai, P. (1984a) Asymptotic for orthogonal polynomials associated with  $\exp(-|x|^4)$ , SIAM, J. Math. Anal. 15, 1177-1187.
- [79] Nevai, P. (1984b).

  Two of my favorite ways of obtain obtaining asymptotics for orthogonal polynomials, in "linear functional Analysis and Approximation" (P. L. Butzer, R. L. Stens, and B. Sz-Nagy, Eds.), ISNM 65, Birkhäuser Verlag, Basel, 417-436.
- [80] Nevai, P. (1985) Exact bounds for orthogonal polynomials associated with exponential weight, J. Approx. Theory 44, 82-85.
- [81] Nevai, P. and freud, G. (1986) Orthogonal polynomials and christoffel functions, J. Approx. Theory 48, 3-167
- [82] Nikiforov, A., Uvarov, V. (1983). Fonctions spéciales de la physique mathématiques. Editions Mir, Moscou.
- [83] Nikiforov, A.F., Suslov, S.K., Uvarov, V.B. (1991). Classical Orthogonal Polynomials of a Discrete Variable. Springer, Berlin.
- [84] Perron, O.(1957). Die Lehre von den Kettenbrüchen, 3eme ed., Leipzig.
- [85] Roman, S.M., Rota, G.C.(1978). The Umbral calculus. Adv. in Math. 27 p. 95-188.
- [86] Ronveaux, A. (1986). Discrete Semi-classical Orthogonal Polynomials: Generalized Meixner, J. Approx. Theory 46, (4), 403-407.
- [87] Ronveaux, A.(1988a). Fourth-order differential equations for numerator polynomials, J. Phys. A: Math. Gen. 21, 749-753.
- [88] Ronveaux, A. (1991). 4th order differential equation and orthogonal polynomials of the Laguerre-Hahn class. *IMACS Ann. Comput. Appl. Math.* 9, Baltzer, Basel, 379-385.
- [89] Ronveaux, A. (1993). Limit transition in Askey tableau. J. Comput. Appl. Math. 48, 334-335.
- [90] Ronveaux, A., Van Assche, W. (1996). Upward extension of the Jacobi matrix for orthogonal polynomials, J. Approx. Theory 86, 3, 335-357.
- [91] Rouveaux, A., Godoy, E., Zarzo, A., and Area, I. (1998a). Fourth-order difference equation for the first associated of classical discrete orthogonal polynomials. Letter, J. Comput. Appl. Math. 90 (1998), 47-52.

[92] Ronveaux, A., Salto, L. (1998b). Discrete orthogonal polynomials: Polynomials modification of a classical functional. to appear in Journal of Difference Equations

- [93] Salto, L.D., (1995). "Polinomios  $D_{\omega}$ -Semiclásicos", Tesis Doctoral, Universidad de Alcalá de Henares.
- [94] Sheen, R. (1984). Orthogonal polynomials associated with  $\exp(-|x|^6/6)$  Ph. D. dissertation, Ohio State Univ., Columbus, Ohio.
- [95] Sherman, J. (1933). On the numerators of the convergents of the Stieltjes continued fractions, *Trans. Amer. Math. Soc.* **35**, 269-272.
- [96] Shohat, J.A. (1938). Sur les polynômes orthogonaux généralisés. C. R. Acad. Sci. Paris, 207, pp. 556-558.
- [97] Smaili, N. (1987). Sur les polynômes E-semi-classiques de classe zero. Thèse de Doctorat 3<sup>eme</sup> cycle, Univ. P. et M. Curie, Paris VI.
- [98] Stone, M.H. (1932). Linear transformations in Hilbert spaces and their application to analysis. Colloq. Publ. no 15, Amer. Math. Soc., New-York.
- [99] Suslov, S. K. (1989) The theory of difference analogues of special functions of hypergeometric type. Russian Math. Surveys 44:2 (1989), 227-278
- [100] Szegö, G. (1939) 'Orthogonal Polynomials. Amer. Math. Soc., Colloq. Publ. vol 23, Amer. Math. Soc., New-York.
- [101] Szegö, G. (1975) "Orthogonal Polynomials", 4th edition, Amer. Math. Soc. Colloq. Public. 23 Providence R.I.
- [102] Wimp, J. (1987). Explicit formulas for the associated Jacobi polynomials and some applications, Can. Jour. Math. XXXIX, 4 983-1000.
- [103] Wintner, A. (1929). Spektraltheorie der unendlichen Matrizen, 2eme ed., Leipzig.
- [104] Wolfram Research, Inc, Mathematica version 2.2, Champain, Illinois, 1993.
- [105] Zarzo, A., Ronveaux, A., Godoy, E. (1993). Fourth-order differential equation satisfied by the associated of any order of all classical orthogonal polynomials. A study of their distribution of zeros. J. Comput. Appl. Math. 49, 349-359.